Australian Guidelines on Attention Deficit Hyperactivity Disorder (ADHD)

June 2009



The Royal Australasian College of Physicians

Disclaimer

This document is a general guide to appropriate practice, to be followed subject to the clinician's judgement and the patient's preference in each individual case. The Guidelines are designed to provide information to assist decision making and are based on the best evidence available at the time of development.

Funding

The Royal Australasian College of Physicians gratefully acknowledges the financial assistance for the development of these guidelines, which was provided by the Australian Government Department of Health and Ageing.

TABLE OF CONTENTS

Executive Summary		vii
Recommendations		x
Introduction Aim of the Guidelines Who are the Guidelines for? What do the Guidelines address? Terms of Reference Guideline development process Literature review	1 1 1 2 3	1
PART I: WHAT IS ADHD? CONCEPTUALISATION, CAUSES,		
PREVALENCE AND CONSEQUENCES		6
Chapter 1. Introduction: Definition, History and Current Diagnos	tic	_
Criteria 1.1 Definition 1.2 A historical perspective 1.3 Current conceptualisation and diagnostic criteria Chapter 2. Aetiology and Neurobiology	7 7 8	7
2.1 Aetiology 2.2 Neurobiology	14 18	
Chanter 3 Prevalence and Associated Developmental Problems	-	20
3.1 Epidemiology 3.2 Comorbidities	20 22	20
Chapter 4. Consequences of ADHD 4.1 Natural history 4.2 Impairments and issues across the lifespan	25 26	25
PART II: ASSESSMENT AND DIAGNOSIS		29
Chapter 5. Assessment and Diagnosis		31
 5.1 Recognition 5.2 Overview of Assessment 5.3 Initial screening assessment 5.4 Assessment and diagnosis of ADHD in school-aged children and adolescents 5.5 Assessment and diagnosis of ADHD in preschool-aged children 	31 31 33 35 39	
5.6 Assessment and diagnosis of ADHD in adults 5.7 Psychoeducational assessment 5.8 Allied health assessments 5.9 Issues in differential diagnosis 5.10 ADHD measures and information sources 5.11 Specific populations 5.12 Diagnostic tools	41 45 46 47 48 53 54	
PART III: MANAGEMENT		61
Chapter 6. Overall Management 6.1 Access to services 6.2 Requirements of ADHD services 6.3 Developing an individualised management plan 6.4 Multimodal therapy 6.5 Multidisciplinary care 6.6 Treatment monitoring and review 6.7 Family support / intervention	63 66 68 69 69 70 70 70	63

Chapter 7. Psychosocial Management		71
7.1 Introduction	71	
7.2 Efficacy of psychosocial interventions	73	
7.3 Psychosocial interventions for ADHD when comorbidities are present	78	
Chapter 8. Medication		80
8.1 Introduction	80	
8.2 Stimulant medication (methylphenidate, dexamphetamine and mixe	d	
amphetamine salts)	84	
8.3 Atomoxetine (non-stimulant)	90	
8.4 Other medications used in ADHD	93	
8.5 Medication management of ADHD when comorbidities are present	102	
8.6 Medications for ADHD in children with a developmental disability	112	
8.7 Side effects: special considerations	116	
8.8 Monitoring of medication effectiveness and side effects	124	
8.9 ADHD and substance misuse	124	
8.10 Athletes with ADHD receiving stimulant medication	129	
8.11 Summary of appraches to management of medication in ADH	129	
Chapter 9. Medication Compared to and Combined with Psychosoci	al	
Interventions135		
9.1 Introduction	135	
9.2 Comparing psychosocial and pharmacological interventions	136	
9.3 Combining psychosocial and pharmacological interventions	138	
9.4 Moderators and mediators of outcomes from psychosocial interventi	ons	
and combined interventions	141	
Chapter 10 Educational Management		143
10 1 Educational research	143	145
10.2 Inclusion: a legal requirement	144	
10.3 Inclusive educational strategies	144	
10.4 Teacher knowledge	146	
10.5 Resources	147	
10.6 Educational challenges for students with ADHD	147	
10.7 Collaborative approaches	151	
10.8 School-based interventions	152	
10.9 Resources for teachers	155	
Chapter 11 Complementary and Alternative Treatments for ADHD		158
11 1 Introduction	158	150
11 2 Diet	159	
11.3 Chiropractic	162	
11.4 Behavioural optometry	162	
11.5 Biofeedback	163	
11.6 Homeonathy	164	
11.7 Cerebellar theranies	165	
11.8 Sport, exercise and relaxation	166	
11.9 Sensory integration interventions	167	
PART IV: SOCIAL AND ECONOMIC CONSIDERATIONS		168
		-00
Chapter 12. The Voices of Individuals with ADHD	160	169
12.1 AUDIESCETIL VOICES	170	
12.2 AUUIL VOICES	170	
Chapter 13. ADHD in the Workplace		176
13 1 Employment nattorns		1/0
15.1 Employment patterns	176	170
13.2 Work performance	176 177	170

13.4 Workplace discrimination	179	
Chapter 14. ADHD and Driving 14.1 Obtaining a driver's or pilot's licence 14.2 ADHD and driving ability, accident rates and traffic violations 14.3 Effect of treatment on driving ability in people with ADHD 14.4 Legal issues regarding driving and medications	180 180 181 182	180
Chapter 15. Issues for Families, Parents and Carers 15.1 Families with a child or adolescent with ADHD 15.2 Adult ADHD and its impact on the family 15.3 Children and adolescents with ADHD in out-of-home care 15.4 Issues for families living with ADHD that have not been addressed these Guidelines	183 190 192 in 193	183
Chapter 16. ADHD and Justice 16.1 ADHD and risk of delinquency and criminal behaviour 16.2 Prevalence of ADHD and comorbidities among offenders 16.3 Impact of treatment for ADHD on rates of offending and recidivism 16.4 Particular vulnerabilities of people with ADHD in the justice system	195 196 200 201	195
Chapter 17. Economic Considerations in ADHD 17.1 Economic burden of ADHD 17.2 Cost-effectiveness of treatment 17.3 Conclusions	203 204 205	203
Abbreviations and Acronyms		206
Glossary		208
References		212

TABLES

Table 1. NHMRC levels of evidence according to type of research	4
Table 2. Grading of recommendations	5
Table 3. DSM-IV diagnostic criteria for ADHD	10
Table 4. ICD-10 diagnostic criteria for hyperkinetic disorders	11
Table 5. State and Territory contacts for information of the regulation of stimulant prescribing	81
Table 6. Dosing guidelines for first-line medications.	131
Table 7. Prevalence of ADHD amongst offender populations	197

FIGURES

Figure 1: Prescriptions of medication used in treatment of ADHD in Australia.. 204

Appendices

The following appendices are available as separate documents.

Appendix A: Membership of Guideline Development Groups

Appendix B: Dualities and Conflicts of Interest

Appendix C: Process Report

Appendix D: Systematic Review Process Report

Appendix E: Systematic Review Research Questions

Appendix F: Data- Extraction and Critical Appraisal Tables

Appendix G: NHMRC Evidence Statement Forms

Appendix H: Assessment Scales and Questions Appendix I: Australian ADHD Support and Education Groups



EXECUTIVE SUMMARY

Attention deficit hyperactivity disorder (ADHD) is a common condition that has been defined in the *Diagnostic and Statistical Manual of Mental Disorders IV* (DSM-IV) as a pattern of inattention and/or hyperactivity-impulsivity symptoms that persist for at least 6 months to a degree that is maladaptive and inconsistent with developmental level. There must be some impairment from symptoms that is present in 2 or more settings (e.g. at school or work and at home) and there must be clear evidence of significant impairment in social, school or work functioning.

Not every person with ADHD will have all of the symptoms, and the severity of the symptoms of ADHD and the level of impairment will vary between individuals. In addition, ADHD symptoms and severity can change with age. Some symptoms, such as hyperactivity-impulsivity, diminish abruptly or present differently with age; other symptoms, such as inattention, are more likely to persist into adulthood. Many people with ADHD have one or more associated problems, such as learning difficulties or anxiety. The long-term outcomes for individuals with ADHD are wide ranging. For the majority of people with ADHD the disorder will persist through childhood into adolescence and adulthood. Although many people with ADHD will grow up without persistent problems, individuals with ADHD are at increased risk of a range of adverse outcomes. These include academic underachievement, difficulties with interpersonal relationships and low-self-esteem, all of which have potentially serious consequences for the individual, and the flow-on effects of ADHD can have a significant impact on families, schools, workplaces and the community more broadly.

The Guidelines on ADHD have been developed in accordance with National Health and Medical Research Council (NHMRC) guideline development requirements by a multidisciplinary expert reference group. The reference group was comprised of experts from throughout Australia and included representatives from each of the key professional disciplines involved with ADHD: paediatrics, child and adolescent psychiatry, adult psychiatry, psychology, general practice, education and consumers/carers.

The Guidelines provide a series of evidence-based recommendations for the assessment, management and care of preschoolers, children, adolescents and adults with ADHD. Where possible, the recommendations have been developed on the basis of current research that was identified by systematic review and reflect current scientifically based "best practice". For areas of practice not addressed by current research, recommendations were developed on the basis of the consensus opinion of the clinicians, educators and consumers from the reference group. The recommendations that are derived from expert consensus are extensive, but it is important that they are included in the guidelines in order to provide practitioners with as comprehensive set of directions as possible.

The recommendations should not be used prescriptively; they are intended as a general guide to appropriate practice, to be followed subject to the clinician's judgement and the preference of the person with ADHD and/or his or her parents or guardians. The guideline recommendations are far reaching in scope, covering the areas of assessment and diagnosis of ADHD; treatment and management of ADHD; and social considerations.

Key messages

Assessment and diagnosis of ADHD

- 1. The DSM-IV criteria are the minimum necessary for diagnosis of ADHD.
- 2. The diagnosis of ADHD should only be made after a comprehensive assessment. This includes medical, developmental and psychosocial assessment, and elicitation of evidence of impairment in multiple settings, via gathering information from multiple informants.

Management of ADHD

- 3. Individuals with ADHD and their families and carers should be provided with information and education about ADHD and its impact, and the advantages and disadvantages of potential treatment strategies.
- 4. Multimodal therapy is recommended for the treatment of ADHD in all age groups. This could include psychosocial management strategies, medication and educational interventions.
- 5. An individualised management plan should be drawn up in collaboration with the person with ADHD and their parents/carers and teachers. Taking into account;
 - The specific needs and expressed preferences of the person, and the circumstances of his or her family and culture.
 - The associated psychosocial problems, educational difficulties and comorbid conditions.
 - The suitability of the plan for the individual and their family, considering affordability, accessibility and acceptability.
- 6. Clinicians should be alert to the risk of depression or other psychiatric disorders in parents/caregivers of children or adolescents with ADHD. Parents/caregivers may need referral for support and treatment.
- 7. Medication and ADHD
 - Not all people with ADHD will require pharmacological management.
 - Medications should only be used when symptoms are pervasive across settings (eg. school and home) and causing significant impairment in academic, social or behavioural function, and after careful consideration of non-pharmacological approaches. Clearly defined goals should be identified prior to commencing a trial of medication treatment.
 - Medication should not be used as first-line treatment for ADHD in preschool-aged children.
 - Patients receiving treatment for ADHD should be reviewed regularly (at least 6-monthly) to ensure that the management strategies remain appropriate and effective.
- 8. Other therapies
 - Elimination and restriction diets are not supported as a general treatment for individuals with ADHD. A subset of children may be sensitive to certain foods or food additives and may benefit from careful exclusion diets. Assessment of food sensitivity and initiation of a special diet should be under the care and supervision of a medical specialist and an Accredited Practising Dietitian.

• Alternative treatments (including fatty acid supplements, biofeedback, homeopathy or sensory integration diets) are not currently supported as treatments for individuals with ADHD.



RECOMMENDATIONS

The *Guidelines on Attention Deficit Hyperactivity Disorder (ADHD)* have been developed to provide a series of evidence-based recommendations for the assessment, management and care of preschoolers, children, adolescents and adults with ADHD. The guidelines are intended to provide a framework based on the best available evidence that can be adapted to local needs and resources, and individual circumstances.

The Guidelines on ADHD have been developed in accordance with National Health and Medical Research Council (NHMRC) guideline development requirements by a multidisciplinary expert reference group. The reference group was comprised of experts from throughout Australia from the key professional disciplines involved with ADHD: paediatrics, child and adolescent psychiatry, adult psychiatry, psychology, general practice, education and consumers/carers.

Guideline recommendations have been developed on the basis of current research that was identified by systematic review. These recommendations have been given an overall grading (A, B, C or D) based on NHMRC criteria. For areas of practice not addressed by current research, recommendations were developed on the basis of the consensus opinion of the clinicians, educators and consumers from the reference group. Research recommendations have not been graded.

Grade	Description
Α	Body of evidence can be trusted to guide practice
В	Body of evidence can be trusted to guide practice in most situations
С	Body of evidence provides some support for recommendation(s) but care should be
	taken in its application
D	Body of evidence is weak and recommendation must be applied with caution
Best pr	ractice points
\checkmark	Recommended best practice based on clinical experience and expert opinion, where
	there is inadequate evidence to establish evidence based recommendations.

Note: Refer page 4 – *Table 1. NHMRC levels of evidence according to type of research question* (1).

Key Definitions

DSM-IV: *Diagnostic and Statistical Manual of Mental Disorders*, 4th edition.

ADHD Severity: Determining the severity of ADHD is a matter for clinical judgement that will take into account the level of impairment, pervasiveness and individual factors, as well as family, social and cultural contexts.

Assessment and diagnosis

Initial screening assessment

1. A short scale such as the Strengths and Weaknesses of Attention Scale (SWAN) or the Diagnostic Rating Scale (DRS), or for adults the Short Adult ADHD Screening Scale or Barkley's Adult ADHD Quick Screen should be used as an initial screen.

 \checkmark Recommended best practice based on clinical experience and expert opinion

2. Clinicians, especially primary care physicians, need to consider that there is more than one form of ADHD and many children may not present with the most obvious symptoms of hyepractivity/impulsivity. The inattentive type may not present until secondary school with increased demands for organisation and independent study.

Recommended best practice based on clinical experience and expert opinion

Assessment and diagnosis of ADHD in school-aged children and adolescents

- 3. Full assessment for possible ADHD requires a comprehensive medical, developmental and psychosocial assessment by the best qualified clinician available. This would usually be a paediatrician or child and adolescent psychiatrist with the training and skills required to assess and treat ADHD. *Recommended best practice based on clinical experience and expert opinion*
- 4. The DSM-IV criteria are the minimum necessary for diagnosis of ADHD. ✓ Recommended best practice based on clinical experience and expert opinion
- 5. Information from and discussion with multiple informants, including the child/adolescent, parents/caregivers, and educational and health professionals, needs to inform a diagnosis of ADHD in school-aged children.
 ✓ Recommended best practice based on clinical experience and expert opinion
- Diagnosis requires evidence of moderate to severe impairment across settings, including home and school.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 7. A thorough psychosocial assessment of the child/adolescent and family and medical assessment of the child/adolescent are part of a comprehensive assessment.

✓ Recommended best practice based on clinical experience and expert opinion

8. Assessment should cover the presence and functional significance of comorbidities, including learning disabilities, anxiety/depression and disruptive behaviour disorders. This could include use of broad-based screening questionnaires.

✓ Recommended best practice based on clinical experience and expert opinion

Assessment and diagnosis of ADHD in preschool-aged children (3–5 years)

- 9. For children under 6 years of age, a stage when child development is rapid, it is essential to distinguish ADHD symptoms from normal developmental variation in impulsivity and attention.

 recommended best practice based on clinical experience and expert opinion
- 10. Assessment of children under 6 years of age should be undertaken especially thoroughly by paediatricians or child psychiatrists with expertise in developmental assessment, paying particular attention to identification of comorbidities and understanding of family dynamics and of cultural/religious diversity.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 11. The DSM-IV criteria are the minimum necessary for diagnosis of ADHD. *Recommended best practice based on clinical experience and expert opinion*
- 12. Information from and discussion with multiple informants, including the child, parents/caregivers, kinder and preschool teachers, childcare staff, preschool staff and health professionals, needs to inform a diagnosis of ADHD in preschool-aged children.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 13. Diagnosis requires evidence of moderate to severe impairment across settings, including home and kinder/preschool.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 14. Assessment of children under 6 years of age should include, as a minimum, a screening developmental measure, such as the Ages and Stages Questionnaire, and, when developmental delay is suspected, a formal developmental assessment such as the Griffiths Mental Developmental Scales.

✓ Recommended best practice based on clinical experience and expert opinion

- 16. In this age range, where it is difficult to differentiate ADHD from other common problems, including developmental delays and specific speech and motor disorders, allied health assessments should be considered, especially when developmental difficulties are evident.

✓ Recommended best practice based on clinical experience and expert opinion

Assessment and diagnosis of ADHD in adults

- 17. The DSM-IV criteria are the minimum necessary for diagnosis of ADHD. ✓ Recommended best practice based on clinical experience and expert opinion
- 18. ADHD needs to be considered in adults who present with longstanding symptoms suggestive of ADHD (inattention, impulsivity, disorganisation) that appear to have started in childhood and are persisting into adult life. *Recommended best practice based on clinical experience and expert opinion*
- 19. People with personality disorder and/or substance abuse accompanied by a significant level of impulsivity accompanied by inattention should be referred for evaluation of ADHD.

 Create Recommended best practice based on clinical experience and expert opinion
- 20. Assessment of adults with suspected ADHD should include a comprehensive medical and psychosocial assessment.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 21. Assessment of adults with suspected ADHD should be undertaken by the best qualified clinician available. This would usually be an adult psychiatrist or a comprehensive psychiatric service with the training and skills required to assess and treat ADHD. This is due to the high incidence of comorbid psychiatric conditions and the overlap of clinical features of ADHD with other conditions such as bipolar disorder and personality disorders. *Recommended best practice based on clinical experience and expert opinion*
- 22. Other possible diagnoses or comorbidities should be considered, via history; for example, acquired brain injury, neurological condition or other DSM-IV diagnosis such as anxiety disorder or pervasive developmental disorder.

√Recommended best practice based on clinical experience and expert opinion

23. Adult presentations may occur in the context of problems encountered with work and study. In such instances, vocational/intellectual assessments may be useful, not for diagnostic purposes, but to clarify the functional consequences of the diagnosis.

✓Recommended best practice based on clinical experience and expert opinion

Psychoeducational assessment

- 24. Most children with ADHD have intellectual development within the normal range. A significant proportion, however, present with cognitive deficits, learning difficulties and social adaptive difficulties, and in these individuals a comprehensive psychoeducational assessment is particularly necessary. *Recommended best practice based on clinical experience and expert opinion*
- 25. Psychoeducational assessment should cover all aspects of academic performance. At a minimum this should consist of an individually administered intelligence test such as the age-appropriate version of the Wechsler series.

✓ Recommended best practice based on clinical experience and expert opinion

26. Educational difficulties are common in people with ADHD. Where these problems are suspected (e.g. via parent/caregiver or teacher report, or national/State/Territory literacy and numeracy assessments),

psychoeducational assessment should be conducted to assess learning potential and attainment.

✓ Recommended best practice based on clinical experience and expert opinion

27. Assessment of educational difficulties is important in people of all ages with suspected ADHD and/or suspected educational difficulties, to inform diagnosis and identify learning difficulties that should be targeted for intervention. Such assessments are not diagnostic of ADHD.
 ✓ Recommended best practice based on clinical experience and expert opinion

Allied health assessments

28. Allied health assessment may be helpful in many children and adolescents with suspected ADHD, to identify comorbid speech and language difficulties, motor co-ordination and sensory difficulties that can be targeted with intervention.

✓ Recommended best practice based on clinical experience and expert opinion

29. The need for allied health assessment should be determined on a case-bycase basis.

✓ Recommended best practice based on clinical experience and expert opinion

Assessment when acquired brain injury and other neurological conditions are present

- 30. A thorough medical history and examination are required to identify any acquired brain injury or other neurological condition that may contribute to the presenting symptoms.

 Recommended best practice based on clinical experience and expert opinion
- 31. Individuals with ADHD symptoms and a history of acquired brain injury and other neurological conditions should received comprehensive neuropsychological evaluation.

 Recommended best practice based on clinical experience and expert opinion
- 32. In consultation with the neurologist or neurosurgeon, ADHD medications may be one component of a comprehensive management plan.
- ✓ Recommended best practice based on clinical experience and expert opinion

Measuring impairment

- 33. While the DSM-IV diagnosis of ADHD does require impairment, impairment at home, school/work and in social relationships is poorly defined. Careful discussion is needed with the family as to the extent to which problems are specifically related to ADHD symptomatology. Recommended best practice based on clinical experience and expert opinion
- 34. Measures of impairment specific to ADHD for preschoolers, children and adolescents are not yet sufficiently robust for routine use. (Grade D)
- 35. The Current Behaviour Subscale could be used as a measure for functional impairment in adults with ADHD. (Grade C)

Retrospective self-reports

36. The reports provided by adults with ADHD on the symptoms they had in childhood are accurate and valid for informing diagnosis and assessment. (Grade B)

Information from multiple informants

37. For diagnosis of ADHD in children and adolescents, input from multiple informants (e.g. parents/caregivers, teachers) should be considered in evaluating the chronicity and pervasiveness of impairment. (Grade B)

- 38. Diagnosis of ADHD should not solely rely on the use of parent *or* teacher information, as this may lead to over-diagnosis. Information from both parents *and* teachers should be included. ✓ *Recommended best practice based on clinical experience and expert opinion*
- 39. For diagnosis of ADHD in adults, input from multiple informants (e.g. partners, family members) should be considered, where possible, in evaluating chronicity and pervasiveness of impairment. (Grade B)
- 40. When evaluating parent reports describing children with suspected ADHD, consideration should be given to the psychiatric status of the parent, as maternal depression may have an impact on the mother's reporting of symptoms in her child. (Grade D)
- 41. There are insufficient research data to determine whether ADHD in the parent has an impact on their reporting of ADHD symptoms in their children. (Grade D)

Assessment and diagnosis in specific populations

Culturally and linguistically diverse populations

42. Cultural differences in perceived levels of impulsivity, hyperactivity and inattention should be considered in the diagnosis of ADHD. ✓ Recommended best practice based on clinical experience and expert opinion

Indigenous Australians

43. There is need for a culturally appropriate assessment of ADHD in Indigenous people.

✓ Recommended best practice based on clinical experience and expert opinion

Diagnostic tools

Neuropsychological measures

- 44. There are insufficient research data to recommend the inclusion of neuropsychological measures in routine diagnostic assessment for ADHD. (Grade D)
- 45. Where functional difficulties persist in the context of ADHD (e.g. learning problems, organisational difficulties), despite initial educational, pharmacological and psychological strategies, neuropsychological assessment can provide information on cognitive strengths and weaknesses. This information can then be used by the person with ADHD and the teacher and/or parent/caregiver to develop compensatory approaches to learning and daily functions.

Neurophysiological measures

46. There are insufficient research data to recommend the inclusion of neurophysiological measures in routine diagnostic assessment for ADHD. (Grade D)

Neuroimaging

47. There are insufficient research data to recommend the inclusion of neuroimaging in routine diagnostic assessment for ADHD. (Grade D)

Overall management

Developing an individualised management plan

- 48. An individualised management plan should be drawn up in collaboration with the person with ADHD and their parents/carers and teachers as appropriate. The management plan should take into account:
 - the specific needs and expressed preferences of the person, and the circumstances of his or her family and culture.
 - the associated psychosocial problems, educational difficulties and comorbid conditions.
 - the suitability of the plan for the individual and their family, taking into account affordability, accessibility and acceptability.
 Recommended best practice based on clinical experience and expert opinion
- 49. People with ADHD and their families and carers should be provided with information and education about ADHD and its impact, and the advantages and disadvantages of potential treatment strategies.
 ✓ Recommended best practice based on clinical experience and expert opinion

Multimodal therapy

50. A multimodal approach is recommended for treatment of ADHD. This may include medication, psychosocial management strategies and, where appropriate, educational interventions.

Recommended best practice based on clinical experience and expert opinion

Treatment monitoring and review

51. Patients receiving treatment for ADHD should be reviewed regularly (at least 6-monthly) to ensure that the management strategies remain appropriate and effective.
 ✓ Recommended best practice based on clinical experience and expert opinion

Family support/intervention

52. Participation in support groups can be of benefit for parents/carers of children and adolescents with ADHD.

✓ Recommended best practice based on clinical experience and expert opinion

53. Participation in adult ADHD support groups can be of benefit to adults with ADHD.

✓ Recommended best practice based on clinical experience and expert opinion

Psychosocial management

Psychosocial interventions for preschool-aged children (3–5 years)

- 54. Structured parenting programs with demonstrated effectiveness could be considered for preschoolers with ADHD and associated behavioural problems. (Grade B)
- 55. In considering the use of psychosocial interventions, availability, the family's resources and their capacity to adhere to the program should all be taken into account.

 \checkmark Recommended best practice based on clinical experience and expert opinion

Psychosocial interventions for school-aged children

56. Structured parenting programs with demonstrated effectiveness could be considered for children with ADHD and associated behavioural problems. (Grade B)

- 57. Used alone, clinic-based social skills training is not recommended for improving social skills in children with ADHD. (Grade D)
- 58. There are insufficient research data to recommend the use of cognitive behavioural therapy (CBT) or family therapy for the treatment of ADHD symptoms. (Grade D)
- 59. Cognitive behavioural therapy (CBT) or family therapy may, however, be useful in addressing certain comorbidities, including anxiety, ODD and CD. ✓ Recommended best practice based on clinical experience and expert opinion

Psychosocial interventions for adolescents

- 60. The interventions described for children (with the exception of parenting programs) could be applied to young adolescents with ADHD. ✓ Recommended best practice based on clinical experience and expert opinion
- 61. For older adolescents with functional impairments associated with ADHD, the cognitive behavioural therapy (CBT) strategies described for adults may prove beneficial.

✓ Recommended best practice based on clinical experience and expert opinion

Psychosocial interventions for adults

- 62. For adults whose ADHD has been stabilised on medication, a course of either group or individual cognitive behavioural therapy (CBT) could be considered to improve ADHD symptoms further and assist with daily functioning in areas such as organisational skills, self-esteem and social skills. (Grade B)
- 63. Psychoeducation, counselling, psychotherapy or coaching may benefit some adults with ADHD.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 64. Consideration of the use of psychosocial interventions should take into account availability, and the individual's resources and capacity to adhere to the program.

✓ Recommended best practice based on clinical experience and expert opinion

Psychosocial interventions for ADHD when comorbidities are present

- 65. Psychosocial interventions should be considered for individuals with ADHD and comorbidities.
 - ✓ Recommended best practice based on clinical experience and expert opinion
- 66. Psychosocial interventions can be particularly beneficial when the individual with ADHD has comorbid anxiety. (Grade C)

Medication

- 67. Not all people with ADHD require pharmacological management. ✓ Recommended best practice based on clinical experience and expert opinion
- 68. Medications should only be used when symptoms are pervasive across settings (eg. school and home) and causing significant impairment in academic, social or behavioural function, and after careful consideration of non-pharmacological approaches. Clearly defined goals should be identified prior to commencing a trial of medication treatment *Recommended best practice based on clinical experience and expert opinion*
- 69. A multimodal approach is recommended for treatment of ADHD. Psychosocial management strategies and, where appropriate, educational interventions should be used in conjunction with medication treatments. ✓ *Recommended best practice based on clinical experience and expert opinion*

Notes: Also see recommendations on Side effects: special considerations. Informed consent should be gained if off-label prescribing is being contemplated. Refer to Chapter 8.

Stimulant medication (methylphenidate (MPH), dexamphetamine (DEX) and mixed amphetamine salts (MAS))

Preschool-aged children (3–5 years of age)

- 70. Medication should not be used as first-line treatment for ADHD in preschool-aged children.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 71. The use of medication should only be considered in preschool-aged children when there has been poor response to behavioural or psychosocial therapy and the ADHD symptoms are having a severe impact on the child and their family/carers. For preschool-aged children medication should only be initiated following specialist assessment and in the context of multidisciplinary care, preferably in a tertiary setting.
- 72. Where a decision is made to trial medication, MPH-IR should be used at low dose, in conjunction with appropriate behavioural intervention. (Grade C)
- 73. Preschool-aged children on medication need to be monitored closely because of the increased incidence of side effects in this age group. (Grade B)
- 74. Extended-release forms of stimulants should not be routinely used in preschool-aged children.

✓ Recommended best practice based on clinical experience and expert opinion

Note: Use of stimulant therapy in preschool-aged children is regulated underspecific State and Territory legislation.

School-aged children

- 75. Where severe ADHD is present, treatment with MPH or DEX should be considered as a first-line pharmacological treatment. (Grade A)
- 76. The choice of MPH-immediate-release or MPH-extended-release depends on the symptom profile, as well as individual child and parent/caregiver preferences. (Grade A)

Adolescents

- 77. Where severe ADHD is present, treatment with MPH or DEX should be considered as a first-line pharmacological treatment. (Grade A)
- 78. The choice of MPH-immediate-release or MPH-extended-release depends on the symptom profile, as well as the preference of the adolescent with ADHD. (Grade B)

Adults

- 79. Where severe ADHD is present, treatment with MPH or DEX should be considered as a first-line pharmacological treatment. (Grade A)
- 80. The choice of MPH-immediate-release or MPH-extended-release depends on the symptom profile, as well as the preference of the person with ADHD. (Grade B)

All age groups

81. When stimulant treatment is used, it should be continued for as long as it is of assistance to the person with ADHD and is not causing unacceptable

side effects.

✓ Recommended best practice based on clinical experience and expert opinion

Stimulants not available in Australia

 MAS may have a role in the pharmacological management of ADHD in primary school-aged children, adolescents or adults. (Grade B)
 Note: MAS is not currently available in Australia.

Immediate-release versus extended-release

- 84. Immediate-release forms should be the initial treatment, to titrate to the optimal dose, and they may be the preferred maintenance therapy for various reasons, for example, flexibility of dosing. ✓ Recommended best practice based on clinical experience and expert opinion
- 85. If starting on immediate-release stimulants, consideration should be given to changing to an extended-release form once the optimal dose has been established. This can help to avoid the stigma and inconvenience of taking medication at school.

✓ Recommended best practice based on clinical experience and expert opinion

86. In some cases the combined use of immediate-release and extendedrelease forms is required. This should only be considered if there is inadequate symptom control with the extended-release form. ✓ Recommended best practice based on clinical experience and expert opinion

Atomoxetine (non-stimulant)

- 87. Preschoolers: There is no published research to show benefit from ATX in treating ADHD in preschool-aged children.
- 88. Children and adolescents: ATX should be considered for children and adolescents with severe ADHD who do not respond to or are intolerant of stimulant medication, or in whom stimulant medication is contraindicated. (Grade B)
- 89. Adults: Treatment with ATX should be considered for adults with severe ADHD who do not respond to or are intolerant of stimulant medication, or in whom stimulant medication is contraindicated. (Grade B)
- 90. ADHD and comorbidities: ATX may be considered as the first-line medication if there is comorbid substance abuse, severe tic disorder or anxiety disorder.
 - ✓ Recommended best practice based on clinical experience and expert opinion

Other medications used in ADHD Clonidine

91. Clonidine may provide some benefit in modifying ADHD symptoms in children, particularly in the home setting, used either alone or in combination with MPH. It could be trialled in the absence of clinical response to stimulants and ATX. (Grade D)

Modafinil

- 92. Modafinil is not recommended as an alternative medication in school-aged children and adolescents with ADHD due to concerns about the associated side effects in this age group. (Grade C)
- 93. Modafinil may be considered as an alternative medication in adults with ADHD who do not respond to stimulants or ATX. (Grade C)

Selegiline

94. There are insufficient research data to recommend the use of selegiline for the treatment of ADHD. (Grade D)

Guanfacine

95. Guanfacine may have a role in the pharmacological management of ADHD. (Grade C)

Note: Guanfacine is not currently available in Australia.

Nicotine patch

96. There are insufficient research data to recommend the use of nicotine patches in the treatment of ADHD in school-aged children and adolescents. (Grade D)

Bupropion

97. Bupropion may provide some benefit in treating ADHD symptoms in adults. It could be trialled in the absence of clinical response to stimulants and ATX. (Grade D)

Risperidone

98. There are insufficient research data to recommend the use of risperidone for the treatment of ADHD. (Grade D)

Medication when comorbidities are present

99. The RANZCP, RACGP and RACP Clinical Guidance and the Adverse Drug Reactions Advisory Committee (ADRAC) recommendations on the use of antidepressants in children and adolescents must be considered by the physician before prescribing antidepressants. There are no antidepressant medications with TGA-approved indications for treatment of children and adolescents with depressive symptomatology or major depressive disorder. There are significant concerns about serious adverse reactions (increased suicidality and thoughts of self-harm) in a small percentage of this patient group.

Recommended best practice based on clinical experience and expert opinion

Anxiety

100. Psychosocial management of comorbid anxiety should always be considered.

Recommended best practice based on clinical experience and expert opinion

- 101. In individuals with ADHD and anxiety, treatment with stimulant medication or ATX should still be considered to treat the ADHD symptoms. (Grade C)
- 102. If anxiety symptoms do not respond to psychosocial interventions, the use of selective serotonin reuptake inhibitors (SSRI) could be considered, either alone or in combination with a medication to treat the ADHD. Antidepressant medication should only be used in the context of comprehensive patient management and should be combined with psychological interventions. The RANZCP, RACGP and RACP Clinical Guidance and the Adverse Drug Reactions Advisory Committee (ADRAC) recommendations on the use of antidepressants in children and adolescents must be considered by the physician before prescribing antidepressants.

✓ Recommended best practice based on clinical experience and expert opinion

Other disruptive behaviour disorders (oppositional defiant disorder, conduct disorder)

- 103. Non-pharmacological management of comorbid disruptive behaviour disorders should always be considered.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 104. In children and adolescents with ADHD and comorbid ODD, MPH or ATX should be considered to treat the ADHD symptoms. (Grade B)
- 105. In children and adolescents with ADHD and comorbid ODD or CD, MPH plus clonidine could be considered to treat the ADHD symptoms. (Grade C).
- 106. In children with ADHD and comorbid CD, MPH plus risperidone could be considered to treat the ADHD symptoms. (Grade C)

Depression

107. If major depression is suspected in a person with ADHD, consultation with a psychiatrist should be considered to review diagnosis and assist in management.

✓ Recommended best practice based on clinical experience and expert opinion

- 108. Non-pharmacological management of comorbid depression should always be considered.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 109. The relative contribution of any currently prescribed stimulant medication to the onset of depression should be reviewed. Trials off stimulant medication may be appropriate.

 ✓ Recommended best practice based on clinical experience and expert opinion
- 110. Where depression is identified in an adolescent or adult with ADHD, the potential contribution of illicit drugs to the onset of the depression should be considered.

✓ Recommended best practice based on clinical experience and expert opinion

- 111. In people with ADHD and comorbid major depressive disorder, use of ATX could be considered to treat the ADHD symptoms. (Grade C)
- 112. The use of stimulant medication could be considered to treat ADHD symptoms in people with ADHD and comorbid major depressive disorder. Recommended best practice based on clinical experience and expert opinion
- 113. Antidepressant medication may be considered to treat ADHD symptoms in adults with ADHD and comorbid moderate to severe depression. Recommended best practice based on clinical experience and expert opinion
- 114. Antidepressant medication could be considered for adolescents with ADHD and comorbid moderate to severe depression. Antidepressant medication should only be used in the context of comprehensive patient management and should be combined with psychological interventions. The RANZCP, RACGP and RACP Clinical Guidance and the Adverse Drug Reactions Advisory Committee (ADRAC) recommendations on the use of antidepressants in children and adolescents must be considered by the physician before prescribing antidepressants.

 \checkmark Recommended best practice based on clinical experience and expert opinion

Bipolar disorder

115. If bipolar disorder is suspected in a child or adolescent with a diagnosis of ADHD, consultation with a child and adolescent psychiatrist should be considered to review diagnosis and assist in management.
✓ Recommended best practice based on clinical experience and expert opinion

116. In adults with ADHD and non-psychotic comorbid bipolar disorder, use of low dose stimulant medication, in addition to mood-stabilising medication, may be considered. (Grade C)

Epilepsy

- 117. In individuals with ADHD and well-controlled epilepsy, stimulant medication or ATX should be considered to treat the ADHD.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 118. The impact of ADHD medication on seizure frequency should be monitored, especially in the initial medication trial period. ✓ Recommended best practice based on clinical experience and expert opinion

Tic disorders and Tourette syndrome

- 119. In people with ADHD and a comorbid tic disorder or Tourette syndrome, stimulant medication is not necessarily clinically contraindicated and should be considered to treat ADHD symptoms. (Grade C)
- 120. In people with ADHD and comorbid tic disorder or Tourette syndrome, use of ATX should be considered to treat the ADHD. (Grade C)
- 121. In children and adolescents with ADHD and a comorbid tic disorder /Tourette syndrome, clonidine should be considered if ADHD symptoms show poor response to stimulants or ATX. (Grade C)
- 122. If a tic occurs or becomes worse after commencing an ADHD medication, a clinical decision should be made in collaboration with the person with ADHD and/or the parents/caregivers on whether to:
 - o continue the ADHD medication alone
 - o add an anti-tic medication, or
 - trial another ADHD medication.

The decision will be informed by the degree of response to the stimulant and the severity of the tics. (Grade C)

Medications in children with a developmental disability Learning disabilities

- 124. Stimulant medication or ATX should be considered to improve the ADHD symptoms.
 - Recommended best practice based on clinical experience and expert opinion

Intellectual disability

- 125. Where a person with an intellectual disability also has ADHD, management of the intellectual disability should be an integral part of overall management.
 - ✓ Recommended best practice based on clinical experience and expert opinion
- 126. In people with an intellectual disability and ADHD, use of stimulant medication should be considered. (Grade C)

Autism spectrum disorders

127. Where a person with an autism spectrum disorder also has ADHD symptoms, management of that disorder should be an integral part of overall management.

✓ Recommended best practice based on clinical experience and expert opinion

- 128. In children with an autism spectrum disorder and symptoms of ADHD, use of stimulant medication or ATX should be considered to treat ADHD symptoms. (Grade C)
- 129. Careful monitoring is required due to the possibility of exacerbating ritualistic behaviours and stereotypies. ✓ Recommended best practice based on clinical experience and expert opinion

Side effects: special considerations Growth

- 130. If a child is prescribed stimulant medication or ATX, growth parameters should be assessed at baseline and monitored every 3–6 months. (Grade B)
- 131. Monitoring should include plotting of growth parameters (weight, height, growth velocity) on appropriate charts.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 132. For children most at risk of growth attenuation (younger age, higher dose), medication at low doses only, and/or other intervention strategies, should be considered. (Grade C)
- 133. Where there are concerns about reduced growth velocity (crossing centiles or inadequate weight gain), intervention strategies might include medication "holidays" (dose reduction or cessation during weekend and vacation periods).

\checkmark Recommended best practice based on clinical experience and expert opinion

Cardiac events

- 134. Before initiating pharmacological treatment for ADHD, cardiac risk factors should be assessed, including:
 - history of congenital heart disease or arrhythmias, palpitations, exercise intolerance or chest pain
 - family history of early cardiac disease (<50 years) or unexplained sudden death
 - cardiovascular examination. (Grade B)
- 135. Routine monitoring at review appointments should include heart rate and blood pressure, and recent history of palpitations, exercise intolerance and chest pain. (Grade B)
- 136. Specialist cardiologist advice is recommended for people with cardiovascular risk factors (congenital heart disease, arrhythmias, family history) and in whom blood pressure or heart rate exceeds normal limits. (Grade B)

Psychiatric disturbance

All ADHD medications

- 137. For individuals taking ADHD medications there is a rare but serious risk of psychiatric adverse events. Before prescribing ADHD medication, the person with ADHD and/or their caregiver should be specifically told of the risk of emergent psychiatric adverse effects, including aggression, anxiety, mania and psychosis. (Grade B)
- 138. As part of ADHD medication management, the person with ADHD and/or their caregiver should be specifically asked about emergent psychiatric adverse effects. (Grade B)

139. This is especially important for people who are receiving both ADHD medications and other psychotropic medications, have significant psychiatric comorbidities and/or have a family history of psychiatric illness. (Grade B)

Atomoxetine

140. All people for whom ATX is being considered should be told of the possibility of suicidal ideation, and people on ATX should be monitored for this side effect with extra vigilance. (Grade B)

Stimulant medication

141. All people for whom stimulant medication is being considered should be told of the possibility of suicidal ideation, and people on stimulant medication should be monitored for this side effect with extra vigilance. (Grade B)

Monitoring of medication effectiveness and side effects

- 142. Individuals on medication for ADHD should be monitored by their treating doctor for medication effectiveness and side effects. This should occur frequently in the early phase of treatment, and thereafter at intervals of 3–6 months.
 - ✓ Recommended best practice based on clinical experience and expert opinion
- 143. Monitoring should include reports from caregivers, partners and/or teachers (as applicable). Brief parent and teacher behaviour rating scales can also contribute to monitoring of behaviour or performance at home or school.

✓ Recommended best practice based on clinical experience and expert opinion

- 144. Monitoring at each visit should include height, weight and growth in children/adolescents and psychiatric disturbance, heart rate and blood pressure in all age groups. (Grade B)
- 145. Trials off medication should be conducted to evaluate whether the medication is still clinically indicated.
 - ✓ Recommended best practice based on clinical experience and expert opinion

ADHD and substance misuse

Effect of ADHD management on risk of developing substance use disorders

146. The use of stimulant medication to treat people with ADHD does not increase the risk of developing substance use disorder. (Grade B)

Treatments for ADHD when comorbid substance use disorder is present

- 147. A history of substance abuse and previous treatments should be included in the evaluation of patients.
 - ✓ Recommended best practice based on clinical experience and expert opinion
- 148. If the substance use disorder is active the symptoms of the substance use disorder should be addressed prior to attending to the symptoms of ADHD. ✓ Recommended best practice based on clinical experience and expert opinion
- 149. Medication treatment for ADHD comorbid with substance use disorder should only be provided by a medical practitioner with expertise in both conditions.

✓ Recommended best practice based on clinical experience and expert opinion

- 150. ATX should be the first medication trialled if the person with ADHD has a comorbid substance use disorder.
 - ✓ Recommended best practice based on clinical experience and expert opinion

151. Relapse or worsening of substance use should necessitate reassessing the appropriateness of stimulant treatment. ✓ Recommended best practice based on clinical experience and expert opinion

Medication compared to and combined with psychosocial interventions

Comparing psychosocial and pharmacological interventions

- 152. Although the evidence suggests that ADHD symptoms improve more in the short to medium term with pharmacological management than with psychosocial interventions, it is not possible to draw conclusions about the best management of ADHD symptoms in the long term. The likely presence of modifying factors, including comorbidities and the age of the child, warrant the use of psychosocial interventions. (Grade D)
- 153. Psychosocial interventions should be considered in the long-term management of children and adolescents with uncomplicated ADHD, with or without medication.
 - ✓ Recommended best practice based on clinical experience and expert opinion
- 154. Psychosocial interventions should be considered, in addition to pharmacological management, in people with ADHD associated with ODD or CD.
 - ✓ Recommended best practice based on clinical experience and expert opinion

Combined interventions for preschool-aged children, children and adolescents

155. Although the evidence suggests that ADHD symptoms are most improved in the short to medium term with pharmacological management alone, it is not possible to draw conclusions about the best management of ADHD symptoms in the long term. The likely presence of modifying factors, including comorbidities and the age of the child, warrant the inclusion of psychosocial interventions. (Grade D)

Combined interventions for adults

156. The likely presence of comorbidities in adults with ADHD favours the inclusion of psychosocial interventions alongside pharmacological interventions. (Grade D)

Moderators and mediators of outcomes from psychosocial interventions and combined interventions

- 157. Psychosocial interventions can be particularly beneficial in a person with ADHD and comorbid anxiety. (Grade C)
- 158. Individualised treatment strategies should be developed. When variables such as parental depression, severe ADHD symptoms or below-average IQ are present, additional intervention strategies may be required. (Grade C)

Educational management

Inclusion: a legal requirement

159. All professionals supporting students with ADHD should be familiar with their legal responsibilities under the 1992 Disability Discrimination Act (DDA) and Disability Standards for Education (2005). ADHD is recognised as a disability under the DDA. As such, schools are responsible for explicit planning and review of support strategies and services for students with ADHD.

✓ Recommended best practice based on clinical experience and expert opinion

Teacher knowledge

- 160. Pre-service and in-service teacher preparation courses should be designed to prepare all teachers with the knowledge and skills to accommodate students with specific learning needs and to manage students in need of additional support for their learning, behaviour, organisation and concentration difficulties.
 - \checkmark Recommended best practice based on clinical experience and expert opinion

Resources

- 161. Resource allocations to schools should be accessible to teachers and school-based personnel for professional development in areas where established and emerging empirical scientific evidence about academic and social learning in children can inform more effective pedagogical practice. Such upgrading of skills should have an emphasis on practical school-based interventions.
 - ✓ Recommended best practice based on clinical experience and expert opinion

Tertiary/post-school years

- 162. Australian research into the impact of learning and behaviour challenges for adults pursuing tertiary qualifications is needed to identify what reasonable accommodations can be made to enhance their chances of success.
 - ✓ Recommended best practice based on clinical experience and expert opinion

Collaborative approaches

- 163. Medical and education personnel should engage in high-level collaboration (e.g. with Wraparound teams or Positive Behavioural Intervention Support teams, exchange of information, completion of surveys/questionnaires, joint meeting with the family) when a student presents with the characteristics of ADHD to effect the best possible support for both student and family.
 - ✓ Recommended best practice based on clinical experience and expert opinion

School-based interventions

- 164. Schools should have policies and procedures in place to support students experiencing learning, behaviour, organisation and concentration difficulties, for example, pre-referral processes, Wraparound and Positive Behavioural Intervention Support teams. *Recommended best practice based on clinical experience and expert opinion*
- 165. Effective school-based interventions, including peer tutoring, mentoring and peer support (e.g. buddy systems), should be considered for children and adolescents experiencing learning, behaviour, organisation and concentration difficulties to enhance their learning, social and behavioural outcomes. (Grade C)
- 166. Well-designed research into behavioural and school-based academic interventions that teachers can effectively and easily implement for the benefit of students with ADHD is needed.

Complementary and alternative treatments for ADHD

Elimination and restriction diets

- 167. Healthcare professionals should encourage good nutrition and a balanced diet.
 - ✓ Recommended best practice based on clinical experience and expert opinion

- 168. Elimination and restriction diets are not supported as a general treatment for all individuals with ADHD. Consumers considering the use of elimination or restriction diets should be informed about the uncertainties surrounding the efficacy of these diets in treating ADHD and of the potential risks of unsupervised elimination diets. (Grade D)
- 169. A subset of children may be sensitive to certain foods or food additives and may benefit from careful exclusion diets. Assessment of food sensitivity and initiation of a special diet should be under the care and supervision of a medical specialist and an Accredited Practising Dietitian.
 - \checkmark Recommended best practice based on clinical experience and expert opinion

Diet supplements

- 170. There are insufficient research data to recommend the use of diet supplementation with essential fatty acids for the treatment of ADHD. Consumers considering the use of essential fatty acids should be informed about the uncertainties surrounding their efficacy in treating ADHD. (Grade D)
- 171. The use of essential fatty acids in the management of ADHD symptoms warrants further investigation in well-designed randomised controlled trials (RCTs), including clarification of dosage levels and types of essential fatty acid used.

Chiropractic

172. Consumers considering chiropractic should be informed about the current lack of research to assess its efficacy in the treatment of ADHD. Recommended best practice based on clinical experience and expert opinion

Behavioural optometry

173. Consumers considering behavioural optometry should be informed about the current lack of research to assess its efficacy in the treatment of ADHD. ✓ Recommended best practice based on clinical experience and expert opinion

Biofeedback

- 174. There are insufficient research data to recommend the use of biofeedback for the treatment of ADHD. (Grade D)
- 175. Consumers considering biofeedback should be informed about the uncertainties surrounding its efficacy in the treatment of ADHD, the time required to undertake the treatment and its costs.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 176. The use of biofeedback in the treatment of ADHD symptoms warrants further investigation using well-designed RCTs.

Homeopathy

177. There are insufficient research data to recommend the use of homeopathy for the treatment of ADHD. Consumers considering the use of homeopathy should be informed about the uncertainties surrounding its efficacy in treating ADHD. (Grade D)

Cerebellar therapies

- 178. There are insufficient research data to recommend the use of cerebellar therapies for the treatment of ADHD. (Grade D)
- 179. Consumers considering the use of cerebellar therapy should be informed about the uncertainties surrounding its efficacy in treating ADHD, the time required to undertake the treatment and its costs. ✓ Recommended best practice based on clinical experience and expert opinion

Sport, exercise and relaxation

- 180. Yoga, massage and Tai Chi may be of some benefit in treating ADHD. (Grade D)
- 181. While other sport, exercise and relaxation programs may have many mental and physical health benefits, their role in the management of ADHD has yet to be demonstrated.

Sensory integration interventions

182. There are insufficient research data to recommend the use of sensory diets / sensory integration therapies for the treatment of ADHD. Consumers considering the use of sensory diets / sensory integration therapies should be informed about the uncertainties surrounding the efficacy of these programs in treating ADHD. (Grade D)

ADHD in the workplace

183. Career and workplace assessments are often valuable in assisting adults with ADHD in their selection of career or workplace function.
✓ Recommended best practice based on clinical experience and expert opinion

ADHD and driving

Effect of treatment on driving ability in people with ADHD

- 184. People with ADHD should be strongly advised to take their stimulant medication if they are driving. Treatment with MPH improves driving performance in people with ADHD. The use of extended-release stimulants is preferable if the person drives at night. (Grade C)
- 185. There are insufficient research data to recommend the use of ATX for the improvement of driving performance in individuals with ADHD. (Grade D)
- 186. The use of a manual transmission over an automatic transmission should be considered for individuals with ADHD. (Grade C)

Issues for families, parents and caregivers

Families with ADHD in a child or adolescent

- 187. Clinicians should be alert to the risk of depression in parents/caregivers of children or adolescents with ADHD. Parents/caregivers may need referral for support and treatment.
 - ✓ Recommended best practice based on clinical experience and expert opinion
- 188. When a child or adolescent is diagnosed with ADHD, the clinician should consider the impact of marital discord on the children over and above the effects of the ADHD, as ADHD may exacerbate underlying family tensions. Recommended best practice based on clinical experience and expert opinion
- 189. Consideration should be given to the siblings of children with ADHD. Siblings may need additional support at school if there is an expectation that they care for their sibling with ADHD or if they become the focus of peer ridicule because of the behaviour of their sibling with ADHD.

Adult ADHD and its impact on the family

- 190. Treatment planning for adults with ADHD needs to include strategies for daily living and additional support during life transitions such as changes in career or family situation.
 - Recommended best practice based on clinical experience and expert opinion

- 191. Treatment planning for children with ADHD needs to take into account whether the parent/caregiver has symptoms of ADHD, as parents/caregivers with ADHD symptoms may need additional support to implement the program with their child successfully. ✓ Recommended best practice based on clinical experience and expert opinion
- 192. When an adult is diagnosed with ADHD, the clinician should consider the impact on their partner and family as additional support may be needed. ✓ Recommended best practice based on clinical experience and expert opinion

Children and adolescents with ADHD in out-of-home care

- 193. All children and adolescents in out-of-home care should receive a medical assessment that includes a developmental and mental health assessment. ✓ Recommended best practice based on clinical experience and expert opinion
- 194. Special care needs to be taken in establishing or refuting a diagnosis of ADHD in children in out-of-home care, because of the high risks of misdiagnosis, and the equally high risks of doing harm by ceasing established therapy for children already at high risk of long-term disadvantage and disability.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 195. When a child with a diagnosis of ADHD is placed in out-of-home care, every effort should be made to maintain continuity of treatment and to ensure that foster carers are educated about the special needs of the child. Recommended best practice based on clinical experience and expert opinion
- 196. The professionals in Australia's welfare system who are responsible for the care of children and adolescents should be educated about ADHD. ✓ Recommended best practice based on clinical experience and expert opinion
- 197. Research on the management and care of children and adolescents with ADHD in out-of-home care is needed.

ADHD and the justice system

Prevalence of ADHD and comorbidities among offenders

Impact of treatment for ADHD on rates of offending and recidivism

199. More research is needed to determine whether treatment of ADHD can reduce the risk of crime and recidivism.

Particular vulnerabilities of people with ADHD in the justice system

- 200. Officers of the legal and justice systems should be made aware of the potential vulnerabilities and needs of people with ADHD.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 201. Proper assessment and diagnosis of ADHD is important for identifying symptoms that may make an individual vulnerable to standard police and court procedures. Psychoeducational and neuropsychological assessment may be useful in this situation to assess individual vulnerability in comparison to population standards.
 - ✓ Recommended best practice based on clinical experience and expert opinion
- 202. Special provisions may be appropriate for some individuals with ADHD (e.g. regular breaks, repetition of important information).
 - ✓ Recommended best practice based on clinical experience and expert opinion

203. For individuals with ADHD within the prison system, management of ADHD symptoms should be considered in order to improve outcomes in prison, including substance use, antisocial behaviour and critical incidents. Recommended best practice based on clinical experience and expert opinion

INTRODUCTION

Aim of the Guidelines

The aim of these Guidelines is to support and inform the care of individuals with ADHD by providing a series of recommendations to guide assessment, management and care. It is anticipated that the recommendations will provide a guide to best practice that can be followed subject to the judgement of the professional involved and the preference of the person with ADHD and/or his or her parents or guardians.

Who are the Guidelines for?

The Guidelines are written for the range of professionals involved with ADHD, including teachers, psychologists, paediatricians, psychiatrists, allied health professionals and general practitioners. The Guidelines are also intended to inform researchers, policymakers, carers, consumers and interested community members. Accordingly, the expressed views and concerns of consumers and carers were considered throughout the development of the Guidelines.

What do the Guidelines address?

The development of these Guidelines acknowledges the burden of ADHD for individuals and their families and is indicative of a concerted effort by professional and governmental bodies to enhance, provide and sustain equitable clinical care across the country for patients and their families.

ADHD is a condition that affects people across the lifespan. The Guidelines address the specific needs of:

- preschool-aged children (3–5 years)
- children (6–12 years)
- adolescents (13–17 years)
- adults (18 years and above).

The guidelines are not applicable to children under the age of 3 years.

The recommendations are relevant to the diversity of Australian communities. Consideration has been given to rural and remote settings, and community groups such as Aboriginal and Torres Strait Islanders and people from culturally and linguistically diverse (CALD) backgrounds. It is anticipated that the Guidelines will enable a consistent response to ADHD across all Australian States and Territories and will facilitate a co-ordinated approach across sectors, including Health, Education, Justice and Community Services.

Several groups are poorly represented in the research literature on ADHD. For example, ADHD in girls and women has only recently garnered attention and there is a paucity of research on ADHD in individuals over 50 years of age. Research on ADHD in many specific communities, including Aboriginal and Torres Strait Islander peoples, is also rare. Consequently, guidelines and recommendations in relation to these groups can generally be made only on the basis of expert consensus.

Terms of Reference

To revise the 1997 NHMRC *Guidelines on Attention Deficit Hyperactivity Disorder (ADHD)*, incorporating literature published since, to inform health professionals, educators, researchers, policymakers, carers, consumers and interested community members.

Specifically, the revision of the 1997 NHMRC Guidelines on ADHD should:

- 1. Define and describe the current conceptualisation of ADHD.
- 2. Address aspects of ADHD that have generated community and professional interest over the past decade, including:
 - o neurobiology
 - aetiology genetics, environmental factors
 - o comorbid conditions
 - symptom patterns and management at different developmental stages, including preschool-aged children, school-aged children, adolescents and adults
 - issues specific to particular populations, e.g. Indigenous Australians and people from culturally and linguistically diverse (CALD) backgrounds
 - role of newer medications, e.g. long-acting stimulant preparations, non-stimulants
 - long-term outcomes
 - \circ $\;$ quality of life for the individual with ADHD and their family
 - o complementary and alternative therapies
 - o diagnostic practices (differential diagnosis)
 - access to services.
- 3. Synthesise and interpret the current state of knowledge on ADHD for different stakeholders.
- 4. Update the list of available resources to assist clinicians, educators, carers and consumers in the evaluation and management of ADHD.
- 5. Advise on national data collection in the management of ADHD.
- 6. Advise on targets for further research into the causes, management and outcomes of ADHD.
- 7. Provide recommendations for policy and practice for different stakeholders.

Guideline development process

The revision of the NHMRC Guidelines on ADHD has been undertaken in accordance with NHMRC guideline development processes, and following the NHMRC handbooks on guideline development (2-4) and the NHMRC standards and procedures for externally developed guidelines (5).

The Guidelines were prepared by a project officer and scientific writer working closely with a multidisciplinary expert reference group. The reference group brought expertise from the key professional disciplines involved with ADHD, including paediatrics, child and adolescent psychiatry, adult psychiatry, psychology, general practice, education and consumers/carers. The project has been overseen by an NHMRC-appointed Guidelines Assessment Register Consultant, and a project management team has directed the administrative aspects of guideline development. Consultation with other individuals and organisations has been included in the development process in line with NHMRC standards.

Details of the membership of the guideline development groups are provided in Appendix A. The dualities and conflict of interest statements of all individuals involved in the development of the Guidelines are provided in Appendix B. The terms of reference for each guideline development group and a detailed report on the development process is provided in Appendix C. **The next revision:** It is anticipated that these Guidelines will have a lifespan of 5 years and will need to be revised in 2014.

Literature review

The content of the Guidelines is based largely on the results of a systematic review of the literature. The systematic review is presented as a separate document and is available to download from the RACP website. The review presents each research question and a series of summary tables describing the identified studies. Each summary table provides the lead author, a brief description of the participants in the study, the study design, a brief description of the intervention or diagnostic criteria, the outcome measures used and a brief conclusion. The level of evidence for each study has been designated according to the NHMRC levels of evidence (Table 1). Within the text of the Guidelines, unless otherwise specified the level of evidence refers to intervention studies. For studies with levels of evidence based on diagnosis, prognosis, aetiology or screening intervention, this is noted.

The methods used to conduct the review comply with NHMRC requirements (3-5) and are described in Appendix D. The research questions addressed by systematic review are listed in Appendix E. For each included article, data were extracted into a standardised data-extraction / critical appraisal table (Appendix F).

The guideline recommendations have been developed using the procedure outlined` in the *NHMRC additional levels of evidence and grades for recommendations for developers of guidelines: Stage 2 consultation 2008–2010* (1). A standardised form has been used to formulate and grade the recommendations: the NHMRC Evidence Statement Form (Appendix G). When Level I, II, III or IV evidence was available from the systematic review the recommendation was graded as A, B, C or D. In areas where no Level I, II, III or IV evidence was available, best practice points that are based on the clinical experience and expert opinion of the reference group of clinicians, educators and consumers have been provided. Research recommendations have not been graded. Table 2 sets out the evidence gradings.

In this guideline document, when a recommendation is based on evidence derived from the systematic review, the relevant research question(s), summary evidence statement(s) with level of evidence, and resultant recommendation(s) are provided in a box at the start of the appropriate section, followed by an outline of the research evidence.

			- ·		- ·
Level	Intervention	Diagnosis	Prognosis	Aetiology	Screening
I	Systematic	Systematic review	Systematic	Systematic	Systematic
	review of level	of level II studies	review of level	review of level	review of
	II studies		II studies	II studies	level II
					studies
II	Randomised	Study of test	A prospective	All or none	Randomised
	controlled trial	accuracy with an	cohort study		controlled
		independent,			trial
		blinded			
		volid reference			
		standard among			
		consecutive			
		patients with a			
		defined clinical			
		presentation			
III-1	Pseudo-	Study of test	All or none	All or none	Pseudo-
	randomised	accuracy with an	A V		randomised
	controlled trial	independent,			controlled
	(I.e. alternate	blinded			trial (I.e.
	allocation or	comparison with a			alternate
	method)	standard among			some other
	meenou	non-consecutive			method)
		patients with a			meenou)
		defined clinical			
		presentation			
III-2	Comparative	Comparison with	Analysis of	Retrospective	Comparative
	study with	reference	prognostic	cohort study	study with
	concurrent	standard that	factors	w.	concurrent
	controls:	does not meet the	amongst		controls:
	•Non-random,	for Level II and	control		•Noll- random
	trial	III-1 evidence	natients in a		experimental
	•Cohort study	III I Chidenee	randomised		trial
	•Case-control		controlled trial		•Cohort
	study				study
	 Interrupted 				•Case-
	time series with				control study
	a control group	A			
III-3	Comparative	Diagnostic case-	Retrospective	Case control	Comparative
	study without	control study	conort study	study	study without
	controls:				controls:
	•Historical				•Historical
	control study				control study
	•Two or more				•Two or
	single arm				more single
	studies				arm studies
	 Interrupted 				
	time series				
	without a				
IV	Case series	Study of	Case series	Cross-	Case series
1.4	with either	diagnostic vield	or cohort	sectional	
	post-test or	(no reference	study of	study	
	pre-test/post-	standard)	patients at	,	
	test outcomes		different		
			stages of		
			disease		

Table 1. NHMRC levels of evidence according to type of research question (1)

Table 2.	Grading	of recommendations	(1)
----------	---------	--------------------	-----

Grade	Description
Α	Body of evidence can be trusted to guide practice
В	Body of evidence can be trusted to guide practice in most situations
С	Body of evidence provides some support for recommendation(s) but care should be
	taken in its application
D	Body of evidence is weak and recommendation must be applied with caution
Best pr	actice points
\checkmark	Recommended best practice based on clinical experience and expert opinion, where
	there is inadequate evidence to establish evidence based recommendations.

PART I: WHAT IS ADHD? CONCEPTUALISATION, CAUSES, PREVALENCE AND CONSEQUENCES



CHAPTER 1. INTRODUCTION: DEFINITION, HISTORY AND CURRENT DIAGNOSTIC CRITERIA

1.1 Definition

Attention deficit hyperactivity disorder (ADHD) is defined as a persistent pattern of inattention and/or hyperactive and impulsive behaviour that is more frequent and severe than is typically seen at a given stage of development. Symptoms are usually present from early childhood, they tend to become particularly problematic when the child starts school and they may remain troublesome across the lifespan. The symptoms of ADHD are associated with impairment in educational, social and emotional function.

Over the past 30 years there has been extensive scientific research published that examines ADHD causes, associations, outcomes and treatments. ADHD continues to be a subject of great interest in the community, with frequent discussion in the media. Diverse and strong opinions are often expressed, particularly about causes and treatments.

There are essentially two clinical subtypes of ADHD: combined subtype, which includes both inattention symptoms and impulsive/hyperactive symptoms, and predominantly inattentive subtype, in which the impulsive/hyperactive symptoms are minimal. A third subtype, predominantly hyperactive/impulsive, is described, but is rarely diagnosed.

ADHD occurs in all ethnic groups and social classes and is the most common neurobehavioural disorder of childhood (6). The childhood community prevalence of 6.8% found in Australia (7) is similar to that reported in other countries (8-10).

Many individuals with ADHD also have related problems in areas including language, learning, mood, emotional regulation and motor control (11-15) (see section 3.2 Comorbidities, page 22). Academic and social struggles can lead children with ADHD to feel demoralised and depressed, or angry and oppositional. They are at increased risk of a range of adverse outcomes including academic underachievement, difficulties with interpersonal relationships and low self-esteem, with potentially serious consequences for the individual and society (16, 17) (see Chapter 4. Consequences of ADHD, page 25).

1.2 A historical perspective

The earliest account of inattention, recognition of its potential for causing disability and speculation on its nature date from 1798 when Alexander Crichton published "On Attention and its Diseases" as part of a treatise entitled "An enquiry into the nature and origin of mental derangement" (18) Children with problematic hyperactivity, impulsivity and inattention were described in more detail by George Still in the Coulstonian Lectures of 1902 (19).

Other features such as defiance, aggression and emotionality have also often been observed in these children. Professional understanding of these behaviours has evolved over time and diagnostic labelling has followed, with a range of descriptions including defective moral control (19), minimal brain damage, post-encephalitic behaviour disorder (20), brain injured child (21) and hyperkinetic reaction of childhood (22). In the 1970s, the developing understanding of the fundamental neurocognitive problem saw a shift in focus from excessive motor activity to deficits in sustained attention and impulse control (23), and the condition was renamed attention deficit disorder in 1980 (24).

Psychostimulant medication to assist children with behavioural disturbance was first described in 1937 (25). Improved behaviour and school performance was observed
as an unexpected effect of using the amphetamine benzedrine to reduce headaches following lumbar punctures, undertaken to perform investigative pneumoencephalograms in hyperactive children. Stimulant medications began to be used for treating ADHD symptoms in children in the 1970s in North America, and in the 1980s in Australia.

1.3 Current conceptualisation and diagnostic criteria

ADHD is a neurobiological disorder with genetic and environmental contributions. The dominant current paradigm suggests that disordered fronto-striato-cerebellar brain circuitry underpins the executive function deficits at the core of this condition (26). Twin studies have established a strong genetic component (27, 28). This appears to involve polymorphisms in a number of genes, including those coding for dopamine transporters.

Like a number of other psychological characteristics, the features of ADHD vary throughout the entire population. These characteristics are recognisable on a population continuum from minimal signs to dominant characteristics. Individuals in whom a diagnosis of ADHD is made have more marked or more intense features, and represent one end of the continuum (29).

Environmental risk factors for ADHD include antenatal exposures such as tobacco and alcohol, adverse early childhood experiences (including social, family and cultural factors), head trauma, and environmental toxins such as lead (see section 2.1 Aetiology, page 14). The relative contribution of environmental risk factors is the subject of a significant body of research (30).

Children with ADHD are not a homogenous group. While they share the characteristics of poor self-regulation, planning, execution and monitoring of their behaviour (31), they vary widely in the way these characteristics are manifested, and the overall effect on the child is determined by his or her underlying personality, individual strengths and other associated developmental difficulties.

The symptoms of ADHD are not specific to ADHD and a range of differential diagnoses and co-existing conditions need to be considered. These include learning difficulties, sleep deprivation, hearing impairment, attachment deficits and affective disorders (e.g. anxiety and depression). Family conflict, bullying and child abuse can also present with ADHD-like symptoms, and may be alternative explanations for the symptoms, or co-occurring problems in a person who also has ADHD.

The two diagnostic systems most commonly used in both clinical practice and research are the *Diagnostic and Statistical Manual of Mental Disorders* (DSM), published by the American Psychiatric Association, and the *International Statistical Classification of Diseases* (ICD), published by the World Health Organization. The DSM system is used in North America, while the ICD system has been used more in Europe. In Australia the DSM-IV system is generally used.

DSM-IV diagnostic criteria for ADHD (1994) (32) are outlined in Table 3. While earlier editions of the DSM relied on the consensus of experts, based on clinical descriptions to define diagnostic categories, the DSM-IV committee carefully applied psychometric measurement theory (33). Based on extensive literature review, large multi-site field trials were conducted to determine and test the validity and reliability of diagnostic criteria. Factor analysis was used to test the consistency of aggregation of behavioural symptoms and their dimensional fidelity, and threshold cut-off points (required number of symptoms) were determined based on ability to predict impairment (overall, academic and social) and further tested for test-retest and inter-clinician reliability.

This methodical analysis of empirical scientific data resulted in a substantially improved set of diagnostic criteria. A number of issues related to the DSM-IV criteria for ADHD, however, remain unresolved. These include:

- definition of impairment a problem common to many psychiatric diagnoses
- refinement of the term "often" in terms of frequency and intensity
- appropriateness of items for different developmental stages. The DSM symptom set was based on research in children. Some items are not relevant for adults, for example, "leaves seat in classroom", "runs about or climbs excessively"
- need for variation in symptom thresholds for different developmental stages, for example, preschool, school age, adolescent, adult
- the age of onset criterion. Given the lack of evidence that the life course of people with ADHD varies according to symptom onset before or after age 7, it has been suggested that the age-of-onset criterion be broadened to include onset of symptoms during childhood (34)
- Predominantly Inattentive Type people with this type of ADHD present as qualitatively different from those with ADHD combined subtype, have a different pattern of comorbidities and respond less well to stimulant medication. It has been suggested that at least some people in this group may be better described as having a condition distinct from ADHD (35).

The Externalising Disorders Research Group has begun work to develop DSM-V, (scheduled to be published in 2012) and is addressing these issues.

ICD-10 (1992) criteria for hyperkinetic disorder (36), outlined in Table 4, describe similar symptoms to DSM-IV and, like DSM-IV, require early onset of symptoms that are "excessive for the child's age and IQ", symptom presence across situations, the persistence of symptoms over time and the exclusion of certain other psychiatric disorders. However, the ICD-10 system requires, in addition to at least 6 inattention symptoms (from a list of 8), the presence of at least 3 (of 5) hyperactive symptoms and at least 1 (of 4) impulsive symptoms. It also excludes people with autism or other pervasive developmental disorders, anxiety or depression. The application of these more restrictive criteria would clearly result in fewer patients with this diagnosis than with DSM-IV ADHD (37), with people meeting ICD-10 criteria being more severely impaired (38).

Neuroscience research has been investigating genetic polymorphisms, biological markers and neuropsychological endophenotypes in ADHD (39, 40). While this work has greatly enhanced understanding of the biology of ADHD, the prospect of shifting from the present behavioural approach to a laboratory-based diagnostic method remains remote.

Table 3. DSM-IV diagnostic criteria for ADHD (32)

1. Either A or	В	
A. Inattentio	n – 6 or more symptoms persisting for at least 6 months to a degree that is	
maladaptive	and inconsistent with developmental level	
_	Often fails to give close attention to details or makes careless mistakes in schoolwork,	
	work or other activities	
	Often has difficulty sustaining attention in tasks or play activities	
	Often does not seem to listen when spoken to directly	
	Often does not follow through on instructions; fails to finish schoolwork, chores or	
	workplace duties (not due to oppositional behaviour or failure to understand	
	instructions)	
	Often has difficulty organising tasks and activities	
	Often avoids, dislikes or is reluctant to do tasks requiring sustained mental effort	
	Often loses things necessary for tasks or activities	
	Is often easily distracted by extraneous stimuli	
	Is often forgetful in daily activities	
B. Hyperactiv	ity-impulsivity – 6 or more symptoms persisting for at least 6 months to a	
degree that is	s maladaptive and inconsistent with developmental level	
Hyperactivity	Often fidgets with hands or feet or squirms in seat	
	Often leaves seat in classroom or in other situations where remaining seated is expected	
	Often runs or climbs excessively where inappropriate (feelings of restlessness in young	
	people or adults)	
	Often has difficulty playing or engaging in leisure activities quietly	
	Is often "on the go" or often acts as if "driven by a motor"	
	Often talks excessively	
Impulsivity	Often blurts out answers before questions have been completed	
	Difficulty awaiting turn	
	Interrupts or intrudes on others (e.g. butts into conversations or games)	
2. Some hype	eractive-impulsive or inattentive symptoms that caused impairment were present	
before age 7	years.	
3. Some impa	irment from symptoms is present in 2 or more settings (e.g. at school or work	
and at home)		
4. There mus	t be clear evidence of significant impairment in social, school or work	
functioning.		
5. The sympt	oms do not happen only during the course of a pervasive developmental	
disorder, sch	izophrenia or other psychotic disorder. The symptoms are not better accounted	
for by anothe	r mental disorder (e.g. mood disorder, anxiety disorder, dissociative disorder or	
a personality	disorder).	
Based on these criteria, 3 types of ADHD are identified:		
ADHD Combined Type: if both criteria 1A and 1B are met for the past 6 months		
ADHD. Predominantly Inattentive Type: if criterion 1A is met but criterion 1B is not met for the		
past 6 months.		
ADHD, Predominantly Hyperactive-Impulsive Type: if Criterion 1B is met but Criterion 1A is not		
met for the past 6 months.		

Table 4. ICD-10 diagnostic criteria for hyperkinetic disorders (36)

G1 Inatten	tion	
At least 6 of is maladapt	the following symptoms of inattention have persisted for at least 6 months, to a degree that ive and inconsistent with the developmental level of the child:	
0	often fails to give close attention to details, or makes careless errors in school work, work or other activities	
0	often fails to sustain attention in tasks or play activities	
0	often appears not to listen to what is being said to him or her	
0	often fails to follow through on instructions or to finish school work, chores or duties in the workplace (not because of oppositional behaviour or failure to understand instructions)	
0	is often impaired in organising tasks and activities	
0	often avoids or strongly dislikes tasks, such as homework, that require sustained mental effort	
0	often loses things necessary for certain tasks or activities, such as school assignments, pencils, books, toys or tools	
0	is often easily distracted by external stimuli	
0	is often forgetful in the course of daily activities.	
G2 Hypera	ctivity	
At least 3 of that is mala	the following symptoms of hyperactivity have persisted for at least 6 months, to a degree daptive and inconsistent with the developmental level of the child:	
0	often fidgets with hands or feet or squirms on seat	
0	leaves seat in classroom or in other situations in which remaining seated is expected	
0	often runs about or climbs excessively in situations in which it is inappropriate (in adolescents or adults, only feelings of restlessness may be present)	
0	is often unduly noisy in playing or has difficulty in engaging quietly in leisure activities	
0	exhibits a persistent pattern of excessive motor activity that is not substantially modified by social context or demands.	
G3 Impulsi	ivity	
At least 1 of is maladapt	the following symptoms of impulsivity has persisted for at least 6 months, to a degree that ive and inconsistent with the developmental level of the child:	
0	often blurts out answers before questions have been completed	
0	often fails to wait in lines or await turns in games or group situations	
0	often interrupts or intrudes on others (e.g. butts into others' conversations or games)	
0	often talks excessively without appropriate response to social constraints.	
G4 Age of	onset	
Onset of the	e disorder is no later than the age of 7 years.	
G5 Pervasi	veness	
The criteria hyperactivit children are information unlikely to b	should be met for more than a single situation, e.g. the combination of inattention and y should be present both at home and at school, or at both school and another setting where observed, such as a clinic (evidence for cross-situationality will ordinarily require from more than one source; parental reports about classroom behaviour, for instance, are be sufficient).	
G6 Impairment		
The symptoms in GI & G3 cause clinically significant impairment in social, academic or occupational functioning.		
G7 Exclusion criteria		
The dicordo	r dess net meet the suitoris for new raise developmental disorders (FOA), menis enisede	

The disorder does not meet the criteria for pervasive developmental disorders (F84,-), manic episode (F30,-), depressive episode (F32,-) or anxiety disorders (F41,-).

ADHD Subtypes

Three major ADHD subtypes have been described:the combined subtype (inattention symptoms and impulsive/hyperactive symptoms), the predominantly inattentive subtype and the predominantly hyperactive/impulsive subtype. The first formal description of ADHD subtypes was in DSM-III, where the distinction of ADHD with and without hyperactivity was made (24). This distinction between ADD and ADHD which has been confusing for both the public and clinicians has subsequently been removed. In DSM-III-R ADHD was described as a unitary disorder with the requirement that 8 out 14 inattentive and/or hyperactive/impulsive symptoms be present for the diagnosis of ADHD. In DSM-IV three subtypes were described; combined, predominantly inattentive and predominantly hyperactive/impulsive (32). It is generally held that the subtyping of ADHD is clinically useful as prior to the introduction of subtypes clinicians were in the position of giving individuals the same diagnosis despite the very different presenting symptoms (41).

Concurrently ICD-10 developed a different and more stringent approach to the diagnosis of ADHD with the requirement that there be symptoms of inattention, hyperactivity and impulsivity (36). In effect this diagnosis is equivalent to a severe form of the DSM-IV combined type (42). While ICD-10 does not provide subtypes based on ADHD symptomatology, it does provide classifications based on comorbidity and recognises a subtype of ADHD that includes significant conduct disorder (CD).

Although the classification is likely to be modified again in DSM-V which is scheduled for publication in 2012 (see http://www.psych.org/MainMenu/Research/DSMIV/DSMV/DSMRevisionActivities/DSMVTaskForceReport.aspx) these Guidelines will follow the DSM-IV classification.

Four lines of evidence address the heterogeneity of ADHD fundamental to the DSM-IV description of ADHD subtypes; 1. symtomology, 2. patterns of comorbidity, 3. genetics and 4. neuroscience. The current status of this research in each of these areas is discussed.

1. Symptomatology

In contrast to historical attempts to statistically unravel patterns of association among symptoms of ADHD, the DSM approach has focussed on utilising clinical acumen to determine likely symptoms and then testing them in field trials to make the final decisions regarding syndrome definition (43). Studies, including from Australia (44), have used factor analysis to identify two distinct dimensions in the inattention and hyperactivity/impulsivity subtypes with coherence among the items. That is, the items are measuring each particular dimension and have little overlap.

Factor analysis however does not adequately allow for the distinction of the combined subtype and for this reason Latent Class Analysis (LCA), which allows the combined type to be distinguished from the other two subtypes has been utilised. This approach has been promoted for ADHD by Rassmussen et al (45) using twin data from Missouri and the Australian Twin ADHD Project-ATAP (46). These data came from twins recruited and assessed in very different ways and as such provide support for the current subtypes as the classifications were consistent across the two samples and followed DSM-IV fairly closely.

2. Patterns of comorbidity

One clear reason for the subtyping in DSM-IV is the clinical heterogeneity observed in individuals with ADHD. There are now data on a wide range of comorbid conditions, especially from population (47) rather than clinical samples which is important in avoiding biases caused by multiple reasons for seeking clinical help. The population-based study of Levy et al (47) included internalising as well as externalising comorbidities and separation anxiety to be commonly comorbid in girls with the inattentive subtype and generalised anxiety commonly comorbid in girls with the combined subtype (47).

Although the data are much less extensive, subtype differences in psychopathology have also been found in adults. For example, the self-report measure, the Brief Symptom Inventory (BSI), showed surprisingly large differences between the ADHD subtypes on most of the BSI subscales (48).

3. Genetics

ADHD has a substantial genetic component (see Section 2.1.2 Genetics). Consequently, if the subtypes of ADHD are defensible, there should be evidence that they "breed true", that is symptom subtypes should occur in multiple members of the same family. Stawicki et al (49) in a meta-analysis show a modest association of symptom sub-type and family aggregation. The expected pattern of associations is stronger, however, in studies utilising data from same-sex twins (46, 50) which allows much more statistical power for the analysis as data from monozygotic (MZ) and dizygotic (DZ) twins can be combined.

Genetic data also allows us also to address concerns regarding the arbitrariness of the six-item cut off for the two dimensions of inattention and hyperactivity/impulsivity and demonstrate that the distinction between the subtypes is not an artefact of the cut-off score. Concern is raised that individuals miss out on a combined type diagnosis because they have five rather than six symptoms on the other dimension. Martin et al (46, 50) examined the distribution of hyperactive/impulsive symptoms in twins who met the criteria for the inattentive subtype (and vice versa) and their MZ or DZ co-twin. For MZ twins, the data are very clear. If you have the inattentive subtype, you (and your cotwin) are as likely to have 0, 1, 2, 3, 4 or 5 hyperactivity/impulsivity symptoms. The results for those with the hyperactive/impulsive subtype are similar but not as clear cut which may reflect support for Barkley's (31) subtype model which argues that those with hyperactive/impulsive symptoms "grow" into the combined subtype.

4. Neuroscience

There has been a major research emphasis on identifying differences between ADHD subtypes based on brain structure and functioning and this research has included electrophysiological and neuropsychological measures (51). These studies are, however, limited in their clinical validity (52). While there are differences between groups when comparing those with ADHD and a control group on most measures, many of the individuals with ADHD will fall within the normal range which means the measures have only modest validity at a clinical level. Other difficulties in using electrophysiological and neuropsychological measures arise due to differences between subtypes in IQ score and in patterns of relevant comorbidities such as learning problems which confound arguments for or against the existence of subtypes. Overall, while this is a vital area of research, it remains of limited use in terms of clarifying the debate over subtyping.

CHAPTER 2. AETIOLOGY AND NEUROBIOLOGY

2.1 Aetiology

2.1.1 Introduction

The aetiology of ADHD is complex and includes both genetic and environmental components. Current understanding is that ADHD is a heterogeneous neurobiological disorder caused by the interaction of many different types of risk factors. The condition is highly heritable, and environmental risk factors include prenatal exposures such as tobacco, alcohol and lead; perinatal issues such as low birth weight; and postnatal factors such as adverse early childhood experiences, childhood illness and brain injury.

2.1.2 Genetics

Family, adoption and twin studies

Family, adoption and twin studies have demonstrated the importance of genetic factors in the aetiology of ADHD (27). Family studies show that the occurrence of ADHD is elevated amongst the relatives of people with ADHD, with first-degree relatives (parents/siblings/children) having a 4–5-fold increase in risk of ADHD compared to the general population (for review see (53)). Adoption studies have consistently shown that the biological families of people with ADHD are at greater risk of developing the disorder than the adoptive families (54-56).

Twin studies can be utilised to determine general estimates of heritability; that is, the proportion of phenotypic variance that can be attributed to genetic effects. Based on data from 20 twin studies from a number of countries, the heritability of ADHD has been estimated to be 76% (57).

While the heritability of ADHD is high, environmental factors are critical in determining whether the disorder will develop in a particular individual and how it will present. Within a family, not everyone with the susceptibility genes will develop ADHD, and among those family members who do develop the disorder, the clinical presentation may differ widely (58).

Molecular genetic studies

ADHD is a polygenic disorder in which multiple genes contribute to susceptibility. No single gene of large effect has been identified (59); that is, no single gene has been found to be sufficient or necessary to cause ADHD. Rather, the development of ADHD requires a combination of genes of moderate effect interacting with environmental factors.

Molecular genetic studies aimed at identifying susceptibility genes for ADHD generally take one of two approaches: genome-wide linkage analysis or direct targeting of candidate genes. Genome-wide linkage analyses compare the genomes of individuals with and without ADHD to identify chromosomal regions that may include susceptibility genes. Genome-wide scans have been rare and, although they have identified areas of the genome that may be of further interest, no major susceptibility genes have been identified by this approach (59-62).

Candidate gene studies have focused on genes from three neurotransmitter pathways that are thought to be involved in the aetiology of ADHD: the dopaminergic, serotonergic and noradrenergic neurotransmission pathways. There have been numerous candidate gene studies, findings have been independently replicated and meta-analyses have identified significant levels of association for several susceptibility gene variants (for review see (53, 63, 64)). These gene variants include the dopamine receptors D4 (DRD4) and D5 (DRD5), the dopamine transporter (DAT1), the serotonin transporter gene and the synaptosomal-associated protein 25 gene (SNAP-25). Weak association has been found for the dopamine transporter gene DAT1/SLC6A3 and no association has been found with the catechol-O-methyltransferase gene (COMT) (63).

Molecular genetic research is also generating information on how susceptibility genes may interact with the environment. For example, in the Mannheim Study of Children at Risk (65), children with a particular genetic variation (the VNTR polymorphism of DAT1) who had experienced psychosocial adversity were more likely to display both inattentive and hyperactive/impulsive symptoms. Ultimately, these studies not only highlight the complex relationship between genes and environment, but also suggest why susceptibility genes for ADHD are not fully penetrant, as environment is such a critical component in disease development.

The interaction between genetic factors and medications is commonly referred to as pharmacogenetics. Ultimately, the goal of psychiatric pharmacogenetics is to use the patient's genotype to predict both the response to treatment and the development of side effects. As the stimulant medications used to treat ADHD affect the dopaminergic system, the possibility of fruitful results from pharmacogenetic studies seems high. This research is, however, at the earliest of stages and no clear relationship between medication and genotype has been identified for ADHD (for review see (64)).

Links between ADHD and genetic syndromes

Genetic syndromes associated with ADHD include Angelman, Prader-Willi, Smith Magenis, Williams, Turner, Klinefelter, velocardiofacial (22q.11.2 deletion) and fragile X. These syndromes are rare in routine presentations of ADHD, and chromosome analysis or DNA testing for them is warranted only when other signs and symptoms of the syndromes are present (66).

2.1.3 Environment

Prenatal exposure to nicotine and alcohol

Prenatal exposures to nicotine and alcohol are associated with an increased risk of developing ADHD in childhood. Although numerous studies have been conducted, early research was hampered by potential confounding factors and other methodological limitations. The many possible confounding risk factors include psychopathology or ADHD in the parent and psychosocial adversity, and these confounding factors must be carefully controlled for in the study design. Smoking and alcohol consumption could be a mother's way of coping with her own ADHD symptoms or her partner's or other family members' ADHD symptoms, and may indicate a high genetic risk of ADHD in her child. Similarly, both smoking and alcohol use disorders have a large genetic component and there may be an overlap in susceptibility genes between these disorders and ADHD.

Linnet et al (67) considered 24 studies investigating the possible association of maternal smoking during pregnancy and the development of ADHD or ADHD-like symptoms in the child. The majority of studies reported an association between prenatal exposure to nicotine and inattention and hyperactivity; however, 5 found no statistically significant association (67). More recently, 2 well-designed population-based studies that controlled for maternal ADHD and used full diagnostic criteria for ADHD found a significant association between prenatal smoking and an increased risk of ADHD in the offspring (68, 69). In addition, data from twin studies show a significant association between maternal smoking and ADHD in the offspring, in addition to the influence of genetic factors (70).

Several studies have identified an interplay between specific genotypes and maternal smoking in determining the risk of ADHD. Children who were homozygous for a DAT1 polymorphism and were also exposed to prenatal smoking were found to be at significantly increased risk of hyperactivity, impulsivity and oppositional symptoms, compared to children with only one of these factors (71). Similarly, polymorphisms in the DAT1 or DRD4 genes were associated with altered risks of ADHD in individuals also exposed to prenatal smoking (72). In twins exposed to prenatal smoking, the risk of developing ADHD was 2.9-fold greater when the DAT1 polymorphism was present and 2.8-fold greater with the DRD4 polymorphism, compared to unexposed twins without the risk allele (72).

The mechanism underlying any association between maternal smoking and an increased risk of ADHD in the offspring is thought to be driven by the impact of nicotine on the dopaminergic pathways in the brain (67). Animal studies suggest that prenatal nicotine exposure influences brain development and elicits neurobehavioural outcomes such as hyperactivity that are similar to those seen in ADHD (73).

Prenatal nicotine exposure can also be associated with premature birth and low birth weight (74) which, as discussed below, are also linked with the development of ADHD or ADHD-like symptoms in the child.

Alcohol readily crosses the placenta and its teratogenic effects are established. Prenatal exposure to high levels of alcohol can result in the child developing one of a spectrum of disorders termed foetal alcohol syndrome disorders (FASD). FASD are characterised by facial abnormalities and intellectual disability, and can be associated with ADHD-like behaviours (75). The relationship between FASD and ADHD has contributed to the suggestion of a causal role for prenatal alcohol consumption and an increased risk for ADHD in the child.

Studies considering maternal alcohol consumption during pregnancy and ADHD or ADHD-like symptoms in the child have yielded inconsistent results. Linnet et al (67) reviewed nine studies, of which five reported no significant association between alcohol consumption and increased risk of ADHD, and of the studies that did find a link, 2 were based on very high levels of alcohol consumption (67). More recent studies have found a significant association between heavy prenatal alcohol exposure and an increased possibility of ADHD in the child (76, 77).

Using a children-of-twins design, an Australian study has found a significant genetic correlation between maternal alcohol use disorder (AUD) and ADHD in the offspring (78). In this study, the children of twins with a history of AUD and the children of twins without AUD, but whose co-twin had AUD, were significantly more likely to exhibit ADHD than the children of individuals without AUD. AUD was also associated with an increased probability of maternal smoking during pregnancy, but this did not alter the significant contribution of the genetics. In this study, and in a previous twin study that found ADHD was more likely to be diagnosed in adolescent girls with a parent with AUD, the genetic and environmental risks appear to be additive (78, 79). Consequently, children at high genetic risk for ADHD may also be at higher environmental risk for ADHD (78).

Prenatal exposure to maternal stress

A recent systematic review that included four studies addressing ADHD symptoms found a robust association between prenatal maternal stress and aspects of child development including ADHD symptoms (80). These effects were independent of postnatal depression and anxiety. Maternal stress may influence the in utero environment through changes in the level of hormones. Animal and human studies suggest a link between the stress-responsive hypothalamic-pituitary-adrenal axis and its hormonal end-product cortisol, which can affect the serotonergic system and interfere with neurone development (80).

Exposure to toxins

An increased risk of ADHD or ADHD-like behaviours is associated with prenatal exposure to toxins such as lead, mercury, manganese and polychlorinated biphenyls (PCBs, a group of compound used in brake liners, electrical insulation and paints); for review see (81). Recently a large population-based study of 4700 US children aged 4–15 years found a significant dose-dependent association between the level of lead in the blood and a clinical diagnosis of ADHD (82). The children had a 4-fold increased risk of ADHD when blood levels of lead were >2.1 µg/dL.

Birth complications

Children born prematurely (earlier than 37 weeks gestation) are at 2-fold greater risk of ADHD compared to children born at full term (83). Several studies support the association between low birth weight and an increased risk of ADHD (84-88). There is also support for an association between ADHD and perinatal factors such as young age of mother and caesarean delivery (89). The research in this area, however, is inconsistent, with some studies finding no association between ADHD and low birth weight, delivery characteristics or multiple births (90). Perinatal hypoxic-ischemic encephalopathy (damage to cells in the central nervous system from inadequate oxygen during birth) has also been shown to increase the risk of developing ADHD (91).

Acquired brain injury

Secondary ADHD is a common neurobehavioural consequence of acquired brain injury (ABI). ADHD is reported to occur in 20–50% of children following ABI and is predictive of poorer functional outcomes (92, 93). Attentional deficits are not as common following ABI in adults, and the deficits appear to differ. In adult ABI, many attentional skills are intact, but processing speed is reduced. With childhood injury, deficits tend to be more global, incorporating processing speed and sustained and shifting attention (94, 95). Deficits vary according to injury severity and the nature of underlying brain pathology (96, 97). Attentional deficits may persist over time, and attentional and processing speed deficits have been reported in children 5 years after an ABI (98).

Thyroid function

Thyroid hormones are essential to normal brain development and over- or underavailability of these hormones can influence behaviour and cognitive function in children and adults. Prospective screening of 277 children with ADHD found that the prevalence of thyroid abnormalities was higher (5.4%) in children with ADHD than in the normal population (<1%) (99). A cross-sectional study in preschool-aged children reported that high concentrations of thyroid stimulating hormone and low free thyroxine index were associated with ADHD symptoms in healthy preschoolers (100). ADHD has also been associated with generalised resistance to thyroid hormone, a condition where there is a reduced responsiveness of peripheral and pituitary tissues to thyroid hormones (101). However, not all studies have found an association between ADHD and generalised resistance to thyroid hormone (102).

Diet

The proposal that certain foods or food additives might cause ADHD and that the use of modified diets, such as the Feingold diet, can help to reduce ADHD symptoms has been the subject of ongoing debate among researchers and has been widely discussed in the media. Methodological problems and small sample sizes are common in the research in this area (103). Several studies have reported a link between additives and preservatives in the diet and behavioural symptoms (104-107). These studies support the notion that a small subset of children may be sensitive to certain foods or food additives that could be helped by careful exclusion

diets. The research does not support alterations to the diet as a general treatment for all individuals with ADHD (see Section 11.2 Diet, page 159).

Meta-analysis of 23 controlled studies found that sugar does not affect the behaviour or cognitive performance of children (108).

A number of controlled studies have reported zinc deficiencies in children with ADHD compared to control groups or population averages (for review see (109)). There is some support for the beneficial use of zinc sulphate supplements; however, these studies have been conducted in countries where dietary zinc levels are commonly low (Turkey and Iran) (109). Iron deficiency has been suggested as a risk factor for ADHD, but no studies to date have confirmed this (110).

The long chain polyunsaturated fatty acids (PUFA) such as omega-6 arachidonic acid (AA) and the omega-3 eicosapentaenoic acids (EPA) and docosahexaenoic acids (DHA) are critical to the development and function of the brain. It has been suggested that deficiencies in these essential fatty acids (EFAs) have an impact on brain function that causes or worsens some ADHD symptoms. This has been supported by research findings showing that levels of EFAs in the blood plasma of people with ADHD are reduced compared to healthy controls (for reviews see (98–100)). EFA supplements as a treatment for children with ADHD have been trialled in several studies; however, methodological limitations make it difficult to draw clear conclusions on the efficacy of this treatment (see Chapter 11. Complementary and Alternative Treatments for ADHDsection 11.2.2 Diet supplements, page 160).

Psychosocial factors

Increased risk of ADHD has been linked to several psychosocial factors related to an adverse family environment, including family dysfunction, chronic conflict and exposure to maternal psychopathology (111, 112). Controlled studies have also found that the prevalence of ADHD and other behavioural disorders was significantly higher among children who had suffered maltreatment (113) or sexual abuse (114).

The core symptoms of ADHD – inattention, overactivity and impulsiveness – are commonly reported outcomes when children have spent their early years under deprived conditions in institutional care (115-117). Recent evidence from an ongoing study of Romanian children raised in deprived conditions who have been adopted into UK families shows that the outcomes of inattention and overactivity have persisted into early adolescence (115).

These adoptee studies and several animal studies point to the possibility of a neurobiological basis to the ADHD-like symptoms, where an alternative neurodevelopmental pathway has been triggered by the stressful environment during an early critical period (115, 118). What is not clear is whether institutional, deprivation-related inattention and overactivity are phenotypically and neurobiologically distinct from ADHD.

Childhood illness

Viral infections during pregnancy (e.g. measles, varicella, rubella) and in childhood (e.g. enterovirus 71, HIV, encephalitis) have been associated with an increased risk of developing ADHD (for review see (119)). Febrile seizures related to viral infections such as human herpes virus 6 or influenza A have also been identified as risk factors for the development of ADHD and ADHD-like symptoms (119).

2.2 Neurobiology

The precise mechanisms involved in the neurobiology of ADHD are not fully understood. The dominant current paradigm suggests that disordered frontostriato-cerebellar brain circuitry underpins the executive function deficits at the core of this condition. Neuroimaging studies suggest that these deficits arise as a result of changes in the structure and function of the prefrontal cortex, the basal ganglia and the cerebellum (120).

Structural neuroimaging can be used to assess differences in brain structure, such as total brain volume or the size of specific regions of the brain. Functional neuroimaging techniques can also look at differences in regional cerebral blood flow as a measure of brain activation (see section 5.12.3 Neuroimaging, page 59). Several recent reviews and meta-analyses have collated the available data in this area (120-123).

It is now well accepted that children with ADHD have a significantly smaller total brain volume compared to matched healthy controls (124). A recent meta-analysis of structural imaging studies found that the largest differences were observed in the cerebellum, the corpus callosum, total and right cerebral volume, and right caudate (122). The impact of medication status was not assessed, as information on current or lifetime medication use was not well reported across studies (122).

The most consistently significant findings from structural and functional neuroimaging studies of people with ADHD, compared to matched controls, are differences in the frontal cortex, the basal ganglia and the cerebellum (for review see (120)).

A systematic review of functional MRI (fMRI) studies in ADHD identified group differences in activation in many brain areas (123). Individuals with ADHD showed dysfunction in the frontal lobes during tasks of inhibitory control. In analyses that examined inhibition errors, attention processes, motor function and working memory, the group with ADHD showed lower brain activity in temporal and parietal areas in attentional tasks and over frontal areas in motor tasks. Review of the fMRI data from people with ADHD who had not used medication suggests that altered brain activation patterns in children with ADHD are not due to the effects of treatment (123).

Meta-analysis of fMRI and PET studies in children, adolescents and adults found a consistent pattern of decreased frontal activity in those with ADHD compared to controls (121). Group differences encompassed the anterior cingulate and dorsolateral prefrontal and inferior prefrontal cortices, as well as related regions, including the basal ganglia, thalamus and portions of the parietal cortex (121).

Studies investigating the relationship between regional brain volumes and measures of functioning, such as behavioural rating scales and neuropsychological tests, have found that smaller volumes are associated with greater ADHD symptom severity (124). It is not clear whether the different subtypes of ADHD have differences in their neurobiological basis.

Limitations of the current neurobiological research

Inconsistencies in the literature remain, resulting from:

- methodological differences such as variation in reporting and inconsistent use of standardised measures, which make it difficult to compare studies
- small sample sizes, which give low statistical power
- heterogeneity in sample characteristics such as age, gender, medication status and ADHD subtype.

CHAPTER 3. PREVALENCE AND ASSOCIATED DEVELOPMENTAL PROBLEMS

3.1 Epidemiology

3.1.1 Prevalence in children and adolescents

Three recent systematic reviews of studies on the prevalence of ADHD suggest a prevalence of 5% to 10% (125-127).

Early studies suggested differences between North America and England in the prevalence of childhood hyperactivity; however, it was later recognised that these were not true differences in prevalence, but arose from use of different criteria to define ADHD (128, 129).

It is difficult to compare prevalence estimates across the last 30 years, due to changes in the DSM criteria for ADHD. The criteria were operationally described in the DSM-III in 1980 (24), and two revisions of the DSM have subsequently been published with somewhat different conceptualisations of the disorder (32, 130). Faraone et al (125) have described the prevalence of ADHD obtained when these different criteria for the disorder were employed:

- Thirteen studies used DSM-III criteria (125). Four of these studies were of US populations with a mean age of 9–11 years, and prevalence estimates ranged from 9.1% to 12%. Among the studies using non-US populations conducted in children and adolescents aged 4–16 years, prevalence estimates ranged from 5.8% to 11.2%.
- Twenty-two studies used DSM-III-R criteria to estimate the prevalence of ADHD among children aged 5–14 years (125). Prevalence estimates ranged from 7.1% to 12.8%, with the exception of one study which used teacher assessments and reported a prevalence of 26%. One further study (131) reported a prevalence of 2.8%. Twelve studies used DSM-III-R diagnostic criteria in non-US populations. In adolescents, prevalence ranged from 1.8% to 3.9%. Among children, prevalence ranged from 3.9% to 14.4%.
- Nineteen studies used DSM-IV criteria. Among the six studies that examined both male and female children, reported prevalence ranged from 9.5% to 16.1%. In studies of non-US populations, reported prevalence for children with a mean age between 7 and 11 years covered a wide range, from 2.4% to 19.8%. The authors note that in US studies, the highest prevalence had generally been obtained when using DSM-IV criteria.

Skoutini et al (126) also suggested that the estimated prevalence of ADHD was higher when DSM-IV criteria rather DSM-III-R were used. They reported that the majority of studies showed that ADHD predominantly inattentive subtype was the most common subtype, hyperactive-impulsive subtype was the least common and the combined subtype occupied an intermediate position. ADHD was more prevalent among males in all three subtypes.

Polanczyk et al (127) also noted that studies based on DSM-III-R or ICD-10 criteria had significantly lower ADHD/HD prevalence rates than those using DSM-IV criteria. These authors noted that the different definitions of ADHD in DSM-IV would be expected to yield higher prevalence rates than diagnoses based on ICD-10 because the latter requires that all criteria be met in at least two different situational contexts, whereas DSM-IV requires the presence of some impairment in more than one setting. ICD-10 also identifies mood, anxiety and developmental disorders as "exclusion diagnoses", whereas DSM-IV allows these diagnoses to be made as comorbid conditions.

Other factors that may influence estimates of the prevalence of ADHD include the informants used to describe children's problems (children, parents/caregivers and/or teachers), the age of participating children, the survey methodologies, the populations studied, the methods used to recruit participants, the diagnostic criteria used and whether or not the presence of impairment is required for diagnosis. Different interpretations of the meaning of "impairment" may also explain some of the variations in estimated prevalence (126, 127).

In epidemiological studies, children's problems are typically identified by means of questionnaires completed independently by children, parents/caregivers and/or teachers, rather than on the basis of an assessment undertaken by a clinician. The level of agreement between these informants when describing child and adolescent mental health problems is often relatively low (132). It cannot therefore be assumed that information from one informant accurately reflects the views of others. Studies need to describe clearly how discrepancies in information obtained from different informants were reconciled. Different protocols may be used in this situation. For example, recording of a positive symptom may require that the symptom be endorsed by one informant, or by more than one.

The assessment techniques used may influence prevalence estimates. For example, few behaviour checklists are designed to establish whether individuals meet criteria for diagnosis, but rather to identify the number of emotional and behavioural problems a person experiences. The use of checklists rather than structured interviewing may therefore affect the results of surveys. Finally, because there is typically a higher prevalence of males with ADHD, the ratio of males to females in a population may influence the prevalence of ADHD identified.

Estimated prevalence may be influenced by the age of children included in studies. In their review, Skoutini et al (126) reported a prevalence of 12.8% among 10–13 year olds, 9% among 14–16 year olds and 6% among 17–20 year olds. Egger et al (133), in a review of the nosology and epidemiology of ADHD in children aged 2–5 years, found that in studies using DSM diagnostic criteria, ADHD reported prevalence was between 2% and 7.9%. In this age range, they suggested that the hyperactive-impulsive subtype and the combined subtype were significantly more common than the inattentive subtype. Males and older preschoolers were more likely to meet the criteria for ADHD. Skoutini et al (126) also suggested that higher prevalences were associated with lower socio-economic status and urban as compared to rural residency.

The most comprehensive review of studies examining the prevalence of ADHD is that reported by Polanczyk et al (127). This review examined 102 studies comprising 171,756 participants from all regions of the world (the studies are listed at http://www.ufrgs.br/psiq/prodah-e.html). The overall pooled prevalence was 5.29% (95% CI, 5.01–5.56) and the pooled prevalences for children and adolescents were respectively 6.48% (4.62–8.35) and 2.74% (2.04–3.45). The authors reported that studies in children consistently suggest that ADHD is more prevalent in boys than in girls, with a male-to-female ratio ranging from 3:1 to 9:1, depending on the source of the sample.

3.1.2 Prevalence of ADHD among adults

ADHD, previously considered to be a disorder limited to childhood and adolescence, is now recognised to be, in many instances, a lifelong disorder. Follow-up of large cohorts of children with ADHD has demonstrated that, for the majority, clinically significant ADHD symptoms persist into adulthood (134, 135), and a smaller proportion of individuals continue to meet the full diagnostic criteria in early adulthood (136). There are discrepancies between the reported rates of persistent ADHD in adulthood, probably due to methodological differences between studies, which include differing study populations, discrepant diagnostic criteria and differing age at final assessment.

A recent meta-analysis of follow-up studies confirmed a significant age-dependent decline in ADHD symptoms (137) and highlighted that the extent to which the syndrome is found to persist into adulthood depends heavily on how it is defined. In this meta-analysis, about 65% of children with ADHD experienced partial remission in adulthood, and the full ADHD diagnosis persisted in approximately 15%. Such data highlight the need to consider ADHD as a lifelong disorder in some individuals.

Clinical populations represent the "tip of the iceberg", and community prevalence needs to be examined to appreciate the full extent of ADHD. Such prevalence data are rare and, as a result of the methodology applied in such large surveys, are inevitably difficult to interpret. Estimates of the prevalence of adult ADHD vary significantly. A population survey conducted in the United States estimated the current prevalence of adult ADHD to be 4.4% (138). This is comparable with a large cross-national survey which estimated adult ADHD prevalence across 10 countries to be 3.4% (range 1.2-7.3%) (139).

The factors that predict persistence of ADHD from childhood through to adulthood are poorly understood. However, there is evidence that increasing severity of ADHD symptoms in childhood is a predictor of its persistence into adulthood (140).

Such population-based findings highlight the public health importance of adult ADHD and the scope for improving access to appropriate treatments.

3.2 Comorbidities

The term comorbidity can be used in different ways. For the purpose of these guidelines, it is used to indicate that a person with ADHD also meets the criteria for another psychiatric disorder, or is experiencing other significant problems (e.g. cognitive difficulties, learning problems, problems with sleep).

3.2.1 Comorbidities in children and adolescents

There is substantial evidence that many psychiatric disorders are comorbid with each other; that is, the rate of simultaneous occurrence of these disorders among children and adolescents in the community is greater than would be expected by chance on the basis of their individual prevalence. This pattern is evident among children with ADHD (141). Several factors may contribute to this (141), for example:

- Comorbidity rates may be high among children with ADHD who are referred to clinics because those with more than one psychiatric disorder may be more likely to be referred.
- There is overlap between symptoms of ADHD and other disorders and this may lead clinicians to make comorbid diagnoses.
- It is possible that the presence of ADHD increases the risk for a second disorder among children and adolescents.

Current diagnostic criteria preclude the diagnosis of comorbidity between some conditions. For example, children cannot be diagnosed as having ADHD using DSM-IV criteria if their problems are better explained by another diagnosis category, such as autism or other pervasive developmental disorders (142). As such, by definition these conditions cannot be comorbid with each other.

Comorbidities may consist of different disorders present in the same children or adolescents:

- at the same time for example, many children who meet the criteria for ADHD also, at the same time, meet the criteria for ODD or CD (141, 143, 144)
- at different times for example, the presence of ODD is known to increase the risk for subsequent CD (145).

Finally, comorbidities not only occur within individuals, but also within families. For example, there is evidence that children with ADHD have relatives with higher rates of antisocial personality disorder, alcoholism and substance dependence (144).

As noted, children attending clinics may have higher rates of comorbidity than those in the general community. To determine rates of comorbidity in the community, it is important that studies randomly select children from the general community rather than those attending clinical services.

Comorbid psychiatric disorders

Over the last 20 years, substantial evidence has accumulated to show that children with ADHD are at increased risk for other psychiatric disorders. For example, Szatmari et al (11) reported that 44% of children with ADHD had at least one other psychiatric disorder, 32% had two other disorders and 11% had three other disorders.

ODD and CD are the most common comorbid disorders experienced by children and adolescents with ADHD. For example, Barkley (143) reported that between 45% and 84% of children and adolescents with ADHD meet diagnostic criteria for either ODD and/or CD. In their meta-analysis, Angold et al (144) reported a median odds ratio of 10.7 (95% CI = 7.7 to 14.8) between ADHD and CD/ODD. This is important because subsequent problems identified among children with ADHD may reflect the impact of ODD or CD rather than ADHD.

There is also evidence that children and adolescents with ADHD are at increased risk for major depressive disorder or dysthymic disorder. Angold et al (144) reported a median odds ratio of 5.5 (95% CI = 3.5 to 8.5) between ADHD and depressive disorder. The relationship between ADHD and paediatric bipolar disorder remains controversial, and reaching any conclusion about this relationship is difficult because of the lack of internationally agreed diagnostic criteria for paediatric bipolar disorder.

Children and adolescents with ADHD are more likely than others of the same age to meet criteria for anxiety disorder. An odds ratio of 3.0 (95% CI = 2.1 to 4.3) has been reported for this comorbidity (144). Other studies have reported that 25% to 35% of children with ADHD meet the criteria for an anxiety disorder (143).

Finally, there is evidence that children with other conditions may experience higher rates of ADHD. For example, children with Tourette syndrome have been found to be at increased risk for ADHD (146).

There may be gender differences in the types of comorbid disorders seen in children with ADHD. Data from the Australian Twin ADHD Project suggest higher rates of ODD and CD in males, and higher rates of separation anxiety disorder in females (147). This study also found differences between ADHD subtypes, with separation anxiety disorder more common in females with the inattention subtype and generalised anxiety disorder more common in females with the combined subtype (147).

Other comorbid problems

Children with ADHD may experience a range of other problems, including lower levels of intellectual performance than other children of the same age (143), higher rates of learning disorders (148) and problems with academic performance (143). Indeed, Barkley (143) suggests that virtually all clinic-referred children with ADHD perform poorly in school.

Estimated rates of learning disorders among children with ADHD vary across studies, in part at least because different studies use different definitions of learning disorder. For example, in some studies learning disorder is defined on the basis of a specific discrepancy between scores on intelligence tests and achievement tests for reading and maths. Others base the discrepancy on standard deviation units identified in individual studies (143).

Barkley (143) has drawn attention to a range of other problems experienced by children with ADHD. These include language difficulties, problems with speech and verbal problem solving, and deficits in a range of other cognitive functions. Executive function deficits may be present in some individuals with ADHD (149). Barkley (143) also suggested that ADHD is strongly associated with difficulties in emotional regulation, particularly the management of frustration.

Several studies have found that children with ADHD are more likely to have sleep problems than children of the same age without ADHD (150, 151). These problems include difficulties in falling asleep and more disruptions during the course of the night.

Many children with ADHD display movement difficulties consistent with developmental co-ordination disorder (152). There is evidence to suggest differences between subtypes in the type and degree of movement difficulty. One study identified poorer fine motor skills in children with the predominately inattentive subtype, while children with the combined subtype had greater difficulty with gross motor skills (152).

Children with ADHD are at increased risk for accidents. For example, Szatmari et al (11) reported that 7.3% of children with ADHD experienced accidental poisoning and 23.2% had had bone fractures, compared to 2.6% and 15.1% respectively in comparison groups. Another study (153) found that adolescents with ADHD were at increased risk for later accidents involving driving. They were also more commonly caught driving without a licence and reported for traffic violations, even after controlling for the presence of conduct problems, driving experience and gender (153).

3.2.2 Comorbidities in adults

The rates of comorbid psychiatric disorders in adults with ADHD are substantial and have a significant bearing on the clinical approach in this population.

Rates of psychopathology among adults who have been followed through after a childhood diagnosis of ADHD are 2–2.5 times those of controls, with a particularly high risk for antisocial personality disorder (up to 10 times that of controls) and drug or alcohol misuse (4–5 times that of controls) (134) (154). Compared to control populations, adults with ADHD have elevated rates of mood disorder (2–6 times) (154, 155), anxiety disorders (2–4 times) (154, 155), relationship dysfunction (2 times) and learning disorder (154).

Evidence from population studies points to substantially elevated rates of comorbid psychiatric disorders in adults with ADHD. The USA National Comorbidity Survey Replication (138) found, in a community sample of 3,199 individuals, that having ADHD as an adult was significantly associated with past marriage failures and unemployment, and that adults with ADHD were significantly more likely than those without the condition to experience other psychiatric disorders. Odds ratios were: for comorbid mood disorder, 5.0 (95% CI 3.0–8.2); for comorbid anxiety disorder, 3.7 (95% CI 2.2–5.5); and for comorbid substance abuse disorder, 7.9 (95% CI 2.3–27.3). In this survey the vast majority of respondents with ADHD (90%) were untreated.

The high rate of comorbid psychopathology seen in adults with ADHD may in part reflect the impact of longstanding adaptive impairments on development, as well as shared familial, environmental and possibly genetic vulnerabilities.

CHAPTER 4. CONSEQUENCES OF ADHD

4.1 Natural history

ADHD is one of the most common conditions of childhood and the persistence of symptoms into adulthood is well established. The core symptoms of inattention, distractibility, disorganisation, hyperactivity and impulsivity can cause clinically significant impairment in social, academic or occupational functioning. In addition, the disorders that frequently accompany ADHD, such as CD, ODD, tic disorders and learning disorders, add to the morbidity associated with ADHD.

Several longitudinal studies have documented the natural history of ADHD, clearly demonstrating that for the majority of people with ADHD the disorder persists through childhood into adolescence and adulthood. Faraone et al (137) analysed data from 32 follow-up studies of children with ADHD into adulthood. When full criteria for ADHD were used, approximately 15% of people diagnosed as children were still diagnosed with ADHD at age 25. In addition, meta-analysis found that approximately 65% of people by age 25 fulfilled the broader definition of DSM-IV, ADHD "in partial remission", indicating persistence of some symptoms of ADHD associated with significant clinical impairments (137). Consequently, even though they no longer met the full DSM-IV criteria for ADHD, people in partial remission may have significant functional impairment from their remaining symptoms (156).

The presentation of ADHD is highly variable. Some people have only very minor symptoms, while others may have severe impairments. In addition, there are cultural differences in the level of activity and inattention that are regarded as a problem. The core behaviours of ADHD are typically present from before the age of 7 and may be noticed as early as 3 years of age. Late onset of ADHD symptoms in adolescents and adults has also been observed (157-159).

A person's ADHD symptomatology can also vary throughout the lifespan, in both severity and types of symptoms seen. Longitudinal studies show that, on the whole, ADHD symptoms decrease over time (29, 160). Some symptoms, such as hyperactivity-impulsivity, diminish abruptly or present differently with age; other symptoms, such as inattention, are more likely to persist into adulthood in both males and females (161-163).

Symptom continuity appears to be mediated to a large extent by genetic influences (164-166). For the majority of children with ADHD, symptoms appear relatively stable when measured over periods of 2 to 10 years (164-167); but for a minority, symptoms will change over this period. A large population-based twin-study (166) assessed symptoms of ADHD in children at 8 to 9 years of age and then again at 13 to 14 years. For some children the number of ADHD symptoms decreased with age, with the number of children meeting the study's cut-off for ADHD decreasing from 4.7% to 3.1%; however, some children in the study began to display ADHD symptoms for the first time in early adolescence (166).

Changes in symptoms across childhood and adolescence may be a consequence of natural developmental processes seen in all children, but symptoms may also diminish due to learned skills, coping strategies and environmental restructuring. Accordingly, there is evidence to suggest that the change in symptoms between childhood and adolescence is mediated by both genetic and environmental effects that come into play during adolescence (166).

Outcomes over the lifespan cover a wide range. Although children with ADHD are at increased risk of continuing difficulties, many will grow up without persistent problems (168). Analysis of a 4-year longitudinal study of 85 boys with persistent ADHD found heterogeneous outcomes in education and social and emotional adjustment (169). Twenty percent of the boys were functioning poorly in all three

domains, 20% were functioning well in all three domains and 60% had intermediate outcomes (169). Poorer functioning was in part predicted by exposure to maternal psychopathology, larger family size, DSM-III-R psychiatric comorbidity and impulsive symptoms.

As the presentation and challenges of ADHD change over time, clinicians must take a lifespan approach and follow patients closely, modifying their care and treatment according to the individual's present needs.

4.2 Impairments and issues across the lifespan

4.2.1 Preschool-aged children

The early onset of ADHD symptoms and associated impairment has been recognised in both community and clinical samples (170, 171). Longitudinal studies report that preschool-onset ADHD symptoms persist well into adolescence (12, 170, 172-174). Difficulties for preschool-aged children with ADHD include problems with social behaviour and impaired fine motor skills. In one study, preschool-aged children with ADHD exhibited more negative social behaviours and scored significantly lower on a test of pre-academic skills than their healthy counterparts (175). The early onset of ADHD is also associated with an increased risk for accidents or injuries compared to children without ADHD (170, 176).

4.2.2 School-aged children

School-aged children with ADHD may face problems in many different domains, including low self-esteem, learning difficulties, impaired social skills, inappropriate social behaviour, poorer psychosocial adjustment and fewer friends (177).

Many of the difficulties for these children arise from the demands of school. Children with ADHD may perform poorly at school, and there is a well-established link between ADHD and academic outcomes that are poor compared to those achieved by children without ADHD (11, 16, 178-182). For example, compared to a healthy control group, children with ADHD were more likely to have a history of learning disabilities, repeated grades, placement in special education and academic tutoring (183). Academic underachievement may vary according to ADHD subtype (174, 184, 185). A longitudinal study that investigated academic achievement in children over 8 years old with ADHD found that only children with the predominantly inattentive subtype had lower reading, spelling and mathematics scores over time, compared to children without ADHD and children with ADHD combined subtype or ADHD hyperactive-impulsive subtype (174).

Difficulties in parent-child interactions have been identified for both boys and girls with ADHD, with consistent reports of greater adversity and more problems in their styles of interaction (see section 15.1.2 Parent-child interactions, page 185). In addition, greater conflict has been observed in the sibling relationships of children with ADHD, compared to controls (see section 15.1.4 Siblings of children with ADHD, page 188). Social interactions with peers are also often problematic and social rejection by peers is common (for review see (186)). Children with ADHD have been reported to form fewer friendships than those without the disorder (187, 188). Hyperactive and impulsive symptoms may result in unrestrained and overbearing behaviour that is not received well by peers (189) and inattention may hinder a child's ability to attend to social cues (187). These social difficulties may be more extensive if the child also has ODD or CD.

A study addressing the relationship between aggression and ADHD compared the maternal reports of aggressive behaviour in children with ADHD and children with no history of the disorder (190). Results revealed that the children with ADHD were significantly more aggressive than those without the condition. Work is needed to

determine whether the increase of aggression is in fact associated with ADHD symptomatology or is more closely linked to symptoms of ODD and/or CD.

Other difficulties are associated with ADHD in childhood. An increased number of accidents are common in this age group and school-aged children who have ADHD have more emergency department visits than their peers who do not have the disorder (191). There are higher-than-expected rates of ADHD amongst obese children, with two prevalence surveys finding a significant association between ADHD and paediatric obesity (192, 193).

4.2.3 Adolescents and adults

Longitudinal studies show the chronic course of ADHD through adolescence into adulthood. ADHD symptoms continue to be a source of significant impairment for around 60% of people who are diagnosed with ADHD in childhood (137). Although ADHD symptoms tend to change with age, with improvement in hyperactivity and attention span, adolescence and adulthood bring additional demands in education, at work, and in social and family relationships. Consequently, educational difficulties continue, and problems may appear in the areas of employment, driving and sexual relationships.

Compared to controls, fewer people with ADHD go on to higher education and they are less likely to complete their studies (194). One study on college students with ADHD found that these students had a significantly lower mean grade point average, were more likely to be on academic probation and reported significantly more academic problems (195).

Adult ADHD is associated with significant deficits in work performance and greater levels of unemployment compared to control groups (196-199). In addition, employment can turn over rapidly through dismissal or a persistent need for change unless the person is self-employed or in a position that allows autonomy at work (198, 200) (see Chapter 13. ADHD in the Workplace, page 176).

Individuals with ADHD have been found to have poor driving outcomes compared to peers without the disorder (17) (see Chapter 14. ADHD and Driving, page 180). Adolescents and adults with ADHD are more likely to have their licence suspended, to have received traffic infringement notices and to have had motor vehicle accidents (17, 201).

There is evidence for earlier and increased use of alcohol, tobacco and substance abuse in adolescents with ADHD compared to controls (202-204). In addition, several prospective longitudinal studies of children and adolescents with ADHD have found that ADHD is a significant predictor for later substance abuse (205-208). This risk was found to be greater if the individual also has CD (205, 208-210) or mood disorder (205). There is also a consistent association between ADHD and delinquency, criminal behaviour and recidivism (211-214).

ADHD is associated with being overweight and obese. Individuals with ADHD are overrepresented in paediatric and adult obesity clinic populations and obesity is identified as a problem in some paediatric ADHD clinics (192, 215, 216). In addition treatment of ADHD has recently been reported to result in improvement in weight control in subjects with a long history of weight loss failure (217). A large US population based study reported a 1.5 fold risk of being overweight for children and adolescents with ADHD (218), but not all population studies have confirmed this association (219). A US adult population study reported that adults with ADHD had a greater likelihood of being overweight (Odds Ratio 1.5, CI 1.05-2.38) and obese (Odds Ratio 1.81, 1.14-2.64) (220). It has been suggested that impulsivity may overeating with resultant increased weight however biological factors have not been ruled out (221).

Adolescents and adults with ADHD face a range of difficulties in social interactions and psychosocial adjustment. A 5-year prospective study following adolescent girls with ADHD found significant impairments in social skills, peer relationships and selfperception, compared to controls (163). Adults with ADHD have reported having fewer close friends, more trouble keeping friends and more social problems (198, 222). Adults with ADHD have also reported higher rates of separation and divorce (154, 223). In one study, adults with ADHD reported poorer overall marital adjustment and more family dysfunction than controls (224). However, there was no significant difference in reports between the spouses of adults with ADHD and the spouses of controls, suggesting that adults with ADHD have more negative perceptions of the health of their relationships and families than their spouses (224).

Adolescents and adults with ADHD may take greater sexual risks than their peers without ADHD. A survey of young adults found that childhood ADHD predicted earlier initiation of sexual activity and intercourse, more sexual partners, more casual sex and more partner pregnancies (225). Similarly, Barkley et al (198) found, in a longitudinal study from childhood, that young adults with ADHD (mean age 20, range 19–25) had become parents more often (38% versus 4%) and had been more frequently treated for sexually transmitted disease (16% versus 4%).

Low self-concept and low self-esteem are common in adults with ADHD, often resulting from educational and interpersonal difficulties (for review see (226)). People with ADHD are more likely than their peers without ADHD to have been diagnosed with any of a wide range of psychiatric disorders including anxiety and mood disorders (154, 227), and girls with ADHD are more at risk of developing eating disorders (163, 228, 229).

These long-term outcomes have major social implications for individuals with ADHD and are discussed in detail in other areas of this guideline document.

There remain many gaps in the research into the course of ADHD across the lifespan. Although adults with ADHD are increasingly a focus of research (230), some adult groups have not been adequately investigated. In particular, research is needed into the effects of menopause on women with ADHD and the incidence of ADHD in geriatric populations (231). Studies that include culturally diverse groups and Aboriginal and Torres Strait Islander peoples are also lacking, and more research is needed across all age groups.

PART II: ASSESSMENT AND DIAGNOSIS



Key messages

Assessment and diagnosis of ADHD

- 1. The DSM-IV criteria are the minimum necessary for diagnosis of ADHD.
- 2. The diagnosis of ADHD should only be made after a comprehensive assessment. This includes medical, developmental and psychosocial assessment, and elicitation of evidence of impairment in multiple settings, via gathering information from multiple informants.

CHAPTER 5. ASSESSMENT AND DIAGNOSIS

5.1 Recognition

It is clear from the literature that many individuals with ADHD are not initially recognised as having a problem related to attention, activity or impulsivity, even though they or their parents recognise that here are difficulties. These individuals may seek help from a variety of health and education professionals or through social or criminal justice services, all of whom need to be aware of the core features of ADHD and its varied presentations.

Because of under-recognition many children with ADHD do not receive therapy (232-235). In an Australian study Sawyer identified that only 28% of children with ADHD symptoms had attended any service, 23% attending a health service and 17% a school based service (234).

Often these children have been identified by parents or teachers as having problems with behavioural, emotional, learning difficulties or social interactions (235). Comorbid problems such as conduct disorder, anxiety disorder and mood disorder may also be the reason for bringing a child to attention (236).

Adolescents and adults with ADHD may come to notice because of substance abuse, including alcohol and cigarettes, especially if there is associated conduct disorder (237-239). Young women with ADHD run the risk of having the diagnosis overlooked, and instead be labelled with a diagnosis of depression or anxiety (240).

5.2 Overview of Assessment

5.2.1 Steps in assessment and diagnosis

There are five stages to "best practice" assessment and diagnosis of ADHD:

- 1. **Initial assessment by a GP or allied health professional** (see page 33). Identification of the core symptoms that comprise DSM-IV ADHD should form the basis of referral to a paediatrician or psychiatrist.
- Initial assessment by a paediatrician or psychiatrist (see page 36). This should address diagnostic evaluation and consider differential diagnosis and/or comorbidities. Specific medical conditions that may involve attentional problems or hyperactivity, and conditions such as pervasive developmental disorder, posttraumatic stress disorder, attachment disorder and bipolar disorder should be considered.
- 3. **Diagnosis of ADHD.** This should be carried out according to DSM-IV criteria and based on a full developmental, medical, psychosocial and psychiatric history, with data from other informants such as family members and teachers, obtained by rating scales or via interview or report. Assessment must not be too time-consuming (although must be of sufficient detail to consider the complexity of presentation). It should not require specialised training on diagnostic interviews to which many practitioners may not have access, or be restricted to certain professions (as with some of the "broadband" measures of psychological functioning which can only be accessed by certain professional groups). It is recommended that both broadband rating scales and scales more specific to ADHD symptoms be used (see section 5.4.3 Symptom checklists and behaviour rating scales, page 37). The former can aid in recognition of comorbidities and differential diagnosis assessment; the latter, in assessing pervasiveness and severity of symptomatology.

There are particular challenges for diagnosis in preschool-aged children and in adults. These are discussed below.

4. Assessment to inform management. This covers four main areas:

- a) recognition of comorbid conditions such as:
 - in children, anxiety, ODD and developmental co-ordination disorder
 - in adults, anxiety disorders, affective disorders, substance use disorders and personality disorders
- b) medical assessment of potential contraindications to some medications (e.g. pre-existing cardiac problems; see section 8.7 Side effects: special considerations, page 116)
- c) allied health assessment if the person has speech and language or coordination or sensory problems
- d) formal psychoeducational or neuropsychological assessment if the patient has academic difficulties
- e) evaluation of family and social issues, as behavioural management of ADHD generally requires active input from the family.
- 5. **Regular assessment of response to intervention**, whether medication or behavioural.

5.2.2 Approach taken

This chapter takes the position that the physician should follow DSM-IV criteria in assessing symptomatology and impairment in the diagnosis of ADHD (see Table 3, page 10, for the DSM-IV criteria). This position has been endorsed by other published best practice guidelines (241-244).

As discussed in Chapter 1, the ICD-10 diagnostic system is more restrictive and less widely used in Australia. The ICD-10-defined hyperkinetic disorder identifies a severely affected subset of patients with ADHD combined subtype. Such criteria do not encompass the group whose symptoms are predominantly inattention, despite the fact that individuals with these symptoms may experience significant impairments as well as comorbidities.

It is recognised that alternative viewpoints exist, especially from sociology and anthropology; however, discussion of these is beyond the scope of this clinically oriented document.

DSM-IV does not provide specific, developmentally adjusted ADHD criteria; the same diagnostic criteria are applied to patients across the lifespan. Consequently, there are particular issues in the diagnosis and assessment of ADHD for adults and preschool children. The application of the criteria in relation to preschool children and adults is discussed below.

The chapter incorporates some of the issues under discussion for the future DSM-V. The concept of a "sluggish" type of ADHD with hypoactivity remains a matter of speculation and is not included in current clinical guidelines (33).

There is scant information on ADHD in culturally and linguistically diverse (CALD) groups or Indigenous populations (in Australia or elsewhere). What little is available has been incorporated.

5.3 Initial screening assessment

Recommendation

1. A short scale such as the Strengths and Weaknesses of Attention Scale (SWAN) or the Diagnostic Rating Scale (DRS), or for adults the Short Adult ADHD Screening Scale or Barkley's Adult ADHD Quick Screen, should be used as an initial screen.

✓ Recommended best practice based on clinical experience and expert opinion

2. Clinicians, especially primary care physicians need to consider that there is more than one form of ADHD and many children may not present with the most obvious symptoms of hyepractivity/impulsivity. It is easy to overlook the most common subtype, the inattentive one and it may not present until secondary school when there are many additional demands on organisational skills and less support from teachers.

✓ Recommended best practice based on clinical experience and expert opinion

Initial assessment may be by a GP or other health professional. The clinician needs to:

- recognise each of the three core symptoms of ADHD, inattention, impulsivity and hyperactivity, and be aware that a diagnosis of ADHD does not require the presence of all three. Core symptoms need to have been present since a young age; however, during the brief time in a primary care consultation, a child may not show any obvious features of the ADHD, and this is not necessarily a helpful diagnostic approach
- be aware that ADHD without hyperactivity (i.e. predominantly inattentive symptoms) is a diagnosis that needs to be considered, particularly in girls and in adults
- be aware that the family and child may present initially for commonly comorbid conditions such as learning difficulties or oppositional defiant behaviour and be alert to the possibility that ADHD may be the underlying problem or an associated problem
- consider the use of symptom checklists. For children and adolescents, a modified version of the Strengths and Weaknesses of Attention Scale (SWAN – see http://www.adhd.net) is included in Appendix H. The Diagnostic Rating Scale (DRS) (245), a DSM-IV based ADHD screen, could also be used. For adults, the Short Adult ADHD Screening Scale (246) could be used. This scale is included in Appendix H. These measures do not provide a formal cut-off score for diagnosis, but are intended to help the clinician to decide if a full assessment is needed. For children with suspected ADHD, symptom checklists may be of value where there is family disagreement about the extent of a child's problems and the need for referral
- be aware that symptoms should occur in all environments (although they may not be evident or equally impairing in all settings) and discuss with the family any associated impairment in functioning at home, at school or in peer relationships. If a child presents via primary care then consultation with the school or kinder/preschool is critical
- be aware of the comorbid conditions that may occur with ADHD in children and adolescents, such as ODD, anxiety and autism spectrum disorders
- be aware of common comorbid conditions that may occur in adults with ADHD, such as mood disorders, anxiety disorders, substance use disorders and personality disorders

- be aware of family circumstances. In particular, recent changes in behaviour that may be linked to life events are far less likely to be due to ADHD
- be aware of the child's developmental and medical history. Issues such as hearing problems or inadequate sleep may be particularly relevant.

5.3.1 Education and training of GPs, and engagement in management of ADHD

In Australia (unlike some other countries) GPs are the gatekeepers to referral. Studies of Australian GPs have found that there can be a low interest in involvement in the diagnosis or care of children with ADHD (247, 248). GPs will therefore need continued support to facilitate their role in initial assessment and on-referral of children and adults presenting with attentional problems. They will need increased awareness of ADHD and associated conditions and to be able to access appropriate training and supportive materials. This could be enhanced by training by specialist colleges and supported by local divisions of general practice. As the hub of family health, the GP needs a framework to co-ordinate multidisciplinary management of families with ADHD, a role taken on especially by rural GPs who have developed special interest in ADHD. Ideally, GPs should be able to utilise models similar to the Wraparound approach, which provides a framework for collaboration between parents/caregivers and education and health professionals.

5.4 Assessment and diagnosis of ADHD in school-aged children and adolescents

Recommendations

- 3. Full assessment for possible ADHD requires a comprehensive medical, developmental and psychosocial assessment by the best qualified clinician available. This would usually be a paediatrician or child and adolescent psychiatrist with the training and skills required to assess and treat ADHD. *Recommended best practice based on clinical experience and expert opinion*
- 4. The DSM-IV criteria are the minimum necessary for diagnosis of ADHD. ✓ Recommended best practice based on clinical experience and expert opinion
- 5. Information from and discussion with multiple informants, including the child/adolescent, parents/caregivers, and educational and health professionals, needs to inform a diagnosis of ADHD in school-aged children. Recommended best practice based on clinical experience and expert opinion
- Diagnosis requires evidence of moderate to severe impairment across settings, including home and school.

 Kecommended best practice based on clinical experience and expert opinion
- 7. A thorough psychosocial assessment of the child/adolescent and family and medical assessment of the child/adolescent are part of a comprehensive assessment.
 - ✓ Recommended best practice based on clinical experience and expert opinion
- 8. Assessment should cover the presence and functional significance of comorbidities, including learning disabilities, anxiety/depression and disruptive behaviour disorders. This could include use of broad-based screening questionnaires.
 - ✓ Recommended best practice based on clinical experience and expert opinion

5.4.1 Criteria for diagnosis

As a minimum, the DSM-IV diagnostic criteria should be met before a diagnosis of ADHD can be made. The DSM-IV criteria break down symptoms into two groups: inattention and hyperactivity/impulsivity.

Diagnosis of ADHD require the presence of 6 of 9 symptoms of inattention or 6 of 9 symptoms of hyperactivity/impulsivity from the 18 ADHD symptom criteria set down in DSM-IV. Symptoms must occur "often", not occasionally, and persist for at least 6 months. Symptoms must be severe enough to be "maladaptive and inconsistent with developmental level" and must cause impairment across two or more contexts (e.g. home and school).

Consequently, symptom checklists and rating scales need to be accompanied by questions on:

- the age at which each symptom began
- duration of symptoms
- the severity of the symptom, with examples
- the pervasiveness across situations
- the impairment associated with that symptom at home, school (or work) and in social situations.

DSM-IV requires that six inattentive and/or hyperactive/impulsive criteria be met for a diagnosis of one of the three subtypes of ADHD (inattentive, hyperactive/ impulsive or combined). It is clear that the six-symptom cut-off needs to be more age- and gender-specific (249). There has also been discussion in the literature (27) on what to do when five criteria for each subgroup are met and no formal diagnosis can be made. Overall, suitable discretion that takes into account the issues of pervasiveness and impairment is advised in making a diagnosis.

5.4.2 Assessment by a paediatrician or child and adolescent psychiatrist

Full assessment for possible ADHD requires a comprehensive medical, developmental and psychosocial history by a paediatrician or child and adolescent psychiatrist, accompanied by the administration of ADHD symptom rating scales (parent/teacher/self-report) and an evaluation of impairment associated with ADHD symptoms across a variety of settings. The clinician needs to have a sound understanding of the normal patterns of infant, child and adolescent development, in order to differentiate behaviours/symptoms of ADHD from the normal patterns of cognitive function and behavioural features appropriate for the developmental age.

Medical, developmental and psychosocial history

- A thorough physical examination is required to exclude other conditions that may mimic or cause some symptoms of ADHD. It is import to check for symptoms and signs of:
 - hearing or vision impairment
 - o epilepsy
 - thyroid dysfunction
 - o allergic history
 - o sleep disorders such as sleep apnoea
 - o dysmorphic syndromes
 - o general medical problems.
- It is important to exclude possible aetiological causes of ADHD symptoms such as Fragile X syndrome, foetal alcohol syndrome, epilepsy or acquired brain injury.
- Factors known to be associated with risk of ADHD such as maternal stress in pregnancy, prematurity and neonatal complications should be considered.
- Developmental evaluation may include the use of a screening tool such as the Brigance (clinician rated) OR Ages and Stages (parent rated).
- Assessment of cognitive functioning and patterns of behaviour is needed to identify other developmental disorders (such as intellectual disabilities, speech and language disability, global or specific learning disabilities, developmental co-ordination disorder, autism and related spectrum disorders, and Tourette syndrome).
- A history of psychiatric problems should be taken, including attachment disorders, anxiety, depression, bipolar disorder and psychosis.
- Primary learning disability should be considered where problems emerged after school commencement.
- Any family history of medical and psychiatric problems should be considered.
- Social history and family dynamics should be discussed, and the contribution of family and social adversity, including neglect and abuse, assessed.
- If medications are being considered, the clinician must check for contraindications to the use of specific medications for ADHD, including cardiac disease and substance abuse/dependence in adolescents.

- Any co-existing disorders and risk factors need to be identified and their contribution to the behavioural/symptom profile and level of impairment assessed.
- Observation during the clinical consultation is important, but people vary in their behaviour during a consultation and the presence or absence of hyperactivity in this setting does not reflect the pervasiveness of symptoms.

Further assessment may be necessary for selected children to exclude other diagnoses:

- Comprehensive audiological assessment, including auditory acuity, "speech-innoise" discrimination and auditory memory (250) may be useful in some cases, for example, where there are learning difficulties and teachers concerned about auditory processing (see section 5.8.3 Auditory processing difficulties, page 47).
- Psychoeducational assessment is recommended for children who have educational or learning difficulties. The clinician may be aided in decision making here by reviewing previous school reports and results of standardised achievement tests such as the National Assessment Program Literacy And Numeracy (NAPLAN) completed in Years 3, 5, 7 and 9. Similarly, a range of assessment tests such as the Wide Range Achievement Test (WRAT) and Peabody Picture Test may provide some guidance in clinical assessment as to which children may require a more formal psychoeducational assessment.

5.4.3 Symptom checklists and behaviour rating scales

Data from Latent Class Analysis (251, 252) indicate the robustness of the 18 DSM-IV questions, given in different formats. Twin studies in Missouri and Australia (251) showed very similar latent structures – that is, the same constructs were being understood by the ADHD informants – despite different ways of asking the same questions (telephone interview, questionnaire).

Many symptom checklists for children and adolescents (e.g. as described in (27) or (253)) include the symptoms of ODD, which often co-occurs with ADHD (27).

Behaviour rating scales can provide useful information from the point of view of multiple informants, in a standardised fashion. The patient's symptoms are compared with age- and sex-matched normative data to provide a profile of behavioural symptoms.

While extensive diagnostic interviews such as the Diagnostic Interview Schedule for Children may be useful in research, the length of such interviews makes them unsuitable for routine clinical settings.

Three basic types of assessment measures may be used in assessing ADHD:

- "Broadband" measures of behavioural problems, including:
 - the Child Behaviour Checklist (CBCL)
 - the Behavioural Assessment Schedule for Children (BASC)
 - the Strengths and Difficulties Questionnaire (SDQ).

All these exist in multiple forms by age and by informant (parent, teacher or child). None is specific to ADHD and they should not be used to diagnose ADHD or any other disorder. They can, however, be an appropriate first step in identifying comorbidities and are widely used in Australia to assess the mental health of children attending agencies. Studies in clinically referred children with ADHD support the usefulness of the CBCL to identify patterns of comorbidities in the context of ADHD (254).

Note: The CBCL is now available, through the Australian Council for Educational Research, for paediatricians and psychiatrists experienced in ADHD.

- **Narrow-band questionnaires** that are specific to the symptoms of ADHD. For review see (255). The Conners' series is most widely used in Australia, with versions available for parents, teachers and older children (as well as for adults). It is available for experienced paediatricians through the Australian Council for Educational Research.
- **Interview and other schedules** that are based specifically on the DSM-IV ADHD items.

5.4.4 Evaluation of impairment

In formulating an individualised management plan, the severity of the person's ADHD needs to be taken into account. In determining the severity, the symptoms of hyperactivity, impulsivity or inattention and the level of subsequent impairment in multiple settings must be assessed. Issues that should be considered include academic performance, self-esteem, personal distress from the symptoms, social interactions and relationships, behavioural problems, and the development of psychiatric syndromes.

The question of how best to measure impairment has been the subject of debate and a number of instruments have been evaluated (see section 5.10.1 Measuring impairment, page 48).

5.5 Assessment and diagnosis of ADHD in preschool-aged children

Recommendations

9.	For children under 6 years of age, a stage when child development is rapid, it is essential to distinguish ADHD symptoms from normal developmental variation in impulsivity and attention. Recommended best practice based on clinical experience and expert opinion
10.	Assessment of children under 6 years of age should be undertaken especially thoroughly by paediatricians or child psychiatrists with expertise in developmental assessment, paying particular attention to identification of comorbidities and understanding of family dynamics and of cultural/religious diversity.
11.	The DSM-IV criteria are the minimum necessary for diagnosis of ADHD. Recommended best practice based on clinical experience and expert opinion
12.	Information from and discussion with multiple informants, including the child, parents/caregivers, kinder and preschool teachers, childcare staff, preschool staff and health professionals, needs to inform a diagnosis of ADHD in preschool-aged children. Recommended best practice based on clinical experience and expert opinion
13.	Diagnosis requires evidence of moderate to severe impairment across settings, including home and kinder/preschool. ✓ Recommended best practice based on clinical experience and expert opinion
14.	Assessment of children under 6 years of age should include, as a minimum, a screening developmental measure, such as the Ages and Stages Questionnaire, and, when developmental delay is suspected, a formal developmental assessment such as the Griffiths Mental Developmental Scales.
15.	Regular review is critical to review the course of symptoms over time. <pre> </pre> <pre> <pre> <pre> <pre> <pre> <pre> <pre> <pre> <pre> </pre></pre></pre></pre></pre></pre></pre></pre></pre>
16.	In this age range, where it is difficult to differentiate ADHD from other common problems, including developmental delays and specific speech and motor disorders, allied health assessments should be considered especially when developmental difficulties are evident. Recommended best practice based on clinical experience and expert opinion

5.5.1 ADHD in preschool-aged children

The presence of ADHD symptoms and associated impairment in preschoolers has been reported in both community and clinical studies. Prevalence estimates from community studies range from 2.8% to 6.3%, depending on the method of diagnosis (for review see (256)). Longitudinal studies report that preschool-onset ADHD symptoms persist over time (12, 170, 172-174). For many children, the effects continue well into adolescence and should not be dismissed as temporary or minor problems.

Accurate diagnosis of ADHD in preschool-aged children can be difficult as the range of normal behaviour is wide and the differential diagnosis for the constituent symptoms of ADHD is broad (257). Assessment should include consideration of other causes of behavioural dysregulation, including family contextual patterns, anxiety processes and medical problems. As early childhood development is particularly sensitive to the quality of the caregiver–child relationship, as well as the family, childcare, community and cultural contexts, assessment should include a review of a child's relationship patterns and developmental and attachment history, as well as parental and other contextual stressors and supports (258). That decisions about diagnosis and treatment planning are reached only after multiple appointments, reports from multiple informants and, in many cases, access to a multidisciplinary team are particularly important in this age group.

There is some evidence that both DSM-IV and ICD-10 criteria are valid for the assessment of young children with suspected ADHD (for review see (256)), though ICD-10 does not adequately recognise inattentive features.

5.5.2 Key considerations for diagnosis

Both ICD-10 and DSM-IV indicate the need to consider preschool behaviour in both its developmental and social context. Assessment should be informed by a thorough knowledge of normal development: young children typically have a shorter attention span than older children, impulsivity is more common and self-regulation in social settings is often not developed until around 3.5 to 4 years. A recent Dutch study found that symptoms of ADHD were reported in over 40% of typically developing preschool children (259). Furthermore, children with developmental delay, sensory deficits or language delay or disorder, or early acquired brain injury, for example, commonly exhibit symptoms which are seen in ADHD. Children who have been exposed to domestic violence, child abuse or other trauma can develop symptoms similar to those of ADHD. Similarly, disturbed patterns of attachment between a young child and his or her caregiver can be associated with difficulties in affective and behavioural regulation; a review of attachment relationships therefore is relevant in the assessment of any abnormal pattern of behaviour (260). Careful assessment of cognitive/developmental status, as well as emotional health and the child's social circumstances, is imperative and frequently will need the services of several professionals, preferably working collaboratively. The Ages and Stages screening tool may be useful as a minimum to determine if the child needs to be referred to a speech pathologist and/or other allied health professionals.

Assessment of children less than 4 years of age with impairing ADHD symptoms should be undertaken especially thoroughly, with particular attention to identification of comorbidities and understanding of family functioning. Management should be guided by a multi-axial approach to assessment, and is likely to include behavioural therapy, family therapy and developmental therapy.

5.6 Assessment and diagnosis of ADHD in adults

Reco	ommendations
17.	The DSM-IV criteria are the minimum necessary for diagnosis of ADHD. Recommended best practice based on clinical experience and expert opinion
18.	ADHD needs to be considered in adults who present with longstanding symptoms suggestive of ADHD (inattention, impulsivity, disorganisation) that appear to have started in childhood and are persisting into adult life. Recommended best practice based on clinical experience and expert opinion
19.	People with personality disorder and/or substance abuse accompanied by a significant level of impulsivity accompanied by inattention should be referred for evaluation of ADHD. Recommended best practice based on clinical experience and expert opinion
20.	Assessment of adults with suspected ADHD should include a thorough medical and psychosocial assessment. Recommended best practice based on clinical experience and expert opinion
21.	Assessment of adults with suspected ADHD should be undertaken by the best qualified clinician available. This would usually be an adult psychiatrist or a comprehensive psychiatric service with the training and skills required to assess and treat ADHD. This is due to the high incidence of comorbid psychiatric conditions and the overlap of clinical features of ADHD with other conditions such as bipolar disorder and personality disorders.
22.	Other possible diagnoses or comorbidities should be considered, via history; for example, acquired brain injury, neurological condition or other DSM-IV diagnosis such as anxiety disorder or pervasive developmental disorder. Recommended best practice based on clinical experience and expert opinion
23.	Adult presentations may occur in the context of problems encountered with work and study. In such instances, vocational/intellectual assessments may be useful, not for diagnostic purposes, but to clarify the functional consequences of the diagnosis.

✓ Recommended best practice based on clinical experience and expert opinion

5.6.1 ADHD persistence into adulthood

Depending on the diagnostic criteria or thresholds used, as many as 67% of people with ADHD symptoms in childhood continue to have their disorder in adulthood (253). A systematic review and a meta-analysis of longitudinal studies of ADHD (137) found that about 15% of people who have ADHD in childhood will continue to have it as a full disorder into adulthood, and a further 40-60% experience partial remission, with persistence of some symptoms and significant clinical impairments.

5.6.2 Pathways to diagnosis

Adults presenting for assessment of ADHD do so in a variety of contexts including:

- as a planned transition from paediatric services for ongoing management of ADHD on reaching adulthood
- for reassessment following a variable gap in treatment for the childhood disorder
- for initial assessment following diagnosis in a child or other relative
- after identification of symptoms following assessment or treatment of another (comorbid) mental disorder

• *de novo* following concern about possible ADHD symptoms by the individual, family or primary care physician.

As with most psychiatric disorders, a diagnosis of Adult ADHD is made on clinical grounds. Case identification is based on systematic assessment of the symptom profile and exclusion of alternative psychiatric or medical causes. As ADHD symptoms (e.g. inattention, forgetfulness, motor restlessness) occur commonly in the community, a systematic and hierarchical approach to diagnosis is advocated.

5.6.3 Criteria for diagnosis

There are no universally accepted criteria for diagnosis of adult ADHD and further research is needed in the development of reliable and valid diagnostic criteria for adults.

The DSM-IV was the first set of criteria for ADHD to use wording applicable to both adults and children and it provides a useful guide for the clinician. The DSM-IV criteria, however, have not been validated in adults. They do not include developmentally appropriate symptoms and thresholds for adults and fail to identify some significantly impaired adults who may benefit from treatment.

ADHD symptoms generally undergo some diminution with age and it remains uncertain whether the DSM-IV "threshold" (i.e. number of symptoms endorsed) should be adjusted downward for adults to take into account this age-dependent reduction in symptoms. At present, this issue is dealt with in DSM-IV by labelling those adults with clinically significant sub-threshold symptoms as "ADHD in partial remission". The clinical status of this less severely affected group awaits clarification.

Diagnosis of adult ADHD according to DSM-IV requires onset of symptoms before the age of 7; however, this criterion may not be helpful. A study comparing 27 adults with full ADHD who met all DSM-IV criteria for childhood-onset ADHD with 79 adults with late-onset ADHD who met all the criteria except age of onset showed similar patterns of psychiatric comorbidity, functional impairment and familial transmission (158).

Clinicians therefore need to be flexible in applying current ADHD criteria to adults, and measures of functional impairment such as the Current Behaviour Scale should be used in conjunction with DSM-IV.

5.6.4 Steps in diagnosis

Steps in the diagnostic assessment of adults with possible ADHD include:

- 1. evaluation of presenting symptoms to determine whether they lie in the key domains affected by the disorder
- 2. exclusion on history, physical examination and laboratory testing of other disorders that could possibly mimic ADHD
- 3. evaluation of psychiatric factors contributing to or present in addition to ADHD symptoms
- 4. evaluation of the longitudinal course of symptoms, including verifying their presence or absence during childhood
- 5. evaluating the clinical impact of ADHD symptoms across a variety of settings.

Each of these steps is discussed below.

1. Evaluate presenting symptoms

The symptoms of ADHD in adults are an extension of those seen in children. As in childhood, symptoms may not be present across all domains. Furthermore, there may have been differential improvement in some symptoms (particularly

hyperactivity/impulsivity) with age. Individuals may have difficulty sustaining attention in a number of settings, particularly when performing demanding cognitive tasks. Hyperactivity manifests physically, but may also have a mental component (having accelerated thoughts or many thoughts simultaneously). Individuals may be verbally impulsive (e.g. making tactless comments or interrupting others), or may engage in impulsive behaviour or high-risk activities. Adults with ADHD often have poor organisational skills, and these may affect social or vocational performance. Adults with ADHD may not show marked external symptoms of ADHD (e.g. fidgeting, restlessness, poor attention span) during brief clinical assessment, although they may report such problems in their daily lives.

Many symptoms checklists and rating scales have been designed for adults to assess current symptoms of ADHD. None of the scales, however, are sufficient for diagnostic purposes when used alone. Although the reliability and sensitivity of the instruments are good, specificity can be low, which can give rise to false positive or false negative diagnosis. For example, individuals with disorders such as anxiety or depression are also likely to score high on measures of inattention using these instruments. The scales are most useful when used as part of a formal clinical assessment.

Symptoms checklists and rating scales for adult ADHD include:

- the Conners' Adult ADHD Rating Scales (CAARS) (261)
- the Wender-Reimherr Adult Attention Deficit Disorder Scale (WRAADS) (262)
- Wender Utah Rating Scales (WURS) (263)
- the Brown Attention Deficit Disorder Scale (BADDS) (264)
- the Copeland Symptom Checklist for Attention Deficit Disorders Adult Version (265)
- the ADHD Rating Scale (266, 267)
- the Adult ADHD Self-Report Scale (ASRS). This 18-item self-report scale reflects DSM-IV emphasis on symptoms. It is widely used and has been validated in the National Comorbidity Replication Survey. Subsequently, a six-item ASRS screener has been shown to outperform the full version (246)
- the Barkley Adult ADHD Quick Screen based on the DSM-IV checklist for ADHD symptoms
- the Jasper/Goldberg Adult ADHD Screening Examination.

2. Exclude other medical conditions

Medical conditions that can mimic ADHD include, but are not limited to:

- past acquired brain injury (e.g. head injury, meningitis, cerebral anoxia)
- sleep disorders (e.g. sleep apnoea)
- non-acute effects of viral infection, including past encephalitis or HIV
- chronic medical illnesses such as renal or liver failure
- seizure disorders (e.g. complex partial seizures)
- some endocrine disorders (e.g. hypothyroidism, hyperthyroidism, hypoglycaemia)
- comorbid or previous substance abuse.

Exclusion of possible medical causes of symptoms is particularly important in a person presenting for the first time at a relatively late age (e.g. over 35 years). Assessment should be tailored to the person's needs and may require detailed history, physical examination (including assessment of baseline blood pressure) and
relevant investigations. Routine blood tests (urea, electrolyte, creatinine levels, a full blood count, liver and thyroid function tests) are of use only when the presentation suggests an underlying medical disorder. An electrocardiogram (ECG) is performed in older adults or those with a history or signs of cardiac disease, particularly when treatment with tricyclic antidepressants is being considered.

3. Evaluate other psychiatric factors

The symptomatology of ADHD overlaps with other psychiatric conditions including cluster B personality disorders (antisocial and borderline personality disorders) and bipolar mood disorders. Symptoms seen in all these conditions include mood instability, impulsivity and poor anger control. Chronic low self-esteem, volatile mood and poor frustration tolerance are commonly seen in adults with ADHD and should not exclude its diagnosis.

There is also a high incidence of comorbid psychiatric illnesses, particularly affective disorders, anxiety disorders and substance use disorders, in adults with ADHD. It is common for mood and anxiety disorders to be treated by a psychiatrist, sometimes for years, without diagnosis or treatment of the underlying ADHD.

Chronic use of illicit drugs (e.g. cannabis, cocaine, amphetamines) and alcohol should be considered as a possible cause of cognitive and behavioural symptoms. Where possible ADHD symptoms and drug misuse occur together, it is usually prudent to reassess for ADHD symptoms after treatment of the drug misuse. Random urinary drug screening may be performed to detect or monitor illicit drug use in selected patients.

4. Evaluate the longitudinal course of symptoms

Longitudinal assessment of symptoms usually reveals the consistent impact of symptoms across a variety of settings. The typically episodic nature of some conditions, such as bipolar disorder and major depression, can assist in distinguishing these disorders from ADHD.

Retrospective identification of childhood ADHD symptoms is a central component in the assessment of an adult presenting with ADHD symptoms. A retrospective assessment of childhood symptoms should be attempted regardless of whether a childhood diagnosis has been made in the past by a treating clinician. In addition to a retrospective self-report, a retrospective diagnosis needs to be supported by consistent reports by a parent/caregiver or other informant (e.g. an older sibling) of symptoms of ADHD in one or more settings, as well as objective accounts of aberrant behaviour recorded in past school or educational reports, or past documentation of symptoms from childhood clinical assessments (see section 5.10.2 Retrospective self-reports, page 50).

Use of a standardised rating scale for retrospective assessment of symptoms may aid in standardising the clinician's approach. A limited number of scales are available. These include:

- the self-report Wender Utah Rating Scale (WURS) (263)
- the informant-reported Conners' Abbreviated Symptoms Questionnaire (ASQ).

5. Evaluate the clinical impact of ADHD symptoms

Where a diagnosis of ADHD is being considered, careful evaluation of the impact of symptoms across a variety of settings is required. As with the childhood disorder, clinically significant symptoms will usually impact on functioning in several domains (work, family and/or social spheres).

Many adults presenting with ADHD symptoms have managed adequately through school, with their symptoms only causing functional difficulties for them in the more demanding contexts of further study or employment. While not required for diagnosis, in these instances it may be of assistance to the patient to undergo

vocational or cognitive assessments to identify areas of difficulty (e.g. poor organisational skills, difficulties taking in multi-step instructions) in order to understand the problem better and develop compensatory skills.

A subscale of the Current Behavior Scale (CBS) (see Appendix H) may be of use to identify adults at greater risk of functional impairments (268, 269) (see section 5.10.1 Measuring impairment, page 48).

If there are concerns that ADHD symptoms are progressive or that function has deteriorated, a neurological basis should be considered and this is best done by a neurologist.

5.7 Psychoeducational assessment

Recommendations 24. Most children with ADHD have intellectual development within the normal range. A significant proportion, however, present with cognitive deficits, learning difficulties and social adaptive difficulties, and in these individuals a comprehensive psychoeducational assessment is particularly necessary. ✓ Recommended best practice based on clinical experience and expert opinion Psychoeducational assessment should cover all aspects of academic 25. performance. At a minimum this should consist of an individually administered intelligence test such as the age-appropriate version of the Wechsler series. ✓ Recommended best practice based on clinical experience and expert opinion 26. Educational difficulties are common in people with ADHD. Where these problems are suspected (e.g. via parent/caregiver or teacher report, or national/State/Territory literacy and numeracy assessments), psychoeducational assessment should be conducted to assess learning potential and attainment. ✓ Recommended best practice based on clinical experience and expert opinion 27. Assessment of educational difficulties is important in people of all ages with suspected ADHD and/or suspected educational difficulties, to inform diagnosis and identify learning difficulties that should be targeted for intervention. Such assessments are not diagnostic of ADHD. ✓ Recommended best practice based on clinical experience and expert opinion Educational difficulties are common in individuals with ADHD, with learning

Educational difficulties are common in individuals with ADHD, with learning difficulties being one of the most common comorbidities documented. Where these problems are suspected (e.g. via parent/caregiver or teacher report, or NAPLAN results), psychoeducational assessment should be conducted to assess learning potential and attainment. Common avenues for such assessments include school psychologists and child psychologists.

Psychoeducational assessment should cover all aspects of academic performance and be interpreted in the context of educational opportunity and intellectual level.

At a minimum this should consist of an individually administered intelligence test such as the age-appropriate version of the Wechsler series. There are some subtests of the Wechsler which are sensitive to attentional or impulsivity problems (270), although these are not diagnostic of ADHD. ADHD symptoms may confound the accurate estimation of the person's IQ because of frequently encountered distrust of IQ testing, the difficulties in testing individuals with inattention and impulsivety, and the effects of ADHD on some specifc measures (e.g. digit memory span).

If learning difficulties are suspected, IQ and academic achievement testing should be supplemented with specific measures of cognitive aspects of attention (e.g. vigilance), although such problems are not specific to ADHD and can be seen in many childhood disorders. Identification of such attention deficits is relevant in tailoring educational interventions and may also be useful in monitoring cognitive effects of medication.

As comorbid speech and language difficulties and auditory processing difficulties may also impact on learning, consideration should be given to allied health assessments in these areas (see section 5.7.1 Speech and language difficulties, page 46, and section 5.7.3 Auditory processing difficulties, page 47).

5.8 Allied health assessments

Recommendations

28. Allied health assessment may be helpful in many children and adolescents with suspected ADHD to identify comorbid speech and language difficulties, motor co-ordination and sensory difficulties that can be targeted with intervention.

✓ Recommended best practice based on clinical experience and expert opinion

29. The need for allied health assessment should be determined on a case-bycase basis.

✓ Recommended best practice based on clinical experience and expert opinion

Some children with ADHD present with normal development in other areas and will not require further evaluation and intervention. However, where there is concern about other aspects of a child's development, additional assessment may be helpful. There are a number of functional difficulties that are commonly identified for children with a diagnosis of ADHD:

- speech and language difficulties, including more subtle problems with effective communication (e.g. pragmatics, inference)
- gross and fine motor difficulties (e.g. developmental co-ordination disorder, deficits in attention, motor control and perception, subtle clumsiness and handwriting problems)
- auditory processing difficulties.

5.8.1 Speech and language difficulties

ADHD and language impairments frequently occur together (271), and speech pathology assessment is necessary for children with significant language difficulties. While traditionally perceived as an issue with younger children, a Western Australian study (Hagemann, unpublished PhD thesis, Curtin University) indicates that speech and language problems are still common in adolescents with ADHD. Children with comorbid speech sound disorder and specific language impairment have also been found to be at increased risk of ADHD (272).

5.8.2 Gross and fine motor difficulties

Occupational therapy and/or physiotherapy assessment is necessary, especially for younger children who have problems with motor function, handwriting and spatial and body awareness difficulties. Australian data (152) indicate that some 50% of children with ADHD may meet the criteria for developmental co-ordination disorder, while a further 30% may have some degree of motor problems. Occupational therapy may also have a role in providing support for classroom modification for children with attention issues.

Occupational therapy and/or physiotherapy assessment may also be considered for children with sensory processing difficulties. It has been suggested that poor sensory processing may contribute to the behavioural profile of children with ADHD and there is an overlap in behaviours described for sensory modulation disorders

and those of ADHD, such as inattention, poor impulse control and hyperactivity (273-275). Sensory modulation disorders arise from impairments in receiving, modulating and integrating sensory input (274). Accordingly, when sensory integration does not develop as efficiently as it should, the child may have resultant problems in learning, development or behaviour (276).

Research examining sensory processing difficulties in ADHD has been relatively rare. However, two studies have found that children with ADHD have different patterns of sensory processing compared to children without ADHD (273, 277), which may indicate that a subgroup of children with ADHD have sensory processing difficulties (273).

5.8.3 Auditory processing difficulties

Children who appear to have auditory problems may require additional comprehensive audiological assessment.

Auditory processing assessments, over and above basic hearing tests, may be helpful in order to describe a child's functional difficulties with processing auditory information. Of note, auditory processing disorder and ADHD overlap in diagnostic criteria and behavioural presentation, particularly in symptoms such as distractibility, inattentiveness and associated language and academic problems (278, 279). Auditory processing assessment alone should not be used to diagnose ADHD (280).

5.9 Issues in differential diagnosis

5.9.1 Pervasive developmental disorders

DSM-IV does not recognise the co-occurrence of ADHD and pervasive developmental disorders (PDD) such as autism spectrum disorders (ASD). However, it is well established that individuals with PDD/ASD may also have symptoms of inattention, hyperactivity and impulsivity (281). The symptoms of ADHD can cause significant impairment in individuals with PDD/ASD. The management of ADHD symptoms in the context of PDD/ASD is discussed in section 8.6 Medications for ADHD in children with a developmental disability (page 112).

5.9.2 Acquired brain injury and other neurological conditions

Recommendations		
30.	A thorough medical history and examination are required to identify any acquired brain injury or other neurological condition that may contribute to the presenting symptoms. ✓ Recommended best practice based on clinical experience and expert opinion	
31.	Individuals with ADHD symptoms and a history of acquired brain injury and other neurological conditions should receive comprehensive neuropsychological evaluation. Recommended best practice based on clinical experience and expert opinion 	
32.	In consultation with the neurologist or neurosurgeon, ADHD medications may be one component of a comprehensive management plan.	
Brain brain	insult, particularly that which causes damage or disruption of areas of the reported to be involved in mediating attention (e.g. frontal regions, white	

brain insult, particularly that which causes damage or disruption of areas of the brain reported to be involved in mediating attention (e.g. frontal regions, white matter, parietal lobes), increases the risk of ADHD-like symptoms, in particular, inattention and impulsivity. In fact, such attention impairments may be the hallmark features of such conditions. Specifically, children with traumatic brain injury, prematurity, cancers (treated with chemotherapy and radiotherapy), tumours, epilepsy, lead exposure, foetal alcohol syndrome and developmental malformations are all at elevated risk of attentional difficulties. Of note, these difficulties usually exist in the context of broader cognitive and social problems that will need assessment and intervention.

While these conditions do not necessarily fit all the criteria for diagnosis of ADHD (e.g. age at onset of symptoms), they often receive the diagnosis, and early research is suggesting that they may benefit from similar medical/pharmacological management; see, for example, research on use of Ritalin in children with leukaemia (282) and traumatic brain injury (283). Where there is no medical contraindication, it may assist these individuals to trial such interventions.

5.10 ADHD measures and information sources

5.10.1 Measuring impairment

Research questions

- In individuals with ADHD does the inclusion of measures of impairment, in addition to DSM-IV/ICD10, further inform diagnosis and assessment?
- Sub-question: Can measures of impairment distinguish between impaired and unimpaired individuals with ADHD?

Evidence statements

- **Preschoolers:** One study found that the use of DSM-IV ADHD symptoms plus impairment in two situations predicted greater global, academic and social impairment (284). One study found that the Hillside Behaviour Rating Scale made an independent contribution to the prediction of functional impairment above parent and teacher ratings (285). (Level III-3 diagnosis)
- Children and adolescents: Three studies found that the instruments utilised to measure impairment could discriminate impaired from non-impaired children above information on ADHD symptoms alone (286-288). (Level III-3 – diagnosis)
- Adults: Two studies found that the Current Behavior Scale (CBS) (269) or CBS subscale (268) could identify a subgroup of adults with ADHD who were at significantly higher risk of functional impairment. (Level III-3, III-1 diagnosis)

Recommendations

- 33. While the DSM-IV diagnosis of ADHD does require impairment, impairment at home, school/work and in social relationships is poorly defined. Careful discussion is needed with the family as to the extent to which problems are specifically related to ADHD symptomatology. *Recommended best practice based on clinical experience and expert opinion*
- 34. Measures of impairment specific to ADHD for preschoolers, children and adolescents are not yet sufficiently robust for routine use. (Grade D)
- 35. The Current Behaviour Subscale could be used as a measure for functional impairment in adults with ADHD. (Grade C)

Background

The DSM-IV requires impairment as well as the presence of symptoms for the diagnosis of ADHD, but fails to define impairment clearly. Implicit in this requirement is that the construct of functional impairment is distinct from the ADHD symptoms. A recent review of impairment scales indicates that this is a general problem when diagnosing any psychiatric disorder (289).

While several measures of impairment specific to ADHD have been developed and appear promising, none is established and more research is required.

For many Australian children attending a Child and Adolescent Mental Health Service, the Health of the Nation Outcome Scales for Children and Adolescents (290) will be routinely completed (this varies across jurisdictions). This provides a basic measure of impairment that correlates well with parent and teacher reports on more extensive measures, such as the Strengths & Difficulties Questionnaire (291).

The Children's Global Assement Scale has been utilised to define impairment in ADHD and other psychological disorders, but has not gained wide acceptance as a clinical tool because of its subjectivity (292, 293)

A subscale of the Current Behavior Scale (CBS) (268) (see Appendix H) provides a short checklist of impairments associated specifically with adult ADHD.

Summary of research evidence

Preschoolers

Two studies met our inclusion criteria:

- One study looked at the validity, in preschoolers, of the requirement for crosssituation impairment in the diagnosis of ADHD (284). Children in this age group who met the full diagnostic criteria (DSM-IV ADHD plus impairment in two situations) showed significantly greater global, academic and social impairment in the three annual follow-up assessments, compared to either controls or the group that met DSM-IV symptom criteria but had impairment in only one situation. However, the group with impairment in one situation had significantly greater global and academic impairment at each follow-up than the no-ADHD control group, and 34% later met full diagnostic criteria for ADHD compared to 3.1% of the controls.
- The Hillside Behaviour Rating Scale (285) was found to make an independent contribution, above parent and teacher ratings, to the prediction of functional impairment.

Children and adolescents

Three studies conducted in children and adolescents met our inclusion criteria. Each assessed a different measure of impairment:

- One study (288) identified two algorithms generated from parent and teacher reports, which the researchers called "subtype ROC" and "ADHD ROC", that were more efficient than DSM-IV symptom counts in discriminating children with functional impairment from children who do not have functional impairment.
- Assessment of 6 individual measures of impairment and a combined measure termed Global Impairment Index (GII) (287) found low to moderate correlations with symptoms (Maternal SNAP-IV and CBCL Attention Subscale). The researchers also found that gender, CBCL Attention subscale, PPVT (Peabody Picture Vocabulary Test) score and maternal SNAP-IV symptom count were significant predictors of impairment. Altering the number and type of scales used in diagnosis dramatically altered the number of positive diagnoses. Inclusion of impairment (GII) and symptom scales reduced the number of positive diagnoses.
- The Parent and Teacher Impairment Rating Scale (IRS) was found to make an independent contribution to the prediction of impairment, beyond the prediction from ADHD symptoms alone (286).

Adults

Two studies conducted in adults with ADHD met our inclusion criteria:

- The Current Behavior Scale (CBS), using the cut-off of the 50th percentile, was able to identify adults with ADHD who were at significantly higher risk of functional impairment, including global functioning, social adjustment, number of comorbidities and socio-economic status (269).
- A short (8-item) scale derived from the CBS correlated well with the 99-item CBS and predicted poor functional outcomes in global functioning and social adjustment (268).

5.10.2 Retrospective self-reports

Research question

 In adolescents and adults with ADHD, are retrospective self-reports of symptoms in childhood accurate?

Evidence statement

• Four studies supported the validity of using adult self-reports of childhood symptoms for the diagnosis of ADHD (294-297). (Level III-3 – diagnosis)

Recommendation

36. The reports provided by adults with ADHD on the symptoms they had in childhood are accurate and valid for informing diagnosis and assessment. (Grade B)

Background

The diagnosis of adult ADHD requires that individuals meet both the childhood and the adult criteria for the disorder. Meeting those criteria ordinarily requires a retrospective report of ADHD symptoms, usually by the individuals.

Summary of research evidence

Four studies were identified that met our inclusion criteria:

- A longitudinal study found good agreement between self-reports of childhood symptoms collected at age 21 and again at age 30, despite the 9-year gap, suggesting stability in the way symptoms are recalled over time (294).
- Another longitudinal study found good agreement between retrospective diagnosis of childhood ADHD based on adult self-reports and the actual childhood diagnosis (295).
- In another longitudinal study, there was moderate agreement between selfreports of childhood ADHD in adulthood and the parent report at study entry in childhood (296).
- When adults with ADHD and their parents were asked to report on the ADHD symptoms that were seen in childhood, good agreement was found between the two (297).

5.10.3 Information from multiple informants

Research questions

- For preschoolers, school-aged children or adolescents with ADHD, is there consensus between parent and teacher reports of symptoms and classification of subtype?
- For individuals with ADHD, is there consensus between third-party reports and self-reports of ADHD symptoms when assessed with parallel instruments?
- For preschoolers, children or adolescents with ADHD, does the psychiatric status of the parent influence the content of parent/caregiver reports?

Evidence statements

• Parent versus teacher reports

- Preschoolers: Two studies showed low agreement between parents and teachers for ADHD symptoms and the classification of subtypes (298, 299). (Level III-2 diagnosis)
- Children: Eleven studies showed low agreement between parents and teachers for ADHD symptoms and the identification of subtypes (266, 299-308). (Level II, III-2, III-3, IV – diagnosis)

• Self-report versus third-party report

- Adolescents: Three studies in adolescents showed low to moderate agreement between self-reports and third-party (parent) reports (309-311). (Level III-2 – diagnosis)
- Adults: Six out of seven studies showed low agreement between self-reports and third-party reports (parents, partners, friends, family members) (294, 296, 312-315), and one found good agreement between self-reports and third-party reports (parents, partners, friends) (297). (Level III-2 diagnosis)

• Parent psychiatric status

- One study of children with ADHD found that parental ADHD did not have an impact on maternal reports of ADHD symptoms (316). (Level IV – diagnosis)
- Three studies found that, using the CBCL rating scale, mothers with a history of depression reported significantly more symptoms in their child, compared to mothers with no history of depression (317-319). (Level III-2, IV diagnosis)
- Two studies found that, when using the DBDRS, a history of depression in the mother was not significantly associated with maternal reports of symptoms (308, 319). (Level IV – diagnosis)

Recommendations

- 37. For diagnosis of ADHD in children and adolescents, input from multiple informants (e.g. parents/caregivers, teachers) should be considered in evaluating the chronicity and pervasiveness of impairment. (Grade B)
- 38. Diagnosis of ADHD should not solely rely on the use of parent or teacher information as this may lead to over-diagnosis. Information from both parents and teachers should be considered. *Recommended best practice based on clinical experience and expert opinion*
- 39. For diagnosis of ADHD in adults, input from multiple informants (e.g. partners, family members) should be considered, where possible, in evaluating chronicity and pervasiveness of impairment. (Grade B)

- 40. When evaluating parent reports describing children with suspected ADHD, consideration should be given to the psychiatric status of the parent as maternal depression may have an impact on the mother's reporting of symptoms in her child. (Grade D)
- 41. There are insufficient research data to determine whether ADHD in the parent has an impact on their reporting of ADHD symptoms in their children. (Grade D)

Background

ICD-10 requires information from a second respondent, most often a teacher, when diagnosing ADHD in children and adolescents. DSM-IV does not, but it does ask the primary informant about whether the symptoms are present in a second situation such as school or, for adults, work.

The question of how to combine information from different sources remains unresolved, and the difficulties are exacerbated for children in secondary school, where there are multiple teachers. In some cases, teachers may be unwilling to provide information, or principals to allow information to be provided. With adults, it may be difficult to ask for information from employers or work colleagues. There are also sufficient data to show that teachers may assess a young person's behaviour on grounds other than just ADHD. For example, the teacher may also consider the child's academic performance or behaviour relative to the child's peers.

There is no consensus on how to integrate information from parents/caregivers and teachers. There is some argument in the UK to allow for an "or" decision, so symptoms reported by the parent/caregiver *or* the teacher count towards the final symptom number. This could lead to over-diagnosis; however an "and" approach might be equally unacceptable as it is restrictive and could result in under-diagnosis. Consequently, when parent *and* teacher information gives clear evidence of impairment in two settings, then diagnostic criteria for ADHD are obviously met. If impairment is seen in only one setting, clinical judgement is required to determine whether a diagnosis of ADHD should be made. The clinician could consider what level of functional impairment is present and whether any symptoms are seen in other domains.

It may be more appropriate to view the information from additional informants as a way of confirming the pervasiveness of impairment. The SKAMP Rating Scale (see Appendix H) provides a quick way for a teacher to indicate how a young person with ADHD is managing in the classroom.

A preferred instrument for gathering data from teachers is the Conners' Teacher Rating Scale – Revised (320, 321). This is formally classed by the Australian provider, the Australian Council of Educational Research, as a "specialised" measure only available to psychologists with advanced training in psychometrics. However, it is such a key measure for ADHD information from schools that it is now available to paediatricians and psychiatrists who can demonstrate sufficient experience in working with young people with ADHD.

Summary of research evidence

Agreement between parents and teachers on ADHD symptoms and on the identification of subtypes has been found to be low in preschoolers (298, 299) and children (266, 299-308). Only one study (305) included adolescents, with participants aged 5–17, but data were not presented separately and it is not possible to draw conclusions on parent-teacher agreement for adolescents.

The identified studies used a variety of different ADHD rating scales, including both narrow band and broadbrand scales. The majority of studies were in the USA, with the others carried out in Spain (299), Belgium (303), the Netherlands (308) and

Australia (300), providing some support that the low agreement between parent and teacher ratings holds true across different cultures. Many of the studies failed to include a measure of internal consistency to enable confidence in the parent and teacher ratings themselves before comparing them.

Three studies with adolescents (309-311) all demonstrated low to moderate agreement between self-reports and parent reports.

Six out of seven studies showed low agreement between self-reports and thirdparty reports (parents, partners, friends, family members) using a variety of rating scales (294, 296, 312-315). The seventh study found good agreement between self-reports and third-party reports (parents, partners, friends) (297). One possible reason for the difference is that participants in this last study (297) were recruited from the general community, rather than clinic samples as in the other six studies.

Discrepancies between self-report and third-party informants may also be influenced by factors such as conflict between the two over behaviour, communication problems or bias in partner choice (313).

Four studies were included that addressed the impact of maternal depression on maternal reports of child symptoms (308, 317-319). The studies used two rating scales: the CBCL, which is a broadband rating scale designed to screen for a variety of symptoms, including ADHD symptoms; and the DBDRS, which is more specific to ADHD and considers symptoms of inattention, hyperactivity/impulsivity and CD/ODD behaviour.

Using the CBCL, mothers with a history of depression reported significantly more symptoms in the child compared to mothers with no history of depression (317-319). Using the DBDRS, a history of depression in the mother was not significantly associated with maternal reports of symptoms (308, 319). One of the studies using the DBDRS (308) found that high levels of parent stress were significantly related to parents reporting higher levels of inattention, hyperactivity/impulsivity and ODD behaviour, compared to teachers' reports of this behaviour.

One study of children with ADHD found that parental ADHD did not have an impact on maternal reports of ADHD symptoms (316).

5.11 Specific populations

5.11.1 Culturally and linguistically diverse populations

Recommendation

42. Cultural differences in perceived levels of impulsivity, hyperactivity and inattention should be considered in the diagnosis of ADHD.
 ✓ Recommended best practice based on clinical experience and expert opinion

While there is growing evidence that rates of ADHD are similar throughout the world, most data come from widely used "broad-based" assessments such as the CBCL, where there are only modest differences both in prevalence and in the factor structure, a measure that the tests are assessing the same constructs in different populations (322, 323).

A study has, however, shown significant differences in tolerance of ADHD symptoms between US and New Zealand teachers (324). The most extensive work on differences in how symptoms are perceived comes from a study in which Asian and Western mental health staff rated the same children based on videotapes (325). Asian staff were much more likely than Western staff to rate behaviours such as distractibility and fidgeting as signs of ADHD. As clinical thresholds can be socially and culturally influenced, it is important to consider each individual's level of functioning in relation to his or her usual social and cultural environment.

It is important that information on ADHD and related conditions is available in languages other than English.

5.11.2 Indigenous Australians

Recommendation

43. There is need for a culturally appropriate assessment of ADHD in Indigenous people.

✓ Recommended best practice based on clinical experience and expert opinion

No adequate data were identified on the incidence of ADHD in the Australian Indigenous population. The "broadband" Strengths and Difficulties Questionnaire was modified for use in the Western Australian Aboriginal Child Health Survey and, while it does have a hyperactivity scale, this is far removed from the DSM-IV standards for ADHD. There has been one study of Canadian Indigenous people using the standard version of the Conners' Rating Scale (326). While a slightly elevated rate of ADHD was found, the authors could not determine whether this was a real difference, an artefact of the inappropriateness of the scale or differences in what are acceptable behaviours. The last may apply in Australia, where, for example, it is culturally appropriate for Indigenous children to move around the classroom to check with each other (T Westerman, Indigenous Psychological Services, personal communication). Thus it is not a symptom of hyperactivity.

Given the high rate of suicide in Australia's Indigenous population and the association of impulsivity with suicidal ideation among Indigenous youth (Westerman, unpublished PhD thesis), there is an urgent need for culturally appropriate assessment of ADHD. The issue is compounded by two related issues:

- The recent Western Australian study showed differences in stimulant prescription rates for ADHD based on differences in socio-economic status and remoteness (327).
- People with ADHD are over-represented in the justice system (see Chapter 16. ADHD and Justice, page 195), and rates of incarceration are high among Indigenous peoples (328).

Any planning for ADHD assessment services for Indigenous Australians must be done with appropriate consultation with Indigenous elders, so that ADHD is not added to the list of problems used to stigmatise Indigenous populations. Assessment of ADHD in Aboriginal children should be performed through the Aboriginal Medical Service and involve Aboriginal healthcare workers (329).

5.11.3 The family with ADHD

The substantial genetic component to ADHD (64) (see section 2.1 Aetiology, page 14) and the disruption that ADHD can bring to a family (see Chapter 15. Issues for Families, Parents and Carers, page 183) mean that family functioning is frequently impaired. This can have significant implications for the contributions families can make to the management of ADHD. The General Scale of the Family Assessment, developed as part of the Ontario Child Health Study (330) and used extensively in Australia, including the Western Australian Child Health Survey, is provided in Appendix H.

5.12 Diagnostic tools

Laboratory measures such as neuropsychological testing, electrophysiological techniques and neuroimaging have been advocated to aid in the diagnosis of ADHD. It is important to consider how much is added to the accuracy of the diagnosis by

the incremental information generated by these measures (331). In some cases, some of these tests may help guide specific interventions.

5.12.1 Neuropsychological measures

Research question

• In individuals suspected of having ADHD, does the inclusion of neuropsychological measures in addition to DSM-IV/ICD-10 further inform assessment and diagnosis?

Evidence statements

- Fifteen studies were identified that addressed the use of individual neuropsychological measures for the diagnosis of ADHD compared to clinical diagnosis or rating scales (332-346). The resultant sensitivities and specificities were mixed (low, moderate and good). (Level III-3 diagnosis)
- Seven studies were identified that combined multiple neuropsychological measures in a testing battery for the diagnosis of ADHD (347-353). Each looked at different combinations of measures, and the resultant sensitivities and specificities were mixed (low, moderate and good). (Level III-3 – diagnosis)
- No studies met the inclusion criteria that specifically addressed the use of neuropsychological measures alongside DSM-IV/ICD10 to further inform diagnosis.

Recommendations

- 44. There are insufficient research data to recommend the inclusion of neuropsychological measures in routine diagnostic assessment for ADHD. (Grade D)
- 45. Where functional difficulties persist in the context of ADHD (e.g. learning problems, organisational difficulties), despite initial educational, pharmacological and psychological strategies, neuropsychological assessment can provide information on cognitive strengths and weaknesses. This information can then be used by the person with ADHD, the teacher and/or parent/caregiver to develop compensatory approaches to learning and daily functions.

Recommended best practice based on clinical experience and expert opinion

Background

Specified batteries of neuropsychological tests have been proposed as a basis for diagnosis of ADHD, with particular emphasis on tests of executive functioning and working memory. A number of laboratory tests of attention and executive control, such as the Continuous Performance Task (CPT), have been investigated in the diagnosis of ADHD. The Conners' Continuous Performance Test is quite sensitive to central nervous system dysfunction. A number of disorders can result in impaired performance on the CPT. Whether such testing has sufficient specificity and utility in diagnosing ADHD is open to question. The utility of available data is complicated by such measures usually being included with diagnostic assessment, so the additional information they provide is difficult to ascertain. A reliance on the CPT as a primary diagnostic tool in determining the presence of ADHD could result in an unacceptably high number of false positives (i.e. over-diagnosis of ADHD). The CPT provides a quick standardised method of assessing attention and executive control. It shows promise in the monitoring of treatment response of those with ADHD; however, further research is needed on its utility in this area.

To date, studies examining neuropsychological tests of executive functioning have not been able to demonstrate positive predictive power, and especially negative predictive power, that is sufficiently high to recommend the use of these tests in clinical settings with children or adults. Neuropsychological testing has a value in the assessment of learning disabilities and other cognitive impairments but, despite widespread use, has no demonstrated value in the diagnosis of ADHD.

Experts consulted as part of the DSM-V Externalising Disorders Research Planning Conference (354) emphasised four key issues that apply to psychological, neuropsychological and also to the neurophysiological measures discussed below (page 57):

- Siblings of people with ADHD may share some of the same neuropsychological and neurophysiological impairments, so these difficulties are not diagnostic.
- There is growing recognition of heterogeneity, so that many people with ADHD will have no impairment on any of these measures.
- Sensitivity and specificity are very variable.
- Effects can be confounded by comorbidity.

Some studies have attempted to define ADHD subtypes (combined, predominately inattentive and hyperactive/impulsive) on the basis of neuropsychological patterns of performance. Neuropsychological distinctions between ADHD subtypes are few and highly dependent on which domain of executive functioning is assessed. The combined type has been associated with greater difficulty with inhibitory control (355-358), whereas the predominantly inattentive subtype has been associated with a greater deficit in processing speed on visual-motor or visual search tasks (355, 358, 359).

Notably, some studies (e.g. (355, 359)) have found similar patterns of neuropsychological impairment across all three subtype groups. Chhabildas et al (359) identified symptoms of inattention as the best predictor of performance across all neuropsychological measures and ADHD types. Consequently, individuals with symptoms of inattention may be more likely to warrant neuropsychological assessment.

Summary of research evidence

Fifteen studies were identified that addressed the use of individual neuropsychological measures for the diagnosis of ADHD compared to clinical diagnosis or rating scales (332-346). Both the sensitivities and specificities and the positive/negative predictive values found were mixed (low, moderate and good).

Among these studies were two systematic reviews which looked at the ability of neuropsychological measures to discriminate between individuals with ADHD and other clinical groups (339, 340). Review of use of the Wisconsin Card Sorting Test (WCST) (339) found that, compared to controls, groups with ADHD consistently performed poorly on this test. However, impairment on the WCST was not specific to ADHD, as other clinical groups such as those with learning disability were also impaired. The second review (340) found that the Stroop Colour Word Test could discriminate between individuals with ADHD and controls, but did not consistently discriminate between individuals with ADHD and other clinical groups.

Seven studies looked at combining multiple neuropsychological measures in a testing battery for the diagnosis of ADHD in preschoolers (347), children (348-351), adolescents (348) and adults (352, 353). Each looked at a different combination of measures, and the sensitivities and specificities found were mixed (low, moderate and good). Of particular note, in the two studies of adults (352, 353) the neuropsychological tests were examined in individuals with ADHD and individuals with other psychiatric disorders, and both studies found the ability of the tests to distinguish between the two groups was poor.

The majority of studies used clinic-referred samples; however, one (350) took a random sample of participants from a general school population. In this large and methodologically robust study, the neuropsychological tests could distinguish between ADHD and controls without ADHD at the group level, but at the individual level needed for diagnosis, specificity and sensitivity were low to moderate (48% and 72% respectively).

Overall, while the use of individual neuropsychological measures can discriminate between those with ADHD and those without ADHD at the group level, they do not perform consistently well at the individual level. The combination of neuropsychological measures appears to increase the diagnostic utility of these tests, but not to a reliable level. Importantly, both individual and combined measures appear to be very poor at distinguishing between individuals with ADHD and individuals with other psychiatric disorders.

5.12.2 Neurophysiological measures

Research question

• In individuals suspected of having ADHD, does the inclusion of neurophysiological techniques, in addition to DSM-IV/ICD-10, further inform assessment and diagnosis?

Evidence statements

- One study found that event-related potential (ERP) data distinguished between children and adolescents with and without ADHD with moderate accuracy when compared to clinical diagnosis (360). (Level III-3 diagnosis)
- Six studies found that electro-encephalograph (EEG data) classified children, adolescents and adults with and without ADHD with moderate to good accuracy when compared to clinical diagnosis (361-366). (Level III-1, III-2, III-3 – diagnosis)
- No studies met the inclusion criteria that addressed the use of neurophysiological measures alongside DSM-IV/ICD-10 to inform diagnosis further.

Recommendation

46. There are insufficient research data to recommend the inclusion of neurophysiological measures in routine diagnostic assessment for ADHD. (Grade D)

Background

Electrophysiological measures such as event-related potentials (ERP) and electroencephalography (EEG) have been used to study brain process in children with ADHD for more than 3 decades (first reviewed in (367)). EEG measures the electrical activity of the brain, allowing researchers to map the level of electrical activity occurring in certain brain regions. ERPs provide information on electrical activity taking place in the brain in response to stimuli.

Electrophysiological measures as a diagnostic tool

The most consistent finding in the ADHD electrophysiological literature has been that clinical groups of children and adolescents with ADHD demonstrate increased low-frequency activity, primarily theta (or slow wave) power, when compared to healthy control groups (for review see (368-370)). The increase in theta activity has also been reported in adults with ADHD (371). The question remains, however, whether electrophysiological measures can be used as a diagnostic tool to identify correctly those individuals with ADHD and those without. It is also important to

consider whether EEG/ERP measures can discriminate between individuals with ADHD and individuals with other psychiatric disorders, and what impact the conditions that are comorbid with ADHD will have on EEG/ERP data. These questions were addressed in the systematic review and are discussed below.

Electrophysiological measures and response to stimulant medication

It has been asserted in the literature that EEG measures may be used to differentiate between individuals with ADHD who will respond well to stimulant medication and non-responders (for review see (370)). Children with ADHD who respond to medication have been reported to have specific EEG findings, in particular, excessive slow wave activity, which suggests that they are more cortically hypo-aroused (372). Further research and validation are required to establish the utility of these findings. In addition, stimulant medication has been reported to alter the EEG patterns and ERPs of children with ADHD such that the findings are similar to those of children without ADHD (reviewed in (370)). Although a number of studies have reported the correct identification of 70-80% of stimulant responders when using either electrophysiological measures alone or electrophysiological measures plus behavioural measures (373-375), it has been observed that this figure is no better than estimates solely based on clinical trials of stimulant medication which report that ~70% of children will respond to medication (370).

Summary of research evidence

One study, conducted in Australia, investigated the use of ERP data to classify children and adolescents relative to a clinical diagnosis of ADHD (360). For children aged 8–12, sensitivity was 71.4% and specificity 76.9%, but the values for adolescents aged 13–18 years were considerably lower (sensitivity 56.9%, specificity 62.5%).

Six studies investigated the utility of EEG measures in the diagnosis and assessment of ADHD in children, adolescents and adults (361-366). The studies varied in the actual EEG measures used and the components of the clinical diagnosis used as the reference test for comparison.

Four studies used theta-beta ratio as the EEG measure:

- One described a QEEG-derived attentional index that was based on mean thetabeta power ratio across 4 tasks (eyes open resting, reading, listening and drawing) (362). This yielded a sensitivity of 86% and specificity of 98% when tested in people aged 6–30 years against clinical interview and various behaviour scales and continuous performance tasks.
- The same researchers repeated this study in participants aged 6–20 years (363) and found a sensitivity of 90% and a specificity of 94%.
- The EEG theta-beta ratio yielded a sensitivity of 50% and specificity of 36% when measured against diagnosis based on an 18-item DSM-IV-based ADHD scale that is part of the parent-rated Coolidge Personality and Neuropsychology Inventory (361).
- The theta-beta ratio was compared to clinical diagnosis in people aged 6-21 years referred to a clinic for their ADHD symptoms (366). Of 26 participants, 16 were clinically diagnosed with ADHD and 10 with no ADHD. The sensitivity of the EEG measure was 94% and the specificity 100%. The accuracy of the ADHD Rating Scale-IV was included as a comparator in this study and a sensitivity of 81% and specificity of 22% were reported.

One study used a specific set of EEG measures to provide a "consistency index" (CI) and compared this with clinical diagnosis (365). Across all the participants, aged 6–25 years, the CI measure correctly classified 82% of the participants with

ADHD and 77% of controls. Among participants aged less than 16 years, overall accuracy increased from 80 to 90%.

One study found that total, absolute and relative power EEG measures in eyesclosed conditions yielded a sensitivity of 89.0% and specificity of 79.6% compared to clinical diagnosis (364).

Five studies were identified that showed that the presence of comorbidities (tics, learning disabilities, ODD and CD) does have an impact on EEG and ERP data in children with ADHD (376-380). No studies addressing the ability of ERP or EEG to discriminate between ADHD and other psychiatric disorders met the inclusion criteria.

Overall, these studies report good sensitivity (90–100%) and specificity (84–94%) for electrophysiological measures. However, the possibility of misdiagnosis when using these measures alone remains unacceptably high and there are several limitations in the research to date that must be addressed. In particular, there are few data on test-retest and inter-rater reliability, and rarely have studies been undertaken using blinded procedures (381). More recent studies (e.g. (366)) have begun to address these issues, but additional replication is required. Further, as it is clear that EEG and ERP profiles can vary when comorbidities are present, careful consideration must be given to how this will affect the ability to discriminate between individuals with ADHD and those without.

5.12.3 Neuroimaging

Research question

• In individuals suspected of having ADHD, does the inclusion of neuroimaging techniques, in addition to DSM-IV/ICD-10, further inform assessment and diagnosis?

Evidence statements

- Two meta-analyses have demonstrated the ability of neuroimaging techniques to distinguish between individuals with ADHD and those without ADHD at a group level (121, 122). (Level III-2 diagnosis)
- No studies met the inclusion criteria that address the use of neuroimaging techniques for the diagnosis of individuals.

Recommendation

47. There are insufficient research data to recommend the inclusion of neuroimaging in routine diagnostic assessment for ADHD. (Grade D)

Background

The primary neuroimaging techniques are magnetic resonance imaging (MRI), functional magnetic resonance imaging (fMRI), diffusion tensor imaging (DTI), positron emission tomography (PET) and single photon emission computed tomography (SPECT). Structural neuroimaging (MRI) can be used to assess differences in brain structure, such as total brain volume or the size of specific regions of the brain. Functional neuroimaging techniques (fMRI, SPECT and PET) can also look at differences in regional cerebral blood flow. For a review of these techniques and how they have been used in the research of ADHD see Bush et al 2005 (382).

To date, the research into the use of these techniques in relation to ADHD has been limited to comparisons of imaging data averaged across a group of individuals (e.g. a group with ADHD versus a group without ADHD). To be useful for diagnosis, it must be possible to differentiate between those with and without ADHD at the

individual level. Neuroimaging techniques are not yet robust enough to be used for diagnosis.

Summary of research evidence

Two meta-analyses have addressed neuroimaging for individuals with ADHD:

- One meta-analysis included 22 MRI studies in children and young adolescents (mean age 11 years) with and without ADHD (122). The inclusion of all brain regions measured across all studies found a significant difference between the participants with ADHD and the healthy controls. The heterogeneity across the regions assessed and the measures used was significant. The researchers also looked at individual brain areas and identified several regions that were significantly different between the ADHD and the control groups or in at least three studies. As the authors note, one of the key limitations of the research is that only children and young adolescents were studied, with no participants over 15 years. Only one small study was conducted in adults (383). Similarly, the study population was primarily male, with only half the studies including females. Another important limitation of this work is the variability in the MRI measures, as brain regions were not identically defined across studies.
- A specific meta-analysis technique, activation likelihood estimation (ALE), was used to compare brain activation between healthy controls and individuals with ADHD (121). The included studies used fMRI or PET and were conducted primarily in children and adolescents, with five studies in adults. A consistent pattern of decreased frontal activity was identified in the ADHD group compared to the control group. The limitations in research are similar to those for the MRI studies above. Variation in reporting and the inconsistent use of standardised measures make it difficult to compare the results of individual studies. Few studies have considered large numbers of females, and only 9 out of 16 make a direct comparison between the ADHD group and a control group.

No studies were identified addressing the positive and negative predictive power of brain imaging in diagnosing ADHD in children or adults in clinical settings. As such, the findings to date do not support the use of structural brain imaging in the diagnosis of ADHD.

PART III: MANAGEMENT



Key Messages Management of ADHD

- 3. Individuals with ADHD and their families and carers should be provided with information and education about ADHD and its impact, and the advantages and disadvantages of potential treatment strategies.
- 4. Multimodal therapy is recommended for the treatment of ADHD in all age groups. This could include psychosocial management strategies, medication and educational interventions.
- 5. An individualised management plan should be drawn up in collaboration with the person with ADHD and their parents/carers and teachers Taking into account
 - the specific needs and expressed preferences of the person, and the circumstances of his or her family and culture.
 - the associated psychosocial problems, educational difficulties and comorbid conditions.
 - the suitability of the plan for the individual and their family, considering affordability, accessibility and acceptability.
- 6. Clinicians should be alert to the risk of depression or other psychiatric disorders in parents/caregivers of children or adolescents with ADHD. Parents/caregivers may need referral for support and treatment.
- 7. Medication and ADHD
 - Not all people with ADHD will require pharmacological management.
 - Medications should only be used when symptoms are pervasive across settings (eg. school and home) and causing significant impairment in academic, social or behavioural function, and after careful consideration of non-pharmacological approaches. Clearly defined goals should be identified prior to commencing a trial of medication treatment.
 - Medication should not be used as first-line treatment for ADHD in preschool-aged children.
 - Patients receiving treatment for ADHD should be reviewed regularly (at least 6monthly) to ensure that the management strategies remain appropriate and effective.
- 8. Other therapies
 - Elimination and restriction diets are not supported as a general treatment for individuals with ADHD. A subset of children may be sensitive to certain foods or food additives and may benefit from careful exclusion diets. Assessment of food sensitivity and initiation of a special diet should be under the care and supervision of a medical specialist and an Accredited Practising Dietitian.
 - Alternative treatments (including fatty acid supplements, biofeedback, homeopathy or sensory integration diets) are not currently supported as treatments for individuals with ADHD.

6.1 Access to services

6.1.1 Services for children and adolescents

There are differences across Australia in the ways in which medical services for ADHD are provided. A range of public and private sector practitioners deliver behavioural and supportive therapy, but these services are frequently thought to be difficult to access and are very limited in the public sector for adults with ADHD. Access to stimulant medication is controlled by legislation in all Australian jurisdictions, with requirements for registration of public and private sector prescribers and of stimulant prescription recipients. The ability to initiate a stimulant prescription is in general limited to specialists in the fields of paediatrics and psychiatry. In most jurisdictions there is provision for a co-prescriber, typically a general practitioner, who can issue prescriptions when the registered prescriber is unavailable, or in some instances as routine after an initial period of observation, but with periodic specialist review. Currently, New South Wales and Western Australia are the only two Australian jurisdictions with formal processes for auditing stimulant medication. Uniform national regulations should be established for the prescribing of stimulant medications. In addition, a uniform national system of recording prescribing information should be adopted.

Use of services

There are limited published data on use of services for the management of ADHD in Australia. Sawyer et al (234) studied a subset of the children and adolescents recruited for the National Survey of Mental Health and Wellbeing in Australia. This study was a representative, population-based survey that assessed the pattern and prevalence of ADHD symptoms, the type of help provided to these children and the factors associated with the use of services, including barriers to service access (234). Sawyer et al reported that 28.1% of children with ADHD symptoms had attended a health- or school-based service in the 6 months prior to the study (234). Of these children, 41.1% had attended both health and school services, 39.3% had attended only health services and 19.6% only school services. The most commonly utilised health services were paediatricians, followed by family doctors, and 12.6% had attended a mental health service (234).

Similarly, a 2005 school-based community survey of North Sydney found that the diagnosis of ADHD was made by a paediatrician in 73% of cases, by a psychiatrist in 9%, by a psychologist in 8% and by a combination of these in the remaining 10% (384). The distribution of service providers probably varies in different settings within Australia. In Western Australia, for children and adolescents under the age of 18 years, over 90% of those prescribed stimulants were treated by paediatricians and 8.2% by psychiatrists (385).

Sawyer et al found that counselling was the most frequently provided service by school services and by health services. Eighteen percent of children with ADHD symptoms were being treated with medication (stimulants, antidepressants and antipsychotics). Of those who had attended a health service in the preceding 6 months, 58.9% were receiving medication, comprising 25.6% receiving only medication and 33.3% receiving medication and counselling (234).

6.1.2 Transition of care from adolescent services to adult services

There are few examples within Australia of functioning transition programs for adolescents with ADHD moving into adult sector services. This is despite acknowledgement in the literature that this is a demanding time for young people with ADHD and wider acceptance that adolescents with ADHD need the types of supports recommended for all young people with a chronic condition (386-388). International experience confirms that management of ADHD during this transition period is important if young people with ADHD are to achieve optimal educational outcomes (387, 389). The lack of public sector services for adults with ADHD further compounds the difficulty of effecting a smooth transition between child and adult services. The approach taken in Western Australia is to have adolescent services available to individuals with ADHD up until the age of 25.

6.1.3 Services for adults

For adults, the assessment of ADHD and any initial prescribing of psychostimulants is limited to psychiatrists. Where the person has received treatment for ADHD before the age of 18, a neurologist or paediatrician may also have this role.

There are fewer Australian data on treatment of adults with ADHD than for children. For adults (>18 years of age) with ADHD in Western Australia during 2006, 49.3% of patients prescribed stimulants were treated by child and adolescent psychiatrists, 38.1% by adult psychiatrists and 12.1% by paediatricians. Overall, there were equal numbers of child and adult patients (385). The opportunity for a co-prescriber has proven especially important for patients living in rural and remote settings (385).

A US survey suggests that the majority of adult patients with ADHD are diagnosed outside the paediatric period and hence are likely to need specialist assessment services and then long-term follow-up (390). In Australia, the majority of services for adults exist within the private sector, with limited adult public mental health services willing or able to take on the management of these clients.

6.1.4 Differences in medication availability and usage across Australia

There are a number of differences in the systems governing regulation of stimulant prescriptions across Australia, including differences in the age criteria for prescription, the presence of criteria that allow prescription in special circumstances, the requirements for notification and the requirements for screening for drug use.

There are marked variations in the use of prescription medications for ADHD across Australia (391). In 2003 the rate of dispensed dexamphetamine (DEX) prescriptions ranged from 3.6 per 1,000 population in the Northern Territory to 44.2 per 1,000 population in Western Australia. Rates of DEX prescriptions dispensed under the PBS between 2001 and 2003 showed high rates of stimulant prescribing in outer metropolitan areas and low rates of prescription in rural Western Australia (391). The reasons for these differences are complex, but include lack of recognised expertise in some geographical areas, lack of availability of services in general and in some places lack of alternatives to medication prescription.

6.1.5 Predictors of referral for assessment of ADHD

Children are more likely to attend services if they have the ADHD combined subtype than if they have inattentive or hyperactive/impulsive symptoms (234). Other factors associated with attendance at school or health services are parental perception of need and the presence of comorbid symptoms (234).

6.1.6 Barriers to service access

Sawyer et al's study (discussed in section 6.1.1) found that, of the parents of children with ADHD symptoms who did not attend services, 20.3% reported that they felt they could manage their problems on their own, 14.7% claimed the cost of services was excessive, 14% did not know where to get help and 10.5% were inhibited by the length of waiting lists for treatment (234).

Indigenous Australians

There are differences in the rates of presentation of ADHD to medical services in Indigenous Australians compared to the rest of the population. In North Queensland (392) ADHD was diagnosed in only 0.19% of the Indigenous people assessed at rural outreach clinics, compared to 3.51% of those assessed from non-Indigenous communities, an 18-fold difference (392). This, however, does not reflect the real prevalence of symptoms of ADHD in Aboriginal and Torres Strait Islander children and adolescents. The Western Australian Aboriginal Child Health Survey identified that 15.3% of a broad community survey of Indigenous 4–17 year olds were at high risk of clinically significant hyperactivity, while a further 9.3% were at moderate risk (393). These figures were higher than in the non-Aboriginal population where 9.7% were at high risk of clinically significant hyperactivity. The risk was greatest in children aged 4–11 years, with 17.1% of this cohort identified as being at high risk of hyperactivity, and in males (18.1% compared with females 12.3%) (393). The degree to which hyperactivity is perceived as a problem in Aboriginal communities is not clear from these data. Symptoms of hyperactivity were reported more commonly amongst urban children in comparison with remote children (18.8% of children in Perth, 5.2% of children in extreme isolation) (393). There may be a number of explanations for this discrepancy. For example, the prevalence of ADHD between the communities may be genuinely different. Alternatively, some Indigenous communities may have different beliefs about what constitutes normal behaviour and attention span, and to what extent any differences constitute a problem.

There is a need for improved understanding of how to deliver effective mental health services in Indigenous communities as well as for an increase in the mental health services provided. As noted in section 5.10.2 (page 54), as well as a need for the development of culturally acceptable assessment procedures, there is a need to tailor services so that they can be delivered through programs acceptable to Aboriginal communities, such as Aboriginal Medical Services, with elder endorsement.

Australians from culturally and linguistically diverse backgrounds

The rates of presentation of ADHD in populations throughout the world may not reflect the rate at which migrants present with ADHD in Australia. This may be due to language difficulties, migration stresses, and a lack of familiarity with Australian systems and behavioural norms. Research is needed to determine the prevalence of ADHD amongst CALD populations in Australia. It is also important that we improve our understanding of how to deliver effective mental health services for Australians from CALD backgrounds.

Place of residence and socio-economic status

Location of residence and socio-economic status influence access to services and the nature of services used, with rural residence being a particular disadvantage for individuals with ADHD and their families. The low rates of referral in rural Aboriginal communities is noted above. In Western Australia, people with ADHD living in major centres are 2.3 to 5.3 times more likely to be prescribed stimulant medications, compared to residents of remote areas (327). This is likely to reflect difficulty in accessing a paediatric or mental health practitioner. In some parts of the country there are no resident paediatric services, so that many communities are dependent upon visiting paediatricians. In Western Australia, socio-economic advantage is associated with a lower likelihood of use of stimulant medication for children, but an increased likelihood of stimulant prescription for adults (327). Geographic differences in rates of prescription have also been observed in South Australia (394, 395) where socio-economic disadvantage was associated with increased likelihood of children receiving stimulant medication (394, 395). Studies in Australia and the US indicate that the prevalence of ADHD is increased in families with social disadvantage (7, 396, 397).

In a Florida study, female gender, non-white status and rural residence were associated with lower probability of ADHD service use (398). The same group also identified poverty as a barrier to service utilisation (399).

Failure to recognise symptoms

Failure to recognise the nature of ADHD, by either parents or professionals, has been recognised as a service access barrier in Australia and elsewhere. Australian general practitioners have reported limitations to their willingness to be involved in aspects of the care of children with ADHD (248). A United Kingdom survey of children identified through the British Child and Adolescent Mental Health Survey found that, while 80% of parents recognised that their child had a problem, only 35% identified the problem as hyperactivity (235). Parents were more likely to seek educational services, and hence did not gain access to medical services (400). General practitioners' failure to recognise ADHD was identified as the main barrier to referral (400).

Service availability for children and adolescents

An important issue in Australia is the availability of comprehensive, multidisciplinary services for children with ADHD and their families. In Sawyer et al's study (234), parents of 68.5% of the children attending health services wanted more help, most commonly counselling. Of the group wanting more help, 53.6% of the parents requested additional counselling for their children, and 57.1% requested additional counselling for themselves (234). Analysis of data from the Western Australian Stimulant Register indicates that, despite the availability of some public sector clinics, the majority of prescriptions for stimulant medication are still generated by private practitioners.

The Western Australian Parliamentary Inquiry into ADHD (401) concluded that access to mental health services was a major issue, stating:

The current arrangements in terms of state services lean toward the medication first model, largely due to difficulty of access to appropriate multidisciplinary treatment.

In terms of education support, families cannot easily access services for children diagnosed with ADHD unless the child is also identified as having one of the major categories of disability. Provision is not specifically provided for a single diagnosis of ADHD.

They also identified the difficulty in accessing educational initiatives:

The shortage of qualified health professionals who are approved to diagnose ADHD and to then administer stimulant medication was a common theme in submissions. This view was particularly evident in non-metropolitan submissions, where the shortage was thought to impact on the ability to gain a second opinion and to access the full range of treatment options. Often in regional areas, medication is seen as the only option for treatment. (401)

These issues are not unique to Australia. In the United Kingdom, attitudinal and skill barriers amongst primary carers are a potential limit to implementation of shared models of care (402). In the US, it is difficult to implement complex treatment guidelines in primary care, and there is a need for additional training of primary care physicians and improved access to resources if ADHD treatment guidelines are to be implemented (403, 404).

The Clinical Excellence Commission of the Government of New South Wales, in its 2007 report, identified on the basis of a questionnaire completed by clinicians treating individuals with ADHD major gaps in the services available in that state. Clinicians specifically identified deficiencies in educational support (62%), behavioural therapy (76%), family therapy (74%) and psychological services such as psychometric assessment (51%) (405).

Other barriers identified by health professionals in Australia, for children and their families, include a lack of consumer health education on ADHD, the stigma associated with seeking help and the duration of treatment.

Barriers for adults

Access is possibly more difficult for adult patients.

Some medications are only available under the Pharmaceutical Benefits Scheme to patients under the age of 18 years, so that adults with ADHD do not have ready access to medications other than DEX, or access is limited to sub-optimal dosage amounts.

In Western Australia it has been noted that not all adult psychiatrists are willing to treat patients with ADHD, so there may be excessive waiting times for treatment. These pressures are noted to lead sometimes to limited specialist review and excessive reliance on co-prescribers for follow-up.

There are also financial barriers because of the lack of public sector adult clinics and the resultant need to seek private treatment. In particular, the perceived high cost of non-medication treatment options in the private sector means that adults with ADHD may not pursue this option.

6.2 Requirements of ADHD services

The Australian geography and population distribution and the scarcity of health professionals in some of the disciplines relevant to ADHD provide challenges to health service administrators in developing workable models of care.

Important issues that have been identified as compromising access to services for ADHD in Australia are:

- failure of recognition of the features of ADHD by health professionals, including failure of recognition in Indigenous populations
- rural residence
- socio-economic disadvantage
- shortage of mental health professionals
- lack of services for adult ADHD
- lack of services for paediatric ADHD.

Considerations in planning services for people with ADHD include recognition of primary, secondary and tertiary tiers of care for both paediatric and adult services, acknowledgement that the competencies required for assessment and management of ADHD are not necessarily disciplinespecific so that there is potential for workforce substitution in some areas, and the need for enhanced undergraduate and postgraduate health professional education regarding the recognition and management of ADHD.

6.2.1 Education and training

An understanding of ADHD by all medical, nursing and allied health professionals is the basis for improving access to appropriate healthcare for people with ADHD. Staged training should be routinely available for these professionals, starting in the undergraduate period, so that all health professionals are able to identify children, adolescents and adults who may have ADHD, and all have access to extra training should they wish to provide more intensive management of the condition. Similar education needs exist for other professional staff in the fields of education, social work, justice and law to help ameliorate the disadvantages experienced by individuals with ADHD.

6.2.2 Child-oriented ADHD services: skills and competencies

Primary care providers

Primary care service providers for children (community health agencies and general practitioners) should be able to recognise the core symptoms of ADHD and be aware that these are not situation-specific but can occur in all environments. They should be familiar with symptom checklists as an aid to diagnosis. They need to be aware of differential diagnosis and the potential for comorbid conditions to occur in conjunction with ADHD. General practitioners as the gatekeepers to other service levels need to be aware of the requirements for therapy and of the locally available options.

Secondary care providers

Secondary service providers need detailed knowledge and understanding of normal childhood development and the ability to differentiate these sequences from symptoms of ADHD and other neuro-developmental disorders, such as specific learning disorders, co-ordination disorders and autism, and mental health disorders such as anxiety, depression and obsessive compulsive disorder. They also need to be able to determine the contribution to behavioural disorders of neurological problems such as epilepsy and foetal alcohol spectrum disorder, and of family functioning and socio-economic background (244).

Service providers at this level will initiate stimulant medication and will be able to manage the medications, including recognising and managing side effects. They will also be able to access the behavioural and other therapies that are required.

Tertiary care providers

Tertiary care providers of paediatric care will work in multidisciplinary teams, providing detailed diagnostic assessments that delineate the problems outlined in the paragraph above and multimodal therapies for children with disabling, difficult-to-manage ADHD.

6.2.3 Adult-oriented services: skills and competencies

Primary care providers

Primary care providers of adult services for ADHD need to be aware of the natural history of childhood ADHD and the differences in symptoms that adults may manifest compared to children. Familiarity with screening tools is important.

Secondary care providers

Secondary-level care providers of adult services for ADHD need to be able to utilise diagnostic instruments and to differentiate adult ADHD from mental health disorders such as anxiety, bipolar disorder and obsessive compulsive disorder. They will be able to manage stimulant medication and its consequences. Secondary-level providers require skill in working with adolescents in order to be able to manage the transition from paediatric to adult services. The competencies for secondary-

level healthcare providers of services for adults are most likely to have been acquired through training in psychiatry or clinical psychology, but are potentially achievable by general practitioners.

Tertiary care providers

Tertiary adult services require multidisciplinary diagnostic and therapeutic skills to deal with complex ADHD, often in the setting of comorbid diagnoses.

6.3 Developing an individualised management plan

Rec	Recommendations				
48.	An per ma	An individualised management plan should be drawn up in collaboration with person with ADHD and their parents/carers and teachers as appropriate. The management plan should take into account:			
	0	the specific needs and expressed preferences of the person and the circumstances of his or her family and culture			
	0	the associated psychosocial problems, educational difficulties and comorbid conditions			
,	∘ ✓ Reco	the suitability of the plan for the individual and their family, taking into account affordability, accessibility and acceptability.			
49.	 People with ADHD and their families and carers should be provided with information and education about ADHD and its impact, and the advantages and disadvantages of potential treatment strategies. <i>Kecommended best practice based on clinical experience and expert opinion</i> 				

ADHD is a persistent condition that requires ongoing management and monitoring in all age groups. A management plan for the care of a person with ADHD should be individualised and will vary from person to person. The choice of treatment strategy must be influenced by the specific needs and circumstances of the person with ADHD, their family and their culture. Consequently, a comprehensive assessment is essential.

In formulating an individualised management plan, the severity of the person's ADHD needs to be taken into account. In determining the severity, the symptoms of hyperactivity, impulsivity or inattention *and* the level of subsequent impairment in multiple settings must be assessed. Issues that should be considered include academic performance, self-esteem, personal distress from the symptoms, social interactions and relationships, behavioural problems, and the development of comorbid psychiatric syndromes.

In addition, there may be unique circumstances. For example, is the school considering suspension of the student unless there is a rapid improvement in their behaviour towards teachers or other students? Are the parents seriously considering separation because of their child's difficult behaviour? Or in the case of an adult with ADHD, is the person at risk of losing their employment as a result of disorganisation or disinhibited behaviour? After careful assessment by the clinician, such examples may be considered to be a severe impairment and may warrant the immediate start of medication for two reasons. Firstly, it may change the person's behaviour to the extent that behavioural intervention becomes more possible and more effective. Secondly, the consequences of not altering the behaviour immediately may be serious in terms of the impact of school suspension, the family separating or the adult losing their job, all of which would make subsequent behavioural intervention more difficult.

The individualised management plan should also take into account the person's global health and associated problems such as learning difficulties, peer relationships, low self-esteem and family problems. In addition, as described previously (section 3.2 Comorbidities, page 22), many individuals with ADHD will have comorbid disorders that may require specifically targeted management strategies. The management plan should be developed in accordance with the individual's/family's resources and their capacity to adhere to the plan.

Developing an effective management plan also involves educating the person with ADHD and his or her family and carers about the disorder and how it may affect behaviour, self-esteem, social skills, learning and functioning within the family. In addition, it is important to provide information on the advantages and disadvantages of potential treatment strategies. The provision of written materials can be particularly helpful, as can contact with local and national support groups (see Appendix I: Australian ADHD Support and Education Groups).

Families should be advised not to stop and start interventions without the advice of their managing clinician.

6.4 Multimodal therapy

Recommendation		
50.	A multimodal approach is recommended for treatment of ADHD. This may include medication, psychosocial management strategies and, where appropriate, educational interventions. Recommended best practice based on clinical experience and expert opinion 	

Multimodal therapy is advocated for the management of ADHD in all age groups. Multimodal treatment strategies typically combine medication and psychosocial management strategies such as behaviour therapy, psychoeducation, counselling or support. For children and adolescents, educational interventions that are designed to help with learning difficulties should also be included. Multimodal therapy has been endorsed in other best practice guidelines for ADHD (241, 243, 406, 407), however there is little research that specifically addresses the efficacy of this approach.

As discussed in Chapter 9 (see page 137), the Multimodal Treatment Study of ADHD (MTA) is the largest trial of combined medication and psychosocial treatment for ADHD (408). The MTA study compared four treatment groups: medication management, behavioural management, combined treatment, and community care. After 14 months, the effects of medication (methylphenidate [MPH]) were found to be equivalent to combined treatment, and both were more effective than behavioural management alone or community care (408). Importantly, the dose of MPH utilised in the combined group was significantly lower than the dose used in the medication-only group to achieve the same impact on ADHD symptoms (408). There was no significant difference between the treatment groups in the longer term (24 and 36 months); however, it is difficult to compare between groups because, as the study progressed, the participants were followed up according to their original treatment allocation, regardless of what treatment they subsequently chose to use (409, 410).

6.5 Multidisciplinary care

The management of ADHD may be complex and time consuming and may require extensive collaboration between disciplines. Many agencies and professionals may need to be involved in delivering and supporting an individual management plan.

For children and adolescents, effective multidisciplinary care requires co-operation amongst health professionals as well as effective consultation and liaison with families, schools, early childhood settings and support services. It is important to establish a therapeutic alliance with the child's parents/caregivers and other significant caregivers (e.g. day care providers, school teachers), to promote consistent management between home and school and consistent implementation of interventions. Thought and planning must go into the transition of adolescents into adult services and supports.

For adults with ADHD, there is a paucity of adult mental health or psychologist services in Australia. The majority of adults receiving treatment and care for ADHD do so through adult psychiatrists working in private practice. Psychological support should be available, targeted at the particular problems related to ADHD and comorbid conditions.

6.6 Treatment monitoring and review

Recommendation

- 51. Patients receiving treatment for ADHD should be reviewed regularly (at least 6-monthly) to ensure that the management strategies remain appropriate and effective.
 - ✓ Recommended best practice based on clinical experience and expert opinion

Regular review is important to ensure that the management strategies remain appropriate and effective. Frequency of review will depend on age, stage and complexity of treatment, as well as on educational, work and family factors. Review will be frequent early after initiation of therapy and less frequent with established therapy. At a minimum, patients receiving treatment for ADHD should be reviewed every 6 months by the physician initiating drug therapy. Comprehensive assessment is essential and information should be collected from a range of sources.

Children and adolescents

Many factors can influence symptoms in children and adolescents, such as normal developmental changes, greater academic demands, changes in the home environment and availability of resources. Review should cover medication and other interventions, educational progress, and behaviour in the home and other settings. Information should be sought from multiple sources (e.g. parents or other significant caregivers, teachers).

Adults

Factors that influence symptoms in adults can include changes in the home and family environment, marriage or a new partner, children, loss or change in employment and returning to study. Review should cover medication and other interventions, and work and home life. Again, information should be sought from multiple sources (e.g. partners, other family members).

6.7 Family support / intervention

Recommendations

52. Participation in support groups can be of benefit for parents/carers of children and adolescents with ADHD.

✓ Recommended best practice based on clinical experience and expert opinion

- 53. Participation in adult ADHD support groups can be of benefit to adults with ADHD.
 - ✓ Recommended best practice based on clinical experience and expert opinion

Relationship and family issues are well documented for children with a diagnosis of ADHD and their families (see Chapter 15. Issues for Families, Parents and Carers, page 183). Parents/caregivers often feel that they are unable to manage the complexity of their child's difficulties and this places a strain on the parents themselves as well as the family members and siblings who do not have ADHD. Parents of all children with ADHD require education regarding the nature of ADHD, the progression of symptoms across the lifespan, the ways in which they can access services and support, and the need for parental self-care. This support and education should be initiated by the clinician when developing an individualised management plan.

Local support groups can provide peer support and an opportunity for parents to exchange experiences and advice about caring for a child or young person with ADHD on a day-to-day basis. Support groups can also let families know what services are available to them. For example, families may benefit from services such as respite programs which allow the child with ADHD and the parent/carer to have a short-term break from one another.

Adults with ADHD and their partners may benefit from having written information about ADHD, the services and support offered by local and national support groups, and other sources of advice and advocacy.

Contact details for local support groups should be provided by professionals to families. A list of ADHD support groups is provided in Appendix I.

7.1 Introduction

Psychosocial interventions used in the management of ADHD include cognitive behavioural therapy (CBT), behaviour modification and social skills training. The intervention used will depend on the specific needs of the person with ADHD and his or her family. The presentation of ADHD and the cognitive and psychosocial strengths and weaknesses of people with ADHD vary widely, and treatment planning always needs to be individualised.

As well as addressing the core ADHD symptoms of inattention, hyperactivity and impulsivity, psychosocial interventions for people with ADHD may focus on improving day-to-day functioning at home and at school or work. While many psychosocial interventions are directed solely at the individual with ADHD, interventions such as parenting programs and family therapy require the direct involvement of the parents/caregivers of children with ADHD and help parents/caregivers to develop strategies to cope with and manage their child's difficult behaviour. It is important to note that there is no research support for the use of traditional psychotherapeutic approaches for the treatment of ADHD (411).

Prior to recommending a particular psychosocial intervention, consideration should be given to the current needs of individual families and their ability to implement psychosocial management strategies. This is particularly important for families of children with ADHD who are considering parenting programs. As discussed in Chapter 14 Issues for Families, Parents and Carers, page 183, the parents of children with ADHD often face increased levels of stress, low parenting self-efficacy, low self-esteem, marital discord and depression. In many cases, practical supports for families such as respite, educational support or in-home support may be necessary before psychosocial management is implemented (412, 413).

7.1.1 Psychosocial interventions for preschoolers, school-aged children and adolescents

Psychoeducation

Psychoeducation regarding the nature, causes and natural history of ADHD is important in aiding parents and carers to cope with the behaviours associated with ADHD, to understand and implement management plans, and to anticipate difficulties.

Behaviour modification

Behaviour modification (also known as contingency management) uses strategies such as structured reward systems, response costs and discipline techniques to encourage behaviour change. Rewards can include social rewards such as approval and praise, concrete rewards such as extra recreation time, or more complex reward schemes that involve "tokens" (stars or points) that can be earned. Discipline techniques can include verbal reprimands or time-out from positive reinforcement strategies. The time-out strategies involve removing the individual from a reinforcing situation for a set period of time following inappropriate behaviour. Response-cost techniques involve the loss of a reward as a result of inappropriate behaviour. This can be a loss of earned rewards or a loss of rewards from an agreed set that have been given in advance.

Cognitive behavioural therapy

CBT combines cognitive therapy, which aims to modify or eliminate unwanted thoughts or beliefs, and behavioural therapy, which aims to bring about changes in behaviour. The goal of CBT is to enhance functioning at home and at school. Individual programs may focus on one or more of the following: problem-solving techniques, coping strategies, social skills, goal-directed approaches to tasks or cognitive restructuring. CBT can be conducted individually or within a group.

Social skills training

Children with ADHD can experience difficulties in their social competence, peer relationships and family interactions. The aim of social skills training is to develop and reinforce the use of appropriate social skills (414). Social skills training uses techniques from cognitive and behavioural approaches;

the actual content and procedures vary across programs. It is usually conducted within groups, in either an educational or a clinic setting.

Parenting programs

Parenting programs (often called parent-training programs) aim to teach parents/caregivers strategies for managing disruptive behaviour in their child and improving parent-child relationships. Parenting programs make use of behaviour modification techniques such as structured reward systems and discipline techniques. They also commonly involve cognitive behavioural techniques, which require identifying problem behaviours, analysing their cause, developing a consistent response and modifying it on the basis of feedback. Several different strategies are used in parenting programs, and they can be conducted with individual families or in group settings. Parenting programs have been developed as an intervention for ADHD and other disorders that have an impact on behaviour, such as ODD and CD. Parent-based approaches for treatment of ADHD are most successful in younger children, with little research support for this approach with adolescents (411).

Family therapy

The goal of family therapy is to bring about positive changes in the way families function. Family therapy can be used to help families learn how best to support a child with ADHD and how to avoid parenting practices and parent-child interactions that may intensify problems. The focus of family therapy may be conflict resolution, effective communication, reducing anger in family interactions, problem-solving strategies, or developing clear roles, rules and routines (415).

7.1.2 Psychosocial interventions for adults

Cognitive behavioural therapy

CBT for adults with ADHD is primarily focused on developing strategies to resolve the skills deficits and day-to-day problems associated with ADHD, such as low self-esteem, disorganisation and procrastination (416). CBT programs vary dramatically, but generally centre on teaching problemsolving skills, techniques to reduce distractibility and stress management skills. CBT can be delivered in individual or group formats or may be self-directed.

Psychoeducation

Psychoeducation is thought to be particularly important for individuals who have not received their diagnosis of ADHD until adulthood (388). Psychoeducation at the time of diagnosis should aim to deliver information on the nature of ADHD, its natural history and prognosis. The goal of psychoeducation is to provide support and knowledge that can assist individuals with ADHD in reframing how they see their symptoms. Psychoeducation may also be useful in improving self-esteem and preventing unrealistic expectations from treatment.

Coaching

While coaching has not yet been formally defined or studied (417), its aim is to help people with ADHD to identify strengths, negotiate problems and work on specific goals. Regular contact and the formation of a partnership with the coach provide a support framework and help to build confidence in taking on specific tasks (416).

7.2 Efficacy of psychosocial interventions

7.2.1 Psychosocial interventions for preschool-aged children

Research question

• For individuals with ADHD, do psychosocial interventions, compared to no intervention, affect outcomes?

Evidence statements

• Parenting programs:

- Two trials found that structured parenting programs conducted by specialist therapists, compared to a wait-list control group, significantly improved behavioural symptoms in preschool-aged children with ADHD (418, 419). (Level II)
- One trial found that a parenting program conducted by non-specialist nurses, compared to a wait-list control group, did not have an impact on ADHD symptoms in preschool-aged children with ADHD (420). (Level II)

Recommendations

- 54. Structured parenting programs with demonstrated effectiveness could be considered for preschoolers with ADHD and associated behavioural problems. (Grade B)
- 55. In considering the use of psychosocial interventions, availability, the family's resources and their capacity to adhere to the program should all be taken into account.

✓ Recommended best practice based on clinical experience and expert opinion

Background

Parenting programs comprise the only psychosocial approach for preschool-aged children with ADHD that have been rigorously investigated. Parenting programs use primarily cognitive behavioural techniques and are designed to teach parents/caregivers strategies for understanding and managing disruptive behaviour in their child and improving parent-child relationships.

A recent meta-analysis assessing the efficacy of the various components of parenting programs for children aged 0–7 years was conducted using studies on ADHD and other childhood behavioural disorders (421). Program components consistently associated with positive outcomes in behaviour and adjustment included increasing positive parent–child interactions and emotional communication skills, teaching parents/caregivers to use time out and the importance of parenting consistency, and requiring parents to practise new skills with their children during sessions (421).

This study also found that for child outcomes, parenting programs had a greater positive impact on internalising behaviours than externalising behaviours and cognitive or educational skills. The lowest effect was seen on social skills and prosocial behaviours (421).

Summary of research evidence

Parenting programs

Three RCTs looked specifically at parenting programs for preschool-aged children with ADHD. Two studies found that structured parenting programs delivered individually were effective in reducing child behavioural problems in preschool-aged children (418, 419). In a third study (420), where the parenting program described in the previous study (419) was conducted by non-specialist nurses rather than specialist therapists, no differences were found between the group that had undergone 8 weeks of structured parenting and the control group. These results highlight the importance of considering who is delivering the program.

Other psychosocial interventions

No studies were identified that addressed the use of psychosocial interventions other than parenting programs in preschool-aged children.

7.2.2 Psychosocial interventions for school-aged children

Research question

• For individuals with ADHD, do psychosocial interventions, compared to no intervention, affect outcomes?

Evidence statements

Parenting programs

 One meta-analysis (422) and three trials (423-425) favoured parenting programs over no intervention for the consistent improvement of child– family function, academic performance and behavioural symptoms, with moderate improvement in ADHD symptoms. (Level I and II)

• Social skills training

 One meta-analysis supported social skills training over no treatment for improving some aspects of social skills in children with ADHD (426). (Level III-1)

• Family therapy

 One systematic review concluded that more research was required to determine if family therapy was an effective treatment for children with ADHD (427). (Level I)

Cognitive behavioural therapy

 One trial reported more appropriate coping skills in children with ADHD following a stress management intervention, but no significant change in acquisition of coping skills compared to the control group (428). (Level II)

• Multi-component behavioural interventions

- One study found that the multi-component behavioural intervention used in the summer treatment program improved behaviour, compared to no intervention (429). (Level 4)
- One study found that the multi-component behavioural intervention used in the MTA study did not improve ADHD symptoms, compared to the community control group (408, 410, 430). (Level III-2)

Recommendations

- 56. Structured parenting programs with demonstrated effectiveness could be considered for children with ADHD and associated behavioural problems. (Grade B)
- 57. Used alone, clinic-based social skills training is not recommended for improving social skills in children with ADHD. (Grade D)
- 58. There are insufficient research data to recommend cognitive behavioural therapy (CBT) or family therapy for the treatment of ADHD. (Grade D)
- 59. Cognitive behavioural therapy (CBT) or family therapy may, however, be useful in addressing certain comorbidities, including anxiety, ODD and CD. ✓ Recommended best practice based on clinical experience and expert opinion

Background

ADHD is often associated with behavioural problems and comorbid psychiatric disorders, such as depression or anxiety. Poor self-esteem, difficulty forming friendships with peers and learning problems are also common. Psychosocial interventions for ADHD aim to improve the daily functioning of the child by improving his or her behaviour and family and peer relationships. Strategies such as parenting programs or family therapy also involve parents/caregivers, to help them to develop sustainable skills to cope with and manage difficult behaviour.

Psychosocial interventions for children can consist of a single component, such as parenting programs, positive reinforcement, daily report cards, time-out procedures and social skills training, or can combine any number of behaviour modification strategies in a multi-component approach. A multi-component intervention strategy was adopted for the MTA study (408). Similar multi-component approaches have been used in programs such as the "summer treatment program", an outpatient program, developed in the USA, that runs for 8 weeks in the style of a school camp.

Summary of research evidence

Parenting programs

One meta-analysis and three RCTs were identified.

The well-conducted meta-analysis (422) combined data from 16 studies that addressed parent involvement in the treatment of ADHD. It was not limited specifically to parenting interventions, but included seven studies focused on parenting programs. The meta-analysis found that family-based interventions had a greater impact on child–family functioning and academic performance than on ADHD symptoms.

A small Australian RCT compared a parenting program (Triple P positive parenting program) with a wait-list control group in children aged 5–9 years (423). Parents in the parenting program reported lower levels of child disruptive behaviour compared to the control group, and these benefits were maintained when assessed at 3 months after the end of the trial. Teacher-rated measures were not different between the groups.

In another RCT (424), parenting programs formed part of a three-armed intervention that also included consultation with teachers and skills and knowledge training with the child. Children were aged 7–11 years and were diagnosed with ADHD inattentive type. A positive impact of the intervention on DSM-IV symptom numbers and severity was seen at the end of the treatment period and at the 3–5 month follow-up period. Other outcome measures showed improvement at the end of treatment, but this did not hold at the time of follow-up.

The third RCT (425) compared a behavioural parenting program plus routine clinical care to routine clinical care alone over 5 months. Both groups showed improvement over time in behavioural, internalising and ADHD symptoms, and parenting stress. Those in the parenting program showed significant improvement in behavioural and internalising symptoms compared to those receiving routine clinical care only. ADHD symptoms and parenting stress did not differ between the groups.

Parent-based approaches for treatment of ADHD are most successful in younger children, with little research support for this approach with adolescents (411).

Social skills training

A systematic review (426) examined the efficacy of social skills training (SST) programs. Four out of four studies reported that SST improved aspects of social functioning in children with ADHD. The content of the interventions and the duration varied in each of the studies. There were also methodological limitations with each of the studies, including the concurrent use of parent groups and/or stimulant medication. Overall, there was no clear support for the use of clinic-based social skills programs, which are usually held weekly, using group contexts, and tend to focus on discussion and role playing of various social skills.

Social skills training has also been studied as part of multi-component behavioural interventions as part of summer treatment programs or other recreational-based programs. Social skills training in these settings are thought to yield positive results in these settings (411). The time and financial commitments for recreational training (300–400 hours) are prohibitive compared to clinic-based social skills training interventions (10–20 hours).

Family therapy

One systematic review (427) has assessed the efficacy of family therapy (without medication) for ADHD symptoms. Only two studies met their inclusion criteria and the review concluded that more research was required to determine whether family therapy is an effective treatment for children with ADHD.

Cognitive behavioural therapy

One study compared a stress management program led by therapists, a parent-led program and a no-treatment control group (428). There were no significant differences between the three groups

for acquisition of coping skills. The therapist-led group reported more appropriate coping strategies. Also of note was that the parent-led group reported the greatest change in self-concept, although the difference was not significant.

Multi-component behavioural interventions

One study looked at behaviour modification as part of a multi-component psychosocial management strategy in a summer treatment program (429). It found that behaviour was improved when behaviour modification strategies were in use in the camp, compared to the two-day withdrawal period when the behaviour strategies were not in use. The study had several major limitations that make the results difficult to interpret, such as no assessment with standardised ADHD outcome measures.

The MTA study compared the use of multi-component psychosocial management with a communitycare control group. At 9 months, when the behavioural management was still being intensively administered, no significant difference was found between the multimodal psychosocial management group and the community-care control group (430). At 14 months (408) and 24 months (409), the behavioural management group did not differ significantly from the community control group on any outcome, and both strategies were significantly less efficacious than medication management and combined management. At 36 months, there was no significant difference between any of the groups (410).

Changes in parenting behaviour have also been examined in the MTA study (431). When observed parenting skills were evaluated, behavioural management was found to enhance positive aspects of parenting significantly, compared to the other treatment groups (431). There was a greater use of proactive parenting strategies in the behavioural group and this was attributed to the parenting program component of the combined behavioural management strategy (431).

It is important to note that the community-care group is not necessarily relevant to the Australian setting as there are many differences between the healthcare settings in Australia and those in the USA. Further, although the multi-component interventions include elements of behavioural management and CBT, the study was not set up in a way that allows any judgement on the individual efficacy of these approaches.

7.2.3 Psychosocial interventions for adolescents

Research question

• For individuals with ADHD, do psychosocial interventions, compared to no intervention, affect outcomes?

Evidence statement

• No studies were identified that addressed the use of psychosocial interventions in adolescents.

Recommendations

- 61. For older adolescents with functional impairments associated with ADHD, the cognitive behavioural therapy (CBT) strategies described for adults may prove beneficial.

✓ Recommended best practice based on clinical experience and expert opinion

Background

ADHD in adolescence can be particularly disruptive, both in school and social contexts, with increasing academic demands and the growing importance of peer relationships.

If difficulties with school curriculum are not identified, secondary behavioural symptoms can occur (e.g. school refusal, declining grades). However, if these problems are identified, behavioural interventions (e.g. contracts) may be employed to provide the young person with structure and support. Communication and consistency between home and school are important for the success of these interventions.

Summary of research evidence

No studies were identified that addressed the use of psychosocial interventions in adolescents.

7.2.4 Psychosocial interventions for adults

Research question

• For individuals with ADHD, do psychosocial interventions, compared to no intervention, affect outcomes?

Evidence statement

• Seven small trials support the use of CBT over no treatment for improving ADHD symptoms and daily functioning in adults with ADHD (383, 432-437). (Level II, II-2 and IV)

Recommendations

- 62. For adults whose ADHD has been stabilised on medication, a course of either group or individual cognitive behavioural therapy (CBT) could be considered to improve ADHD symptoms further and assist with daily functioning in areas such as organisational skills, self-esteem and social skills. (Grade B)
- 63. Psychoeducation, counselling, psychotherapy or coaching may benefit some adults with ADHD.

✓ Recommended best practice based on clinical experience and expert opinion

64. Consideration of the use of psychosocial interventions should take into account availability and the individual's resources and capacity to adhere to the program.

✓ Recommended best practice based on clinical experience and expert opinion

Background

At the time of first diagnosis with ADHD, adults go through a complex process of adjustment (438). Commonly, they have managed to cope with their symptoms through childhood, either because of a supportive environment or because they have developed a range of compensatory strategies. Diagnosis in adulthood often occurs when increasing demands of study or employment prompt the person to seek help. These issues may be coupled with psychological adjustment problems including low self-esteem, depression and anxiety.

General psychosocial support and psychoeducation are beneficial when the person first presents with these difficulties, when treatment is initiated and on regular review (388). Assessment of the person's particular needs and what coping strategies he or she has developed can guide the requirements for psychosocial interventions. For example, intervention may need to focus on the treatment of comorbid psychiatric disorders, psychological problems or skills deficits. Neuropsychological assessment may be of particular benefit in this age group, providing the person with an understanding of his or her cognitive strengths and weaknesses, which can be used to inform the provision of strategies and guidelines to manage these issues in a work or study context.

Counselling or client-based psychotherapies may have a role in helping some people with ADHD come to terms with the way the condition has affected their personal and emotional lives. Similarly, coaching interventions could be used to support people with ADHD in rehearsing newly learnt skills. Although there have been no formal studies of the effectiveness of psychoeducation, counselling, psychotherapy or coaching, clinical experience suggests that some adults with ADHD may benefit from these approaches. The psychosocial interventions for adults with ADHD that have been more rigorously assessed are primarily CBT-based.

Summary of research evidence

Seven small studies were identified that addressed different forms of CBT for adults with ADHD:

• Two RCTs were conducted in Australia. One compared a therapist-directed cognitive remediation program with an untreated control group (432). The program targeted attention problems, poor motivation, poor organisational skills, impulsivity, reduced anger control and low self-esteem. Significant improvements in ADHD symptoms were found following the treatment. In a second

study, comparison of a self-directed self-help program with a control group also found significant improvements in ADHD symptoms following the intervention (433).

- Comparison of a psychoeducational intervention with no treatment (434) showed benefits in the areas of disorganisation, inattention and self-concept/esteem for the group receiving a psychoeducational intervention.
- Two German studies using a CBT intervention based on dialectic behavioural therapy reported improvement in ADHD symptoms (383, 435).
- In a Finnish study of a group-based CBT intervention, there was improvement on self-reported symptoms of ADHD following treatment; however, reports from significant others showed no improvement (436).
- A group-based CBT intervention designed to target self-management skills resulted in improvements on self-reported symptoms of inattention, executive function skills and time management, and organisation and planning following treatment (437).

Overall, the studies of psychosocial interventions in adults with ADHD show benefits for the management of ADHD symptoms; however, all the studies were rated as poor quality and the results should be interpreted cautiously. Several studies included participants who were concurrently taking medications for ADHD, but did not necessarily conduct a separate analysis to take this potential confounder into account. Two studies did consider the impact of medication status on treatment outcome (432, 435) and both reported that medication had no significant impact on the outcomes of the psychosocial treatments under investigation.

7.3 Psychosocial interventions for ADHD when comorbidities are present

Research question

• For individuals with ADHD and comorbid disorders, do psychosocial interventions, compared to no intervention, affect outcomes?

Evidence statements

- One trial in adolescents with ADHD and comorbid ODD found no statistically significant improvement in ADHD or ODD symptoms when using psychosocial interventions (problem-solving communication training and/or behaviour management) (439). (Level III-3)
- One trial in children with ADHD and comorbid ODD or CD showed that behavioural management, compared to community care, did not significantly improve ADHD or ODD/CD symptoms (440). (Level III-2)
- One trial in children with ADHD and comorbid anxiety showed that behavioural management, compared to community care, significantly improved ADHD and anxiety symptoms and social skills (408, 440, 441). (Level III-2)
- No studies were found that addressed psychosocial interventions for ADHD and comorbid disorders in preschool-aged children or adults.

Recommendations

- 65. Psychosocial interventions should be considered for individuals with ADHD and comorbidities.
 - \checkmark Recommended best practice based on clinical experience and expert opinion
- 66. Psychosocial interventions can be particularly beneficial when the individual with ADHD has comorbid anxiety. (Grade C)

Background

There are high levels of comorbidity between ADHD and a number of other conditions, including anxiety, depression and disruptive behavioural disorders such as ODD and CD (see section 3.2 Comorbidities, page 22). However, research addressing the impact of psychosocial interventions for ADHD and comorbid conditions is very rare. The available research considers children and adolescents and no studies in preschool-aged children or adults met the inclusion criteria.

Summary of research evidence

Three studies were identified that investigated the efficacy of psychosocial interventions for ADHD when comorbid disorders are present.

One compared two intervention strategies: problem-solving communication training and behaviour management training plus problem-solving communication training (439). The participants were adolescents aged 12–18 years diagnosed with ADHD and ODD. At the 2-month follow-up, only 25% of the participants in either group showed improvement. There were no significant differences between the two intervention strategies.

The remaining two studies were derived from the MTA study. The original MTA study compared four treatment groups: medication management, behavioural management, combined treatment and community care (408). At 14 months, the behavioural management group did not differ significantly from the community control group on any outcome. Both behavioural management and community care were significantly less efficacious than medication management and combined management (408).

Analysis of the MTA study to consider the impact of comorbidities on treatment efficacy found that behaviour management, compared to community care, improved core ADHD symptoms and social skills outcomes for participants with ADHD plus anxiety, compared to those with ADHD only (440). Behaviour management also improved core ADHD symptoms in those with ADHD plus ODD/CD plus anxiety. There was no differential response to treatment for the ADHD plus ODD/CD group compared to the ADHD only group.

Further analysis of the MTA outcomes at 14 months found that, for children with ADHD and anxiety, behavioural treatment was significantly better than community care on parent-rated measures, irrespective of the presence or absence of comorbid conduct problems (441).
CHAPTER 8. MEDICATION

8.1 Introduction

Recommendations

Treatment with medication is not indicated for all individuals with ADHD. Medications should only be used when ADHD symptoms and impairments are considered severe or when individuals with moderate symptoms and impairments have refused non-pharmacological interventions or when their symptoms have not responded to psychosocial treatments.

As discussed above (see section 6.3 Developing an individualised management plan, page 68), the diagnosis of ADHD and the determination of severity takes into account the symptoms of hyperactivity, impulsivity or inattention *and* the level of subsequent impairment in multiple settings. Consideration must be given to the level of impairment in terms of academic performance, self-esteem, personal distress from the symptoms, social interactions and relationships, behavioural problems and the development of comorbid psychiatric syndromes. Circumstances unique to the individual and their family must also be evaluated.

Multimodal therapy is advocated for the management of ADHD in all age groups. Accordingly, when medication treatment is suggested it should be as part of a comprehensive management plan that includes psychosocial and educational management strategies.

The decision to use medication for the treatment of ADHD requires careful consideration of all the usual issues that must be weighed in a decision to initiate medication. These include the level of indication for drug therapy, the balance of benefit versus risk from use of medication, the legal requirements surrounding use of medications and the ability of the individual, family or community to afford the medication. The Therapeutic Goods Administration (TGA), a division of the Australian Government Department of Health and Ageing, is responsible for the assessment and monitoring of the quality, safety and efficacy of medicines marketed in Australia.

When granting marketing approval the TGA approves the use of medications for specific indications. These indications are listed in the approved Product Information document for each prescription medication.

Prescribing medications for indications other than those stated in the Product Information, while not illegal, is not endorsed by the sponsor of that medication or by the TGA. Nevertheless, it is recognised that clinical practice is not always consistent with the indications stated in the Product Information.

When prescribing medications outside the TGA-approved indications it is important that this is made clear to the patient and, where appropriate, their parent/carer. Informed consent should be obtained and documented. The clinician should discuss the reasons for using the medicine, possible side effects, possible alternative therapies and any additional information to address specific concerns (442).

8.1.1 Available medications

In Australia the medications that are registered for use in individuals over 6 years of age with ADHD are methylphenidate (MPH) in immediate-release (IR) (Ritalin 10 and Attenta) and extended-release (ER) formulations (Ritalin LA, Concerta); dexamphetamine sulphate (DEX); and atomoxetine (ATX) (Strattera).

Other medications that have been used or studied in the treatment of ADHD include clonidine, bupropion, selegiline, modafinil, imipramine, risperidone, tricyclic antidepressants and nicotine patches. These medications are available in Australia but are not currently licensed for the treatment of ADHD. Notably, the tricyclic antidepressants have had a diminishing role in the treatment of ADHD since the advent of other non-stimulant treatments. No research studies addressing the use of tricyclic antidepressants for the treatment of ADHD met our systematic review inclusion criteria. As a result, these medications have not been included in the Guidelines.

There are also several medications that are currently not available in Australia, but are commonly used to treat ADHD in other countries. These include mixed amphetamine salts (MAS) and guanfacine, which have both been approved by the FDA to treat ADHD in the USA.

In this review we have considered any medication that has been subject to a randomised controlled trial in individuals with ADHD that also meets the inclusion criteria of the systematic review.

Australian prescribing regulations

MPH and DEX are Schedule 8 controlled drugs for which additional prescribing restrictions apply in most Australian States and Territories. Statutory regulations vary; for example, in some States (e.g. New South Wales, Western Australia) prescribers need to apply for permission to prescribe MPH and DEX. The use of stimulant therapy in preschool-aged children in particular is heavily regulated under specific State and Territory legislation.

The relevant State or Territory department of health (Table 5) can provide details on compliance with the provisions of State or Territory law when prescribing MPH and DEX.

Table 5. State and Territory contacts for information of the regulation of stimulant prescribing

Australian Capital Territory	
ACT Department of Health & Community Care	
Pharmaceutical Services, Population Health Divis	sion
New South Wales	
Pharmaceutical Services Branch	
NSW Health	
Northern Territory	
Poisons Control	
Department of Health and Community Services	
Queensland	
Pharmaceutical Advisory Services	
Queensland Health	
South Australia	<u>^</u>
Public & Environmental Health Service	
Department of Human Services	
Tasmania	
Pharmaceutical Services Branch	
Department of Health and Human Services	
Victoria	
Drugs & Poisons Unit	
Department of Human Services	
Western Australia	
Pharmaceutical Services Branch	
Department of Health	

Monitoring of adverse drug reactions in Australia

Reports of suspected adverse reactions to prescription medicines should be submitted to the TGA Office of Medicines Safety Monitoring (http://www.tga.gov.au/adr). Reporting by Australian doctors, dentists and pharmacists is voluntary.

8.1.2 Variations from approved Product Information

In some instances, the Guideline recommendations differ from the relevant medication Product Information sheets. These differences are noted in the text under the sub-heading "Variations from approved Product Information". Where a variation exists, it has been made on the basis of scientific evidence and expert opinion and reflects current expert practice.

8.1.3 Limitations in the current research on ADHD medications

Efficacy

The vast majority of studies addressing the efficacy of medications for the treatment of ADHD are short-term studies of 3 months or less. The focus of the research has been on the impact of medications on the core symptoms of ADHD: inattention, impulsivity and hyperactivity. Few studies have looked at quality of life, school or work performance, or social function.

The majority of the identified studies on ADHD medications have been sponsored, at least in part, by the manufacturers of the medications.

Side effects

For the three most common medications used for ADHD – MPH, DEX and ATX – side effects were addressed in detail in the systematic literature review. In addition to the randomised controlled trials utilised to assess the efficacy of medications, published articles representing lower levels of evidence were included to give a greater overview of possible side effects.

There are a number of limitations in the reporting of side effects and adverse events in the literature. As highlighted in a review by King et al (443), many studies provide only limited information about adverse events, and standardised reporting between studies is lacking. Measures of adverse events vary widely. The most frequently used measures are open-ended questions and

rating scales that are limited to the most common side effects associated with the medication in question. When only one of these methods is used, there is considerable room for bias in reporting. In addition, younger children may not have the communication skills to convey side effects clearly. Finally, the long-term safety effects of ADHD medications are not clearly established, as long-term studies are rare.

Stimulants and Atomoxetine

The combining of stimulant medication (MPH or DEX) with ATX to treat ADHD has not been adequately researched. No studies were identified that allowed the assessment of the safety or efficacy of combining stimulant medication (MPH or DEX) with the non-stimulant medication ATX. One open pilot study of 25 children aged 6 to 12 years reported the safe combination of these drugs (444). In this small study 16% of participants withdrew because of side-effects. There have been a number of controlled trials describing the combination of MPH with other medications for treating ADHD when comorbidities are present. These individual studies are discussed below in Sections 8.5 and 8.6.

8.1.4 Mangement of medication

The approaches to the use of the different medications for different age groups are summarised in the table at the endof the medication section..

Guidelines on Attention Deficit Hyperactivity Disorder

8.2 Stimulant medication (methylphenidate, dexamphetamine and mixed amphetamine salts)

Research questions

- For individuals with ADHD, do stimulant medications, compared with placebo, affect outcomes?
- For individuals with ADHD, does use of stimulants for 1 year or more, compared with placebo or no intervention, affect outcomes?
- For individuals with ADHD, does any stimulant medication confer an advantage over any other stimulant medications?
- For individuals with ADHD who are taking medication, what are the main side effects?

Evidence statements

- Short-term (weeks/months) MPH
 - o **Preschoolers**

Two RCTs (445, 446) have demonstrated benefit of MPH compared to placebo in reducing the symptoms of ADHD. (Level II)

o Children and adolescents

One systematic review (25 RCTs MPH; 5 RCTs DEX) has demonstrated benefit of MPH and DEX compared to placebo in reducing the symptoms of ADHD (443). (Level I)

Adolescents

One RCT has demonstrated benefit of MPH-ER over placebo in reducing the symptoms of ADHD (447). (Level II)

One RCT has demonstrated benefit of MAS over placebo in reducing the symptoms of ADHD (448). (Level II)

o Adults

One meta-analysis (6 RCTs) (449) and four RCTs (450-453) have demonstrated benefit of MPH over placebo in reducing ADHD symptoms . (Level II)

Three RCTs have demonstrated benefit of DEX over placebo in reducing ADHD symptoms (454-456). (Level II)

Two RCTs have demonstrated benefit of MAS over placebo in reducing ADHD symptoms (457, 458). (Level II)

Immediate-release versus extended-release stimulants

o Children and adolescents

One systematic review (2 RCTs) found no difference in efficacy between immediate-release MPH and extended-release MPH for the improvement of ADHD symptoms (443). (Level I)

• MPH versus DEX

• Children and adolescents

One systematic review (2 RCTs) found no difference in efficacy between MPH and DEX for the improvement of ADHD symptoms (443). (Level I)

• Long-term (>1 year) use of stimulants

 One study (MTA) found the use of stimulant medication (MPH) over placebo significantly improved ADHD symptoms when used for 14 months (408), but there was no significant difference in the longer term (at 24 and 36 months) (409, 410). (Level III-2)

Recommendations

Preschool-aged children (3–5 years of age)

- 70. Medication should not be used as first-line treatment for ADHD in preschool-aged children.
 - \checkmark Recommended best practice based on clinical experience and expert opinion
- 71. The use of medication should only be considered in preschool-aged children when there has been poor response to behavioural or psychosocial therapy and the ADHD symptoms are having a severe impact on the child and their family/carers. For preschool-aged children medication should only be initiated following specialist assessment and in the context of multidisciplinary care, preferably in a tertiary setting. ✓ Recommended best practice based on clinical experience and expert opinion
- 72. Where a decision is made to trial medication, MPH-IR should be used at low dose, in conjunction with appropriate behavioural intervention. (Grade C)
- 73. Preschool-aged children on medication need to be monitored closely because of the increased incidence of side effects in this age group. (Grade B)
- 74. Extended-release forms of stimulants should not be routinely used in preschool-aged children.

✓ Recommended best practice based on clinical experience and expert opinion

Note: Use of stimulant therapy in preschool-aged children is regulated under specific State and Territory legislation.

School-aged children

- 75. Where severe, impairing ADHD is present, treatment with MPH or DEX should be considered as a first-line pharmacological treatment. (Grade A)
- 76. The choice of MPH-immediate release or MPH-extended-release depends on the symptom profile, as well as individual child and parent/caregiver preferences. (Grade A)

Adolescents

- 77. Where severe ADHD is present, treatment with MPH or DEX should be considered as a first-line pharmacological treatment. (Grade A)
- 78. The choice of MPH-immediate-release or MPH-extended-release depends on the symptom profile, as well as the preference of the adolescent with ADHD. (Grade B)

• Adults

- 79. Where severe ADHD is present, treatment with MPH or DEX should be considered as a first-line pharmacological treatment. (Grade A)
- 80. The choice of MPH-immediate-release or MPH-extended-release depends on the symptom profile, as well as the preference of the person with ADHD. (Grade B)

• All age groups

81. When stimulant treatment is used, it should be continued for as long as it is of assistance to the person with ADHD and is not causing unacceptable side effects.

✓ Recommended best practice based on clinical experience and expert opinion

Stimulants not available in Australia

83. MAS may have a role in the pharmacological management of ADHD in

	primary school-aged children, adolescents or adults. (Grade B) Note: MAS is not currently available in Australia.	
Immediate-release versus extended-release		
84.	Immediate-release forms should be the initial treatment, to titrate to the optimal dose, and they may be the preferred maintenance therapy for various reasons, for example, flexibility of dosing. Recommended best practice based on clinical experience and expert opinion 	
85.	If starting on immediate-release stimulants, consideration should be given to changing to an extended-release form once the optimal dose has been established. This can help to avoid the stigma and inconvenience of taking medication at school. Recommended best practice based on clinical experience and expert opinion 	
86.	In some cases the combined use of immediate-release and extended- release forms is required. This should only be considered if there is inadequate symptom control with the extended-release form. Recommended best practice based on clinical experience and expert opinion 	

8.2.1 Background – stimulant medications

Stimulant medications are thought to influence ADHD symptoms because they alter the availability of dopamine and noradrenaline in the regions of the brain involved with behaviour inhibition, impulse control and attention. The precise mode of action on ADHD symptoms is not fully understood.

In Australia, the stimulant medications DEX and MPH are used for the treatment of ADHD.

MPH is available in both immediate-release (IR) and extended-release (ER) formulations. The immediate-release formulations (such as Ritalin 10 and Attenta) have a short half-life, and 2-3 daily doses are usually required. The extended-release formulations (such as Ritalin LA and Concerta (OROS MPH)) need only be taken once a day.

Ritalin LA was registered by the TGA in 2002 and listed on the Pharmaceutical Benefits Scheme (PBS) in April 2008. Ritalin LA is reimbursed for children and adolescents (6–18 years) who require continuous coverage over 8 hours. Concerta was approved by the TGA in 2003 and listed on the PBS in April 2007. Concerta is reimbursed for children and adolescents (6–18 years) who require continuous coverage over 12 hours. An MPH patch has been developed but has not yet been approved for use in Australia.

The PBS authority regulations require a successful trial of MPH-IR before approval will be given for ER forms. Anecdotal feedback from clinicians suggests that a higher dose of MPH is required when using an ER form of MPH. This is supported by Pelham et al's (2001) study (459), which showed when using OROS MPH, a 20% higher dose was required than when using MPH-IR.

Amphetamines are either a single-salt formulation (such as Dexadrine) or a mixed-salt formulation (such as Adderall and Adderall-XR). At this stage only an immediate-release form of DEX is available in Australia. The mixed-salt formulations, Adderall and Adderall-XR, are used to treat ADHD in the USA and Europe but are not available in Australia.

The issue of stimulant medication use in the preschool age group is contentious, although the numbers of children involved are very small. The recent NSW Department of Health Clinical Excellence Commission Review documents the following numbers of preschoolers on stimulant medication in 2007: 2 year olds – 1 (0.001%); 3 year olds – 7 (0.008%); 4 year olds – 56 (0.05%) (460). In addition, the 2007 NSW Department of Health review of prescribing practices in paediatricians and child psychiatrists showed that over 95% were adhering to the strict NSW Department of Health stimulant prescribing guidelines. In Western Australia, the following numbers of preschoolers were on stimulant medication in 2007: children under age 4 – 0; 4 year olds – 14 (0.2%).

There is, however, a very small group of preschool-aged children who do not respond to nonmedical interventions. It is this group with severe symptoms who are suitable for a trial of stimulant medication. Such trials should only be considered in an expert, multidisciplinary tertiary setting. This is consistent with other guidelines which recommend such children be referred to tertiary services (244).

Side effects of stimulant medications are discussed in detail in section 8.7 Side effects: special considerations, page 116. The abuse potential of ADHD medications is discussed in section 8.9 ADHD and substance misuse, page 124.

Variations from approved Product Information

The Product Information for MPH-IR (Attenta and Ritalin 10) states that MPH should not be used in children under 6 years of age as safety and efficacy in this age group have not been established. The Guidelines recommend that MPH-IR can be used in children under 6 years in conjunction with appropriate behavioural intervention when there has been poor response to behavioural or psychosocial therapy alone and the ADHD symptoms are having a severe impact on the child and their family/carers. It is also recommended that preschool-aged children on medication need to be monitored closely because of the increased incidence of side effects in this age group. Historically, DEX has been approved for the treatment of ADHD without age limits and as a result it is approved for use in preschoolers. However, there has been no research identified to support the use of DEX in preschool-aged children.

8.2.2 Summary of research evidence

Preschoolers

Two short-term studies were identified that addressed the use of MPH for ADHD in preschool-aged children:

- A poor-quality study (446) found improvement in core ADHD symptoms, but no significant improvement in measures of child compliance and attention tasks.
- A more recent study (445) found that 2.5, 5 and 7.5mg doses of MPH produced significant reductions in ADHD symptoms compared to placebo, although the number and type of side effects were greater, and the effect sizes smaller, than in older children. Several limitations in reporting signal a need for caution in interpreting these results.

No other types of stimulants have been tested in this population.

Children and adolescents

King et al (443) reviewed the efficacy of MPH and DEX in children and adolescents with ADHD. The quality of the included studies was mixed, with many being rated as poor quality. Based on 25 controlled trials of MPH and 5 of DEX, the overall conclusion was that stimulant medication may improve symptoms and quality of life in this population.

The vast majority of studies of stimulant medications in children and adolescents are of short to medium duration (less than 3 months) and the long-term impact of stimulants is not clear.

Two additional studies were included that addressed the use of stimulants in adolescents, one investigating extended-release mixed amphetamine salts (MAS-ER) (448), the other, extended-release methylphenidate (MPH-ER) (447). Both studies reported improvement in core ADHD symptoms compared to placebo, but both had limitations in methodology that signal caution in interpreting the results.

Adults

Five studies that investigated the use of MPH in adults with ADHD were included. Faraone et al (449) described a meta-analysis of 6 RCTs, which supported the conclusion that MPH was effective in reducing symptoms of ADHD in adults. Methodological limitations resulted in this study being rated as poor quality. Four primary studies (450-453) reported consistent improvement in core ADHD symptoms for adults treated with MPH but the results for quality of life measures were less clear.

Two studies (457, 458) reported benefit from MAS, in both the IR and ER formulations, in reducing ADHD symptoms compared to placebo.

Three studies comparing DEX to placebo reported improvement in core ADHD symptoms (454-456). Measures of the impact of medication on quality of life were not included.

Comparing stimulant medications

No studies were identified that compared stimulant medications in preschoolers or adults.

For children and adolescents with ADHD, King et al's (443) systematic review compared the efficacy of the immediate-release and extended-release stimulant medications. Two studies were discussed, both short term (4–8 weeks) and of poor quality. Both compared MPH-IR and MPH-ER and found no differences between the two in hyperactivity or quality of life measures. One study found a higher incidence of headaches in the group taking MPH-ER.

King et al (443) also compared MPH and DEX in children and adolescents with ADHD. Again, two studies were discussed. The first, which was rated as poor quality, looked at the impact of DEX and MPH on ADHD symptoms and found MPH to be superior on teacher ratings, but no difference between MPH and DEX in parent ratings was found. In addition, there were no differences in adverse events between the groups. The second study reported no differences between the two medications on hyperactivity measures. This study did not adequately report on adverse events. Neither study reported on quality of life measures.

Long-term use of stimulants

The long-term safety and efficacy of stimulant medication has not been established.

No studies addressing this issue in preschoolers or in adults met the inclusion criteria.

In primary school-aged children, the only study addressing this issue that met the inclusion criteria was the Multimodal Treatment Study of ADHD (MTA), which compared four treatment groups: medication management, behavioural management, combined treatment and community care. Data have now been analysed at 14 months (408), 24 months (409) and 36 months (410). At the 14-month and 24-month time points, medication management was superior to community control in improving core ADHD symptoms, but there were no significant differences between groups on measures of academic achievement or social skills (408, 409). At 36 months there was no significant difference between the medication group and the community control group on any measure.

The MTA study has two critical methodological limitations that make it difficult to draw clear conclusions: it was not placebo-controlled, and after the 14-month analysis participants were allowed to change treatment but continued to be followed up according to their initial treatment allocation. For example, a proportion of families whose children were initially assigned to the behavioural management group subsequently opted for stimulant medication, making the long-term outcome of the initial MTA treatment groups difficult to assess.

"Medication holidays", during which stimulant medication is ceased, are used by many practitioners, despite a lack of evidence to guide this practice. Potential benefits of medication holidays are the opportunity to determine continuing benefit of medication and minimisation of side effects. Regular weekend holidays have been shown to reduce side effects of insomnia and appetite suppression without a significant increase in ADHD symptoms (461). The long-term benefits of medication holidays on growth have been questioned (462).

Side effects

Preschoolers

Two clinical trials were identified that reported on adverse events in preschool-aged children receiving MPH. The first was a crossover trial of short duration (7–10 days each on placebo, low-dose MPH and high-dose MPH) (446, 463). The higher dose of MPH resulted in a significant increase in the number and severity of adverse events, including decreased appetite, nightmares, feeling sad/unhappy and social ratings such as "talks less with others", compared with both placebo and the lower dose of MPH. There were no significant differences between low-dose MPH and placebo. Wigal et al (464) described the adverse events identified in the PATS protocol, a 17-month study. Side effects that were significantly increased in the MPH group included decreased appetite, trouble sleeping, weight loss, emotional outbursts and social withdrawal.

In both studies the profile of unwanted side effects differed from those seen in school-aged children, where decreased appetite, delay of sleep onset, headaches and stomach ache were generally the most frequently reported side effects. Wigal et al (464) reported that 11% of the participants withdrew from the PATS study because of unwanted side effects. This figure is high compared with

the school-aged children in the MTA study, which reported less than 1% withdrawals because of unwanted side effects.

Children and adolescents

A thorough review of adverse events in children and adolescents was conducted by King et al (443) with the following conclusions:

- MPH-IR in higher dosages is associated with headache, loss of appetite, stomach ache and insomnia.
- MPH-ER is associated with loss of appetite and increased insomnia.
- DEX in high doses is associated with loss of appetite and sleeping problems.

Adults

MPH-IR in adults was investigated in two studies:

- In a trial involving medication for a total of 3 weeks (452) no participants withdrew because of adverse effects and the only side effect that was more common with MPH than placebo was loss of appetite.
- A 6-week trial (453) had a withdrawal rate due to adverse events of 13% for MPH-IR and 5% for placebo. High blood pressure was significantly increased in the MPH-IR group, and other common side effects were decreased: appetite, dry mouth and moodiness.

The increased side effects in the second study (453), compared to the first (452), may reflect the longer study duration and higher dose (1.1/mg/kg/day compared to 0.9mg/kg/day).

MPH-ER in adults was investigated in two studies:

- A 6-week study (450) found more tension/jitteriness, decreased appetite, gastrointestinal problems, sleep problems, dizziness, cardiovascular complaints, anxiety and weight loss with MPH-ER compared to placebo.
- Another extended-release formulation, multilayer-release methylphenidate (MPH-MLR), was examined in a 3-week RCT (451). Anxiety, nervousness, weight loss and anorexia, and the severity of these side effects, were increased with the MPH-MLR compared to placebo, but there were no withdrawals due to adverse events over the duration of the study (~7 months).

MAS in adults was investigated in two studies:

- In a crossover trial with 3 weeks on active medication (457), appetite suppression, weight loss and agitation were increased in the MAS group compared to placebo.
- In a 4-week RCT of MAS-ER (458), 24 of the 191 participants on active drug withdrew from this study as a result of side effects, which included dry mouth, anorexia/decreased appetite, insomnia, headache and weight loss.

Two studies on DEX in adults (454, 455) found no significant difference in the number of side effects with DEX compared to placebo. The most common side effects were insomnia, irritability and muscle tension (454, 455).

8.3 Atomoxetine (non-stimulant)

Research questions

- For individuals with ADHD, does ATX, compared to placebo, affect outcomes?
- For individuals with ADHD, does ATX confer an advantage over stimulant medications?
- For individuals with ADHD, does ATX for 1 year or more, compared to placebo or no intervention, affect outcomes?
- For individuals with ADHD who are taking medication, what are the main side effects?

Evidence statements

Atomoxetine (ATX)

- Preschoolers: No studies met the inclusion criteria.
- Children and adolescents: One systematic review (3 RCTs) has demonstrated improvement in ADHD symptoms with ATX over placebo (443). (Level I)
- **Adults:** Two RCTs have demonstrated benefit from ATX over placebo for ADHD symptoms (465). (Level II)
- **Long-term use:** One study found that the use of ATX over placebo improved ADHD symptoms in children and adolescents when used long term (1 year or more) (466, 467). (Level II)
- Comparison to stimulant medications
 - Children and adolescents, ATX versus MPH: Five studies addressed this issue. Two reported no difference between MPH and ATX (443, 468), one reported ATX's superiority on behavioural symptoms and no other difference (469), and two reported MPH's superiority (443, 470). (Level II)
 - **Children and adolescents, ATX versus MAS-ER:** One study found that MAS was superior for teacher-rated symptoms and academic performance, but no other differences (471). (Level II)

Recommendations

- 87. **Preschoolers:** There is no published evidence to show benefit from ATX in treating ADHD in preschool-aged children.
- 88. **Children and adolescents:** ATX should be considered for children and adolescents with severe ADHD who do not respond to or are intolerant of stimulant medication, or in whom stimulant medication is contraindicated. (Grade B)
- 89. **Adults:** Treatment with ATX should be considered for adults with severe ADHD who do not respond to or are intolerant of stimulant medication, or in whom stimulant medication is contraindicated. (Grade B)
- 90. **ADHD and comorbidities:** ATX may be considered as the first-line medication if there is comorbid substance abuse, severe tic disorder or anxiety disorder.
 - ✓ Recommended best practice based on clinical experience and expert opinion

8.3.1 Background

ATX (Strattera) is a non-stimulant pharmacological agent which offers an alternative to stimulant medication where response to stimulants is poor, or where there are side effects that cannot be tolerated. ATX is classified as a "noradrenaline reuptake inhibitor". The mechanism of its impact on ADHD symptoms is not known. Atomoxetine was registered by the TGA for the treatment of ADHD in 2004 and became available on the PBS in Australia in July 2007 for the treatment of children, adolescents and adults with ADHD (472).

ATX is currently reimbursed for the treatment of ADHD (diagnosed at age 6–18) where (472):

- treatment with DEX or MPH poses an unacceptable medical risk because the patient has a history
 of substance abuse or misuse (other than alcohol) and/or comorbid motor tics or Tourette
 Syndrome and/or comorbid severe anxiety diagnosed according to the DSM-IV; or
- treatment with DEX or MPH has resulted in the development or worsening of a comorbid mood disorder (diagnosed according to the DSM-IV criteria, i.e. anxiety disorder, obsessive compulsive disorder, depressive disorder), either:
 - o of a severity that necessitates permanent withdrawal of stimulant treatment; or
 - where the combination of stimulant treatment with another agent would pose an unacceptable medical risk; or
- treatment with DEX and MPH has resulted in adverse effects on growth and weight and/or sleep (including insomnia) and/or appetite (including anorexia) of a severity that necessitates permanent withdrawal of these treatments.

For prescribing ATX for patients, the PBS requires a diagnosis of ADHD made when the patient was aged 6–18 years. ADHD must be diagnosed by a paediatrician or psychiatrist according to the DSM-IV criteria, and psychostimulants (DEX and MPH) must pose an unacceptable medical risk.

Side effects of atomoxetine are discussed in detail in section 8.7 Side effects: special considerations, page 116. The abuse potential of ADHD medications are discussed in section 8.9 ADHD and substance misuse, page 124.

8.3.2 Summary of research evidence

Preschoolers

The safety and efficacy of ATX has not been established in children less than 6 years of age (473) and no studies were found that addressed its use in preschool-aged children.

Children and adolescents

King et al (443), reviewing the efficacy of ATX in children and adolescents with ADHD, found consistent support for the conclusion that ATX may reduce symptoms and improve quality of life. This conclusion is based on 9 placebo-controlled double-blind RCTs of short duration using either a low/medium dose (<1.5mg/kg/day – three studies) or a high dose (>1.5mg/kg/day – six studies). The quality of the included studies was mixed.

Adults

Two RCTs with adults (reported in one article (465)) met the inclusion criteria. These studies found that ATX was effective in reducing symptoms associated with ADHD; however, its effect on the Sheehan disability measures, such as work life, social life and family life, was less clear.

Long-term use

The long-term safety and efficacy of ATX have not been established.

One study has addressed the long-term use of ATX compared to placebo in children aged 6–15 years (466, 467). This study has two distinct parts. In the first part, participants who had responded to ATX in a 10-week open-label trial were then randomised to receive ATX or placebo for 9 months. Some, but not all, measures showed a benefit of ATX in improving core ADHD symptoms (466). In the second part of the study, participants who had completed 1 year of double-blind ATX treatment

were then randomly assigned to continued ATX or placebo for 6 months. At the end of this time, ATX was found to be effective compared to placebo for two outcomes measures of ADHD symptoms (467).

While this research provides some support for the long-term effectiveness of ATX for children and adolescents with ADHD, the study included only participants that initially responded to ATX and therefore the study design allowed for bias towards a favourable outcome for the ATX group compared to placebo.

Comparison with stimulant medications

Preschoolers

No studies meeting the inclusion criteria were identified.

Children and adolescents

MPH versus ATX

A systematic review of MPH, DEX and ATX in the treatment of ADHD (443) included two shortduration studies, of poor quality, comparing MPH with ATX. One found no significant differences in efficacy between MPH-IR and ATX; the other found that MPH-ER was superior to ATX on hyperactivity and quality of life measures.

Three additional studies compared MPH and ATX in children and adolescents. One found that ATX was superior to MPH in improving behaviour but there were no other significant differences in ADHD symptoms between the groups (469). Another found no significant difference between the ATX and MPH groups on any measure of ADHD symptoms (468). The third (470), with more than 500 participants, found that both MPH and ATX were superior to placebo, and MPH was superior to ATX.

Compared to placebo, those receiving MPH reported significantly higher rates of decreased appetite and insomnia (469, 470), increased heart rate and diastolic blood pressure, and decreased weight (470). Those receiving ATX reported higher rates of "any adverse event", decreased appetite and decreased weight (470), and increase in diastolic blood pressure and heart rate (469, 470).

In summary, the study results vary, with two out of five reporting no difference between MPH and ATX, one reporting ATX's superiority, and two reporting MPH's superiority. All studies have been short-term assessments of the two medications. Additional longer term studies are required.

MAS versus ATX

A study comparing ATX to MAS-ER in children (6–12 years) (471) found that MAS-ER was superior on teacher-rated and academic achievement measures, but there were no differences in parent-rated measures, quality of life measure or adverse events.

For MAS-ER, the most common adverse events were insomnia, appetite decrease, upper abdominal pain, anorexia and headache; and for ATX, the most common side effects were somnolence, appetite decrease, upper abdominal pain, vomiting and headache (471).

Side effects

Preschoolers

No studies were identified that met the inclusion criteria.

Children and adolescents

A review of adverse events in children and adolescents (443) concluded that ATX at any dose may reduce appetite.

Wilens et al (474) addressed the use of ATX in adolescents, reviewing 13 clinical trials of up to 2 years in length. Reporting of adverse events in this review was limited. The review reported that 31 out of 601 adolescents discontinued ATX use due to adverse events, most commonly headache, nasopharyngitis, nausea and upper abdominal pain.

Adults

The side effects of ATX in adults were investigated in two studies:

- In two identical RCTs conducted over 10 weeks (465), withdrawal due to adverse events was less than 10% overall. Adverse events including dry mouth, insomnia, nausea, decreased appetite and erection difficulties were significantly more common with ATX than placebo in one study only.
- A study of poor quality compared 80mg of ATX given either once or twice a day (475). The most frequent adverse events were headache, insomnia, nausea, dry mouth and appetite decrease. Nausea was more common with ATX once a day than twice a day. Men, but not women, from both groups showed a significant change in sexual dysfunction scores.

8.4 Other medications used in ADHD

8.4.1 Clonidine

Research questions ٠ For individuals with ADHD, does clonidine, compared with placebo, affect outcomes? For individuals with ADHD who are taking clonidine, what are the main side • effects? For individuals with ADHD, does clonidine confer an advantage over any other . pharmacological interventions? **Evidence statements Preschoolers:** No studies met the inclusion criteria. . Primary school-aged children: One RCT found parents but not teachers . reported that clonidine had some benefits in reducing ADHD symptoms compared to placebo (476). (Level II) Adolescents, adults: No studies met the inclusion criteria. . Long-term use: No studies met the inclusion criteria. • **MPH versus clonidine:** . Children: One RCT found no difference between MPH and clonidine for ADHD symptoms (476). (Level II) • Adults: No studies were identified. Recommendation Clonidine may provide some benefit in modifying ADHD symptoms in 91. children, particularly in the home setting, used either alone or in combination with MPH. It could be trialled in the absence of clinical response to stimulants and ATX. (Grade D)

Background

Clonidine is an alpha-2 noradrenergic agonist which is used in the treatment of hypertension, migraine and menopausal flushing. It has also been used in the treatment of ADHD. The precise mechanism of its impact on ADHD symptoms has not been comprehensively studied, but it is thought to work by affecting noradrenaline transmission in the frontal cortex.

Common side effects include drowsiness, dry mouth, gastrointestinal upsets, hypotension and dizziness. There has been some controversy about potential cardiovascular risk, but safety data are limited to date.

Variations from approved Product Information

Clonidine is not currently licensed for the treatment of ADHD in Australia.

Summary of research evidence

Preschoolers

No studies that met the inclusion criteria were identified.

Children and adolescents

A 16-week RCT in children with ADHD compared MPH, clonidine, MPH and clonidine combined, and placebo (476). Comparison of clonidine to placebo found a significant improvement in parent-rated ADHD symptoms and the global assessment scale for the clonidine over placebo. However, there was no difference in teacher-rated ADHD symptoms.

Adults

No studies that met the inclusion criteria were identified.

MPH versus clonidine

A 16-week RCT in children with ADHD compared MPH, clonidine, MPH and clonidine combined, and placebo (476). Direct comparison between the MPH and clonidine groups found no significant differences on any scale. Those receiving MPH (combined or MPH) showed greater improvement in ADHD symptoms than those not receiving MPH (clonidine or placebo) on the Conners' ASQ – Teachers, while those receiving clonidine (clonidine or combined) showed greater improvement than those not receiving clonidine (MPH or placebo) on the Conners' ASQ – Parents and CGAS scales.

Participants receiving clonidine had a higher incidence of dull/tired/listless and drowsiness/sedation side effects (477). Bradycardia was also more frequent in the clonidine groups, but no other significant group differences were seen for electrocardiogram and other cardiovascular outcomes. Moderate or severe adverse events were more common in the clonidine group, but were not associated with higher rates of early study withdrawal.

8.4.2 Modafinil

_			
Research questions			
•	 For individuals with ADHD, does modafinil, compared with placebo, affect outcomes? 		
•	For effe	in in	dividuals with ADHD who are taking modafinil, what are the main side s?
•	For individuals with ADHD, does modafinil confer an advantage over any other pharmacological interventions?		
Evi	Evidence statements		
•	Modafinil versus placebo:		
		0	Children and adolescents: Six double-blind RCTs showed benefit of modafinil for ADHD symptoms (478-483). (Level II)
		0	Adults: One small RCT showed benefit of modafinil over placebo in reducing the symptoms of ADHD (454). (Level II)
		0	Long-term use: No studies over 3 months have investigated the safety and efficacy of modafinil in the treatment of ADHD.
•	Modafinil versus MPH:		
		0	Children and adolescents: One RCT found no difference in efficacy between MPH and modafinil in reducing the symptoms of ADHD (484). (Level II)
•	Мо	da	finil versus DEX:
		0	Adults: One RCT found no difference in efficacy between DEX and modafinil in reducing the symptoms of ADHD (454). (Level II)
Recommendations			
92.		Moo chil side	dafinil is not recommended as an alternative medication in school-aged dren and adolescents with ADHD due to concerns about the associated effects in this age group. (Grade C)
93.	. I	Mod not	lafinil may be considered as an alternative medication in adults with ADHD who do respond to stimulants or ATX. (Grade C)

Background

While modafinil is a stimulant medication, research in mice shows it to be unrelated in structure and function to central nervous system stimulants such as MPH and DEX that are commonly used in the treatment of ADHD (485, 486).

Modafinil is a wakefulness promoting medication. In Australia, it has been approved for use in sleep disorders such as narcolepsy, chronic shift work sleep disorder and obstructive sleep apnoea. Its use in psychiatry, including the treatment of ADHD symptoms, has also been investigated. Modafinil has been shown to affect the brain's catecholamine neurotransmitter systems, but the precise biochemical action of its impact on waking and behaviour is unknown (for review see (487)).

The FDA (August 2007, (488)) has not approved modafinil for use in any paediatric patients, as it found that safety and effectiveness in patients below age 16 have not been established, and that serious skin rashes, including erythema multiforme major (EMM) and Stevens-Johnson Syndrome (SJS), have been associated with its use in paediatric patients. An investigation of the abuse potential of modafinil (200, 400 and 800mg) relative to methylphenidate (45 and 90mg) with inpatients experienced with drugs of abuse, found that modafinil produced psychoactive and euphoric effects and feelings consistent with other scheduled central nervous system stimulants (methylphenidate).

For discussion of the abuse potential of modafinil, see section 8.9 ADHD and substance misuse, page 124.

Variations from approved Product Information

Modafinil is not currently licensed for the treatment of ADHD in Australia.

Summary of research evidence

Preschoolers

The safety and efficacy of modafinil has not been examined in preschool-aged children.

Children and adolescents

Six double-blind RCTs of short duration (4–9 weeks) found that modafinil reduced the number and severity of ADHD symptoms in children and adolescents with ADHD, compared to placebo (478-483). The most common side effects in the modafinil group were insomnia, headache, loss of appetite and weight loss.

Adults

One RCT (454) met the inclusion criteria. This small study, with 22 participants over 2 weeks, found that modafinil reduced the symptoms associated with ADHD as measured by self-report. The most common side effects were insomnia, muscle tension and irritability.

Modafinil versus stimulant medications

Modafinil was compared with MPH in children and adolescents with ADHD in a 6-week, paralleldesign RCT (484). Both the MPH and the modafinil groups showed a significant improvement in ADHD symptoms over time. There were no differences between the groups in parent- or teacherrated ADHD symptoms. The incidence of decreased appetite and difficulty falling asleep was increased in the group taking MPH.

Modafinil was compared with DEX in adults in a 6-week crossover RCT (454). Both medications were effective compared to placebo in reducing ADHD symptoms and there was no significant difference between the treatment outcomes for modafinil and DEX. There was no significant difference in the frequency of side effects between either of the medications and placebo.

General comments

The long-term safety and efficacy of modafinil has not been established for the treatment of individuals with ADHD in any population group, as no clinical trials of longer than 3 months have been reported. The outcome measures in trials to date have focused exclusively on ADHD symptoms, with no attempt to evaluate the impact on quality of life or social or school/work functioning.

8.4.3 Selegiline

Research questions		
•	For individuals with ADHD, does selegiline, compared with placebo, affect outcomes?	
•	For individuals with ADHD who are taking selegiline, what are the main side effects?	
•	For individuals with ADHD, does selegiline confer an advantage over any other pharmacological interventions?	
Εv	idence statements	
•	Preschoolers: No studies met the inclusion criteria.	
•	Children and adolescents : One small study demonstrated benefit of selegiline over placebo for ADHD symptoms of inattention and hyperactivity but not impulsivity (489). (Level II)	
•	Adults: No studies met the inclusion criteria.	
•	MPH versus selegiline:	
	 Children and adolescents: Two studies in children (490, 491) and one in adolescents (490) found no significant difference in efficacy between MPH and selegiline MPH versus selegiline. (Level II) 	
Recommendation		
94	 There are insufficient research data to recommend selegiline for the treatment of ADHD. (Grade D) 	

Background

Selegiline is a monoamine oxidase type B inhibitor which is used alone or in combination with levodopa to help control the symptoms of Parkinson's disease. It is metabolised to amphetamine and methamphetamine stimulant compounds, which may be useful in the treatment of ADHD by inhibiting the breakdown of dopamine and increasing synaptic dopamine levels.

Side effects can include stomach upset, loss of appetite, nausea, heartburn, dry mouth, dizziness, weakness, insomnia, back pain and constipation.

Variations from approved Product Information

Selegiline is not currently licensed for the treatment of ADHD in Australia.

Summary of research evidence

Preschoolers

No studies that met the inclusion criteria were identified.

Children and adolescents

One study, a 4-week crossover RCT with 11 participants (489), compared selegiline to placebo in children with ADHD. Selegiline significantly improved symptoms of inattention and hyperactivity (parent and teacher rated) but not impulsivity. Reporting of side effects was limited, but there was no significant difference in numbers of side effects for selegiline compared to placebo.

Adults

No studies that met the inclusion criteria were identified.

MPH versus selegiline

Two short-term RCTs compared selegiline with MPH in children (490, 491) and adolescents (490) with ADHD. Both studies found that both medications improved ADHD symptoms, with no significant difference in treatment efficacy between the medication groups. Those on MPH experienced decreased appetite, difficulty falling asleep and headaches more frequently than those on selegiline

(491). Neither study used a placebo control group, so it is not possible to draw direct conclusions regarding medication efficacy.

8.4.4 Guanfacine

Research questions

- For individuals with ADHD, does guanfacine, compared with placebo, affect outcomes?
- For individuals with ADHD who are taking guanfacine, what are the main side effects?
- For individuals with ADHD, does guanfacine confer an advantage over any other pharmacological interventions?

Evidence statements

- **Preschoolers:** No studies met the inclusion criteria.
- **Children and adolescents:** One study demonstrated benefit of guanfacine over placebo for the improvement of ADHD symptoms (492). (Level II)
- **Adults:** One study demonstrated benefit of guanfacine over placebo for the improvement of ADHD symptoms (455). (Level II)

• Guanfacine versus DEX:

• **Adults:** One study found no difference in efficacy between DEX and guanfacine for improvement of ADHD symptoms (455). (Level II)

Recommendation

95. Guanfacine may have a role in the pharmacological management of ADHD. (Grade C)

Note: Guanfacine is not currently available in Australia.

Background

Guanfacine is an alpha-2 adrenoreceptor agonist, and is most commonly used as an antihypertensive. It activates an inhibitory neuron, reducing sympathetic outflow and producing a decrease in vasomotor tone and heart rate.

Guanfacine is also used in the treatment of ADHD, and exists in both IR and ER forms.

Side effects include dry mouth, sedation, headache, dizziness, gastrointestinal effects, constipation, xerostomia and impotence.

Note: Guanfacine it is not currently available in Australia.

Summary of research evidence

Preschoolers

No studies that met the inclusion criteria were identified.

Children and adolescents

An 8-week, parallel-design RCT (492) compared 3 doses of guanfacine-ER (2, 3 and 4mg/day) to placebo. There was a significant improvement in ADHD symptoms with all guanfacine doses, compared to placebo. Adverse event reporting was limited, with no tests of significance conducted, but the most frequently reported side effects in the guanfacine group were somnolence, fatigue, upper abdominal pain and sedation. There was a dose-related increase in mean blood pressure and pulse rate.

Adults

In a 6-week crossover RCT (455), adults with ADHD received guanfacine, DEX and placebo for 2 weeks each. Guanfacine was effective compared to placebo in reducing ADHD symptoms, with no significant difference in the number of side effects.

Guanfacine versus DEX

In a 6-week crossover RCT (455), adults with ADHD received guanfacine, DEX and placebo for 2 weeks each. Both medications were effective compared to placebo in reducing ADHD symptoms, and there was no significant difference between the treatment outcomes for guanfacine and DEX. Numbers of side effects were similar for guanfacine and DEX, and there was no significant difference in the frequency of adverse events between either of the medications and placebo.

8.4.5 Nicotine patch

Re	Research questions	
•	For individuals with ADHD, do nicotine patches, compared with placebo, affect outcomes?	
•	For individuals with ADHD who are using nicotine patches, what are the main side effects?	
•	For individuals with ADHD, do nicotine patches confer an advantage over any other pharmacological interventions?	
Evidence statements		
•	Preschoolers: No studies met the inclusion criteria.	
•	Children and adolescents: One small study demonstrated benefit of nicotine patches compared to placebo for the improvement of ADHD symptoms (493). (Level II)	
•	Adults: No studies met the inclusion criteria.	
Recommendation		
96	. There are insufficient research data to recommend nicotine patches for the treatment of ADHD in school-aged children and adolescents. (Grade D)	

Background

Nicotine patches are widely used by people attempting to quit smoking, and they have recently been examined for their ability to relieve ADHD symptoms.

It has been hypothesised that nicotinic receptor stimulation may be useful in treating ADHD, and this is supported by evidence that activation of these receptors enhances dopaminergic neurotransmission (493).

Side effects may include dizziness, headache, nausea and sleep disturbances. The patch can also cause skin irritation.

Variations from approved Production Information

Nicotine patches are not currently licensed for the treatment of ADHD in Australia.

Summary of research evidence

Preschoolers

No studies that met the inclusion criteria were identified.

Children and adolescents

Use of a nicotine patch was compared to placebo in a 7-day pilot parallel-design RCT with 10 participants (493). The learning problems and hyperactivity subscale of the Conners' Parent Rating Scale (CPRS) was significantly improved in the nicotine group compared to placebo, with no other significant differences between groups (e.g. CPRS impulsivity, conduct and anxiety, and the Clinical Global Impressions of Severity scale). Reporting of adverse events was limited, but the most common side effects were nausea, stomach ache, itching under the patch and dizziness.

The nicotine patch was compared to placebo in a 7-day pilot parallel-design RCT with 10 participants (493). Only the learning problems subscale of CPRS was significantly improved in the nicotine group over placebo, with no other significant differences between groups (e.g. CPRS impulsivity,

hyperactivity, conduct and anxiety and CGI). Reporting of adverse events was limited, but the most common side effects were nausea, stomach ache, itching under the patch and dizziness.

Adults

No studies that met the inclusion criteria were identified.

8.4.6 Bupropion

Research questions

•	For individuals with ADHD, does bupropion, compared with placebo, affect
	outcomes in the short term (weeks/months)?

- For individuals with ADHD who are taking bupropion, what are the main side effects?
- For individuals with ADHD, does bupropion confer an advantage over any other pharmacological interventions?

Evidence statements

- Preschoolers, children, adolescents: No studies met the inclusion criteria.
- **Adults**: Two studies demonstrated benefit of bupropion compared to placebo for the improvement of ADHD symptoms (494, 495). Two studies demonstrated no difference in efficacy between bupropion and placebo (496, 497) (Level II)

• Bupropion versus MPH:

• **Adults:** One study found no difference in efficacy between MPH and bupropion for the improvement of ADHD symptoms (496). (Level II)

Recommendation

97. Bupropion may provide some benefit in treating ADHD symptoms in adults. It could be trialled in the absence of clinical response to stimulants and ATX. (Grade D)

Background

Bupropion was first marketed in Australia late in 2000 as a short-term aid to giving up smoking. It was initially developed as an antidepressant. It is a selective inhibitor of the neuronal reuptake of catecholamines (noradrenaline and dopamine) in the brain; however, the precise mechanism is unknown.

Several side effects, some of which may be severe, have been reported. The more commonly reported problems were skin reactions, neurological effects (headache, dizziness/ataxia, convulsions/twitching, tremor, paraesthesia/hypoesthesia), psychiatric effects (insomnia, agitation, depression, anxiety) and gastrointestinal effects (nausea, vomiting). Facial oedema, chest pain, serum sickness, shortness of breath and increased sweating were also reported. Bupropion is contraindicated in patients with epilepsy, and should be used with great caution in those with a predisposition to seizures. Care is also needed in prescribing bupropion for patients with a history of psychiatric conditions, and especially those utilising medication therapy.

Variations from approved Product Information

Bupropion is not currently licensed for the treatment of ADHD in Australia.

Summary of research evidence

Preschoolers, children and adolescents

No studies that met the inclusion criteria were identified.

Adults

Four trials compared the use of bupropion-ER to placebo in adults with ADHD. Two of these studies (494, 495) reported that bupropion improved symptoms of ADHD compared to placebo, and two (496, 497) reported no difference between the two groups. All were short-term assessments (6–8

weeks) with relatively small numbers of participants. Reporting of adverse events was limited, but side effects reported included headache, gastrointestinal problems, insomnia, aches or pains, dry mouth, chest pain, nausea and nasopharyngitis. One of the studies (495) also reported a significant increase in pulse over the course of bupropion treatment and a significant decrease in weight, while another (497) reported no significant difference in side effects between bupropion and placebo.

MPH versus bupropion

Bupropion-ER was compared to MPH and placebo in adults with ADHD (496). In this 8-week, parallel-design RCT, no significant difference was found between bupropion or MPH over placebo for either of the ADHD symptom measures used. There was no direct statistical comparison of bupropion and MPH. Reporting of adverse events was very limited, but the most common were: for bupropion, dry mouth, headache and insomnia; for MPH, appetite suppression and insomnia, tremor, sweats, jitteriness; and for placebo, tiredness.

8.4.7 Risperidone

Research questions

- For individuals with ADHD, does risperidone, compared with placebo, affect outcomes in the short term (weeks/months)?
- For individuals with ADHD who are taking risperidone, what are the main side effects?
- For individuals with ADHD, does risperidone confer an advantage over any other pharmacological interventions?

Evidence statements

• **Preschoolers, children, adolescents and adults:** No studies met the inclusion criteria.

Recommendation

98. There are insufficient research data to recommend the use of risperidone for the treatment of ADHD symptoms. (Grade D)

Background

Risperidone is an atypical antipsychotic from the chemical class of benzisoxazole derivatives. Risperidone has TGA approval for the following indications: schizophrenia and related psychoses; acute mania associated with bipolar I disorder; behavioural disturbances in dementia; conduct and other disruptive disorders in children (over 5 years), adolescents and adults; and behavioural disorders associated with autism in children and adolescents.

Several side effects are commonly reported, including restlessness, increased appetite, weight gain and metabolic disturbances.

With respect to ADHD, risperidone is usually considered for use in conjunction with stimulants when treatment-resistant aggression (498) or other disruptive behaviour disorders (499) are comorbid. For a discussion of the use of risperidone with ADHD and comorbid disruptive behaviour disorders see section 8.5.3, page 104.

Variations from approved Product Information

Risperidone is not currently licensed for the treatment of ADHD in Australia.

Summary of research evidence

Preschoolers, children, adolescents and adults

No studies that met the inclusion criteria were identified.

8.5 Medication management of ADHD when comorbidities are present

8.5.1 Introduction

As described in section 3.2, there are high levels of comorbidity between ADHD and a number of other conditions, including anxiety, tic disorders and disruptive behavioural disorders such as ODD and CD. For many comorbidities, psychosocial management will be indicated (see section 7.3 Psychosocial interventions for ADHD when comorbidities are present, page 78).

Pharmacological research has focused on children and adolescents, and no studies that addressed the efficacy of medications in adults with ADHD and comorbid disorders met the inclusion criteria.

Decisions about the best medication for a given patient will depend on:

- full assessment and consideration of the individual's impairing symptoms
- the effect size of the medication on the symptoms of the comorbidity
- the relative merits of medication use in relation to side effects.

Antidepressant medications

Recommendation

- 99. The RANZCP, RACGP and RACP Clinical Guidance and the Adverse Drug Reactions Advisory Committee (ADRAC) recommendations on the use of antidepressants in children and adolescents must be considered by the physician before prescribing antidepressants. There are no antidepressant medications with TGA-approved indications for treatment of children and adolescents with depressive symptomatology or major depressive disorder. There are significant concerns about serious adverse reactions (increased suicidality and thoughts of self-harm) in a small percentage of this patient group.
 - Recommended best practice based on clinical experience and expert opinion

Psychiatric disorders such as depression and anxiety are often comorbid with ADHD. In 2004 the ADRAC reviewed data on the safety and efficacy of selective serotonin reuptake inhibitors (SSRIs) (citalopram, escitalopram, fluoxetine, fluvoxamine, paroxetine and sertraline, and the related medicine, venlafaxine) in the treatment of major depressive disorder (MDD) and other psychiatric disorders in children and adolescents (500). In 2005, the RANZCP, RACGP and RACP provided clinical guidance on the use of antidepressant medications in children and adolescents that endorsed the ADRAC recommendations (501).

The published and unpublished data available for SSRI use in children and adolescents indicate that there is evidence of an increased risk of suicidality, including suicidal ideation, suicide attempts and self-harm events, associated with each of the SSRIs.

None of the SSRIs, and indeed no antidepressant, currently has any indication in its Product Information (i.e. is approved) in Australia for the treatment of depressive symptomatology or MDD in children and adolescents (persons aged less than 18 years). The recommendations of the ADRAC are that:

- Any use of SSRIs in children and adolescents with MDD and other psychiatric conditions should be undertaken only within the context of comprehensive management of the patient. Management should include careful monitoring for the emergence of suicidal ideation and behaviour which may particularly develop early in therapy, or if therapy is interrupted or irregular because of poor compliance. Cognitive behaviour therapy, if it is available, may enhance the outcome in MDD.
- 2. The choice of an SSRI for a child or adolescent with MDD or other psychiatric condition should be made only after taking into account the recent evaluations of clinical trial data and the Australian Product Information. Prescribers should be aware that the marketers of fluvoxamine and sertraline (indicated for OCD) advise against use in children and adolescents with MDD, and marketers of citalopram, escitalopram, paroxetine, venlafaxine and fluoxetine warn or caution against use in patients aged less than 18 years for any indication.

3. Children and adolescents being treated for MDD with an SSRI should not have their medication ceased abruptly.

8.5.2 Anxiety

Research question

• When anxiety is present in individuals with ADHD, do pharmacological interventions for ADHD, compared to placebo, affect outcomes?

Evidence statements

- **Preschoolers:** No studies met the inclusion criteria.
- Children and adolescents:
 - **MPH:** One study, comparing two groups with ADHD, with and without anxiety, found no significant differences in treatment response to MPH or adverse events between the groups (502). (Level II)
 - **ATX:** One study found a statistically significant reduction in all ADHD and anxiety measures with ATX compared to placebo (503). (Level II)
- Adults: No studies met the inclusion criteria.

Recommendations

- 101. In people with ADHD and anxiety, treatment with stimulant medication or ATX should still be considered to treat the ADHD symptoms. (Grade C)
- 102. If anxiety symptoms do not respond to psychosocial interventions, the use of selective serotonin reuptake inhibitors (SSRI) could be considered, either alone or in combination with a medication to treat the ADHD. Antidepressant medication should only be used in the context of comprehensive patient management and should be combined with psychological interventions. The RANZCP, RACGP and RACP Clinical Guidance and the Adverse Drug Reactions Advisory Committee (ADRAC) recommendations on the use of antidepressants in children and adolescents must be considered by the physician before prescribing antidepressants.

Variations from approved Product Information

None of the SSRIs, and indeed no antidepressant, currently has any indication in its Product Information (i.e. is approved) in Australia for the treatment of anxiety in children and adolescents (persons aged less than 18 years). Also see "Antidepressant medications" above (page 102).

Summary of research evidence

There are gaps in the evidence regarding long-term efficacy and side effects, relative benefits in anxiety subtypes and the impact of conjunctive non-pharmacological management of anxiety.

Preschoolers

No studies met the inclusion criteria.

Children and adolescents

Two RCTs have looked at the treatment of ADHD and comorbid anxiety disorder.

- In a four-month trial of MPH (502), the presence of anxiety did not influence the response to the medication or the numbers of adverse events.
- In a 12-week trial of ATX in the treatment of ADHD and comorbid anxiety (503), there was a reduction in symptoms of both ADHD and anxiety, but ATX treatment was associated with significantly more reports of decreased appetite and weight loss compared to placebo.

Adults

No studies met the inclusion criteria.

8.5.3 Other disruptive behaviour disorders (ODD, CD)

Research question

• When disruptive behaviour disorders are present in individuals with ADHD, do pharmacological interventions, compared to placebo, affect outcomes?

Evidence statements

- Preschoolers: No studies met the inclusion criteria.
- Children and adolescents:
 - MPH: One study demonstrated benefit of MPH compared to placebo in reducing ADHD symptoms in children and adolescents with comorbid ODD (504). (Level II)
 - **MPH plus clonidine:** One study found that MPH plus clonidine reduced ADHD symptoms and mildly reduced ODD symptoms compared to placebo in children and young adolescents with ADHD and ODD or CD (505). (Level II)
 - **MPH plus risperidone:** In one short-term study in children, the addition of risperidone to MPH led to further reductions in ADHD symptoms and reduced some symptoms of conduct disorder (499). (Level II)
 - **ATX:** One meta-analysis (three RCTs) (506) and two RCTs (507, 508) found that ATX reduced ADHD symptoms in children and adolescents with comorbid ODD compared to placebo. (Level II)

Recommendations

- 103. Non-pharmacological management of comorbid disruptive behaviour disorders should always be considered.

 Kecommended best practice based on clinical experience and expert opinion
- 104. In children and adolescents with ADHD and comorbid ODD, MPH or ATX should be considered to treat the ADHD symptoms. (Grade B)
- 105. In children and adolescents with ADHD and comorbid ODD or CD, MPH plus clonidine could be considered to treat the ADHD symptoms. (Grade C).
- 106. In children with ADHD and comorbid CD, MPH plus risperidone could be considered to treat the ADHD symptoms. (Grade C)

Summary of the evidence

There are gaps in evidence regarding long-term efficacy and side effects and the impact of conjunctive non-pharmacological management of other disruptive behaviour disorders such as ODD and CD.

Preschoolers

No studies met the inclusion criteria.

Children and adolescents

Six studies were identified that addressed the use of various medications for the treatment of children and adolescents with ADHD and comorbid disruptive behaviour disorders – ODD or CD.

A 9-week crossover RCT (504) investigated use of MPH to treat children (boys only) with ADHD plus ODD or CD. ADHD symptoms and conduct improved with both high and low doses of MPH, compared to placebo. Two participants withdrew due to side effects.

An Australian RCT conducted over 6 weeks (505) compared the addition of clonidine or placebo to ongoing stimulant treatment in children and young adolescents with ADHD plus CD or ODD. For children taking clonidine, compared to placebo, there was a significant reduction in parent- but not teacher-reported conduct and hyperactive symptoms. Improvement in symptoms became significant

in the clonidine group in weeks 5 and 6 of the trial. Side effects with clonidine included decreased pulse rate and systolic blood pressure, and increases in parent-reported drowsiness and dizziness.

Post-study analysis of two 6-week RCTs comparing the use of risperidone to placebo (either alone or with concurrent stimulant therapy) in children with ADHD and comorbid CD or other disruptive behaviour disorders (499) found that risperidone was more effective than placebo for measures of conduct, hyperactivity and irritability. The impact of risperidone did not vary with the presence or absence of stimulants. The most common adverse events in risperidone-treated patients were somnolence, headache, dyspepsia, rhinitis and vomiting. Increase in weight and BMI was also reported in the risperidone groups compared to placebo.

Three studies addressed the use of ATX compared to placebo in the treatment of children and adolescents with ADHD and ODD:

- Meta-analysis of three RCTs of children and adolescents with ADHD, with and without comorbid ODD (506), found that ATX significantly reduced ADHD symptoms, regardless of the presence or absence of ODD. There was no change in ODD symptoms. Adverse events were not described.
- For children and adolescents with ADHD plus ODD, there was a significant improvement in ADHD symptoms with ATX at 1.3mg/kg/day, and children with ADHD but not ODD experienced significant improvement with ATX at 1.2mg/kg/day (507). When ODD was present, treatment with all ATX doses reduced ODD symptoms compared to placebo.
- A 42-week RCT looked at the risk of relapse in children and adolescents with ADHD, with and without ODD, treated with ATX (508). Participants had previously shown improvement in ADHD symptoms in an 8-week open-label trial with ATX. ATX compared to placebo was effective in maintaining decreased ADHD symptoms, whether ODD was present or absent. In addition, ODD had no impact on the rate of relapse in ADHD symptoms during treatment. The main limitation of this study is that, at the start, participants were known responders to ATX treatment.

Adults

No studies met the inclusion criteria.

8.5.4 Depression

Research question		
When depression is present in individuals with ADHD, do pharmacological interventions, compared to placebo, affect outcomes?		
Evid	ence statement	
• Ir p in	n one 9-week study there was significant improvement with ATX compared to lacebo for global rating and parental ADHD rating, but there was no significant nprovement in depression symptoms (509). (Level II)	
Reco	mmendations	
107.	If major depression is suspected in a person with ADHD, consultation with a psychiatrist should be considered to review diagnosis and assist in management. Recommended best practice based on clinical experience and expert opinion 	
108.	Non-pharmacological management of comorbid depression should always be considered. ✓ Recommended best practice based on clinical experience and expert opinion	
109.	The relative contribution of any currently prescribed stimulant medication to the onset of depression should be reviewed. Trials off stimulant medication may be appropriate. ✓ Recommended best practice based on clinical experience and expert opinion	
110.	Where depression is identified in an adolescent or adult with ADHD, the potential contribution of illicit drugs to the onset of the depression should be considered. Recommended best practice based on clinical experience and expert opinion 	
111.	In people with ADHD and comorbid major depressive disorder, use of ATX could be considered to treat the ADHD symptoms. (Grade C)	
112.	The use of stimulant medication could be considered to treat ADHD symptoms in people with ADHD and comorbid major depressive disorder. Recommended best practice based on clinical experience and expert opinion 	
113.	Antidepressant medication may be considered to treat ADHD symptoms in adults with ADHD and comorbid moderate to severe depression. Recommended best practice based on clinical experience and expert opinion 	
114.	Antidepressant medication could be considered for adolescents with ADHD and comorbid moderate to severe depression. Antidepressant medication should only be used in the context of comprehensive patient management and should be combined with psychological interventions. The RANZCP, RACGP and RACP Clinical Guidance and the Adverse Drug Reactions Advisory Committee (ADRAC) recommendations on the use of antidepressants in children and adolescents must be considered by the physician before prescribing antidepressants. Recommended best practice based on clinical experience and expert opinion 	

Variations from approved Product Information

None of the SSRIs, and indeed no antidepressant, currently has any indication in its Product Information (i.e. is approved) in Australia for the treatment of MDD in children and adolescents (persons aged less than 18 years). Also see "Antidepressant medications" above (page 102).

Summary of research evidence

There are gaps in the evidence regarding the long-term efficacy, side effects and impact of conjunctive non-pharmacological management of depression.

Preschoolers

No studies met the inclusion criteria.

Children and adolescents

A 9-week, parallel-design RCT compared ATX to placebo in the treatment of ADHD and comorbid major depression. With ATX, compared to placebo, there was a significant improvement in ADHD symptoms, but not in depression symptoms, but ATX treatment was associated with significantly more nausea and decreased appetite.

Adults

No studies met the inclusion criteria.

8.5.5 Bipolar disorder

Research question

• When bipolar disorder is present in individuals with ADHD, do pharmacological interventions, compared to placebo, affect outcomes?

Evidence statements

- In one 4-week study there was significantly greater improvement in ADHD and bipolar symptoms with MPH, compared to placebo, for most outcomes (510). (Level II)
- In one 4-week study, following stabilisation with valproate, the addition of MAS was found to decrease ADHD symptoms (511). (Level II)

Recommendations

- 116. In adults with ADHD and non-psychotic comorbid bipolar disorder, use of lowdose stimulant medication, in addition to mood-stabilising medication, may be considered. (Grade C)

Background

At the time of writing, bipolar disorder of childhood has emerged as a possible comorbidity of ADHD, but there is controversy about the validity of this diagnosis in children.

Bipolar disorder in adolescents and adults is a well-established diagnostic entity.

Summary of research evidence

There are gaps in evidence regarding the long-term efficacy, side effects and impact of conjunctive non-pharmacological management of bipolar disorder.

Preschoolers

No studies met the inclusion criteria.

Children and adolescents

Two studies addressed the treatment of ADHD and comorbid bipolar disorder in children and adolescents:

- In a 4-week crossover RCT trialling MPH in 16 participants (510), symptoms of both ADHD and bipolar disorder improved with MPH compared to placebo. Side effects occurring more frequently with MPH than placebo were insomnia, decreased appetite and stomach ache.
- An initial 8-week open-label treatment with divalproex sodium to stabilise mania symptoms was followed by a 4-week crossover RCT comparing MAS to placebo in children and adolescents with ADHD and comorbid bipolar disorder (511). ADHD symptoms improved with MAS compared to placebo. Reporting of adverse events was poor.

Adults

No studies met the inclusion criteria.

8.5.6 Epilepsy

Research question

• When epilepsy is present in individuals with ADHD, do pharmacological interventions, compared to placebo, affect outcomes?

Evidence statement

• **Preschoolers, children, adolescents and adults:** No studies met the inclusion criteria.

Recommendations

- 117. In individuals with ADHD and well-controlled epilepsy, stimulant medication or ATX should be considered to treat the ADHD.

 Kecommended best practice based on clinical experience and expert opinion.
- 118. The impact of ADHD medication on seizure frequency should be monitored, especially in the initial medication trial period.
 - Recommended best practice based on clinical experience and expert opinion.

Background

The rate of ADHD in children with epilepsy is 3–5 times that of the general population (512). No double-blind, placebo-controlled studies have been reported that address the efficacy of medications for ADHD in this population. However, the available open-label and pre-clinical trials suggest that both MPH and ATX may be safe and efficacious for ADHD symptoms in children with epilepsy (for review see (512, 513)).

There has been some concern that stimulant medication could precipitate seizures or increase the number of seizures. The supporting information for this claim is limited, and data from several openlabel trials in children with ADHD and well-controlled epilepsy suggest that the risk of stimulantinduced seizures in people with stable epilepsy is minimal at therapeutic doses (514-517). Similarly, initial studies with ATX report improvement in ADHD symptoms and no elevated risk of seizures (518, 519). However, gaps in knowledge regarding long-term efficacy and side effects remain and larger double-blind placebo controlled trials are needed to establish safety and efficacy conclusively. Bupropion and some of the tricyclic antidepressants are contraindicated for individuals with epilepsy as they may lower seizure threshold (for review see (512, 513))

Variations from approved Product Information

The Product Information for MPH (Attenta, Ritalin 10, Ritalin LA and Concerta) states that there is some clinical evidence that MPH may lower the convulsive threshold in patients with prior history of seizures, in patients with prior EEG abnormalities in absence of seizures, and very rarely in absence of history of seizures and no prior EEG evidence of seizures. Safe concomitant use of anticonvulsants and MPH has not been established. In the presence of seizures, the drug should be discontinued. The Product Information for DEX states that DEX should not be taken by individuals with epilepsy. The Guidelines recommend that in individuals with ADHD and well-controlled epilepsy, stimulant medication should be considered to treat the ADHD. In other situations the clinician may need to make a judgement, in conjunction with the patient and/or carers regarding the risks and benefits of treating an individual who is severely disabled by ADHD in the presence of seizure risk.

Summary of research evidence

No studies met the inclusion criteria.

8.5.7 Tic disorders and Tourette syndrome

Research questions

- When tic disorders or Tourette syndrome are present in individuals with ADHD, do pharmacological interventions, compared to placebo, affect outcomes?
- For individuals with ADHD who are taking medication, what is the risk of developing first-onset tics or worsening existing tics?

Evidence statements: Efficacy for ADHD symptoms and tic severity

• **Preschoolers:** No studies met the inclusion criteria.

• Children and adolescents:

- MPH: One study demonstrated benefit of MPH over placebo for the improvement of ADHD symptoms but not tic severity in children and adolescents with ADHD and comorbid tics or Tourette syndrome (520). (Level II)
- **ATX:** One study demonstrated benefit of ATX over placebo for the improvement of ADHD symptoms but not tic severity in children and adolescents with ADHD and comorbid tics or Tourette syndrome (521). (Level II)
- ATX: One study demonstrated benefit of ATX over placebo for the improvement of ADHD symptoms and tic severity in children and adolescents with ADHD and comorbid Tourette syndrome (521). (Level II)
- Clonidine and MPH plus clonidine: One study demonstrated benefit of clonidine and MPH plus clonidine over placebo for the improvement of ADHD symptoms and tic severity in children and adolescents with ADHD and comorbid tics or Tourette syndrome (522). (Level II)

Evidence statements: First-onset tics or tic worsening

- Preschoolers:
 - MPH: One study found that MPH compared to placebo did not result in the worsening of existing tics in preschool-aged children with ADHD (446, 463). (Level II)
- Children and adolescents:
 - MPH: Data from five studies showed that treatment with MPH compared to placebo did not increase the risk of first-onset tics or worsening of existing tics in children and adolescents with ADHD (502, 523-526). (Level II (4 trials) and III-1 (1 trial))
 - MPH: Two studies found that treatment with MPH, compared to placebo, did not result in the worsening of existing tics in children and adolescents with ADHD and comorbid tic disorders or Tourette syndrome (520, 522). (Level II)
 - ATX: Two studies found that treatment with ATX, compared to placebo, did not result in worsening of existing tics in children and adolescents with ADHD and comorbid tic disorders or Tourette syndrome (521, 527). (Level II)
 - **Clonidine and clonidine plus MPH:** One study found that treatment with clonidine or clonidine plus MPH, compared to placebo, did not result in the worsening of existing tics in children and adolescents with ADHD and comorbid tic disorders or Tourette syndrome (522). (Level II)

Recommendations

119. In people with ADHD and a comorbid tic disorder or Tourette syndrome, stimulant medication is not necessarily clinically contraindicated and should

be considered to treat ADHD symptoms. (Grade C)

- 120. In people with ADHD and comorbid tic disorder or Tourette syndrome, use of ATX should be considered to treat the ADHD. (Grade C)
- 121. In children and adolescents with ADHD and a comorbid tic disorder / Tourette syndrome, clonidine should be considered if ADHD symptoms show poor response to stimulants or ATX. (Grade C)
- 122. If a tic occurs or becomes worse after commencing an ADHD medication, a clinical decision should be made in collaboration with the person with ADHD and/or the parents/caregivers on whether to:
 - continue the ADHD medication alone
 - o add an anti-tic medication, or
 - trial another ADHD medication.

The decision will be informed by the degree of response to the stimulant and the severity of the tics. (Grade C) $% \left(\left({{\rm{C}}_{{\rm{C}}}} \right) \right)$

Background

ADHD and tic disorders / Tourette syndrome frequently co-occur in children and adolescents. There has been ongoing debate about whether stimulant medications increase the risk of first-onset tics or worsen pre-existing tics (for review see (528)). In Product Information, stimulant medication is contraindicated in patients with motor tics, when siblings have tics or when there is a family history or diagnosis of Tourette syndrome (see below "Variations from approved Product Information"). In more recent expert clinical practice, if a tic disorder is comorbid with significant ADHD, stimulant medication may be used judiciously. The ability of stimulant medications to induce tics is widely known. Early concerns that stimulants were highly likely to induce tics (529, 530) have been shown to be generally unfounded (531). Emergent tics are seen in less than 10% of children treated with stimulant medication, with the vast majority being transient in nature (532). Short-term trials of stimulant medication in children with comorbid tic disorder indicate that the risk of exacerbation of tics is low (533, 534). Long-term follow-up of children with comorbid ADHD and tic disorder also suggest that chronic low-dose stimulant treatment does not exacerbate tic disorder (524, 535). It remains unknown whether the ability of stimulants to induce such tics is dose dependent. Given that the onset of tic disorder usually occurs in early childhood, there is low likelihood of inducing first ever tics in adults commenced on stimulant medication. There is a theoretical risk of exacerbating pre-existing tics or unmasking latent tics. Where severe ADHD and tic disorder co-exist, low-dose stimulant treatment remains the most prudent measure, with appropriate clinical monitoring for exacerbation of tics.

Variations from approved Product Information

The Product Information for Attenta, Ritalin 10 and Ritalin LA (MPH) states that these medications are contraindicated in patients with tics, tics in siblings or a family history or diagnosis of Tourette syndrome. The Product Information for Concerta states that Concerta is contraindicated in patients with a family history or diagnosis of Tourette syndrome. The Product Information for DEX states that DEX is contraindicated in patients with tics, a diagnosis of Tourette syndrome or a family history of Tourette syndrome. The Guidelines recommend that in people with ADHD and a comorbid tic disorder or Tourette syndrome, stimulant medication is not necessarily contraindicated and can be considered to treat ADHD symptoms.

Summary of research evidence: efficacy for ADHD symptoms and tic severity

Six studies were identified that addressed the use of various medications for the treatment of children and adolescents with ADHD and comorbid tic disorder or Tourette syndrome.

Clonidine, MPH, and clonidine + MPH were trialled in a 16-week, parallel-design RCT with children and adolescents with ADHD and comorbid Tourette syndrome, chronic motor tics disorder, or chronic vocal tics disorder (522). Compared to placebo, all medication groups showed significant improvement in ADHD symptoms, tic severity and global functioning, but the greatest benefit was with clonidine plus MPH. Medication did not worsen tics. Clonidine increased sedation, but there was no cardiac toxicity with any medication.

Three doses of MPH-IR were trialled in children with ADHD and comorbid Tourette syndrome or chronic motor tics disorder in an 8-week, crossover-design RCT (520). Compared to placebo, all doses of MPH produced significant improvements in ADHD symptoms, but there was no impact on tic severity. Adverse events with MPH-IR included impact on sleep, appetite, headache, upset stomach, dizziness, increased heart rate and diastolic blood pressure, and decreased body weight.

ATX was trialled in children and adolescents with ADHD and comorbid Tourette syndrome or chronic motor tics disorder in a 20-week, parallel-design RCT (521). Compared to placebo, ATX improved ADHD symptoms but not tic severity. Adverse events with ATX included decreased appetite, nausea and decreased body weight. In the group with ADHD and comorbid Tourette syndrome, a subgroup analysis (527) found that, compared to placebo, ATX improved ADHD symptoms and reduced tic severity on most measures other than self-report. Adverse events with ATX included increased mean pulse rate, nausea, and decreased appetite and body weight.

Guanfacine was trialled in children and adolescents with ADHD and comorbid tics in an 8-week, parallel-design RCT (536). Compared to placebo, guanfacine improved tic severity and, on most measures, ADHD symptoms. Adverse events did not differ between groups; however, a relative decrease in heart rate and blood pressure was reported in the guanfacine group.

Summary of research evidence: first-onset tics or tic worsening

Stimulant medication: Preschoolers

One clinical trial, a crossover trial of short duration and rated as poor quality, reported on tics in preschool-aged children receiving MPH (446, 463). The trial found no significant difference between high- and low-dose MPH (0.5mg/kg and 0.3mg/kg) and placebo for the number of participants displaying "tics or nervous movements".

Stimulant medication: Children and adolescents

None of the studies identified considered adolescents older than 14 years.

Five placebo controlled studies addressed the incidence of tics in children with ADHD receiving MPH:

- In a parallel-design RCT over 28 days, comparing OROS MPH, MPH-IR and placebo (526), 4 participants on placebo and 1 on MPH-IR (out of 277 total) reported either the first onset of tics or an increase in pre-existing tics.
- A 6-week crossover RCT that compared three doses of MPH (10, 20 and 30mg) (525) found that the number of participants with moderate or severe tics on at least 1 day was greatest in the placebo group and decreased with increasing doses of MPH.
- A 4-month crossover trial (502) found that there was no significant difference in the reporting of tics for those on MPH compared to placebo.
- In a 12-month clinical trial (524) there was no significant difference between the MPH and placebo in the number of participants reporting the first-onset of tics, or for those with pre-existing tics, in the number reporting worsening tics. The main limitation of this study was that, although initially randomised into even-sized groups, participants were allowed to change "medication" over the course of the trial.
- Pooled data from three short trials (1–4 weeks) comparing MPH-IR, OROS MPH and placebo (523) showed that the percentage of participants experiencing tics was not significantly different between either of the MPH formulations and placebo.

Two studies specifically looked at the use of tic worsening in children and adolescents with ADHD and comorbid tics or Tourette syndrome:

- A 16-week RCT (522) compared clonidine, MPH, or clonidine plus MPH with placebo. Compared to placebo, tic symptoms significantly improved in the MPH groups compared to placebo and there was no worsening of tics in the MPH groups.
- An 8-week crossover design RCT (520) compared 3 doses of MPH-IR to placebo. There were no differences in tic severity between any of the MPH-IR doses and placebo on physician and parent ratings, and improvement in tic severity on teacher ratings. The only scale showing a significant

increase in tics in the MPH group compared to placebo was the simple motor movements on the 2-Minute Tic/Habit Count.

Non-stimulant medication: Preschoolers

No studies were identified that met the inclusion criteria.

Non-stimulant medication: Children and adolescents

Two primary studies addressed the impact of ATX in children and adolescents with ADHD and comorbid tic disorders or Tourette syndrome:

- An 18-week, parallel-design RCT (521) found no increase in tic severity in participants receiving ATX compared to placebo, and on one of the three scales used to measure tic severity, there was a decrease in tic severity with ATX compared to placebo.
- The findings of a 20-week, parallel-design RCT (527) were similar: no increase in tic severity with ATX compared to placebo, and a decrease in tic severity on two of the three outcome measures that favoured ATX over placebo.

One primary study addressed the impact of clonidine in children and adolescents with ADHD and comorbid tic disorders or Tourette syndrome. A 16-week RCT (522) compared use of clonidine, MPH, clonidine plus MPH, and placebo. In the clonidine group, compared to placebo, tic severity improved and there was no worsening of tics.

Adults

No studies were identified that met the inclusion criteria.

8.6 Medications for ADHD in children with a developmental disability

8.6.1 Introduction

Developmental disabilities include pervasive developmental disorders (autism spectrum disorders), learning disabilities and intellectual disability. As discussed in section 3.2.1 Comorbidities in children and adolescents, current diagnostic criteria preclude these disorders being considered as comorbidities of ADHD. DSM-IV suggests, for example, that ADHD should not be diagnosed in the presence of an autism spectrum disorder. However, ADHD symptoms are commonly comorbid to these developmental disorders and these symptoms may cause significant impairment. There is considerable overlap between symptomatology of ADHD and these developmental disorders. The clinician must distinguish the impact and primary source of these symptoms. For example, when considering ADHD symptoms in a patient with intellectual disability, the clinician must consider whether ADHD symptoms are disproportionate to the level of intellectual disability and the relative merits and disadvantages of managing the ADHD symptoms. The management of ADHD symptoms comorbid to these developmental disorders needs to be considered in the context of overall management.

Although a formal comorbid diagnosis of ADHD is precluded, it is clear that children with developmental disabilities often have additional psychiatric disorders and behavioural problems.

When considering treatments for ADHD symptoms in individuals with autism or learning disabilities, careful assessment is required due to the increased risk of medical issues in this group. Treatment may need to be more cautious if there are significant neurological problems due to the increased risk of side effects. Of particular concern is finding a way to enable inclusive discussions about the medication and the occurrence of side effects with individuals who have difficulty communicating. Carers will need to be actively involved in watching for evidence of side effects.

8.6.2 Learning disabilities

Research question

• When ADHD is present in individuals with learning disabilities, do pharmacological interventions, compared to placebo, affect outcomes?

Evidence statement

• In one small study treatment with MPH had a lesser efficacy in treating children with ADHD and a learning disability, particularly in maths, than children with ADHD alone (537). (Level II)

Recommendations

- 124. Stimulant medication or ATX should be considered to improve the ADHD symptoms.
 - ✓ Recommended best practice based on clinical experience and expert opinion

Background

Learning disorders are common in individuals with ADHD. However, research into the impact of ADHD medications in people with ADHD and a comorbid learning disorder is rare.

There have been some preliminary studies that have looked at the impact of medication on very specific components of learning such as comprehension skills in children with ADHD. In a placebo controlled crossover trial of 16 children aged 7–12 years, MPH improved the child's ability to make inferences from complex explanations (538). However, medication did not impact upon comprehension of basic sentences or factual information (538). MPH has also been found to improve some aspects of story retelling in children with ADHD such as sensitivity to emotional information and actions taken by the characters, but not in retelling errors or grammar or comprehension (539). In a small trial of MPH in children with ADHD and comorbid dyslexia (540), improvement was seen in the number of correctly read words; however, reading level remained below average.

Summary of research evidence

One study was identified that looked at the clinical efficacy of MPH in children with ADHD and a comorbid learning disorder (537). This 2-week, crossover-design RCT, compared children with ADHD only and ADHD plus a learning disorder. Significantly fewer children with ADHD and a learning disorder were classified as clinical responders to MPH (55%) compared to children with ADHD alone (75%). Notably, children with a disability in mathematics responded less well than those with reading disability only (537). Adverse events were not reported.

8.6.3 Intellectual disability

Research question

• When ADHD symptoms are present in individuals with intellectual disabilities, do pharmacological interventions, compared to placebo, affect outcomes?

Evidence statements

- **Preschoolers, adolescents and adults:** No studies were found that met the inclusion criteria.
- **Children:** In two short-term studies, medium-dose MPH decreased ADHD symptoms in children with intellectual disability (541, 542). (Level II)

Recommendations

- 126. In people with an intellectual disability and ADHD, use of MPH should be considered. (Grade C)

Summary of research evidence

Two studies were identified that addressed the use of MPH in children and adolescents with ADHD and intellectual disability:

- A 4-week, crossover-design RCT compared three doses of MPH (541). Children with ADHD and intellectual disability showed improvement in ADHD symptoms, as rated by teachers but not by parents, for the highest MPH dose, compared to placebo. There were significant increases in loss of appetite and sleeping problems with MPH.
- In three 2–4-week, crossover-design RCTs (542) in children and adolescents with ADHD and intellectual disability, low-dose MPH improved ADHD symptoms significantly compared to placebo. The authors noted, however, that the effect sizes were moderate and not as high (~half) as those seen in studies of children with a normal IQ. Adverse events were not reported.

Research question When ADHD symptoms are present in individuals with an autism spectrum • disorder, do pharmacological interventions, compared to placebo, affect outcomes? **Evidence statements** Children and adolescents Three RCTs have demonstrated benefit of MPH over placebo in reducing 0 hyperactivity, inattention and impulsivity symptoms in children and young adolescents with an autism spectrum disorder (543-545). (Level II) One RCT has demonstrated benefit of ATX over placebo in reducing 0 hyperactive/impulsive, but not inattentive, symptoms in children and young adolescents with autism spectrum disorders (546). (Level II) **Recommendations** 127. Where a person with an autism spectrum disorder also has disabling ADHD symptoms, management of that disorder should be an integral part of overall management. Recommended best practice based on clinical experience and expert opinion 128. In children with an autism spectrum disorder and disabling symptoms of ADHD, use of stimulant medication or ATX should be considered to treat ADHD symptoms. (Grade C) 129. Careful monitoring is required due to the possibility of exacerbating ritualistic behaviours and stereotypies. Recommended best practice based on clinical experience and expert opinion

Background

Symptoms of inattention, hyperactivity and impulsivity are frequently observed in children with pervasive developmental disorders, including autistic disorder (autism), Asperger's disorder, and pervasive developmental disorders not otherwise specified (PDD NOS). For example, in a survey of 487 non-clinically referred young people with pervasive developmental disorders, over 50% were found to have moderate to severe problems in areas such as distractibility, attention, hyperactivity and excitability (281). In addition to the RCTs reviewed here, there have been several clinical trials of pharmacological treatments of individuals with autism spectrum disorders and symptoms of inattention, hyperactivity and impulsivity (for review see (547)).

There is some overlap in the symptomatology of autism spectrum disorders and ADHD. As discussed in section 3.2 Comorbidities, the clinician must distinguish the primary source of symptoms and consider the management of ADHD symptoms in the context of overall management of the patient.

The commonly utilised ADHD medications, MPH and ATX, are discussed below. Lofexidine has also been examined in one small study, in boys with autistic disorder and ADHD symptoms (546). While parent and teacher ratings for hyperactivity, irritability, stereotypy and inappropriate speech showed significant improvement, compared to placebo, no clinician ratings were significant. Drowsiness and decreased activity were more frequent side effects seen for lofexidine compared to placebo.

Variations from approved Production Information

MPH, DEX and ATX are not currently licensed for the treatment of individuals with autism spectrum disorders in Australia. The Guidelines recommend that in children with an autism spectrum disorder, use of stimulant medication or ATX should be considered to treat disabling ADHD symptoms.

Summary of research evidence

Three crossover RCTs of short duration (3–14 weeks) were identified that considered the impact of stimulant medication (MPH) on ADHD symptoms in children and adolescents with autism spectrum
disorders (543-545). The three studies found improvements in symptoms of hyperactivity, inattention and impulsivity with MPH compared to placebo (543-545). Significant side effects were reported with MPH compared to placebo; these included social withdrawal, dullness, sadness and irritability (543) and appetite decrease, difficulty falling asleep, irritability and emotional outburst (544, 545). One study reported benefits in hyperactive/impulsive, but not inattentive, symptoms with ATX compared to placebo (546). Side effects of mild upset stomach and/or nausea/vomiting occurred in all participants taking ATX compared to five on placebo. Fatigue and racing heart rate also occurred significantly more often when children were treated with ATX compared to control.

8.7 Side effects: special considerations

8.7.1 Introduction

When considering prescribing medications for ADHD the physician should consider the side effects associated with the medication and consult the relevant approved Product Information documents. In addition to the common side effects and contraindications listed below, special consideration is given to the impact of ADHD medications on growth (section 8.7.2), cardiac events (section 8.7.3) and psychiatric disturbance (section 8.7.3)

Common side effects and contraindications - stimulant medications

The common side effects of stimulant medications include decreased appetite, sleep problems, headaches and irritability/nervousness (548). Less common side effects include stomach aches, nausea, abdominal discomfort, weight loss, tearfulness, stuttering, dizziness, and increased heart rate and blood pressure. Rare but more severe side effects can include psychotic symptoms and sensitivity reactions that will require discontinuation of the medication.

Stimulants are generally contraindicated in individuals with hyperthyroidism, phaeochromocytoma, glaucoma, known hypersensitivity or idiosyncrasy to the sympathetic amines and concurrent treatment (or treatment within 14 days) with monoamine oxidase inhibitors (MAO) Stimulants are contraindicated in people with significant cardiovascular disorders as they increase heart rate and blood pressure and therefore have the potential to increase the risk of sudden cardiac death.

Common side effects and contraindications – atomoxetine

The common adverse effects associated with ATX include decreased appetite, drowsiness, abdominal pain, nausea and vomiting, dizziness, and increased heart rate and blood pressure (548). Less common side effects include dyspepsia and mood swings. There have been 3 published case reports of severe liver injury in children and adults using ATX (549, 550). These individuals recovered when the drug was discontinued. Suicide-related behaviour (suicide attempts and suicidal ideation) has been reported in patients treated with atomoxetine (see section 8.7.4).

Atomoxetine is contraindicated for people with narrow-angle glaucoma and concurrent treatment (or treatment within 14 days) with monoamine oxidase inhibitors (MAO). It is also contraindicated in people with structural cardiac abnormalities, symptomatic cardiovascular disease or other serious heart problems as it increases heart rate and blood pressure and therefore has the potential to increase the risk of sudden cardiac death.

8.7.2 Growth

Res	earch	n question		
•	• For preschoolers, children and adolescents with ADHD who are taking medication, what is the risk of impaired growth?			
Evi	dence	e statements: MPH, DEX and MAS		
•	Presc	hoolers:		
	0	One study reported significant decreases in expected gain in height and/or weight following stimulant treatment (551). (Level II)		
•	Child	ren:		
	0	Eight of 10 studies reported a significant decrease in growth rate in height and/or weight during stimulant treatment (552-559). (Level III-2 – aetiology)		
	0	Four of five studies found increased dose was linked to a decrease in growth rate for height but not weight (462, 553, 554, 556). (Level III-2 – aetiology) One study found increased dose was linked to a decrease in growth rate for weight but not height (559). (Level III-2 – aetiology)		
	0	Four studies reported that the impact of stimulant treatment on growth was greatest at the start of the treatment period, when children are stimulant naïve (462, 556-558). (Level III-2 – aetiology)		
	0	Two studies showed that the risk of impaired growth is greater in younger children (462, 556). (Level III-2 – aetiology)		
	0	One study found no difference in adult height between individuals treated with stimulant medication in childhood (3 years) and untreated controls (560). (Level III-2 – aetiology)		
Evi	dence	e statements: ATX		
•	Child	ren and adolescents:		
	0	Three studies (2 pooled analyses) reported impaired height and weight gain in children following treatment with ATX (561-563). (Level III-2 – aetiology)		
	0	One study that considered children and adolescents showed that the risk is greater in children under 9 years (563). (Level III-2 – aetiology)		
Rec	comm	endations		
130	 If a child is prescribed stimulant medication or ATX, growth parameters should be assessed at baseline and monitored every 3–6 months. (Grade B) 			
131	. Mor gro √R	Monitoring should include plotting of growth parameters (weight, height, growth velocity) on appropriate charts. Recommended best practice based on clinical experience and expert opinion 		
132	. For me con	For children most at risk of growth attenuation (younger age, higher dose), medication at low doses only and/or other intervention strategies should be considered. (Grade C)		
133	. Wh inac "ho per	Where there are concerns about reduced growth velocity (crossing centiles or inadequate weight gain), intervention strategies might include medication "holidays" (dose reduction or cessation during weekend and vacation periods).		
	۲K			

Variations from approved Product Information

Please note: In relation to the issue of growth suppression, the TGA has recently (April 2009) added the following precautionary statement to the Product Information for all products containing methylphenidate:

Careful follow up of weight and height in children aged 7 to 10 years of age who were randomised to either methylphenidate or non-medication treatment groups over 14months, as well as in naturalistic subgroups of newly methylphenidate-treated and non-medication treated children over 36 months (ie to the ages of 10 to 13 years), suggests that consistently medicated children (ie treatment for 7 days per week throughout the year) have a temporary slowing in growth rate (on average, a total of about 2cm less growth in height and 2.7 kg less growth in weight over 3 years), without evidence of growth rebound during this period of development.

Published data are inadequate to determine whether chronic use of amphetamines may cause similar suppression of growth, however it is anticipated that they likely have this effect well. Therefore, growth should be monitored during treatment with stimulants, and patients who are not growing or gaining height or weight as expected may need to have their treatment interrupted.

Background

For all medications, the clinician needs to consider the relative risk of side effects compared to the benefits. The effect of stimulant medications on growth has been a concern for many years and the impact of ADHD medication on growth has been the subject of ongoing debate in the literature (for review see (564)). While there are many studies considering the impact of ADHD medications on growth, most have considerable limitations. Few use concurrent controls of either children with ADHD not taking medication or children without ADHD; instead, the majority use normative data as their controls, but these normative controls are not necessarily well matched with the study population. In several cases, at study inception participants were significantly larger than the normative controls. In addition, many of the studies were conducted retrospectively, analysing data from medical records or clinical trials; however, in retrospective analysis, it may not be possible to consider potential confounding variables such as adherence with medication.

Summary of research evidence

There are gaps in the body of evidence in regard to the effect of medications on ultimate adult height.

Preschoolers and stimulants

The one study identified found that after 1 year of MPH treatment, both height and weight of participants were lower than expected based on predictions from baseline height and weight (551).

School-aged children and stimulants

Ten studies addressed the impact of stimulant treatment on growth in children. Two studies utilised concurrent control groups (552, 554). However, the majority used population-based normative data as a control (462, 535, 553, 555-559).

- A retrospective review of medical records of children with ADHD, using matched siblings as controls, found the height and weight of children with ADHD to be significantly lower than the controls after 3 years of treatment with MPH (552).
- Analysis of height and weight data for children in the MTA study who had been treated with stimulants found the gain in height after 3 years on MPH to be significantly less than in the two control groups (children with ADHD who had never taken stimulant medication and matched classmates) (554). Medication dose was significantly related to deficit in height, but not in weight.
- An Australian group conducted a retrospective review of 3.5 years of medical records of children receiving either MPH or DEX (557). The study reported a significant slowing of growth in both height and weight after 6 months of stimulant treatment, which remained significant at 30 months. Increase in height was slowest in the first 6 months of stimulant medication.
- A retrospective review of medical records found that children receiving MAS, MPH or DEX for 3 years had significant slowing of growth in weight over the first 24 months of treatment, which was no longer significant at 36 months (558). BMI was significantly affected at 3 months and height at 6 months.

- A data model based on retrospective analysis of annual height and weight measures over 5 years of treatment with MPH or DEX predicted a significant slowing of growth in weight after 1 year of treatment at doses over 1.5mg/kg/day, and a significant slowing of growth in height after 4 years at doses over 2.5mg/kg/day (555).
- Children with ADHD and comorbid tics or Tourette syndrome who received MPH treatment for 2 years showed a non-significant slowing of growth in height and weight (535).
- Retrospective analysis of growth measures in children receiving MPH-ER for 3.5 years showed a non-significant slowing of growth in height, weight and BMI (462). Subgroup analysis revealed that children who were stimulant naïve at the start of the study had a significant slowing of growth in weight, and younger children had a significant slowing in both height and weight gain. Dose was significantly related to the deficit in height but not in weight. The quality of this study was rated as poor.
- Among children treated with MPH or MAS for 1–3 years, gains in weight and BMI were significantly decreased, while the slowing in height gain was not significant. After 3 years' treatment, children receiving MAS showed a greater slowing in weight and BMI gain, but not height, than children on MPH (553). The quality of the study was rated as poor.
- In children on MAS-ER for 6–30 months (556), decreases in rate of growth in height and weight were significant at 14 months, but not in the second year of treatment. Medication dose was significantly related to the deficit in height but not in weight. Larger weight and height deficits were seen in the younger children when results were stratified for age. The quality of the study was poor.
- Among children on MPH-transdermal for 6–36 months, there was significant slowing of growth in height, weight and BMI (559). Medication dose was significantly related to the weight and BMI deficits but not height. Larger weight and BMI deficits were seen in children when they were stimulant naïve.

Although the studies are disparate in methodology, and so difficult to compare, there do appear to be several trends emerging. Eight of the ten studies reported significant deficits in height and/or weight following stimulant treatment (552-559). The remaining two studies reported a non-significant deficit in height and weight (462, 535). In the four out of five studies that considered the impact of dose, increased dose was linked to a deficit in height but not weight (462, 553, 554, 556). The fifth study found that dose was related to a deficit in weight but not height (559). When results were analysed by age, the impact of treatment on growth was found to be greatest in younger children (462, 556). Four studies reported that the impact of stimulant treatment on growth was greatest at the start of the treatment period (462, 556-558).

Adults

One study addressed the impact of ADHD medications on ultimate late adolescent or adult height (560). The study found no significant difference between individuals treated with stimulants in childhood and the untreated control group; 4.4% of variation in height was attributable to medication side effects such as nausea and vomiting, and 3.2% of variation in weight, to medication dose.

Non-stimulant medication

Three studies addressed the impact of the non-stimulant medication ATX on growth, using normative control data:

- A meta-analysis of weight and height data drew on 13 clinical trials involving children aged 6 and 7 who had received treatment with ATX for 2 years or more (561). Both height and weight were lower than expected. The growth rate differences primarily occurred in the first 18 months of treatment.
- Weight and height data were assessed from 13 clinical trials involving children and adolescents aged 6–16 who had received treatment with ATX for 2 years or more (563). Overall, there was a reduction in expected weight and height. The impact was greater in children starting treatment before 9 years of age.
- In a long-term open-label trial with 61 participants who had received ATX continuously for 5 years (562), significant deficits in height and weight were seen in the first 2 years of the study,

but there were no significant differences between participants and the control normative data set at the 5-year time point.

8.7.3 Cardiac events

Research question

• For individuals with ADHD who are taking medication, what is the risk of cardiovascular adverse events?

Evidence statements

- The research data do not support a clear relationship between sudden death, cardiovascular and cerebrovascular events and use of pharmacological agents to treat ADHD in children with no pre-existing cardiac risk factors (565, 566). (Level III-3 – prognosis)
- There is a significant risk of mildly increased heart rate and systolic and diastolic blood pressure in patients treated with stimulants (567-571) or atomoxetine (572). (Level II, IV)
- The short-term and long-term clinical significance of these changes is uncertain in most patients.
- One retrospective study found current stimulant use was associated with an increase in risk of hospital admission for cardiac causes (573). (Level III-2 – aetiology)

Recommendations

- 134. Before initiating pharmacological treatment for ADHD, cardiac risk factors should be assessed, including:
 - history of congenital heart disease or arrhythmias, palpitations, exercise intolerance or chest pain
 - family history of early cardiac disease (<50 years) or unexplained sudden death
 - cardiovascular examination. (Grade B)
- 135. Routine monitoring at review appointments should include heart rate and blood pressure, and recent history of palpitations, exercise intolerance and chest pain. (Grade B)
- 136. Specialist cardiologist advice is recommended for people with cardiovascular risk factors (congenital heart disease, arrhythmias, family history) and in whom blood pressure or heart rate exceeds normal limits. (Grade B)

Background

The risk of adverse cardiovascular effects of medication, and the modifying factors in relation to this risk, require particular vigilance by the clinician.

The American Academy of Pediatrics and the American Heart Association have recently released a joint statement on the cardiovascular evaluation and monitoring of children and adolescents with heart disease who are also receiving medications for ADHD (574). This statement is consistent with the recommendations above. It should be noted that further assessment and an FDA review is awaited at the time of publication.

Summary of research evidence

ADHD medications and sudden death

The USA Food and Drug Administration (FDA) has conducted two post-marketing safety reviews of sudden deaths during treatment with ADHD medications (565, 566). The reviews include episodes of sudden death, non-fatal cardiovascular events and cerebrovascular events which have been submitted to the FDA. Cases of sudden death are rare and in many cases structural cardiovascular

abnormalities or predisposing factors for sudden death were present prior to commencing medication. The authors point out that the review is limited by a number of factors (565). First, the adverse event reporting system involves unsolicited reporting which can produce incomplete reporting, under-reporting or greater reporting with recently approved drugs (565). Second, medication exposure is estimated, based on prescription rates, and is not true exposure (565).

MPH, DEX and MAS and cardiovascular function

Five studies directly addressed the impact of stimulant medication (MPH, MAS and DEX) on cardiovascular function (567-571). Three were conducted with children, one with adolescents and one with adults. All studies reported increases in heart rate and blood pressure when taking stimulant medications compared to placebo.

A retrospective cohort analysis looked at Medicaid data over a 10-year period for children and adolescents with ADHD (573). Current stimulant use was associated with a 20% increase in risk of emergency department admissions for cardiac-related incidents and a 21% increase in risk for hospital admissions for cardiac causes. The stimulants considered were MPH, amphetamines and pemoline (pemoline is no longer used due to concerns about liver toxicity). Data were not stratified by stimulant type.

There is no information on the long-term effects of stimulant medication on cardiovascular function.

ATX and cardiovascular function

Combined data from 5 RCTs in children, adolescents (3 trials) and adults (2 trials) with ATX showed statistically significant increases in mean pulse rate for all groups, mean systolic blood pressure in adults, and diastolic blood pressure in children and adolescents. Cardiac adverse events led to withdrawal of 4 out of 258 adults, but no children or adolescents (n = 335) (572). The longest trial included in this study was 10 weeks.

There is no information on the long-term effects of ATX on cardiovascular function.

8.7.4 Psychiatric disturbance

Recommendations

- 137. For individuals taking ADHD medications there is a rare but serious risk of psychiatric adverse events. Before prescribing ADHD medication, the person with ADHD and/or their caregiver should be specifically told of the risk of emergent psychiatric adverse effects, including aggression, anxiety, mania and psychosis. (Grade B)
- 138. As part of ADHD medication management, the person with ADHD and/or their caregiver should be specifically asked about emergent psychiatric adverse effects. (Grade B)
- 139. This is especially important for people who are receiving both ADHD medications and other psychotropic medications, have significant psychiatric co-morbidities and/or have a family history of psychiatric illness. (Grade B)

Background

Safety concerns relating to ADHD medications include the rare but serious risk of psychiatric adverse events such as suicidal thoughts and behaviours, aggression, unusual thoughts, changes in mood and depression (472, 575). The risk of adverse psychiatric effects of medication, and the factors that modify this risk, will need to be monitored by the clinician.

Psychiatric adverse events such as suicidal ideation are rare, but irrespective of medication use, children and adolescents with mental health problems are at increased risk of suicidal ideation and behaviours. For example, the Australian National Survey of Mental Health and Well-Being found a strong association between mental health problems and suicidal ideation and behaviour in adolescents (576). Other Australian studies have reported high levels of suicidal ideation and suicidal behaviours in a range of population groups. A study of 140 young people referred to psychiatric services found that up to 57% reported considering suicide and 39% reported attempting suicide in the 12 months prior to referral (577). In a general practice population of 15 to 24 year olds, 22% of 3,242 consecutive patients presenting to participating general practitioners during a

specified 6-week period had clinically significant levels of suicidal ideation (578). Finally, in a survey of 91 adolescents (13–17 years old) in home-based foster care, 6.7% reported suicide attempts requiring treatment (579).

There is one published Australian case report of an 11-year-old boy who developed acute agitation and suicidal ideation when taking ATX (580). As of 18 April 2008, the TGA had received reports of 13 incidents of suicidal ideation and 2 suicide attempts in individuals receiving ATX, 2 incidents of suicidal ideation in individuals receiving DEX, and 2 incidents of suicidal ideation and 2 suicide attempts in individuals receiving MPH. This is a cumulative listing of all reactions to these medications reported to the TGA since their approval for use in Australia. It is difficult to put these figures into perspective for several reasons. Incidence is considered unreliable, as reporting is primarily voluntary and accurate data on the numbers of patients receiving these medications and the number of days the medications have been used cannot be calculated from Australia's current systems of prescription and adverse event monitoring. Further, as noted above, the prevalence of suicidal ideation in the population of young people prescribed medication for ADHD is likely to be higher than in the general population and it is difficult to differentiate background risk from medication side effects.

Atomoxetine

Research question

• For individuals with ADHD who are taking ATX, what is the risk of psychiatric adverse events?

Evidence statements

- One review found that psychosis and mania events, suicidal ideation and aggression events were more frequent in children with ATX treatment than with placebo, but the difference did not reach statistical significance. There were no suicides in ADHD trials (581). (Level I aetiology)
- Two reviews (14 RCTs) found incidence of suicidal ideation was significantly greater in ATX-treated paediatric patients compared with placebo (582, 583). (Level I – aetiology)
- No suicidal ideation or behaviour was identified in three trials of adults taking ATX for ADHD (583). (Level I – aetiology)

Recommendation

140. All people for whom ATX is being considered should be told of the possibility of suicidal ideation, and people on ATX should be monitored for this side effect with extra vigilance. (Grade B)

Summary of research evidence

Three meta-analyses address the association of psychiatric adverse events with the ADHD medication ATX.

A review by the FDA of the ADHD medication clinical trial data (581) included trials of preschoolers, children, adolescents and adults. Psychosis and mania events occurred with every compound. The numbers of such events with medication treatment were small; however, no psychosis or mania events have been reported in the placebo groups (581).

Psychosis/mania events (16/7729), suicidal events (153/7729) and aggression events (247/7729) were reported in double-blind and open-label trials of ATX (Strattera). In the double-blind trials these events were considered to be more frequent with ATX treatment than with placebo. The difference did not reach statistical significance. There were no suicides in these trials.

Meta-analysis of available ATX clinical trial data in children drew on 14 RCTs for either ADHD or nocturnal enuresis (582, 583). The incidence of suicidal ideation was significantly greater in ATX-treated paediatric patients (6/1357) compared with placebo (0/851). There was no difference in the frequency of suicidal ideation or behaviour between participants receiving ATX (1/588) or MPH (1/464) (582).

No suicidal ideation or behaviour was identified in three trials of adults taking ATX for ADHD (583).

The available clinical trial data are of short- to medium-term duration. Long-term effects are not known.

Stimulant medication

Research question

• For individuals with ADHD who are taking stimulant medication, what is the risk of psychiatric adverse events?

Evidence statements

- Most RCTs do not show enhanced risk of these outcomes with regard to stimulant medication. Some RCTS, and some open-label and post-marketing information, have reported rare events in relation to stimulant medications. However, the events are of significant impact (581). (Level I – aetiology).
- One study in children and adolescents (582) reported risk of suicidal ideation with MPH was equivalent to the risk with ATX. (Level I aetiology)

Recommendation

141. All people for whom stimulant medication is being considered should be told of the possibility of suicidal ideation, and people on stimulant medication should be monitored for this side effect with extra vigilance. (Grade B)

Summary of research evidence

A review by the FDA of the ADHD medication clinical trial data (581) included trials of preschoolers, children, adolescents and adults. The numbers of psychosis and mania events with medication treatment were small; however, no psychosis or mania events have been reported in the placebo groups (581).

There were no psychiatric adverse events in double-blind trials of 321 participants receiving MPH-ER (Concerta) and 317 participants receiving placebo. Psychosis/mania (8/284), suicidal events (5/284) and aggression events (52/284) were reported in open-label trials. There were no suicides in these trials.

In double-blind and open-label trials of MPH-ER (Ritalin), psychosis/mania (2/508), suicidal events (1/508) and aggression events (2/508) were reported. There were no suicides in these trials.

Comparison of MPH and ATX found no difference in the frequency of suicidal ideation or behaviour between participants receiving ATX (1/588) or MPH (1/464) (582).

The available clinical trial data are of short- to medium-term duration. Long-term effects are not known.

8.8 Monitoring of medication effectiveness and side effects

Recommendations

142. Individuals on medication for ADHD should be monitored by their treating doctor for medication effectiveness and side effects. This should occur frequently in the early phase of treatment, and thereafter at intervals of 3–6 months.

✓ Recommended best practice based on clinical experience and expert opinion

- 143. Monitoring should include reports from caregivers, partners and/or teachers (as applicable). Brief parent and teacher behaviour rating scales can also contribute to monitoring of behaviour or performance at home or at school. Recommended best practice based on clinical experience and expert opinion
- 144. Monitoring at each visit should include height, weight and growth in children/adolescents and psychiatric disturbance, heart rate and blood pressure in all age groups. (Grade B)
- 145. Trials off medication should be conducted to evaluate whether the medication is still clinically indicated.
 ✓ Recommended best practice based on clinical experience and expert opinion

8.9 ADHD and substance misuse

8.9.1 Diversion and misuse of prescription medications for ADHD

There is a risk that a person prescribed medication for ADHD might misuse the drug/s prescribed and/or distribute them for illicit use by others.

A systematic review published in 2008 considered 21 studies addressing these issues (584). Two studies were included that addressed the diversion of medication from those with ADHD to others (584). The review found that:

- 5–9% of school students and 5–35% of college students had used a non-prescribed stimulant over the previous 12 months.
- Between 16 and 29% of students with stimulant prescriptions had at some time been asked to give, sell or trade their medications.
- The reasons individuals reported for misusing stimulants were to enhance performance, selfmedicate for ADHD symptoms (IR and ER stimulants) and for their euphorogenic effects (IR stimulants only).

Australian data on the misuse and diversion of ADHD medications are limited. A review by the Queensland Crime and Misconduct Commission found that the evidence was primarily anecdotal, and concluded that illicit diversion and abuse of ADHD prescription medications constituted a minor problem (585). The triannual Australian School Students' Alcohol & Drug Survey included questions about the use of prescription stimulants for the first time in 2005 in Western Australia only (586). The survey found that 8% of students had ever used DEX or Ritalin without a doctor's prescription, and students who were not prescribed these medications by their doctor reported that they were given them, bought them or traded something for them (586). The results of the survey are consistent with the findings of the 2008 systematic review discussed above (584).

The stimulant prescription notification system established by the Western Australian Stimulants Regulatory Scheme is designed to detect deliberate excessive access to stimulant medication (587). Recently, a discrepancy was identified between the notified and the dispensed dose of stimulants that had been prescribed for ADHD treatment in Western Australia (588). As the authors point out, this discrepancy could be ascribed to stimulant misuse and diversion; however, it might also reflect poor compliance by prescribers in renotifying dose increases or prescriptions dispensed before the full amount of the previous prescription was consumed (587).

8.9.2 Alcohol and tobacco use

ADHD may predict early use of alcohol and cigarettes. In the MTA study, for example, at 36 months when the participants were aged 11–13 years, alcohol and tobacco use were more prevalent amongst the group with ADHD than the control group (202).

It is less clear whether ADHD predicts the development of alcohol use disorders. One recent study found that childhood ADHD predicts heavy drinking, symptoms of alcohol use disorders, and alcohol use disorders for 15–17 year olds, but not 11–14 or 18–25 year olds (589). Long-term follow-up studies conducted in adolescents and young adults found no increased risk for alcohol use disorders in individuals with ADHD (205, 208, 590).

Studies in both adults and adolescents have found ADHD to be associated with earlier initiation of regular cigarette smoking and higher rates of lifetime smoking (203, 591-593). Current ADHD symptom levels may influence smoking (592, 594). One study, for example, found a significant association between clinically significant ADHD inattention symptoms and rates of both current smoking and lifetime smoking (592). A subgroup of individuals with ADHD are likely to be self-medicating with cigarettes, and in fact nicotine is being trialled as an intervention for ADHD (493, 595) (see section 8.4.5 Nicotine patch, page 99).

8.9.3 Abuse liability of prescription medications for ADHD

"Abuse liability" or "abuse potential" is a measure of the likelihood that a medication could be abused, that is, self-administered for non-medical reasons. To determine abuse liability, preclinical studies are conducted in animals, followed by human trials to evaluate the following: 1) Discriminate-stimulus effects – can the medication be used to substitute for drugs of abuse? 2) Reinforcing effects – is the medication used preferentially over placebo? 3) Subjective effects – what are the self-reported effects of the medication?

Studies in both animals and humans have found that IR stimulants and injected stimulants can generate physiological profiles consistent with abuse liability, as they produce both reinforcing effects and subjective effects and can substitute for cocaine in drug discrimination studies (for review see (596, 597)). The abuse liability of ER stimulants has not been as well studied, but findings in people without ADHD suggest that the slower rate of delivery associated with these formulations may reduce abuse potential (598). Stimulants have also been shown to be potentially addictive in animals; however, there are few documented cases of addiction to stimulant medication in humans (for review see (597)).

The abuse potential for other medications used in ADHD has been assessed and compared to stimulant medication. Both human (599) and primate (600) studies have shown ATX to have different physiological effects to MPH and it has thus been suggested that ATX has limited potential for abuse (601). Similarly, the physiological effects of modafinil differ from those of MPH and the abuse potential is considered to be limited (for review see (602))

8.9.4 ADHD as a risk factor for substance use disorders

Prevalence of substance use disorders in people with ADHD

Substance abuse disorders (SUDs) are more common among people with ADHD than in the general population. A higher prevalence has been demonstrated both in population-based cross-sectional surveys, and in prospective studies following children with ADHD and matched controls.

A recent community survey conducted in US adults that screened for ADHD and other DSM-IV disorders found significant comorbidity between ADHD and SUDs (138). In addition, a community-wide survey of Brazilian adolescents established that those with ADHD were more likely than those without the condition to have comorbid SUD (603). Several prospective longitudinal studies of children and adolescents with ADHD have also shown that those with ADHD are at greater risk of developing SUD than matched controls (205-208).

While ADHD alone increases the risk of developing SUD, the risk increases further when CD (205, 208-210) and bipolar disorder (205) are also present.

Prevalence of ADHD in people with substance use disorders

It is well documented that the prevalence of ADHD is high among adolescents and adults with SUD who are seeking treatment (604-611). Notably, the co-occurrence of SUD and ADHD has generally been found to be higher in these clinical samples than in community samples such as the US community survey described above (138).

An Australian study found high rates of ADHD in a sample of 95 male drug users: 46% met criteria for adult ADHD and 36% reported childhood ADHD. Also prevalent was CD, which was reported in 68% of participants, but there were significant effects of ADHD on drug use, over and above the effects of CD (611).

8.9.5 Effect of ADHD management on risk of developing substance use disorders

Rese	earch	question		
• I c d	In individuals with ADHD, does the use of pharmacological interventions, compared to no intervention, alter the risk for later substance use or the development of substance use disorders?			
Evid	ence	statements		
Substance use disorders:				
	0	One meta-analysis (612) (six studies) and four primary studies (613- 616)found that among people with ADHD, those treated with stimulants in childhood and/or adolescence have an equivalent or lower incidence of substance use disorders compared to those who did not receive treatment with stimulant medications. (Level III-2 – aetiology)		
• 5	Subst	ance use:		
	0	Two prospective follow-up studies showed that children diagnosed with ADHD had significantly higher rates of substance use than children without ADHD (202, 617). (Level III-2 – aetiology)		
	0	Two prospective follow-up studies showed that the rates of substance use did not differ between children with ADHD treated with stimulant medication and children with ADHD who were untreated (202, 617). (Level III-2 – aetiology)		
Reco	ommo	endation		
146.	The the	use of stimulant medication to treat people with ADHD does not increase risk of developing substance use disorder. (Grade B)		

Substance abuse

One meta-analysis and four primary studies were identified that addressed the question of whether treatment with medications could alter the risk for substance use disorders in individuals with ADHD.

The meta-analysis (612), using data from six studies, indicated a 1.9-fold reduction in risk for SUD in people with ADHD who have taken stimulant medication, compared to those with ADHD without stimulant medication. The risk reduction was greater in adolescents than in young adults. However, the range of odds ratios suggests that there is no single treatment effect across the studies. The meta-analysis had several limitations, including the small number of studies available and inherent confounding factors. The quality of this study was rated as poor.

The primary studies support the finding that stimulant use in childhood or adolescence does not increase the risk of developing substance use disorders but do not confirm the suggestion of risk reduction:

- A small retrospective study of poor quality found that adults with ADHD who had been treated with stimulants in childhood and/or adolescence had a significantly lower incidence of substance abuse than those not treated (614).
- Another, again a retrospective study of poor quality, found a significant reduction in substance abuse among boys, but not girls, who had been treated with stimulant medication (615).

- A retrospective cohort study found that stimulant medication had no significant effect on the development of substance use disorders (613).
- A prospective cohort study found no significant difference in the risk of substance abuse for individuals with ADHD who had been treated with stimulant medication compared to individuals with ADHD who were untreated, and no significant association between the age of starting stimulant treatment, or the duration of that treatment, and the risk for subsequent substance use disorder (616).

Substance use

Two studies on the impact of medication treatment for ADHD on the risk for substance use met the inclusion criteria. Both also considered the prevalence of substance use in individuals with ADHD, compared to a matched control group:

- Children with ADHD enrolled in the MTA study were compared with a control group without ADHD (202). At the 24- and 36-month time points, the ADHD group had significantly higher rates of substance use than the controls, but there was no significant link between substance use and days of prescription medication. At the 36-month time point, the children in the MTA study were aged between 11 and 13.
- A study rated as poor quality described a birth cohort followed up at age 6 and 11 (617). At 11, those diagnosed with ADHD had significantly higher rates of substance use than the group without ADHD. Within the ADHD group, there was no significant link between treatment with stimulants and substance use.

8.9.6 Treatments for ADHD when comorbid substance use disorder is present

Research question

• For individuals with ADHD and substance use disorders, do pharmacological or psychosocial interventions affect outcomes?

Evidence statements

- **MPH:** Four trials showed no statistically significant difference in the efficacy of MPH compared to placebo for ADHD symptom severity in adults with ADHD and comorbid SUD (618-621). (Level II)
- **MPH:** Two out of three trials showed no statistically significant difference in the efficacy of MPH compared to placebo for the frequency of illicit drug use in adults with ADHD and comorbid SUD (618, 620). The third trial showed a decrease in the probability of illicit drug use in the MPH group compared to placebo (621). (Level II)
- **Bupropion:** One trial showed no statistically significant difference in the efficacy of bupropion compared to placebo for ADHD symptom severity or frequency of illicit drug use in adults with ADHD and comorbid SUD (620). (Level II)

NB: No studies of psychosocial interventions were identified.

Recommendations

147. A history of substance abuse and previous treatments should be included in the evaluation of patients.

Recommended best practice based on clinical experience and expert opinion

- 148. If the substance use disorder is active the symptoms of substance use disorder should be addressed prior to attending to the symptoms of ADHD.
 ✓ Recommended best practice based on clinical experience and expert opinion

The treatment of ADHD with comorbid SUD is complicated by the fact that the presence of ADHD influences the characteristics of SUD in numerous ways. In people with ADHD, SUD may have an earlier onset (622) and it may be more severe (610, 623) and harder to treat (624, 625). The duration of SUD may be longer (625-627) and the rate of remission from substance use may be slower. In addition, individuals with comorbid ADHD and SUD may be at increased risk for other psychiatric disorders, compared to individuals with either condition alone (628).

Best practice in the treatment of ADHD and comorbid SUD is summarised as follows (629, 630):

- Careful evaluation and assessment should include a history of substance abuse and previous treatments.
- Abstinence from the drug of abuse is required to evaluate the ADHD symptoms properly.
- If the SUD is active, the symptoms of SUD should be addressed prior to attending to the symptoms of ADHD.
- Education on ADHD should be provided to the individual and family.

- There are no research data to show that ADHD medications are effective in the treatment of ADHD when comorbid with SUD, and no studies have been conducted that consider the use of psychosocial interventions.
- Non-stimulants, long-acting stimulants or antidepressants are preferred methods of treatment as they have lower abuse liability and abuse potential.
- Pharmacotherapy should be trialled for effectiveness in individual patients, carefully monitored for benefits and adverse effects, and modified as necessary.
- Urine tests should be used to monitor adherence to treatment.
- Relapse or worsening of substance use may necessitate reassessing the appropriateness of stimulant treatment.

Summary of research evidence

Psychosocial intervention

No studies of psychosocial interventions were identified.

MPH

Four studies were identified that addressed the efficacy of MPH treatment in adults with ADHD and comorbid SUD (618-621). All four studies recruited participants from treatment centres, so the results may not be representative of the general community. Reporting of side effects was very limited. The study time ranged from 8 to 14 weeks, and participants were being treated for various substance use disorders.

The studies suggest that MPH treatment is not effective for the treatment of ADHD when SUD is comorbid. All four found no significant difference in ADHD symptoms with MPH treatment compared to placebo, although in one study (618), physician- and self-ratings taken at various times showed a significant improvement in ADHD symptoms in the MPH group compared to placebo in some but not all ADHD symptom ratings.

Regarding the impact of MPH on substance use, one study (619) did not address the question; two ((618, 620)) found no significant differences in SUD outcomes between MPH and placebo groups; and the fourth found a significant decrease in the probability of cocaine-positive urine samples with MPH compared to placebo (621).

One study (619) noted that there were significantly more side effects for MPH compared to placebo; one (618) found that insomnia or trouble sleeping were more frequent with MPH compared to placebo; and two ((620, 621)) found no significant differences in side effects between groups.

Bupropion

The one RCT identified that looked at the efficacy of bupropion treatment for ADHD and comorbid SUD (620) found no significant improvement in ADHD symptoms and no differences in cocaine (or other drug) use, compared to placebo. There was also no difference in the number of side effects between the two groups.

8.10 Athletes with ADHD receiving stimulant medication

8.10.1 Current regulations on stimulant use in high performance athletes

Stimulants are banned in elite sport (631), and people with ADHD are currently required to stop their stimulant medication several days before competing (632). The reasons for this ban are that:

- stimulants in high doses are performance enhancing and may be harmful
- in the view of the International Anti-Doping Agency (IADA) there are no objective tests to confirm the presence of ADHD.

At present, it is possible that individuals with ADHD taking stimulant medication may be granted an exemption by a standing committee of the Therapeutic Use Exemption (TUE) committee. This process requires extensive documentation regarding the diagnosis of ADHD from a young age, supported by an assessment using validated instruments; documented improvement on stimulant therapy and worsening without it; regular documented supervision by a doctor with a specific interest in the condition; consideration and exclusion of alternative causes of the symptoms; and

regular updating to the committee of any changes in the patient's circumstances relative to the condition (633).

Clonidine, antidepressants and antipsychotic agents are not banned substances.

Full documentation of the requirements of drugs in sport and the TUE scheme are available at the IADA website, <u>http://www.wada-ama.org</u>.

8.10.2 Impact of stimulants and other pharmacological treatments on athletic performance

There has only been one RCT on the effects of stimulants on sporting performance (634). This showed that stimulant therapy improved the ability of boys with ADHD playing softball to pay attention to the state of the game, but did not improve their batting skills.

A second trial compared children with ADHD and age-matched controls, to test the effect of stimulants on the ability to track a ping pong ball and hit it to a designated target (635). Stimulants improved the ability of subjects to track the ball and accurately aim for the target, but not to hit a target if the target was announced in a short time frame (½ second after the ball was served from the opposite end of the table), compared to 2 seconds before the ball was served. There was no difference in arm movement between participants with ADHD and the control group, and stimulants made no difference to arm movement.

8.11 Summary of appraches to management of medication in ADHD

Dosing guidelines for first-line medications

Issues:

Stimulant medications (MPH and DEX) are S8 classed medications. Therefore, state health regulations are applied. Unfortunately different states in Australia have different requirements in relation to prescription authorisation, patient notification, patient age restrictions, and dosage guidelines. Some states also require notification of each individual patient. Clinicians must be observant and compliant with the statutory guidelines within their state.

In addition, the clinician must be aware of PBS prescribing information, which is sometimes different to state guidelines and different to the clinical recommendations made in this document.

International consensus statements and other national guidelines have been consulted in order to prepare this brief guideline to the practical use of the first-line medications in treatment of impairing ADHD.

General Principles for use of all medications in the treatment of ADHD

- Medical practitioners need to be aware of pharmacokinetics of medication.
- Measure and document possible adverse family or patient medical history; height, weight, resting heart-rate and blood pressure (see recommendations 134 and 135).
- Consider further investigations or specialist assessment in children less than 6 years of age, in the presence of adverse family or patient medical history and adverse physical findings.
- Informed patient and family decision making concerning benefits and side-effects, controversies, uncertainties.
- Emphasise potentially life-threatening or severe side-effects.
- Consider the impact of comorbidities and adjust medication dose and expectations of efficacy.
- Seek child assent for medication treatmentthrough a discussion at a developmentally appropriate level.
- Provide written information about medication effects and side-effects, and direct family to further resource material eg. books, websites, support groups.
- Consider initiating medication during school holidays or weekend in order to permit parents/carers to make observations.

- Consider drug interactions and impact of use of complementary and alternative treatments.
- Use mono-therapy where possible.
- At each medical review re-consider the diagnosis, co-morbidities, necessity for medication and overall management of the patient.
- •Start medication at a low dose and aim for lowest total daily dose to provide optimal response considering side-effects.
- Monitor efficacy and side-effects in multiple settings closely in initiation phase (weekly to second-weekly).
- Consider use of standard rating scales for behavioural efficacy, side-effects and overall benefit.
- Recommend additional ongoing non-medicational management of behavioural, educational, social problems.

Table 6. Dosing guidelines for first-line medications. REFER TO GENERAL PRINCIPLES BEFORE COMMENCING MEDICATION				
Medication	Age Group	Initiation	Titration and Dose	Monitoring
	Preschool-aged children <6 years	Refer to text for specific recommendations in this age group. NOTE: State variations	Refer to text for specific recommendations in this age group. NOTE: State variations	Refer to text for specific recommendations in this age group. NOTE: State variations
late	Children 6-12 years (20-40kg)	 Start low in simple fractions or multiples of dose Begin at 5mg morning, 5 mg lunch-time for one week 	 Increase to 10mg morning, 5 mg lunch-time for one week Consider further increase to 10mg morning and 10 mg lunch-time Consider further dose increases with aim to stablise at 4-6 weeks after initiation Total daily dose typically varies between 0.5-2.0 mg/kg/day, to maximum 80mg daily – it is uncommon to require larger doses (NOTE: refer to specific State guidelines with regards to maximum dose) 	 Monitor heart rate, blood pressure and growth Consider compliance Consider further dose decreases or increases Consider transition to longer-acting MPH after 1-2 months to approximate daily dose equivalent Review medication effectiveness and emergent side-effects Consider ongoing need for medication at each review
Methylphenic	Adolescents 13-18 years (40-70kg)	 Start low in simple fractions or multiples of dose Begin at 10mg morning, 10mg lunch-time for one week 	 Increase to 15-20mg morning, 15-20mg lunchtime for one week Consider further dose increases with aim to stablise at 4-6 weeks after initiation Total daily dose typically varies between 0.5-2.0mg/kg/day, to maximum 100mg daily – it is uncommon to require larger doses (NOTE: refer to specific State guidelines with regards to maximum dose) 	 Monitor heart rate, blood pressure and growth Consider compliance Consider further dose decreases or increases Consider transition to longer-acting MPH after 1-2 months to approximate daily dose equivalent Review medication effectiveness and emergent side-effects Consider ongoing need for medication at each review
	Adults	Start at 5mg-10mg morning on day 1, then begin 5mg-10mg morning, 5mg-10mg lunchtime for one week.	 Thereafter consider further stepwise increases in dosage with aim to stabilise at 4-6 weeks after initiation Total daily dose typically varies between 10mg and 60mg a day It is uncommon to require dosages larger than 80mg a day. A second opinion from an independent clinician should be sought if dosages above 60mg are thought to be necessary. 	 Monitor heart rate, blood pressure and weight Consider compliance Consider transition to longer acting MPH after 1-2 months to approximate daily dose equivalent Review medication effectiveness and monitor for emergent side effects paying particular attention to sleep, appetite and mood Consider ongoing need for medication at each review

Medication	Age Group	Initiation	Titration and Dose	Monitoring
	Preschool-aged children <6 years	Refer to text for specific recommendations in this age group. NOTE: State variations	Refer to text for specific recommendations in this age group. NOTE: State variations	Refer to text for specific recommendations in this age group. NOTE: State variations
mine	Children 6-12 years (20-40kg)	 Start low in simple fractions or multiples of dose Begin at 2.5mg morning, for one week 	 Increase to 5 mg morning for one week Increase to 10mg morning Consider further dose increases with aim to stabilise at 4-6 weeks after initiation Typical total daily dose 0.5-1mg/kg/day, to maximum 40mg daily – it is uncommon to require larger doses (NOTE: refer to specific State guidelines with regards to maximum dose) 	 Monitor heart rate, blood pressure and growth Consider compliance Consider further dose decreases or increases Review medication effectiveness and emergent side-effects Consider ongoing need for medication at each review
Dexampheta	Adolescents 13-18 years (40-70kg)	 Start low in simple fractions or multiples of dose Begin at 5mg morning, for one week 	 Increase to 5 mg morning for one week Increase to 10mg morning Consider further dose increases with aim to stabilise at 4-6 weeks after initiation Typical total daily dose 0.5-1mg/kg/day, up to 40mg daily – it is uncommon to require larger doses (NOTE: refer to specific State guidelines with regards to maximum dose) 	 Monitor heart rate, blood pressure and growth Consider compliance Consider further dose decreases or increases Review medication effectiveness and emergent side-effects Consider ongoing need for medication at each review
	Adults	Start at 2.5mg -5.0mg morning on day 1, then 2.5mg-5.0mg morning, 2.5mg-5.0mg lunch- time for one week	 Thereafter consider stepwise increases in dosage with aim to stabilise at 4-6 weeks after initiation Total daily dose typically varies between 5mg and 30mg a day It is uncommon to require dosages larger than 40mg a day A second opinion from an independent clinician should be sought if dosages above 30mg a day are thought to be necessary 	 Monitor heart rate, blood pressure and weight Review compliance Consider further dosage increases or decreases. Review medication effectiveness and emergent side-effects, paying particular attention to sleep, appetite and mood Consider ongoing need for medication at each review

Medication	Age Group	Initiation	Titration and Dose	Monitoring
	Preschool-aged children <6 years	Atomoxetine not approved for use in children < 6years	Atomoxetine not approved for use in children < 6years	Atomoxetine not approved for use in children < 6years
Φ	Children 6-12 years (20-40kg)	 Start at nearest capsule size to approximate 0.5mg/kg/day, morning 	 Increase to approximately 1.2mg/kg/day morning after one week 	 Efficacy may not be established till 5-6 weeks after initiation Place particular emphasis at review on possibility of self-harm and liver dysfunction Monitor heart rate, blood pressure and growth Consider compliance Review medication effectiveness and emergent side-effects Consider ongoing need for medication at each review
Atomoxetin	Adolescents 13-18 yrs (40-70kg)	 Start at nearest capsule size to approximate 0.5mg/kg/day, morning 	 Increase to approximately 1.2mg/kg/day morning after one week 	 Efficacy may not be established till 5-6 weeks after initiation Place particular emphasis at review on possibility of self-harm and liver dysfunction Monitor heart rate, blood pressure and growth Consider compliance Review medication effectiveness and emergent side-effects Consider ongoing need for medication at each review
	Adults	 Start at 25-40mg a day for a minimum of 7 days. 	 Increase thereafter to a target dose of 60- 80mg a day Maximum dosage 100mg a day 	 Efficacy may not be established until 5-6 weeks after initiation Place particular emphasis at review on possibility of mood disturbance, genitourinary symptoms and liver dysfunction Monitor heart rate, blood pressure and weight Consider compliance Review medication effectiveness and emergent side-effects Consider ongoing need for medication at each review

CHAPTER 9. MEDICATION COMPARED TO AND COMBINED WITH PSYCHOSOCIAL INTERVENTIONS

9.1 Introduction

An individual treatment and care plan may typically combine medication and psychosocial interventions. The potential benefits of a combined intervention approach include the possibility of:

- increasing the impact and duration of treatment effects by using multiple interventions
- lowering the dose of medication required to have a clinical impact, which minimises the side effects of medication
- targeting both the core symptoms of ADHD through medication and specifically intervening to improve daily functioning with psychosocial interventions.

9.2 Comparing psychosocial and pharmacological interventions

Research question

• For individuals with ADHD, do psychosocial interventions, compared to pharmacological interventions, affect outcomes?

Evidence statements

- One systematic review comparing MPH to psychosocial treatments identified three studies that reported a significant benefit of MPH over psychosocial interventions for symptoms of hyperactivity (443). (Level I)
- One study found that medication management was superior to behavioural treatment for ADHD core symptoms, but not for academic achievement or social skills, at 9 months (430) and 14 months (408). At 24 months (409) and 36 months (410), the difference was not significant. (Level III-2)
- Another study found that behaviour modification and medication management were equivalent for classroom behaviours (rule violations and seat-work completion) (636). (Level II)
- One meta-analysis (637) found:
 - For ADHD symptoms, psychosocial interventions yielded smaller effect sizes than either MPH or combined treatment.
 - For ODD symptoms, MPH was significantly more effective than the psychosocial interventions.
 - For social behaviour, the psychosocial, MPH and combined interventions were equally effective.
 - For academic functioning, no treatment was clearly effective. (Level I)
- No studies were found that compared psychosocial interventions to pharmacological interventions in preschool-aged children or adults.

Recommendations

- 152. Although the evidence suggests that ADHD symptoms improve more in the short to medium term with pharmacological management than with psychosocial interventions, it is not possible to draw conclusions about the best management of ADHD symptoms in the long term. The likely presence of modifying factors, including comorbidities and the age of the child, warrant the use of psychosocial interventions. (Grade D)
- 153. Psychosocial interventions should be considered in the long-term management of children and adolescents with uncomplicated ADHD, with or without medication.
 - \checkmark Recommended best practice based on clinical experience and expert opinion
- 154. Psychosocial interventions should be considered, in addition to pharmacological management, in people with ADHD associated with ODD or CD.

✓ Recommended best practice based on clinical experience and expert opinion

9.2.1 Summary of research evidence

Very few studies have compared pharmacological interventions with psychosocial interventions in the treatment of ADHD in children, and no studies were identified that considered preschool-aged children, adolescents or adults.

A systematic review identified eight studies comparing various psychosocial treatments with MPH in children (443). Only three reported on hyperactivity or quality of life. Different psychosocial interventions were utilised in each of the studies: parenting programs, cognitive therapy and child behaviour modification, plus parenting programs and teacher training. Three studies found that MPH was superior to psychosocial intervention for symptoms of hyperactivity. One study demonstrated variable results depending on the scale used to measure outcomes. None of the studies compared DEX or ATX to psychosocial interventions.

The MTA study compared the use of medication with multimodal psychosocial management. At 14 months, medication management was superior to behavioural treatment in improving ADHD core symptoms, but not academic achievement or social skills (408). Analysis of measures at 9 months, when the behavioural management was still being intensively administered, also showed equivalent findings (430). At 24 months the difference between the medication group and the combined group was still evident, although not as great (409). In the most recent assessment, at 36 months, there was no significant difference between the effectiveness of psychosocial interventions and medication (410). It is, however, difficult to compare between groups because, as the study has progressed, the participants have been followed up according to their original treatment allocation, regardless of which treatment they subsequently chose to use.

A study based on the summer treatment program (636) reported on classroom behaviours in a comparison between behaviour modification and MPH. Behaviour modification was equivalent to MPH on measures of classroom rule violations and seat-work completion.

A meta-analysis of RCTs compared effect sizes between MPH, psychosocial interventions and a combination of the two (637). Studies of psychosocial interventions were much fewer in number than those investigating MPH. The meta-analysis found that:

- Both MPH and psychosocial treatments were effective in reducing ADHD symptoms, but psychosocial treatment yielded smaller effect sizes than either MPH or combined treatment.
- For ODD symptoms, both psychosocial interventions and MPH generated significant improvements; however, MPH was significantly more effective than the psychosocial interventions.
- For social behaviour, the psychosocial, MPH and combined interventions were equally effective.
- For academic functioning, the psychosocial interventions generated a marginally significant improvement, with no observed improvement when MPH or combined treatments were employed. The authors concluded that no treatment was effective in improving academic function.

9.3 Combining psychosocial and pharmacological interventions

9.3.1 Combined interventions for preschool-aged children, children and adolescents

Research question			
• For provide the providence of the providence o	For individuals with ADHD, do psychosocial interventions used alongside pharmacological interventions, compared with pharmacological interventions alone, affect outcomes?		
Evid	ence	statements	
• C	hildr	en:	
	0	One meta-analysis of studies conducted in children aged 6–12 years found that combined treatment did not improve outcomes (ADHD and ODD symptoms and social behaviour) over those achieved with stimulant medication alone. Neither combined treatment nor MPH was effective for improving academic function (637). (Level I)	
	0	One study in children aged $6-12$ years in a summer treatment program found that combined treatment was superior to MPH alone for reducing ADHD symptoms (638). (Level II)	
	0	One poor-quality study in children aged 6–7 years in a summer treatment program found that combined treatment was equivalent to MPH alone for point system measures (ADHD symptoms were not directly assessed) (639). (Level II)	
	0	One study in children aged 5–12 years in a summer treatment program found that behaviour modification plus low-dose MPH was equivalent to a higher dose of MPH alone for classroom rule violations and seat-work completion (ADHD symptoms were not directly assessed) (636). (Level II)	
	0	One study found no significant difference between combined treatment compared to medication alone in reducing ADHD symptoms at 14 months (408), 24 months (409) and 36 months (410). (Level III-2)	
4	•	One study found no significant difference between combined treatment compared to medication alone in reducing ADHD symptoms or improving social behaviour over a 2-year period (640-642). (Level III-2)	
	0	One poor-quality meta-analysis found that the calculated effect sizes for combined treatment were greater than those previously reported in the literature for treatment with medication alone (643). (Level III-2)	
	0	No studies were found that compared psychosocial interventions to pharmacological interventions in preschool-aged children or adolescents.	
Reco	mm	endation	
155.	. Although the evidence suggests that ADHD symptoms are most improved in the short to medium term with pharmacological management alone, it is not possible to draw conclusions about the best management of ADHD symptom in the long term. The likely presence of modifying factors, including comorbidities and the age of the child, warrant the inclusion of psychosocial interventions. (Grade D)		

Summary of research evidence

Six primary studies were identified that directly compared the use of a combined intervention to medication alone in children with ADHD.

Comparison of a multi-component psychosocial intervention plus MPH to MPH alone found significant improvements within both the combined treatment and medication only groups (644). Combined treatment did not appear to affect outcomes over those achieved with stimulant medication alone.

A summer treatment program study conducted in children 6–12 years of age found that combined treatment was superior to MPH alone for improvement of ADHD symptoms (638). A second summer treatment program study in younger children (6–7 years) found a significant difference between the two MPH groups (MPH alone and MPH plus behavioural interventions) and placebo in all point system measures and 2 of 3 classroom measures (639). These results suggest that the behavioural intervention added little benefit. The direct impact on ADHD symptoms as measured via DSM-IV criteria or related scales was not assessed. More recently, a summer treatment program study reported on classroom behaviours following treatment with either behaviour modification (low or high intensity) alone or in combination with MPH at three doses (0.15, 0.3 or 0.6mg/kg) (636). Behaviour modification plus low doses of MPH (0.15mg/kg) was equivalent to a high dose of MPH (0.60mg/kg) alone on measures of classroom rule violations and seat-work completion.

Three assessments were derived from the MTA study, which included a combined intervention arm of medication and multimodal psychosocial management, and a medication only arm. Over the course of the study, the reports at 14 months (408), 24 months (409) and 36 months (410) found no significant difference in the effectiveness of combined treatment compared to medication alone in any of the five ADHD symptom domains they assessed. It is, however, difficult to compare between groups because, as the study has progressed, the participants have been followed up according to their original treatment allocation, regardless of which treatment they chose to use.

A multimodal psychosocial treatment study was conducted over 2 years in children aged 7–9 years with ADHD (640-642). Entry was restricted to children who responded to MPH in a 5-week open-label treatment period. Three groups were compared: MPH alone, MPH plus multimodal psychosocial treatment, and MPH plus a psychosocial treatment control. While significant improvement in ADHD symptoms and social behaviour occurred in all treatment groups and continued over 2 years, there were no differences between groups. The biggest limitation of this study is that the included children were known responders to MPH, so the results cannot be assumed to reflect accurately a more general population of children.

A meta-analysis of eight studies using combined treatments found that the calculated effect sizes for combined treatment were greater than those previously reported in the literature for treatment with medication alone (643). This review included both RCTs and studies with lower levels of evidence, and was rated poor in quality.

Another meta-analysis compared effect sizes between MPH, psychosocial interventions and a combination of the two, drawing on results of 26 RCTs (637). There were very few studies addressing combined interventions. MPH and combined treatment were equally effective in reducing ADHD symptoms and ODD symptoms and improving social behaviour. Neither combined treatment nor MPH was effective for improving academic function. Effect sizes of MPH and combined treatment were not significantly different, suggesting no additive effect of psychosocial interventions.

9.3.2 Combined interventions for adults

Research question

• For individuals with ADHD, do psychosocial interventions used alongside pharmacological interventions, compared with pharmacological interventions alone, affect outcomes?

Evidence statement

• Two studies of poor quality support the use of combined psychosocial and medication interventions compared to medication alone (645, 646). (Level II and IV)

Recommendation

156. The likely presence of comorbidities in adults with ADHD favours the inclusion of psychosocial interventions alongside pharmacological interventions. (Grade D)

Summary of research evidence

Two primary studies met our inclusion criteria, and both found in favour of the use of combined interventions for adults with ADHD compared to medication alone:

- A parallel RCT found that the combination of medication and cognitive behavioural therapy was superior to medication alone (645). The quality rating of the study was poor as blinding was not used due to the difficulty of blinding of participants to this behavioural intervention.
- A retrospective study found that the combined intervention of medication and cognitive therapy was greater than the use of medication alone (646). This study had major methodological limitations, with no clear comparator or control group.

9.4 Moderators and mediators of outcomes from psychosocial interventions and combined interventions

Research question

• For individuals with ADHD, what are the moderators and mediators of treatment response with psychosocial interventions?

Evidence statements

- One study demonstrated that children with ADHD plus anxiety showed particular benefit from psychosocial interventions (647) irrespective of the presence of comorbid CD (441). (Level III-2)
- One study found that parental depressive symptoms, high severity of child ADHD and below-average child IQ were associated with a poorer response to medication or combined treatment (648). (Level III-2)
- One study found that both parent education and parent occupation had an impact on treatment outcomes (649). (Level III-2)
- No studies were found that addressed moderators and mediators of outcomes in preschool-aged children, adolescents or adults.

Recommendations

- 157. Psychosocial interventions can be particularly beneficial in a person with ADHD and comorbid anxiety. (Grade C)
- 158. Individualised treatment strategies should be developed. When variables such as parental depression, severe ADHD symptoms or below-average IQ are present, additional intervention strategies may be required. (Grade C)

Background

A number of factors may influence treatment outcomes:

- Mediator variables have the potential to influence outcomes after a treatment has been initiated. They include treatment intensity, treatment duration, adherence to medication and parenting style.
- Moderator variables are factors that are in place before treatment is initiated. They include gender, age, ADHD subtype, comorbid conditions, race/ethnic factors, education levels and socio-economic status.

Research into these variables can help to guide clinicians' decisions about what type of treatment is best for each individual.

Summary of research evidence

Four primary studies were identified that investigated the moderators and mediators of outcomes for psychosocial interventions. All were derived from the MTA study. The original MTA study compared four treatment groups: medication management, behavioural management, combined treatment, and community care (408). At 14 months, the behavioural management group did not differ significantly from the community control group on any outcome. Both behavioural management and community care were significantly less efficacious than medication management and combined management (408).

The MTA Cooperative Group initially assessed five potential moderator variables (gender, prior medication, comorbid ODD or CD, comorbid anxiety and public assistance (financial support for low-income families)) to determine whether these factors would have an impact on treatment response (647). The study found that for children with anxiety disorders, behavioural treatment was significantly better

than community care, and was no longer statistically different from medication management or combined treatment. In other words, children with both ADHD and anxiety responded equally well to behavioural management, medication management or combined treatment.

A further analysis of the MTA results looked at whether the presence or absence of CDs in addition to anxiety would alter the impact of anxiety as a moderator variable (441). For children with anxiety disorders, behavioural treatment was significantly better than community care for parent-rated measures, irrespective of the presence or absence of comorbid conduct problems.

Analysis of treatment outcomes in the behavioural management group showed that none of the potential variables examined (public assistance, gender, anxiety, CD/ODD, maternal education, parental depressive symptoms, initial child ADHD severity and child IQ) had an impact on the treatment response (648). These results contradict the two studies reported above (441, 647), which found that comorbid anxiety moderated the treatment response. It is not clear whether the findings of those two studies were not robust enough when considering optimal treatment response, or whether the inconsistency relates to a difference in how the analysis was conducted.

Notably, parental depressive symptoms, high severity of child ADHD and belowaverage child IQ were associated with a less efficacious response in the medication management and combined treatment groups (648).

Analysis of the impact of parent education and occupation found that both affected treatment outcomes (649). Children from more educated families showed greater reduction of ADHD symptoms with combined treatment, while children from less educated homes showed no differential treatment response. For oppositional-aggressive symptoms, children from blue-collar households benefited most from the combined treatment, while children from white-collar homes generally showed no differential treatment response. Variables such as household income and marital status did not have an impact.

While not assessed as a moderator of ADHD in the same ways in which the above factors have been analysed, it is clear that the effect of parental ADHD can effect the management of children with ADHD, including the ability to implement treatment plans (see section 15.2.2 Adult ADHD and parenting)

CHAPTER 10. EDUCATIONAL MANAGEMENT

10.1 Educational research

As a result of the complex socio-political nature of the educational enterprise, the field has yet to establish systematic guidelines for evaluating research practices (650). The What Works Clearinghouse (WWC), established by the US Department of Education's Institute of Education Science, applies a gold standard of fully randomised research with a control group (651), but as Carter and Wheldall (652) observe, such research is rare in education, and there is controversy over some of the research accepted by the WWC (651).

The ongoing concern about the quality of research in education (653) is articulated by those who challenge the validity of qualitative research (654-656). The April 2008 issue of *The Australasian Journal of Special Education*, for example, is devoted to such criticism.

The complexity and variability of schools, children and teachers, and the interactions among them, however, prevent educational researchers from applying the usual scientific principles. In medicine, researchers are able to quantify and control variables and isolate data from contexts in order to predict, generalise and control their studies with precision and accuracy. It is much more difficult to do so when studying teachers and children in classrooms, with multiple variables and contexts which cannot be controlled or manipulated. The researcher must not only determine whether a practice is effective, but for whom and in what circumstances, and this prevents findings from being generalised (650).

The US National Academy of Sciences identified three types of scientific research questions in education: (1) description; (2) cause; and (3) process (650). The Academy concluded that experimental group, correlational, mixed methodology, RCT, large scale and single subject and qualitative designs are all valid as long as the research methodology matches the design question, and the research is rigorously, ethically and empirically applied (650).

In contrast to the research in health management, medicine and other fields, which is dominated by a positivist ontology, objective epistemology and quantitative methodology, much of the research in education has a relativist ontology, subjective epistemology and qualitative methodology (657). Qualitative methods are ideally suited to providing an understanding of the complex context in the classroom and school, and a detailed description of how practices actually work (657). A key contribution of qualitative research is the development of theories and concepts that can aid understanding of education and disability (657). For example, numerous promising approaches have emerged in recent years to contribute to the education of people with severe disabilities, such as curriculum adaptations, inclusive education, positive behaviour supports, supported employment and community-based instruction (657).

An additional problem is the very limited funding for educational research and the publication and distribution of research findings. Peer-reviewed education journals, therefore, accept position papers, studies from a wide range of methodological categories, including single-subject experimental research, descriptive research (e.g. quantitative surveys, qualitative studies based on interviews, observations or document analysis), and evaluation research that relies on both quantitative and qualitative data (657). Many available studies, therefore, are marked by small sample sizes, lack of controls or inadequate description of methodology or interventions, which prevent them from being included in any quantitative collection.

10.2 Inclusion: a legal requirement

Recommendation

159. All professionals supporting students with ADHD should be familiar with their legal responsibilities under the 1992 Disability Discrimination Act (DDA) and Disability Standards for Education (2005). ADHD is recognised as a disability under the DDA. As such, schools are responsible for explicit planning and review of support strategies and services for students with ADHD. ✓ Recommended best practice based on clinical experience and expert opinion

Students with ADHD have a legal right to access schools and the curriculum, as ADHD meets the criteria for disability under the *Disability Discrimination Act 1992* (Cth). The Act makes it unlawful to discriminate directly or indirectly against a person with a disability. It is unlawful to treat a person with a disability less favourably than one without the disability, on the basis of that disability, and in circumstances that are the same or not materially different. A need for additional services or facilities does not make circumstances materially different.

In relation to education, it is unlawful for an educational authority to discriminate against a person on the ground of disability by refusing or failing to accept an application for admission as a student or, in the terms or conditions of admission, by denying or limiting access to any benefit provided by the authority, or by expelling the student or subjecting the student to any other detriment. These provisions are qualified by the exception provided via the defence of *unjustifiable hardship*. There have been cases when the rights of the majority of students outweigh the right of the individual student with a disability, particularly when extreme violence has been involved, but these are rare.

The Disability Standards for Education (2005) provide guidance to education providers on their obligations under the DDA. The standards specifically address the areas of enrolment, participation, curriculum development, accreditation and delivery, student support services, and the elimination of harassment and victimisation. The Disability Standards emphasise the right to curriculum access through differentiation of curriculum, teaching and assessment.

Schools thus have a legal obligation to ensure the inclusion of students with a disability, and teachers are legally required to adapt their teaching as necessary to meet the needs of children with ADHD. Examples of cases involving students with ADHD brought before the Human Rights and Equal Opportunity Commission (HREOC) can be located on the HREOC website

(http://www.hreoc.gov.au/disability_rights/dda_guide.htm). For educational systems to meet their legal obligations, however, schools and teachers must have the capacity to support and effectively teach and accommodate students with ADHD.

10.3 Inclusive educational strategies

Inclusion in the educational setting means equality of treatment for all students, regardless of ability or disability, as well as equity (specific measures such as making reasonable accommodations) for those with additional needs. An inclusive school is one that caters for the needs of all learners, and where all learners are valued and respected. ADHD is just one of many conditions in children and adults in education which must be considered and supported in diverse teaching environments.

Shaddock et al (658) define inclusive practice as:

any and all efforts made by a school and its community to make students and their parents feel welcome. Inclusive practice implies that if participation becomes an issue for any student, whether arising from disability, gender, behaviour, poverty,

culture, refugee status or any other reason, then the desirable approach is not to establish special programs for the newly identified individual or group need, but to expand mainstream thinking, structures and practices so that all students are accommodated. (p. 4)

In this, the largest study of teaching practices for students with disabilities in mainstream classrooms in Australia, the researchers found that the teaching techniques, classroom adaptations and curricular differentiations effective for students with disabilities tend to be effective for other students. Furthermore, they found that principals played a pivotal role in supporting inclusive practices and that learning outcomes for students with disabilities were dependent on the cultures and policies of mainstream schools and school systems.

Shaddock et al (658) also reported that successful teachers tended to view all their students as having individual needs, and capitalised on the strengths and interests of each student. The study found that:

- In facilitating curriculum access for their students, inclusive teachers adopted a wide range of strategies that involved whole-school, paired-class, within-class and individual student strategies, i.e. they routinely involved colleagues, parents and other students in assisting them to deliver a differentiated curriculum.
- Inclusive teachers preferred to assist students to participate in the work of the class rather than to work on individually tailored programs or modified curriculum.
- The teachers "experimented", tested hunches about what might work, and took a reflective and problem-solving approach to their teaching.
- The teachers planned thoroughly and extensively and wanted more time for essential consultation.
- School culture and flexible school policies were strong influences on the success and ease of curriculum differentiation.
- Inclusive teachers capitalised on the strengths and interests of each student, and viewed all of their students as having individual needs – not just those with a disability.
- Schools that utilised teaching aides / classroom assistants provided clear and effective guidance on roles and responsibilities and ensured that the teacher directed the teaching assistant. They facilitated joint professional development for teachers and teaching assistants, and arranged workloads to allow teachers and teaching assistants time to build their relationship, to plan, and to reflect on and evaluate their work.
- Literacy and numeracy outcomes were enhanced by early identification, early intervention and the use of a variety of teaching methods.

Shaddock et al (658) also found that there was a range of research-based approaches used in effective classroom practice to accommodate students with ADHD. Such approaches usually involved planning instruction around differentiations to (a) curriculum content; (b) classroom processes (including teaching techniques and student groupings); (c) learning products and outcomes; and (d) the learning environment.

The study found that effective practitioners universally believed that there is no single "best model" for including students with disabilities in the mainstream. Schools need to use the available evidence-based strategies, adapting them for their unique contexts to serve the complex and often unique needs of learners with additional needs such as ADHD (658).

It also should be recognised that students with ADHD may not have special needs in all areas of learning. For example, ADHD is considered a valid diagnosis in children who have a high IQ (659). Therefore, individuals with ADHD may be gifted in one area of performance but have a learning disability in other areas of performance.

10.4 Teacher knowledge

Recommendation

160. Pre-service and in-service teacher preparation courses should be designed to prepare all teachers with the knowledge and skills to accommodate students with specific learning needs and to manage students in need of additional support for their learning, behaviour, organisation and concentration difficulties.

✓ Recommended best practice based on clinical experience and expert opinion

With the current prevalence of ADHD in Australia, it is likely that there are students with ADHD in every school and educational facility, and that teachers interact with students with ADHD in their classes and the playground, during sport, on excursions or during extracurricular activities. It is imperative, therefore, that teachers have a sound understanding of ADHD.

Research suggests that teachers have some knowledge about ADHD, but there are gaps in their knowledge and some misconceptions (660-662). The technical nature of medical research, the proliferation of alternative remedies and the focus on controversial aspects of ADHD in the media are as confusing to teachers as they are to parents. In one study, teachers were found to know significantly more about the symptoms and diagnosis of ADHD than about its nature, course and treatment (663). ADHD knowledge was related to teacher self-efficacy, prior exposure to a child with ADHD and years of teaching experience (663). Another study found that teachers and parents knew significantly more about the causes of ADHD than its characteristics, and their knowledge of its characteristics was significantly higher than their knowledge of ADHD treatment (662). Parents knew more than teachers about the cause and treatment of ADHD. A review of the literature on children with ADHD and their teachers found that teachers tend to hold negative beliefs about externalising behaviour problems such as CD in students with ADHD. They tend to be pessimistic about teaching these children, and feel that this requires extra time and effort (664).

In general, teachers regard ADHD as a valid diagnosis that has implications for the school setting, and they express a desire for comprehensive training but complain that their training is inadequate for addressing the needs of many students with disabilities, especially those with challenging behaviours (665-670).

A major area of concern for students with ADHD and their parents/caregivers is the high school sector because the organisational structure of schools often impedes communication regarding vulnerable learners (671). As well, there can be a lack of communication and collaboration amongst parents/caregivers, school personnel and medical professionals (672), which can prevent teachers acquiring critical student knowledge about their students with ADHD.

Teachers of students with learning disabilities or emotional/behavioural disorders have reported that "research-based" was not important to them as a criterion for their selection of interventions with students (673). Rather, teachers sought instructional practices that were feasible, appropriate for their students, accompanied by all necessary materials and professional development support, and that could be individualised for multilevel classrooms (674).

10.5 Resources

Recommendations

161. Resource allocations to schools should be accessible to teachers and schoolbased personnel for professional development in areas where established and emerging empirical scientific evidence about academic and social learning in children can inform more effective pedagogical practice. Such upgrading of skills should have an emphasis on practical school-based interventions. ✓ Recommended best practice based on clinical experience and expert opinion

Shaddock et al (658) found that teachers' capacity to implement effective differentiation strategies necessary for supporting learners with special needs depended to a large extent on school cultures, policies, resources and support. When inclusive practice was facilitated by system or whole-school change and support, individual teachers and groups of teachers could have a positive impact on school practices. The time and support allocated to teachers for planning, collaboration and professional development was shown to contribute significantly to positive student outcomes.

In their review of the literature, however, Shaddock et al (658) found that teachers generally said they did not have the time or the resources to make adaptations for students with significant individual needs and that these concerns had been expressed for many years. This finding was confirmed in their own study of classroom teaching practices across Australia. The authors found that students with complex needs did not meet funding eligibility criteria or attract additional resources. Further, the needs of some students could not be adequately met with the level of resourcing provided. Some traditional funding arrangements, such as allocating classroom support on the basis of teacher assistant hours, "lock in" practices and discourage flexible and creative use of resources. Teachers' participation in necessary professional development was also hampered by lack of funds.

10.6 Educational challenges for students with ADHD

10.6.1 Early years

During the preschool years, children acquire the social, behavioural and academic skills that allow them to navigate successfully through primary school (675). They learn how to focus their attention on teacher-directed activities, regulate their actions and inhibit inappropriate responses, interact appropriately with peers and authority figures, and follow spoken and unspoken rules in the classroom, gradually gaining the skills they will need to function in a formal classroom setting (675). In addition, they acquire the basic building blocks for later academic success in literacy, maths and communication skills. Difficulties with impulse control, attentional capacity and hyperactivity hinder this complex, essential learning (675).

Longitudinal studies have found that children who are inattentive, impulsive and hyperactive in preschool are more likely than other children to experience continuous problems through primary school and into adolescence (675). A significant percentage of children with these behaviours are eventually diagnosed with ADHD; however, the studies generally show that preschool children are at risk for the development of a range of disorders, ADHD being only one possible diagnosis (675).

Research also shows that problems with disinhibition (hyperactivity/impulsivity) become evident around 3–4 years of age, and problems related to inattention emerge later, around 5–7 years of age, with entry to formal schooling (143). The rapid developmental changes occurring between the ages of 2 and 6 years, however, make it difficult to determine when hyperactivity, inattentiveness and

impulsivity are developmentally appropriate and when a diagnosis of ADHD should be considered (675).

For many children with behaviour problems in preschool, clinically significant behaviour problems do not persist into later life. Factors that are associated with persistence include poor cognitive and language skills, adaptive disability, comorbid conduct problems, family stress and maternal depression (675).

Skilled observers rating the behaviour of 571 low-birthweight children aged 30 months, based on a 10-minute video, found that high scores on inattentiveness predicted a physician's diagnosis of ADHD and school difficulties at 8 years (676). These findings suggest the possibility of earlier intervention, perhaps before the behaviours begin to affect other areas of development and family dynamics.

The mechanisms that lead preschool children with ADHD-type symptoms to become academic underachievers years later are not clear; however, studies have shown that there are multiple pathways to the development of related learning and behavioural problems (675).

10.6.2 Primary years

Although there is an increased likelihood of medical diagnosis as children with ADHD symptoms engage with the formal settings for learning, the problems that children with ADHD experience in the early years continue to compound as they move through primary school. As many as 80% of students with ADHD have been found to exhibit academic performance problems (677) and about 20–30% have been classified as having learning disabilities due to deficits in the acquisition of specific academic skills (678). Learning difficulties can exacerbate behavioural problems and affect the social and emotional wellbeing of the student, resulting in underachievement, antisocial behaviour, a sense of failure, alienation from peers, and school and social exclusion (679, 680).

The structure of the primary classroom is at odds with the behaviours of many students with ADHD (671). Students are expected to be organised, attentive, quiet and compliant, and to sit still for long periods of time, complete multi-step tasks, and work independently and co-operatively – behaviours that students with ADHD typically lack (679).

Students vary in the type of symptoms they display, from dreamy and inattentive to impulsive, loud and constantly in motion, and some will show both inattentive and hyperactive/impulsive behaviour. Their behaviours vary day to day, minute to minute, and across different learning contexts (679).

Cognitive difficulties underpin both inattentive behaviour and underachievement (681). In particular, poor working memory is impaired by ADHD, and is linked with inattentive behaviours (681). Working memory is the ability to hold and manipulate information temporarily (for a few seconds only), despite ongoing distractions such as conversation or other classroom activities (682).

Working memory is necessary for performing complex tasks such as mental arithmetic, listening and reading comprehension, and reasoning, and the quality of working memory predicts future academic achievement in literacy, maths and science (683-685). For example, in three separate studies of primary school children of different ages, the expressive writing of children who exhibited symptoms of ADHD was worse than that of the control groups in adequacy, structure, grammar and lexicon (686). Students with ADHD wrote less, their texts were disorganised, their vocabulary was limited and they made more errors than their peers (686).

Inattentive behaviours, poor working memory and poor academic achievement have been described as an interrelated "risk triad" for learning, and intervention

that aims either to reduce inattentive behaviour or to improve cognitive function would be expected to enhance a child's academic outcomes (681).

Poor self-regulation is a major contributor to academic and behavioural problems in children with ADHD. Self-regulation has three components: an attentional component, an inhibitory component, and an organisational, strategic component, which directs cognitive processing – notably preparation and planning, working memory and "set-shifting" (687).

Children with a reduced capacity for self-regulation of affect have a low tolerance of frustration, and a tendency to emotional outbursts and to view things in personal terms. They cannot use everyday language effectively as a cognitive and social tool for passing on information and resolving conflicts (688).

Lack of social skills, such as forming friendships and relating to peers, creates increasingly significant problems for children in the primary school years. Hyperactive and impulsive behaviours contribute to unrestrained and overbearing behaviours that make children with ADHD unpopular with their peers, while inattention may mean that children do not pick up subtle and obvious social cues and fail to learn social skills through observation alone (186).

A recent review of the research (186) indicates that:

- having positive peer relationships is developmentally important for all children
- low acceptance or rejection by peers places children at risk for a host of negative outcomes
- both boys and girls with ADHD have difficulty relating to peers
- once rejected, overcoming a negative reputation with peers is extremely difficult, and the label of "ADHD" exacerbates the problem
- addressing relationships with peers with children with ADHD is extremely difficult and it may be more beneficial to address relationships with peers directly.

A study of 133 male and 42 female primary school students with ADHD (689) confirmed results of earlier research in finding relatively few gender differences in impairment in school functioning across the academic, behavioural and social domains. Although girls are less likely than boys to have ADHD, their impairments are as severe, and in some cases even more severe, than for boys. The authors noted that the referral rate for boys with ADHD is higher than for girls (143), maybe because boys exhibit more disruptive and aggressive behaviour; but they emphasised the importance of early identification and treatment for all children with ADHD (689).

10.6.3 Secondary years

Evans et al (690) described the problems faced by adolescents with ADHD that have been identified in research. In adolescence, the problems experienced at younger ages – academic difficulties, discipline problems at school and at home, conflict with peers – often have more serious consequences, such as school dropout and entering the juvenile justice system. Physical and social maturation may also bring new problems such as car accidents, driving infringements, difficulty in relationships, vocational problems and substance abuse. Although medication can provide short-term benefits for adolescents with ADHD, use of medication drops steeply throughout adolescence (690).

Furthermore, the school setting, routines and expectations change as students move from primary to secondary school. Students are expected to be independent learners and they move from the security of one teacher and one class to many (671). Many adolescents with ADHD have significant problems related to homework

completion and class preparation (678, 685, 691). They often come to class unprepared, fail to record or recall assignments, and rush through schoolwork making careless mistakes. They may not begin tasks in a timely manner, make prompt decisions, maintain effort, remember responsibilities, organise materials and manage time efficiently. Students with ADHD can also struggle to apply and review the material they have learned. They find it difficult to understand material covered, and often do not allow sufficient time for test preparation (691).

10.6.4 Tertiary/post-school years

Recommendation

162. Australian research into the impact of learning and behaviour challenges for adults pursuing tertiary qualifications is needed to identify what reasonable accommodations can be made to enhance their chances of success. ✓ Recommended best practice based on clinical experience and expert opinion

In adults with ADHD, symptoms of inattention, impulsivity, disorganisation and a lack of self-regulation can make progression through higher levels of education difficult. ADHD continues to be associated with problems in academic functioning at the college level (692). College students with ADHD are easily distracted, restless and prone to act impulsively without regard to consequences. These symptoms often interfere with a student's ability to perform in an academic setting that requires sustained attention, behaviour self-regulation and impulse inhibition (693).

The symptoms of ADHD can manifest differently in adulthood compared to childhood (see Chapter 4. Consequences of ADHD, page 25). For example, it has been reported that the hyperactivity component improves dramatically as children increase in age, and that adults with ADHD have greater difficulty with internal distractions such as daydreaming and a constant flow of irrelevant ideas. This has been described as "mental restlessness", and individuals with diagnosed ADHD consistently endorsed significantly higher ratings on the *Mental Restlessness Scale* than people without ADHD (694). Mental restlessness may involve different components: internal distractibility, internal restlessness, internal impulsivity and internal disorganisation (694). Collectively, these symptoms appear to reflect problems with cognitive disinhibition, which is consistent with the large body of research that suggests that ADHD is characterised by deficits in executive function (695).

Anecdotal evidence suggests that unless students have strong family support throughout their studies, it is likely that many will fail to complete tertiary training. Support organisations such as ADDults with ADHD (NSW) Inc. suggest the following:

- the provision of mentors or coaches for students with ADHD
- training for tertiary education counsellors on ADHD so they know how to support adults with ADHD
- varying presentations and novel ways of presenting material, to assist recall (180)
- seating to minimise sound and visual distractions, preferably close to the front
- Behaviour Management Contracts to moderate difficult classroom behaviours
- a dependable classroom environment with a well-displayed schedule
- workshops for students with ADHD to teach them skills for studying, including note-taking, memorising, how to access support, organisational skills such as prioritising and arranging special exam provisions

• ADHD listed as a separate "condition" on any tertiary institution website, so that students can more easily locate appropriate available support.

10.7 Collaborative approaches

Recommendation

163. Medical and education personnel should engage in high-level collaboration (e.g. with Wraparound teams or Positive Behavioural Intervention Support teams, exchange of information, completion of surveys/questionnaires, joint meeting with family, with Wraparound teams or with Positive Behavioural Intervention Support teams) when a student presents with the characteristics of ADHD, to effect the best possible support for both student and family. ✓ Recommended best practice based on clinical experience and expert opinion

Shaddock et al (658) found that student learning is enhanced by a) good communication among teachers, students, parents/caregivers and the school community, and b) by teachers having specific knowledge (often gained directly from parents and students) about how each student's learning can be facilitated. Knowledge of their students assists teachers in delivering a differentiated curriculum, for example.

"Student knowledge" is defined as knowing the student personally, and having detailed information on the student's academic profile and awareness of all areas of the student's development (i.e. social, behavioural and emotional), and how ADHD affects each area (696). Much of the information is learned from parents/caregivers and the students and cannot be learned during teacher training because it is student-specific. For example, two students in the same class may have ADHD, but they are unlikely to have the same learning profiles, characteristics of ADHD, personalities or adaptive skills.

Inclusive teachers plan thoroughly and extensively, and take a reflective and problem-solving approach to teaching (658). They seek new knowledge by listening to parents/carers and the student, asking questions, finding out which strategies work at home, seeking input from the medical profession, discussing best practices with colleagues, year co-ordinators and the school psychologist, and contacting consultants. Teachers in some schools work collaboratively with Learning Support Teams, special education teachers or Learning Support Co-ordinators (696).

Individual Education Plans (IEPs) or Individual Learning/Behaviour Plans, and Wraparound are collaborative planning approaches that can be effective if all the participants are committed to the processes. In each case, the student with ADHD and everyone involved in his or her education discuss and document long-term goals, short-term objectives, strategies and evaluation methods. The team meets regularly to monitor progress and set new objectives.

Wraparound is a planning process for building constructive relationships and support networks among families, teachers and other caregivers to benefit the inclusion or facilitate the progress of youth with emotional and behavioural problems. The process incorporates a family-centred and strengths-based philosophy of care, and is used to build consensus among a team of professionals, family members and other support providers. Wraparound planning has decreased the numbers of out-of-home and restrictive school placements and improved students' behavioural, academic, social and post-school adjustment indicators (for review see (697)).

Positive Behavioural Interventions and Supports (PBIS) (http://www.pbis.org) addresses the behavioural and discipline systems of schools. PBIS has been widely implemented in the USA and more recently several education systems in Australia have also instigated this program.
10.8 School-based interventions

Research question

- For children and adolescents with ADHD, do school-based interventions, compared to no intervention, affect outcomes?
- For children and adolescents with ADHD, do peer support, tutoring or mentoring programs, compared to no intervention, affect outcomes?
- For adults with ADHD, do university/TAFE-based interventions, compared to no intervention, affect outcomes?

Evidence statements

- One systematic review and six primary research studies found that school-based interventions were of benefit to children in improving ADHD symptoms and classroom behaviour (698-703). (Level II, III-1, III-2)
- In two of three studies that addressed the efficacy of school-based interventions on academic performance in children, no clear benefits were reported (698, 699, 704). (Level II, III-2) One study found improvement in academic performance (704) (Level III-3)
- Three studies found that tutoring programs were of benefit for children with ADHD (705-707). (Level III-2, IV)
- One study found that self or parent monitoring of homework improved homework completion and homework problems for adolescents with ADHD (708). (Level III-2)
- Interventions for adults in university or TAFE settings have not been trialled.

Recommendations

164. Schools should have policies and procedures in place to support students experiencing learning, behaviour, organisation and concentration difficulties; for example, pre-referral processes, Wraparound and Positive Behavioural Intervention Support teams.

✓ Recommended best practice based on clinical experience and expert opinion

- 165. Effective school-based interventions, including peer tutoring, mentoring and peer support (e.g. buddy systems), should be considered for children and adolescents experiencing learning, behaviour, organisation and concentration difficulties to enhance their learning, social and behavioural outcomes. (Grade C)
- 166. Well-designed research into behavioural and school-based academic interventions that teachers can effectively and easily implement for the benefit of students with ADHD is needed.

10.8.1 Background

Behavioural interventions

Explicit teaching and strong teacher–student relationships are the bases of effective education provision. Schools must also establish policies and procedures to support students with special needs, as required by the Disability Discrimination Act. A pre-referral process works from a strengths-based model and determines which strategies are already in place and which need to be established.

"Response to intervention" (RTI) integrates assessment and intervention within a multilevel prevention system to maximise student achievement and to reduce behaviour problems. With RTI, schools can identify students at risk for poor

learning outcomes, monitor student progress, provide evidence-based interventions and adjust the intensity and nature of those interventions depending on a student's responsiveness, and identify students with learning difficulties (<u>http://www.rti4success.org</u>).

School-based behavioural interventions for ADHD generally focus on disruptive behaviour and task engagement and utilise behaviour modification techniques such as support groups, cognitive behavioural therapy conducted by teachers or time-out procedures. Co-operation and co-ordination between the school, the teachers and the parents/caregivers are required as well as the expertise, time and resources to develop and implement the intervention programs. While studies show that interventions have some impact, it is not always sustained, and beneficial impact may be isolated to one area of development or one context.

Whole school strategies aim to prevent disruptive behaviour in schools. One such program, Positive Behavioural Interventions and Supports (PBIS) (http://www.pbis.org), has been the subject of extensive research in the USA, including RCTs (709, 710). PBIS has been implemented in over 5,000 schools in the USA, with encouraging results in reducing referrals and suspensions and improving academic results and social skills (709). This program has been implemented by several State education systems in Australia.

PBIS uses applied behaviour analysis to gain an understanding of the purpose of a child's behaviour (711). As a systems-based approach, it endeavours to fix the context rather than the child by creating a supportive academic and social environment (711). PBIS is a 3-tiered prevention and intervention model of positive behavioural supports: "primary" school-wide preventative supports, "secondary" interventions for groups of at-risk students and "tertiary" intensive support for individual students (709). In this model, a leadership team of school and community personnel works collaboratively to align school policy, political commitment, funding, resources and training, and monitor implementation (709).

The impact of PBIS has not been specifically evaluated for children and adolescents with ADHD. Research is required to determine if this program will be of particular benefit to students with ADHD.

Another school-based program in use in Victoria is CASEA (CAMHS and Schools: Early Action Program). CASEA has been developed in Victoria by the Royal Children's Hospital and is currently being run in some Melbourne schools. CASEA is a multilevel prevention and early intervention program developed to reduce the incidence and impact of behavioural disorders, particularly CD (http://www.rch.org.au/mhs/casea/index.cfm?doc_id=10214).

Academic interventions

School-based academic interventions target academic performance by focusing on academic instruction, materials or environment. These interventions include help with organisational strategies and computer-assisted instruction.

The complex needs of learners with ADHD may mean that a range of interventions has to be trialled by classroom teachers or tertiary instructors to effect the best possible support for these learners with additional needs.

Several small studies have been conducted that show promising results; however, replication in larger well-designed RCTs is required. For example, six male students aged 11–12 years with a diagnosis of ADHD were trained in self-management that focused on the improvement of classroom preparation skills and homework completion, plus self-monitoring skills using a written log (712). After training, the students were monitored individually, with monitoring fading as students increasingly met their targets. All participants showed improvement, and improvements were maintained after written self-monitoring ceased.

In another study (713), a computer-mediated social skills training program was trialled with four primary school children with ADHD. The program presented specific social skill sequences in a variety of computer facilitated formats, with video peer modelling, social problem solving and reinforcement components. All participants showed improved social problem-solving skills, which were maintained at 6-week follow-up.

Peer support and mentoring

Just as collaborative problem-solving and planning processes can support students with ADHD, so too can collaboration among students, in the form of peer support, mentoring and peer tutoring.

A synthesis of 38 studies of children with emotional and behavioural disorders found that peer tutoring was effective with primary, middle school and secondary school students. Benefits in behavioural symptoms, social interactions and academic outcomes were reported (714). Peer tutoring can involve same-age or cross-age tutoring, reciprocal teaching or reverse-role tutoring (714, 715). In reciprocal teaching, students take turns in being tutor and tutee. In reverse-role tutoring, the student with the learning difficulty is the tutor. In a small study of children with emotional and behavioural disorders, reverse role tutoring generated improvements in self-esteem, behaviour and relationships with peers (715).

Glomb et al (716) describe the development and implementation of a school-based mentoring program that matches children and adolescents (both primary and secondary school) with learning disabilities and attention problems with university students who have experienced similar challenges. Program staff hold weekly 1-hour seminars to provide training and feedback to student mentors. Children and mentors meet for 1 hour per week. Parents and teachers reported moderate to significant improvements in school performance – specifically homework completion – and in attitude towards school. The program was most successful with the 9–12-year-old age group.

10.8.2 Summary of research evidence

DuPaul and Eckert (698) reviewed 63 school-based interventions for ADHD and concluded that school-based interventions were effective in enhancing classroom behaviour, but effects on academic performance were less clear. The methodology of the included studies was varied, with many studies having a small number of participants, no control group or no clear diagnostic criteria for the inclusion of participants.

A number of studies have been conducted since that review:

- A well-designed RCT was used to investigate the use of a combination of behaviour modification techniques in the kindergarten classroom by specially trained teachers (699). The behavioural interventions included a token system, time-out procedures, group training in social skills and anger control, and a daily school report card with home-based reinforcement. When compared to the no-treatment group, children receiving the classroom intervention showed improvement in teacher-rated attention, aggression and social skills behaviour, but there was no improvement in academic achievement and the changes in behaviour were not sustained at home. At the 2-year follow-up, Shelton et al (717) found no differences between the treated and untreated groups.
- A "Stop and Think" program of cognitive behavioural therapy that was aimed at increasing concentration and reflection resulted in significant improvements in ADHD symptoms, school problems and antisocial behaviour measures at post-treatment testing and 2-month follow-up (700). However, there was no benefit in social adjustment or academic performance.

- The use of behaviour modification and cognitive behavioural techniques carried out by teachers in the classroom was reported in two studies. In the first of these, Miranda et al (701) reported significant benefit in the ADHD symptom measures of hyperactivity and self-control only. In the second study, Miranda et al (702) reported benefits of the classroom intervention over a no-treatment control group in measures of inattention and hyperactivity.
- Use of computer-based working memory tasks with escalating difficulty was investigated in a well-designed RCT involving primary school children (703). Parent ratings of inattention and hyperactivity/impulsivity measures were improved, compared to the control group. No differences were found between groups for teacher-rated ADHD symptoms. The impact on academic performance was not addressed.
- Another study compared two interventions that involved teachers working with consultants (such as school psychologists) either as usual (TDAI) or more intensively (IDAI) to develop interventions to target maths and reading skills (704). Although there were some improvements in reading and maths for both groups, without a control group it is difficult to determine the efficacy of these programs.
- A controlled study of middle-school children (11–14 years) with ADHD looked at either self-monitoring or parent-monitoring of homework compared to a wait-list control group (708). In addition to monitoring of homework, students were given training in study strategy. Homework completion and homework problems improved significantly in both groups compared to the control group.

Three studies were identified that addressed the use of peer support or mentoring specifically for students with ADHD:

- One study made use of a "buddy system" that was reported to reduce negative behaviour and increase participation and adaptation (705). Standard ADHD symptom scales were not used and the study was poorly reported, making it difficult to interpret the results.
- Another study investigated a class-wide peer tutoring program that was reported to increase the children's ability to engage actively in academic tasks and reduce the time spent in off-task behaviour (707). Half of the children in this study were found to have improved their academic performance.
- Use of individual tutoring by adults led to improvements in ADHD symptoms. This study did not address academic outcomes (706).

No studies addressing the use of university/TAFE-based interventions for individuals with ADHD were identified.

10.9 Resources for teachers

10.9.1 Websites

http://www.ldonline.org

LD OnLine website in the USA offers advice on ADHD and learning disabilities, including definition, symptoms, diagnosis, causes, treatment, as well as information for family and school, teenagers and adults with ADHD.

http://www.wraparoundkids.com

Australian website for Wrap Around Kids, a collaborative program which facilitates communication amongst family, teachers and professionals to support students with medical conditions, ADHD, ASD, LD, hearing and vision impairments, intellectual and physical disabilities, and emotional and behavioural disorders. Provides information on the program, benefits and access to an online resource centre. http://www.chadd.org

US Children and Adults with Attention Deficit/Hyperactivity Disorder website established by parents to disseminate information on ADHD. Non-profit organisation offers membership and information on causes, symptoms, evaluations, treatment, research studies and FAQs.

http://www.nimh.nih.gov

US National Institute of Mental Health website. Focuses on mental illnesses and behavioural disorders; research on mind, brain and behaviour. Offers information on ADHD symptoms, diagnosis, causes, family and ADHD, ADHD in adults, references and resources, scientific studies. Book on ADHD can be downloaded.

http://detcms.det.wa.edu.au/inclusiveeducation/detcms/inclusiveeducation/public/r esource-topics/supporting-students-with-attentional-difficulties/supporting-students-with-attentional-difficulties.en

Western Australian Department of Education and Training website. Includes an online resource: *Supporting Students with Attentional Difficulties*. Provides information on attentional difficulties, what teachers need to know, how to access and provide support, what to do in the classroom.

http://www.add.org.au

ADDults with ADHD (NSW) Inc. website. Addresses the needs of adults with ADHD and other related conditions, and also the needs of their families.

http://www.ais.sa.edu.au/resources/DDA%20text%20pages%20R8.pdf Students with Disabilities: Enrolment Guidelines for Independent Schools. This document from the Association of Independent Schools of South Australia provides advice in the area of disability discrimination and outlines the responsibilities of independent schools.

10.9.2 Books

Barkley RA (1995). Taking Charge of ADHD. New York: Guildford Press.

- Dendy CAZ (2000). *Teaching Teens with ADD and ADHD*. Bethesda, MD: Woodbine House.
- Kohlberg J & Nadeau Brunner K (2002). *ADD-Friendly Ways to Organize Your Life*. New York: Brunner-Routledge.
- Mooney J & Cole D (2000). *Learning Outside the Lines*. New York: Simon & Schuster.
- Nadeau K (1998). Help4ADD@HighSchool. Bethesda, MD: Advantage Books.
- Nadeau K & Dixon E (2004). Learning to Slow Down and Pay Attention: A Book for Kids About ADD, 3rd edn. Washington DC: Magination Press.
- Nadeau K, Littman EB & Quinn PO (2000). *Understanding Girls with AD/HD*. Silver Spring, MD: Advantage Books.
- Quinn P (1995). *Adolescents and ADD: Gaining the Advantage*. Washington DC: Magination Press.
- Quinn PO & Stern J (2001). Putting on the Brakes: Young People's Guide to Understanding Attention Deficit Hyperactivity Disorder, rev. edn. Washington DC: Magination Press.
- Reiff MI & Tippins S (2004). *ADHD: A Complete and Authoritative Guide*. Elkgrove, IL: American Academy of Pediatrics.
- Reimers C & Brunger BA (1999). *ADHD in the Young Child*. Plantation: Specialty Press.

- Rief S (2005). *How to Reach and Teach ADD/ADHD Children*, 2nd edn. New York: The Center for Applied Research in Education.
- Solden S (2005). *Women with Attention Deficit Hyperactivity Disorder*. Grass Valley, CA: Underwood Books.
- Young JL (2000). *ADHD Grown Up: A Guide to Adolescent and Adult ADHD.* New York: W.W. Norton.
- Zeigler-Dendy C. (2000). *Teaching Teens with ADD and ADHD: A Quick Reference Guide for Teachers and Parents*. Bethesda: Woodbine House.

10.9.3 DVDs

Last one picked ... first one picked on. Richard Lavoie. When the chips are down. Richard Lavoie. How difficult can this be? The F.A.T. City Workshop. Richard Lavoie. Beyond F.A.T. City: A Look Back, A Look Ahead. Richard Lavoie. ADHD in the Classroom: Strategies for Teachers. Russell A Barkley. ADHD in the 21st Century. Edward M Hallowell.

CHAPTER 11. COMPLEMENTARY AND ALTERNATIVE TREATMENTS FOR ADHD

11.1 Introduction

A number of other treatment modalities have been advocated and used for the management of ADHD. These are often referred to as "complementary and alternative medicines" (CAMs), although it is difficult to draw a clear line between CAMs and mainstream approaches (718). An alternative treatment in the context of ADHD is any treatment other than prescription medication or standard psychosocial behavioural treatments that claims to treat the symptoms of ADHD with an equal or better outcome.

Just as with mainstream treatments, some CAM therapies may cause adverse effects and can result in adverse drug interactions, some of them potentially serious. It is important that healthcare practitioners treating people with ADHD ask about their patients' use of both traditional and non-traditional approaches. Similarly, it is important that people with ADHD and/or their parents and carers tell their healthcare practitioners about all the therapies they are using, both traditional and non-traditional.

Two studies have found that the use of CAMs for ADHD is quite common in Australia, with 67% (50/75) of surveyed families in a Victorian sample (719) and 64% (186/290) of surveyed families in a Western Australian sample (720) reporting the use of alternative treatments. Notably in both studies the most frequently reported CAM was modified diet (719, 720).

In general, CAM therapies have undergone less rigorous assessment than other therapies for ADHD, making it difficult to provide positive recommendations for them, despite their widespread use and popular support.

11.2 Diet

11.2.1 Elimination and restriction diets

Research question

• For individuals with ADHD, do diets that eliminate or restrict certain foods or food components, compared with no intervention, affect outcomes?

Evidence statements

- One poorly designed trial on the oligoantigenic diet reported improved behaviour in 50% of participants (721). (Level III-2)
- One small trial found that children with ADHD on a restricted elimination diet showed significant improvement in parent- and teacher-rated ADHD behaviours compared to a control group not on the diet (722). (Level III-2)
- Two food challenge trials assessing the impact of sodium benzoate and artificial food colourings on children from the general population reported an increase in hyperactivity (104, 106). (Level II, III-2)

Recommendations

167. Healthcare professionals should encourage good nutrition and a balanced diet.

✓ Recommended best practice based on clinical experience and expert opinion

- 168. Elimination and restriction diets are not supported as a general treatment for all individuals with ADHD. Consumers considering the use of elimination or restriction diets should be informed about the uncertainties surrounding the efficacy of these diets in treating ADHD, and of the potential risks of unsupervised elimination diets. (Grade D)
- 169. A subset of children may be sensitive to certain foods or food additives and benefit from careful exclusion diets. Assessment of food sensitivity and initiation of a special diet should be under the care and supervision of a medical specialist and an Accredited Practising Dietitian.
 - ✓ Recommended best practice based on clinical experience and expert opinion

Background

Elimination or restriction diets for ADHD are based on the assumption that certain foods or food components affect behaviour. The diets most commonly advocated for use in ADHD restrict food additives and/or preservatives, eliminate sugar, or eliminate potentially allergenic foods such as dairy products, wheat, corn, yeast or soy (oligoantigenic diets) (723, 724).

Modified diets to assist symptoms of ADHD came into prominence in the early 1970s, when Feingold published *Why Your Child is Hyperactive* (1975). In this book he proposed that many children were sensitive to dietary salicylates and artificial colours, flavours and preservatives, and that learning and behavioural problems could be improved by eliminating all food additives and naturally occurring salicylates from the diet. Subsequent controlled studies have generally not supported the Feingold diet (725-727).

Other early studies, such as those by Egger et al (728) and Carter et al (729) used open trials to identify which foods affected individual children, and then tested those incriminated foods using a double-blind crossover design. The incriminated foods varied substantially between children, and included natural foods (e.g. cow's milk, wheat flour, citrus fruit, eggs) as well as artificial colourings and preservatives. Both studies indicated that the results of the open trial could be replicated in the double-blind follow-up study. Some children were helped by individually designed elimination diets, at least in the short term. Carter et al (729) suggested that children's responsiveness to incriminated foods was predicted by parents' informal observations.

Summary of research evidence

At the time of writing the 1997 ADHD Guidelines, the authors found that scientific evidence to support the use of restriction diets in the management of ADHD was lacking. Three primary studies that were focused on preschool-aged children and children were identified. No trials involving adolescents or adults were found.

Schmidt et al (721) addressed the use of the oligoantigenic diet (avoiding known food allergens) in children with ADHD. The diet comprised two meats (lamb, turkey), two carbohydrate sources (rice, potatoes), two vegetables (cabbage, carrots) and two fruits (apples, bananas), plus apple juice and mineral water. Improved behaviour was reported in approximately 50% of participants. However, the study was of poor quality, with several limitations in design and reporting that make the results difficult to interpret.

A recent randomised, controlled trial in 27 children with ADHD looked at the use of a restricted elimination diet (722). The diet consisted of rice, turkey, lamb, vegetables, fruits, margarine, vegetable oil, tea, pear juice and water. The number of children who showed improvements in parent- and teacher-rated ADHD symptoms was greater in the diet group than in the control group (722).

Two food challenge trials (104, 106) addressed the impact of artificial food colourings on hyperactivity. The participants were children aged 3 years or 8–9 years drawn from the general population, not specifically children with ADHD. These studies found that a short-term challenge (1 week) with sodium benzoate preservatives and artificial food colourings moderately increased features of hyperactivity in both 3-year-old children and 8–9-year-old children irrespective of their atopic status (as identified by skin prick testing) (104, 106).

Overall, there is insufficient evidence to draw conclusions on the use of elimination or restriction diets for the treatment of ADHD. It is possible that a small subset of children may have true sensitivity to certain foods or food additives that could be helped by careful exclusion diets. The research does not support alterations to the diet as a general treatment for all individuals with ADHD. Further research is required using standardised and validated outcome measures in RCTs with a parallel design.

11.2.2 Diet supplements

Numerous dietary supplements have been suggested to be of benefit for the treatment of ADHD; however, there are few well-conducted RCTs evaluating the efficacy of these treatments. The exception is supplementation with essential fatty acids (fish oils), which has been the subject of several controlled trials.

Research question

• For individuals with ADHD, does diet supplementation with essential fatty acids, compared with placebo or no intervention, affect outcomes?

Evidence statement

• Five trials showed conflicting results when comparing essential fatty acids with placebo in children with ADD/ADHD symptoms (730-734). (Level II-III-1)

Recommendations

- 170. There are insufficient research data to recommend the use of diet supplementation with essential fatty acids for the treatment of ADHD. Consumers considering the use of essential fatty acids should be informed about the uncertainties surrounding their efficacy in treating ADHD.
 ✓ Recommended best practice based on clinical experience and expert opinion
- 171. The use of essential fatty acids in the management of ADHD symptoms warrants further investigation in well-designed randomised controlled trials (RCTs), including clarification of dosage levels and types of essential fatty acids used.

Background

Essential fatty acids (EFAs) are important for brain development and function. They cannot be synthesised by the human body and must be obtained from dietary sources. EFAs include the omega-3 fatty acids, eicosapentaenoic acid (EPA), docosapentaenoic acid (DPA) and docosahexaenoic acid (DHA); and the omega-6 fatty acids, arachidonic acid (AA), linoleic acid (LA), alpha-linoleic acid (ALA).

The NHRMC has recently recommended an increased consumption of omega-3 fatty acids in the average Australian diet to promote general health (735), suggesting a daily intake of 610mg/day for men and 430mg/day for women. In addition to the benefits to general physical health, EFAs are often promoted as an alternative treatment for many medical and psychiatric conditions, including ADHD.

Nerve cell membranes are composed of phospholipids containing large amounts of polyunsaturated fatty acids (omega 3 and 6), and it has been suggested that deficiencies in EFAs may impact upon brain function in such a way as to cause or worsen some ADHD symptoms. This has been supported by research findings showing that levels of EFA in the blood plasma of people with ADHD are reduced compared to healthy controls (for reviews see (723, 736, 737)). This in turn has led to the hypothesis that dietary supplements of EFA may reduce symptoms of ADHD.

It is important to note that the commercially available EFA supplements may include only a single type of EFA or variable ratios of different types of EFA.

Summary of research evidence

Five randomised, double-blind, placebo-controlled trials were identified that addressed the use of EFAs in the treatment of ADHD. The studies involved children aged 6–13 years, and trialled 8–16 weeks of EFA supplementation. No trials were found involving adolescents or adults with ADHD.

The efficacy of EFA supplementation in the treatment of ADHD remains unclear. Two trials found no improvement with EFA compared to placebo (730, 731), while three trials showed significant benefits in a subset of ADHD-related measures (732-734). None of the trials reported any serious adverse events.

The inconsistent findings may be the result of differences in study design, such as inclusion criteria, dosage levels and type of EFA used (DHA, EPA or a combination of the two). Furthermore, the studies did not take into account confounding factors

such as the inclusion of people on stimulant medication. A formal diagnosis of ADHD was made in only two of the studies.

11.3 Chiropractic

Research question

• For individuals with ADHD, does chiropractic, compared with no intervention, affect outcomes?

Evidence statement

• Chiropractic treatment for ADHD has not been evaluated in clinical trials.

Recommendation

172. Consumers considering chiropractic should be informed about the current lack of research to assess its efficacy in the treatment of ADHD. ✓ Recommended best practice based on clinical experience and expert opinion

Background

Chiropractic interventions are based on a belief that spinal problems are the cause of health problems, and spinal manipulations ("adjustments") can restore and maintain health (738). Advocates believe that imbalance in muscle tone can cause an imbalance in brain activity, and that spinal adjustments and other somatosensory stimulation (e.g. exposure to various frequencies of light and sound) can effectively treat ADHD and learning disabilities. Other chiropractors believe that the skull is an extension of the spine and advocate a method called applied kinesiology or neural organisational technique. The premise is that learning disabilities are caused by misalignment of the sphenoid and temporal bones of the skull creating unequal pressures on the brain. Treatment involves restoring the cranial bones to the proper position through specific bodily manipulation.

Summary of research evidence

No studies addressing the use of chiropractic treatment for the management of ADHD were identified. The literature in the area is limited to a small number of case reports.

11.4 Behavioural optometry

Research question	W

• For individuals with ADHD, does behavioural optometry, compared with no intervention, affect outcomes?

Evidence statement

• Use of behavioural optometry, including use of Irlen lenses, for the treatment of ADHD has not been evaluated in clinical trials.

Recommendation

173. Consumers considering behavioural optometry should be informed about the current lack of research to assess its efficacy in the treatment of ADHD. ✓ Recommended best practice based on clinical experience and expert opinion

Background

Behavioural optometry refers to the use of lenses and vision training to improve vision and its impact on behaviour and learning ability. Visual problems such as faulty eye movements, sensitivity of the eyes to certain light frequencies and focus problems may cause problems with reading. Vision therapy may assist some children with learning difficulties if they have existing visual problems. Programs vary widely and may include eye exercises, educational and perceptual training (e.g. general body movements designed to improve visual perception), and the use of tinted or Irlen lenses.

Summary of research evidence

No studies addressing the use of behavioural optometry for the management of ADHD were identified. The literature in the area is limited to a small number of case reports.

11.5 Biofeedback

Research question

• For individuals with ADHD, does biofeedback, compared with no intervention, affect outcomes?

Evidence statement

• Preliminary studies of biofeedback for the treatment of ADHD in children show some benefit for ADHD symptoms (739-744). (Level III-1, III-2, IV) Larger, well-designed trials are needed.

Recommendations

- 174. There are insufficient research data to recommend the use of biofeedback for the management of ADHD symptoms. (Grade D)
- 175. Consumers considering biofeedback should be informed about the uncertainties surrounding its efficacy in the treatment of ADHD, the time required to undertake the treatment and its costs. Recommended best practice based on clinical experience and expert opinion
- 176. The use of biofeedback in the treatment of ADHD symptoms warrants further investigation using well-designed RCTs.

Background

Biofeedback, also called neurofeedback or electroencephalogram (EEG) biofeedback, aims to train people to self-regulate their brain activity so that they achieve a state of focus. Computerised monitoring of the EEG is fed back to the participant in real time via visual and auditory signals. When the person makes desired changes in the targeted behaviour, he or she is rewarded and the change is positively reinforced. Biofeedback is conducted over a series of sessions over several weeks.

The underlying principle behind the use of biofeedback for the treatment of ADHD is based on research findings that suggest that a person's level of alertness and behavioural control is associated with particular EEG frequencies in specific regions of the brain (reviewed in (745)). In addition, quantitative EEG studies have identified differences in EEG rhythms between individuals with ADHD and individuals without ADHD. The resting EEG of children with ADHD shows increased slow wave activity and decreased fast wave activity (reviewed in (368)). The purpose of biofeedback is to train the brain to control specific EEG frequencies, which will theoretically increase attentiveness and improve behavioural control.

Summary of research evidence

One systematic review and five primary studies were identified that addressed the use of biofeedback in the treatment of ADHD. No RCTs were found.

- The systematic review included 18 studies, 17 of which reported benefits for symptoms from biofeedback in ADHD (739). The review highlighted the flaws in the literature, including a predominance of case studies and a dearth of studies using a control group, randomisation or a combination of outcome measures.
- In a case-series involving children aged 8–13, with no control group, the children showed improvements on a number of ADHD symptom measures, but there was no clear benefit when parent-rated DSM-IV criteria were used (740). The changes were stable at 6 months follow-up. The results are difficult to interpret as 5 of the 23 children were also taking stimulant medication.
- A study of 13 children aged 7–13 years with ADHD found significant improvement in parent ratings of ADHD symptoms in the biofeedback group compared to a wait-list control group (741). Only one rating scale was used, and participants were permitted stimulant medication, which may have confounded results.
- In a study of biofeedback as part of a combined treatment strategy that also included medication and psychosocial interventions, inattention and hyperactivity were significantly improved when neurofeedback was included, compared to the control group (742).
- The remaining two studies compared biofeedback to stimulant medication. One study reported improvements with both biofeedback and medication treatment, but found no differences between the groups (743). The second study reported improvement with both biofeedback and medication on three subscales of the TOVA, but again found no differences between the groups (744).

Although no studies reported adverse events, it was not clearly reported that any of the studies had sought to measure adverse events.

Thus, while biofeedback shows some promise in the treatment of ADHD, RCTs are needed to clarify whether it can be recommended instead of or in combination with medication in ADHD.

11.6 Homeopathy

Research question

• For individuals with ADHD, does homeopathy, compared with no intervention, affect outcomes?

Evidence statement

• One good quality systematic review reported that no conclusions could be drawn from the available 3 small RCTs (746). (Level I) Larger, well-designed trials are needed.

Recommendation

177. There are insufficient research data to recommend the use of homeopathy for the treatment of ADHD. Consumers considering the use of homeopathy should be informed about the uncertainties surrounding its efficacy in treating ADHD. (Grade D)

Background

Homeopathy is based on the principle that "like treats like": substances that are capable of causing specific diseases, and that produce symptoms in a healthy individual, are used to treat the same symptoms when they occur as part of a disease or condition. The goal of treatment is to stimulate the body's own defence mechanisms. Homeopathy is individualised through both diagnosis and treatment

strategies. The active ingredients are given in highly diluted form and for this reason are considered reasonably harmless (747).

Homeopathy has been suggested to be of benefit in a variety of conditions including ADHD.

Summary of research evidence

The research articles identified that address the use of homeopathy for ADHD are primarily case reports. There was a dearth of controlled trials, and no studies addressed the use of homeopathy in adults with ADHD.

A review of three double-blind, placebo-controlled RCTs in children aged 0–19 years who had a diagnosis of ADHD reported that two studies found a benefit from homeopathy compared to placebo, while the third found no differences between the intervention and control groups (746). In one of these trials, adverse events led to the withdrawal of 4 out of 62 participants.

The small numbers of participants in these trials and the varying results mean that at present there is insufficient data to draw a conclusion about the use of homeopathy as a therapy for ADHD. Larger, well-designed trials are needed in all age groups.

11.7 Cerebellar therapies

Research question

• For individuals with ADHD, do cerebellar therapies, compared with no intervention, affect outcomes?

Evidence statement

• One preliminary, uncontrolled study of cerebellar treatment for the management of ADHD has shown some improvement in ADHD symptoms (748). (Level IV)

Recommendations

- 178. There are insufficient research data to recommend the use of cerebellar therapies for the management of ADHD symptoms. (Grade D)

Background

Cerebellar therapies, such as those used in the Dore Program, aim to stimulate the cerebellum through a series of exercises focused on motor skills. The exercises are conducted for 5–10 minutes each day over an average of 13 months. Cerebellar therapies have been advocated for use as a treatment for a number of conditions including dyslexia and ADHD.

Summary of research evidence

One primary study was found on the use of exercises designed to stimulate the cerebellum in people with ADHD (748). It involved 891 consecutive clients at the Dore Centres in the United Kingdom, 698 aged under 16 years and 193 aged over 16 years. Improvements were reported in inattention and hyperactivity/impulsivity, which appeared to be sustained for 6–12 months after completing the program. No unwanted side effects were reported.

These results, however, must be interpreted with caution. The report was a poster presentation, not a peer-reviewed publication; it lacked a control group; and it did not use objective outcome measures. Further research is required using well-designed RCTs.

Cerebellar therapies are expensive and as such consumers should be aware of the shortcomings in the existing research literature (for review of the Dore Program, see (749)).

11.8 Sport, exercise and relaxation

Research question

• For individuals with ADHD, does participation in sport, exercise or meditation programs, compared with no intervention, affect outcomes?

Evidence statements

- Two small studies show some benefit of yoga in children and adolescents with ADHD (750, 751). (Level IV)
- Two small studies show some benefit of massage therapy in children and adolescents with ADHD (752, 753). (Level IV)
- One small study showed some benefit of tai chi in children and adolescents with ADHD (754). (Level IV)

Recommendations

- 180. Yoga, massage and tai chi may be of some benefit in treating ADHD. (Grade D)
- 181. While other sport, exercise and meditation programs may have many mental and physical health benefits, their role in the management of ADHD has yet to be demonstrated.

Background

Exercise, relaxation and meditation programs are increasingly being promoted as a treatment for ADHD. While there is strong evidence for the physical health benefits of exercise, research into its effects on mental health is less common. Recent systematic reviews addressing the impact of physical activity on self-esteem (755) and anxiety and depression (756) demonstrated positive effects. However, both reviews noted the small number of studies and the need for further research.

Summary of research evidence

Five case-series were identified. They addressed the use of yoga (750), yoga meditation (751), massage (752, 753) and tai chi (754) for children and adolescents with ADHD. No trials involving preschool-aged children or adults were identified. No studies evaluating exercise or sport programs were identified.

The three identified studies involved pre- and post-test outcome measures using the Conners' Parent or Teacher Rating Scale, and all showed some evidence of improvement in ADHD symptoms. The methodology of the studies was limited and none of the three involved randomisation or use of controls. No adverse effects were reported.

Further research is required with larger numbers of participants in RCTs.

11.9 Sensory integration interventions

Research question

• For individuals with ADHD, do sensory diets / sensory integration treatments, compared with no intervention, affect outcomes?

Evidence statement

• One preliminary study in children with ADHD reported improvements in ADHD symptoms in 11/20 study participants (275) (Level IV)

Recommendation

182. There are insufficient research data to recommend the use of sensory diets / sensory integration therapies for the treatment of ADHD. Consumers considering the use of sensory diets / sensory integration therapies should be informed about the uncertainties surrounding the efficacy of these programs in treating ADHD. (Grade D)

Background

Sensory integration therapy is an intervention used by occupational therapists for children who have difficulties processing sensory information (see section 5.7 Psychoeducational assessment, page 45). Sensory integration therapies have been widely used among occupational therapists working with children with developmental, learning and behavioural problems (for review see (274)). A recent survey of Australian occupational therapists found that approximately 50% of the surveyed therapists had specific training in sensory integration therapy (757). The content of sensory integration therapies varies as they are generally developed to be specific for each individual child. The term "sensory diet" is sometimes used and this describes a daily program of sensory activities for the child at school and at home.

In general, the efficacy of sensory integration therapies has not been rigorously studied (274, 758). Preliminary data from one well-designed controlled trial suggests sensory integration therapy may be of benefit in treating sensory modulation disorders (758).

Summary of research evidence

One study was identified that utilised occupational therapy with a sensory integration approach for the management of children with ADHD. This small preliminary study of 20 children found improvement in ADHD symptoms in 11 of the 20 participants based on either parent or teacher reports. The study did not include a control group and was not blinded, therefore the results should be interpreted with caution. Further research is required with larger numbers of participants in RCTs.

PART IV: SOCIAL AND ECONOMIC CONSIDERATIONS



CHAPTER 12. THE VOICES OF INDIVIDUALS WITH ADHD

Many people, both adults and children who have ADHD, are as yet undiagnosed and unaware of the real cause of their difficulties. But increasingly, people are learning about the condition, joining support associations and seeking treatment. For many people, this newly found access to knowledge which accurately reflects their own life and experience of the world, enhances their sense of personal freedom and autonomy. Here are some stories provided by members of Australian ADHD support associations.

12.1 Adolescent voices

A group of six Melbourne adolescents participated in a panel discussion about their experiences with friends before and after they were diagnosed with ADHD and tried medication (759). The names of the adolescents have been changed.

Starting treatment

Moderator: What things did you notice when you first started the medication?

Peter: People didn't get so annoyed with me as they used to and it was just a lot easier to get on with people.

Robert: I didn't really notice anything, but other people were saying "You're not being such a little shit".

Jane: I noticed that I got a lot more friends than I'd had, and it helped me to concentrate a bit better.

Craig: I've been getting better grades at school since I've been on medication, and it's been a bit easier for me at home with my Mum as well.

John: I got so many more friends after I started taking medication ...

Moderator: Why do you think you had more friends?

John: I wasn't acting stupid.

Robert: My friends all said "Why aren't you being so rebel-ish any more?"

Peter: A lot of people who you thought were your friends were just there to watch you get into trouble. I fell into the mould of doing things to get a laugh out of people, doing stupid things and getting myself into trouble. When that sort of thing stops, you get real friends coming to you.

School

Moderator 1: What aspect do you think the medication helped you most with?

Robert: Schoolwork.

John: Being able to sit down and do ONE thing instead of trying to do half of three million things.

Peter: Not doing stupid things. I just don't feel the need to throw something across the class or flick someone's ear. I don't feel the need any more.

Belinda: You don't feel the need to be the centre of attention any more.

Peter: Yes. Exactly.

Belinda: You feel like you can actually be like everyone else in the classroom and actually do something, and not be there for entertainment's sake.

Peter: It makes you a lot more thoughtful.

Moderator 2: How do you think the medication helps you in the practical day-to-day activities in doing your schoolwork?

John: It works wonders. I've just had a week of almost solid exams and I sat one of those without my tablet. It was a disaster. I just sort of thought "Oh, I can't do that question so I might just draw a little cartoon on the bottom of the page". And you look at it afterwards and you think "My God! What was I doing? Why on earth did I do this?"

Robert: And you get it back and you think "Why did I put that?"

Peter: Yes exactly. And you often overlook really simple things.

Belinda: Or you just keep on reading the same question over and over.

Peter: That's a common one.

Robert: Or you skip questions. You think "I can't be bothered doing that one so I'll go to the next one".

Friends and family

Moderator: How do you feel about yourself, your achievements, about how you get on with your friends and family?

Belinda: The teachers have always said that they don't understand why you're like this, because you've got the ability to do well, but you don't. Now that you're actually getting the grades that they think you should have got, it's so much easier, because they don't think that you're not trying ...

Belinda: It's so much easier to be in a situation where you are not always fighting with the teachers all the time. And fighting for them to listen to you because you feel like you haven't done anything wrong.

Peter: You don't have a fear of school any more, like I used to ...

Peter: The main thing that you hear from the teachers and that I see on [my] reports all the time is "Not working to his full capacity. Has the potential but ... Blah blah blah".

John: If you know what you want to do, it's good, but if you don't know what you want to do and you don't have the concentration to sit down and think about what you want to do, it's a real problem.

Belinda: It opens up a whole new area that you can look at for a career because before [some] areas were too hard to concentrate on like science, but now, it's so much easier.

John: Before: garbo? Afterwards: Um engineer.

12.2 Adult voices

Voice 1

Being diagnosed as an adult with ADHD was the most dramatic and life-changing event and provided an explanation to the difficulties I had experienced since I was a child. My behaviour, poor academic performance and laziness were clustered under two broad headings – environmental problems and, of course, poor parenting. Looking back at this journey I can't help feel that it was so unnecessary and so different had the medical expertise available today been in place then.

Those early school days were filled with educational assessments, tutoring, reading, spelling and mathematics support. The negative impact from these early years on my self-esteem, social development and academic outcomes still create painful and embarrassing memories to this day. It seemed that my mother's efforts were

disproportionate to the outcomes and the criticism and cynicism she received as a parent were unhelpful and unjust.

My journey with ADHD came to an end, or perhaps the beginning, a little over 10 years ago. At that time, after another job change, I found myself facing the reality that I could no longer make excuses for my behaviour. Also my son, who was in Year 3 at the time, had been experiencing a number of learning and behavioural difficulties at school and was subsequently diagnosed with ADHD and medicated.

As I educated myself about my son's condition, read the texts and articles, I found myself leaping from the pages. I wasn't just reading about the condition my son was affected by, but about myself. Whilst I was filled with an enormous sense of relief to finally have a potential explanation for the difficulties I experienced, [there was] a real sense of grief for the life that could have been had I been diagnosed as a child.

I then sought the advice of a specialist in the field of adult ADHD. After a lengthy history and discussion of my behaviours, with input from my wife, a diagnosis was made and medication prescribed. The benefits were almost immediate. Suddenly I could see myself doing the actions that people had commented upon for years. I could now remain focused, complete simple tasks quickly and be less socially and verbally impulsive. My life went from ordinary to good quite quickly, and within two years I was able to describe all aspects of my life as excellent.

My journey to this point has shown me that I had to search out the appropriate professionals, educate myself about the condition and work closely with the people assisting to achieve the best outcome.

Voice 2

What is it like living with ADHD? Every family who has a member with that diagnosis will have a wealth of anecdotes to tell you. One of the compensations of being inflicted with that or Autism, is that those affected do intrinsically "out there" things. It is a bit like gallows humour. I have had to deal with my kids doing things ranging from putting the puppy in the tumble dryer, to putting siblings into the water tank and forgetting about them. Once after I had driven the two hours to Melbourne, I discovered on arrival that one of my offspring had disassembled the entire back seat of the car, removing all of the bolts.

What is living with ADHD? It is eternal vigilance, looking out for the possible dangers and likely impulsive triggers. It is nearly every weekend at the emergency department with fractured limbs and lacerations. It is annoyance at the sanctimonious hand wringers who "tut tut" about using drugs to sedate children. Those who are loudest in the media I suspect are not those who have family members with true ADHD.

We are involved in running a small school for students with medical disorders affecting their schooling. A large percentage of our students have a formal diagnosis of ADHD (using DSM-IV criteria) in addition to their other diagnoses. We provide in our setting as much of the environmental modifications as possible for attentional difficulties. All of our students do yoga, regular physical exercise, work in a small group setting (usually 2–3 in a group, no more than 6–7), use visual learning and computers. I know from personal experience (from the days that they have "forgotten" their tablets) that almost one-third of our group would not be able to attend school and learn anything without their medication.

Do I have ADHD? My son clearly has, but I am (well, relatively speaking) quite functional. One day on impulse, I decided to do the ADHD questionnaire on myself. To my surprise I could tick almost every box for ADHD, save the one that stated that there had to be clear impairment in work or social settings (which begs the question, what is impairment?) To my even greater surprise, when I mentioned this to a couple of close friends, they were not the slightest bit surprised. They had clearly seen these characteristics.

How much better would I function without these traits? I don't have any learning disabilities, have been lucky enough to have a good memory and no language difficulties. I work in a complex environment, juggling a lot of tasks and making quick decisions about courses of action. On the other hand, perhaps this is a good environment for someone with these traits. Let's think about this. How have I structured my life?

I am a GP. I live in the country and several years ago acquired a second-hand bicycle to try to keep my exercise up (good, sound, preventive health advice). I ride to the hospital for ward rounds, then the clinic, then home for lunch, back to the hospital if there are any emergencies or to home visits and finally home at the end of the day. I have 15 minute appointments. That means every 10 or 15 minutes I get up out of my swivel chair –I really like that swivelling! – walk to the front desk and call the next patient in. I see a large number of people each day, and probably get most of the social contact I need for the day. I am presented with a great variety of problems, from delivering babies to palliative care. I will see anyone that can walk, or be wheeled or carried through the door. I occasionally have medical emergencies that can include absolutely anything whatsoever.

My attention span is well suited to short appointments. If something drags on and is not interesting, I can find my attention wandering, thinking about black holes, where I can get some Wasabi plants for the backyard pool, and what I am having for dinner that night. As you do.

If I go to meetings, or even movies, I find it quite difficult to sit still for the required length of time, frequently recrossing my legs and moving about, until I get thumped by my partner and told to settle down.

I pondered on my bike riding. I don't ride sedately by, enjoying the sunshine. I usually ride flat out, for the couple of kilometres each way. There is no good reason for this, it is just what seems to happen. I thus get about 5 minutes strenuous exercise half a dozen times a day, that breaks up my sedentary existence.

I have staff! Think about that! They organise my appointments, make phone calls for me and rearrange my schedule if needed. If I have a weak spot (well, the most obvious one anyway), it is paperwork. My staff can't do everything, but it would be a hell of a lot worse without them.

Voice 3

I am a 45-year-old mother who was incredibly relieved, even joyful, to be diagnosed this year with ADHD; prior to that my life had been a roller-coaster of anxiety, depression, self-loathing and confusion. It was almost two years ago that I had reached a point when I could no longer function at work because the anxiety had become so overwhelming; my whole body shook and my mind felt like shattering. I stopped work, was prescribed antidepressants and saw a psychologist. Deep down I knew there was a specific cause; that there had always been something intrinsically different about me. But no one recognised the ADHD.

When my son's Kindergarten teacher told me he was bright but often unresponsive, easily distracted and slow to get going, I became concerned. I joined a group of parents with kids like my son and it was here that I had the opportunity to hear a guest speaker talk about ADHD. I had been researching relentlessly for months, scouring the internet and reading a slew of books. It became increasingly evident to me that both my son and I had ADHD.

On the whole, school was a miserable place for me, despite having been identified as gifted in Year 7. I often became so hyper-focused on what I was doing I would block out the world and that constantly got me into trouble. I was constantly late and I didn't hear the teacher. In fact, teachers often thought I was deliberately ignoring them and that I had an attitude problem. Another issue for me was remembering multi-step verbal instructions. It was simply impossible for me. I would sit there in a panic too scared to ask and watching everyone else for a clue. "Why weren't you listening; you need to learn to listen; pay attention!" the teachers would say ad nauseum. I would literally shutdown and withdraw into myself in an attempt to spare my heart and my head from more than I could bear. I realise now that the memory and hearing problems can be attributed to the "attention deficit", although I know I did pay attention as best that my brain would allow.

Labels don't help. Yet it does make me sad, angry and ashamed that I'm too fearful to tell my son that he has ADHD, or about the meds he's taking, simply because of the ignorance and narrowmindedness of even the highly educated. My battered self-esteem is still a major obstacle and the sense of loss I feel from not knowing until now the essence of my problem still plagues me. The media has stigmatised and sensationalised ADHD to such an extent I doubt I will ever be able to share it with my friends or my family.

Voice 4

I am a 28-year-old woman who was diagnosed with ADHD and medicated almost 2 years ago. I can honestly say that it has astronomically improved my life in all areas of functioning. I came to be diagnosed almost by accident. Working as a health professional I attended a "Law and the Courts" conference and had the opportunity to hear Dr Caroline Stevenson discuss the clinical presentation and nature of adult ADHD. It didn't take long for me to inch to the edge of my seat as I realised that I too actually had ADHD. I had found the answer that I had spent my life searching for!

As a child in primary school, I was a high academic achiever, however I felt different to my peers, as though I didn't fit in and that I was somehow fundamentally flawed. My family and teachers, however, had a very different perception, they saw me as intelligent, caring and capable.

The transition into high school was difficult. I felt overwhelmed and anxious and had significant difficulty sitting through classes and suffered from significant mood swings. I also had difficulty maintaining friendships in school. It was not long before my grades dropped and I soon began avoiding classes and completing assignments. My report cards displayed a common theme that I was not achieving academically to my potential. This caused significant problems at home with my parents. I completed high school feeling like a failure and believed I was of average intelligence.

I then attended university and chose to study within the health profession. Studying an area that I enjoyed, at a smaller university, I began to thrive and maintained a Distinction average. But the anxiety, depression and mood swings still prevailed. Intimate relationships were difficult. After the completion of my Postgraduate thesis the agitation, anxiety and mood swings worsened and I had a major depressive episode. I was admitted to a private psychiatric hospital and had my first contact with a psychiatrist who told me I had rapid cycling bipolar disorder, borderline personality disorder, generalised anxiety disorder, major depression and/or possibly dysthymia. Needless to say this was a very confusing and difficult time. Hospital staff informed me that I was so unwell it was unlikely I would complete my Masters program, nor would I ever handle full-time employment, and that clearly I suffered from severe mental illness.

My admission to hospital allowed me to gain the professional psychiatric assistance I needed; however, I received no answers. I always felt as though something was missing, something wasn't right. Over the years I have tried seven different

antidepressants, mood stabilisers and benzodiazepines. None of these medications provided me with much relief. I also saw a clinical psychologist and a psychotherapist for extended periods of time, but the mood swings and depression never seemed to lift.

I managed to finish my Masters in 4 years with a Distinction average. During this period I also developed a substance dependence, which was a direct result of needing to escape as a coping strategy, in addition to self-medicating – it was the only way at times I could feel relief.

After hearing [a doctor] speak at the conference, I researched adult ADHD and took the information to my psychiatrist. He had no knowledge of adult ADHD and the symptomatology, even though he had worked in drug and alcohol and mental health for years! We trialled Strattera. Within days the results were significant. I felt calmer and was able to finally fall asleep at night, as my mind was no longer racing with scenarios and ideas. My cravings for substances stopped overnight.

Following a second opinion, I then switched to a stimulant medication. I sleep at night, I don't have any feelings of agitation, restlessness or anxiety and I do not experience any significant mood swings. I feel focused and my work performance has improved to such an extent I can now complete the same amount of work in much less time and to a better standard. My social relationships have improved, my self-esteem is good, I no longer have the feeling as though something isn't right. I feel content.

It is difficult at times to live with this disorder, particularly with public misperceptions and myths. It is even difficult to access clinicians for assistance as most professionals are still not trained in adult ADHD, even though it can have such clinically significant impairments on daily functioning. Hopefully "evidence based research" will finally be acknowledged, and through appropriate education and awareness, clinicians and the community can start to support those adults who struggle all day, every day, to live with ADHD.

Voice 5

With ADHD, whether you are medicated or not, it seems to exacerbate your mood. You have high highs and low lows. Many of those who I know have ADHD also have depression. I have had three lots of depression, The last went for about 3 years.

In some ways my ADHD is an advantage. I work in retail, and it helps that I am sociable. I have done well. Sometimes I am too sociable. The kind of work that I do suits my personality. In some ways, I am a perfectionist.

At home, my room is always messy. Everyone who knows me knows that I spread my things everywhere. I broke up with my ex because he couldn't cope with my messiness. There are things that I keep putting off because I can't face them.

I don't blame myself for this. It's just the way that I am.

I've been lucky because I've always had really good people around me. My parents were really supportive. That was a huge thing. I had a really good childhood. My friends are people who are like me. If I forget something, they don't go off at me.

If I had not been medicated, I would never have finished school. I could never concentrate on my work. Before medication, I was always in trouble, e.g. saying stupid things without thinking. After medication, I was offered scholarships to various prestigious schools, and I got a score of 96 in Year 12.

One thing that really worries me is that ADHD seems to be such a dominant trait. My mother had two ADHD children; my brother and me. I really think that my brother was the worst child ever. He used to beat me up. He urinated on my books. He was just nasty. I would hate to end up like my mother, with a kid like that.

Voice 6

I am a male of 47 years. As long as I can remember I have suffered with anxiety and depression. I have spent most of my life in a living hell, not knowing what was wrong. Suicide attempts and failed relationships have made me a bitter person. I saw everything in a negative way and had very low self-esteem and self-worth. When agitated I would stutter and breathe heavily when I talked. I would easily become angry and frustrated, compulsed to say and do things I knew I shouldn't – and emotional to the point of tears. As an older adult I became paranoid, not wishing to leave the house. I found solace in heavy drinking and gambling.

Nothing I tried helped – not GA, not the numerous psychiatrists and psychologists I attended over the years, nor the various antidepressants I tried.

After a recent suicide attempt, I was finally diagnosed ADHD and prescribed stimulant medication. I became a totally different person. The anxiety and depression lifted. My drinking and gambling literally stopped overnight – I no longer had the urge or the reason.

Whilst I am undergoing strategies to manage my ADHD, and counselling to deal with the grieving for 25 lost years of my life; I am now re-building relationships with my ex-wife and my children. At last I can see sun and blue skies ... I am the happiest man in the world.

CHAPTER 13. ADHD IN THE WORKPLACE

The symptoms of ADHD can create challenges for the adult in the workplace, just as they do for the child in school. People with ADHD may have difficulties in the workplace with staying focused, attending to details, remaining organised, following through on instructions, forgetfulness, restlessness and scheduling and prioritising tasks (760). Inattention and distractibility can be the most prominent symptoms of ADHD in adults, with hyperactivity and impulsivity tending to decrease with age (161, 162).

In addition, adults with ADHD can display symptoms that are related to executive function deficits in attention, behavioural inhibition and verbal long-term memory (761). They may also have deficits in problem-solving ability and planning (762). All of these executive function problems can have a direct impact on skills and abilities that are important at work.

Adults with ADHD frequently report difficulties in social interactions (198, 222) and in the workplace this may affect relationships with employers or other members of staff. Negative preconceptions about ADHD can also be an issue. In one investigation of the social stigma associated with a diagnosis of ADHD in adulthood the researchers found "a fairly subtle, negative bias toward ADHD that contributes to rejection of individuals with the disorder, particularly in academic and work settings" (763).

Workplace safety is also an issue, with ADHD associated with increased risk of workplace accidents and injuries (764-766).

Research looking directly at how individuals with ADHD function in the workplace has been sparse. One study considering the relationship between ADHD and efficacy in working in teams (767) found that adults with ADHD had lower confidence about their ability to work in teams than their counterparts without ADHD (767). A simulated workplace has been used to compare the work performance of 18 adults with ADHD with 18 without ADHD (768). Most notably, those with ADHD had more difficulty in comprehension and speed while reading in this situation.

It is also clear that individuals with ADHD have poorer educational outcomes than their peers, which may in turn affect the types of work they undertake. Long-term follow-up studies have found that individuals with ADHD, compared to controls, completed less formal schooling (198, 769) (about 2 years less on average (769)) and did less well in class than their peers, with a lower grade point average and class ranking (198).

Similarly, a retrospective study (chart review) of college students found that those with ADHD tended to have significantly lower grade point averages and report more academic problems than students without the disorder (195). Finally, a survey of 500 adults with ADHD and 501 community controls found that those with ADHD were less likely to have graduated from high school (83% graduated compared to 93% of controls) or to have obtained a college degree (19% compared to 26%) (770).

13.1 Employment patterns

Adult ADHD is associated with greater levels of unemployment than control groups (196-198). It may also influence patterns of employment.

Individuals with ADHD prefer, and may be better suited to, certain professions, particularly those that permit self-employment or autonomy in work. Preliminary information from a survey of 1,463 US workers using the Adult-Self-Report Scale (ASRS) suggested that individuals with symptomatic ADHD were more likely to be

employed in non-office-based occupations such as tradespeople (plumbers, carpenters, electricians, gardeners, etc.) entertainers (comedians, actors, musicians) and elected officials. Office, bank and retail store clerks were least likely of all the surveyed occupations to show behaviour consistent with Adult ADHD (771). A long-term follow-up study of children with ADHD found that, at average age 24 years, those with ADHD had lower ranking occupational positions than controls, and were more often self-employed (769).

Recently, the prevalence of ADHD amongst workers was found to be 3.5% in a survey of 7,075 18–44-year-olds in paid or self-employment conducted across 10 countries from North and South America, Europe and the Middle East, as part of the WHO World Mental Health Survey Initiative (199). ADHD was less common among professionals (1.7%) than other workers: white collar technical (5.8%), service (2.9%), and blue collar (4%).

People with ADHD are more likely to have been through a rapid turnover of jobs, either as the result of dismissal or through a persistent need for change. A study of 30 adults with ADHD found them more likely than controls to have taken on many different types of occupations, rather than developing a career in a single occupation (772).

Similarly, a survey of 500 adults with ADHD and 501 community controls found that those with ADHD had more job changes over the preceding 10 years (mean of 5.4 jobs compared to 3.4 for controls) (770). Long-term follow-up of children diagnosed as hyperactive (198) found that, compared to controls, these adults had been fired from more jobs and showed greater levels of employer-rated symptoms of ADHD and ODD and lower job performance (198). A significant predictor of being fired was employer-rated ADHD and self-rated ODD in the workplace.

An Australian qualitative study of 10 adults with ADHD provides a more detailed picture (200). All participants recalled changing employment frequently, one reporting 20 different jobs, while another had more jobs than "he could remember". With poor school performance and/or leaving school early, many had started with low-skilled work. Boredom was a common reason for moving from job to job, and often the participants had moved to a new town with each new job, "in search of excitement". Other jobs were lost when the person was fired. Reasons for being fired included a lack of organisational skills, "rushing through the work and mucking it up", "over-focusing on unimportant aspects" of the job and not meeting deadlines, and "personality clashes".

13.2 Work performance

Adult ADHD is associated with significant deficits in work performance compared to control groups (196-198).

The WHO World Mental Health (WMH) Survey Initiative discussed above found that ADHD was associated with an average annual reduction in work quantity (21.7 excess days) and quality (13.6 excess days) compared to workers without ADHD (199). The researchers estimated 143.8 million lost days of productivity associated with ADHD each year in the surveyed countries. This study also considered treatment, and only a small minority of workers reported having been treated for ADHD (199). Only in the Netherlands (2.7%) and the USA (12.6%) did those with ADHD report receiving treatment specifically for ADHD, although treatment for other emotional problems did occur in other countries (199).

A US nationally representative household survey, with follow-up interviews (196), found that 4.2% of US workers had ADHD. ADHD was associated with an average of 35 days lost performance per worker with ADHD per year (a mix of absenteeism and low performance while at work). Figures varied across different employment groups, with 55.8 days for blue collar workers, 32.6 for service workers, 19.8 days

for technical workers, and 12.2 days for professionals. Notably, this study did not find a difference in the prevalence of ADHD between males and females. This was most likely due to a significant association between ADHD and unemployment in men but not women (196).

A survey in a large manufacturing firm found an ADHD prevalence of 1.9% (766). ADHD was associated with a 4–5% reduction in work performance, with twice as many sickness absence and workplace accidents and injuries in the ADHD group. Productivity losses were attributed to low on-the-job performance rather than sickness absence.

Analysis of medical claims databases for a large national US manufacturing company found a higher rate of work loss (773) and accidents (764) in individuals with ADHD.

Barkley et al's (198) long-term follow-up of children diagnosed as hyperactive demonstrated a link between work performance and severity of ADHD, with employer-rated ADHD and parent-rated severity of childhood hyperactivity predicting lower work performance ratings from employers (even after controlling for workplace ODD, childhood conduct problems, lifetime CD, and IQ) (198).

The study also considered the impact of ADHD on personal financial issues (198). The hyperactive group did not differ from controls in their annual income or total savings; however, they owed twice as much to other people and reported having more difficulty saving to pay their bills.

13.3 Workplace solutions

13.3.1 Interventions and strategies

To date no research has been conducted to assess directly the impact of the treatment of ADHD on absences, work performance or workplace injuries. There is, however, evidence to suggest that ADHD medication improves executive functioning in adults with ADHD (774-776).

There is also sparse research on workplace-based interventions that may assist individuals with ADHD. It has been suggested that mentoring early in a person's career may help the individual with ADHD fare better in the workplace (777).

The Australian Government JobAccess website offers workplace adjustments and solutions for individuals with ADHD that may be useful to both the person with ADHD and their employer. The website provides solutions and adjustments for the following job requirements:

- attention and memory
- active listening
- focusing attention
- learning and applying knowledge
- managing emotions at work
- organisation and planning
- problem solving and decision making.

Available at: http://www.jobaccess.gov.au/JOAC/Advice/Disability/Attention_ Deficit_Hyperac.htm

The high prevalence of ADHD in the workplace and the associated losses in work performance have led to suggestions that ADHD could be targeted for workplace screening and treatment programs (199). The cost-effectiveness of such programs

has not been determined and caution is warranted as workers with ADHD should not be stigmatised.

13.3.2 Vocational assessment

Recommendation

183. Career and workplace assessments are often valuable in assisting adults with ADHD in their selection of career or workplace function.

 ✓ Recommended best practice based on clinical experience and expert opinion

It is important that clinicians and career consultants have knowledge and training in how to help adults with ADHD select careers or jobs (760).

Career choice needs to match the person's strengths, weaknesses, ADHD tendencies and co-existing conditions. In particular, people with ADHD will have different degrees of symptom severity and varying functional limitations.

Career and workplace assessments may be of benefit to assist people with ADHD to determine the types of employment to which they are best suited. In adults with ADHD, neuropsychological testing may be helpful to generate a profile of the person's particular impairments/deficits and strengths, to inform career choice.

Vocational assessment should be individualised and may include one or more of the following (760):

- clinical interview to take a work history and assess current workplace functioning
- neurocognitive testing to measure strengths and weaknesses
- psychological testing to evaluate comorbid psychiatric conditions
- personality testing to assess temperament and values related to career
- interest testing to assess appropriateness of fit between career choice and interests.

A key concern is workplace safety, given the link between ADHD and increased risk of workplace accidents and injuries (764). In a recent study of male college students, those with ADHD reported placing less emphasis on job safety, compared to their peers, when considering their professional direction (778). This suggests that it might be of benefit to individuals with ADHD if vocational counsellors broached the topic of work safety when discussing vocational choice (778).

In Australia, there are several employment agencies that provide specialist assistance to individuals with ADHD. ADHD support groups from each State and Territory can provide details (see Appendix I, for contact details).

13.4 Workplace discrimination

In Australia, the Disability Discrimination Act 1992 (Cth) covers:

- recruitment
- terms and conditions of employment
- opportunities for promotion, transfer or training
- other benefits associated with employment
- termination of employment.

See: http://www.humanrights.gov.au/index.htm

CHAPTER 14. ADHD AND DRIVING

14.1 Obtaining a driver's or pilot's licence

Having ADHD is not a contraindication to holding a driver's licence in Australia. Drivers' licences are not required to note a diagnosis of ADHD, nor that a person is receiving stimulant therapy (or any other legally prescribed psychoactive therapy) (779).

National medical assessment guidelines warn that stimulants taken in high doses may increase aggressiveness and risk taking, and people on such medications should be cautioned accordingly (779). However, stimulants in the doses prescribed for ADHD do not pose a problem.

People with a diagnosis of ADHD who wish to gain a pilot's licence are required to have a thorough assessment by a psychiatrist. Applicants will not be eligible for a pilot's licence if they are found to have persistent cognitive deficits or any behavioural aberrancy, or to require regular stimulants for their ADHD (780).

14.2 ADHD and driving ability, accident rates and traffic violations

Studies considering the impact of ADHD on driving performance demonstrate that individuals with ADHD (compared to individuals without ADHD) are at greater risk of adverse driving outcomes such as greater numbers of traffic infringements, more frequent and more severe crashes, and an increased likelihood of having a licence suspended or revoked (for review see (17, 201)). This increased risk has been found in both longitudinal studies of adolescents and young adults diagnosed with ADHD in childhood and studies of clinically referred adults with current symptoms of ADHD.

Inattention, risk taking and distractibility, particularly in-vehicle distraction, are frequent contributors to traffic accidents. Accordingly, executive functioning deficits that contribute to inattention and impulsivity have been suggested to contribute to the poor driving outcomes in individuals with ADHD (17).

14.3 Effect of treatment on driving ability in people with ADHD

Research question

• For individuals with ADHD, does the use of interventions (pharmacological, psychosocial or other), compared with no intervention, improve driving performance?

Evidence statements

- **MPH:** Two systematic reviews (five small studies) reported that treatment with MPH, compared to placebo, in adolescents and adults significantly improved driving performance in simulated driving exercises (17, 201). (Level III-2)
- **ATX:** One study found that treatment with ATX, compared to placebo, in adults improved self ratings of driving performance but not observer rating or test score in simulated driving exercises (781). (Level II)
- **Non-medication intervention:** One small study found that use of a manual transmission compared to an automatic transmission in adolescents improved driving performance in simulated driving exercises (782). (Level II)

Recommendations

- 184. People with ADHD should be strongly advised to take their stimulant medication if they are driving. Treatment with MPH improves driving performance for people with ADHD. The use of extended-release stimulants is preferable if the person drives at night. (Grade C)
- 185. There are insufficient research data to recommend the use of ATX for the improvement of driving performance in individuals with ADHD. (Grade D)
- 186. The use of a manual transmission over an automatic transmission should be considered for individuals with ADHD. (Grade C)

14.3.1 Summary of research evidence

Two systematic reviews addressed the impact of MPH on driving performance in adolescents and adults with ADHD (17, 201).

Five small RCTs were considered. Four of the five used a virtual reality driving simulator and one used on-road testing. MPH was found to improve driver performance on driving simulator tests in all studies. Further studies are needed to support these outcomes, especially with larger sample sizes.

One of the studies on MPH included in the systematic reviews found that driving performance worsened later in the day when MPH-IR was used (783). This contrasted with MPH-ER, where performance was improved at all driving test time points throughout the day (783).

One small crossover trial compared treatment with ATX to placebo in adults and reported improved self ratings of driving performance but not observer rating or test score in simulated driving exercises (781).

One study was identified that considered the use of a manual transmission compared to an automatic transmission in adolescents with ADHD. Improved driving performance was seen when using a manual transmission in simulated driving exercises (782).

14.4 Legal issues regarding driving and medications

In most States and Territories in Australia, roadside drug testing has been, or soon will be, introduced. These roadside screening tests are limited to detecting three substances in saliva: tetrahydrocannabinol (TCH), methamphetamine (speed) and MDMA (ecstasy). The potential problem of identifying psychoactive substances that have been prescribed for therapeutic use (including stimulants) has been recognised, and testing will not detect the presence of prescription medications for ADHD.



CHAPTER 15. ISSUES FOR FAMILIES, PARENTS AND CARERS

When a child, adolescent or adult has ADHD, the impact of the disorder may be felt by every member of the family. In addition, the high heritability of ADHD means that a number of family members may have the disorder. Many studies have investigated the psychosocial impact on families of children and adolescents with ADHD. ADHD in children may have a substantial impact on parents/caregivers and siblings, in addition to its impact on the child with the disorder. Children with ADHD experience significant difficulty in their relationships with peers (186), so it is not surprising that these children will also have problems in relationships with family members. The impact of adult ADHD on family function and parenting has also been investigated, and here too the effects tend to be negative, with impacts on relationships with spouses/partners and children. It is also important to consider the special needs of children and adolescents with ADHD in out-of-home care.

a. Families with a child or adolescent with ADHD

Recommendations

Clinicians should be alert to the risk of depression in parents/caregivers of 187. children or adolescents with ADHD. Parents/caregivers may need referral for support and treatment.

✓ Recommended best practice based on clinical experience and expert opinion

188. When a child or adolescent is diagnosed with ADHD, the clinician should consider the impact of marital discord on the child or adolescent over and above the effects of the ADHD, as ADHD may exacerbate underlying family tensions.

Recommended best practice based on clinical experience and expert opinion

189. Consideration should be given to the siblings of children with ADHD. Siblings may need additional support at school if there is an expectation that they care for their sibling with ADHD or if they become the focus of peer ridicule because of the behaviour of their sibling with ADHD. ✓ Recommended best practice based on clinical experience and expert opinion

15.1.1 Family functioning

Much of the research on families with a child or adolescent with ADHD has addressed the wellbeing of individual family members and specific interactions within the family, such as mother-child interactions, and the impact on the parents/caregivers. Fewer studies have investigated the overall functioning of families with a child with ADHD and the impact of the disorder on the entire family system.

Overall, the research in this field consistently shows that families with a child or adolescent with ADHD experience considerable discord and greater deficits in family functioning than controls.

One of the few studies to look at family functioning in a community sample found higher levels of family dysfunction in families with children with ADHD compared to families with children without the disorder (784). An increase in the severity of ADHD symptoms was directly associated with higher levels of family dysfunction, even after controlling for the presence of comorbid disorders (784).

There are similar findings from clinic-referred samples of children and adolescents with ADHD (175, 785-787). For example, families with a child with ADHD were found to have less adaptive coping styles, more problematic relationships and increased levels of family dysfunction compared to controls (175). Other studies have suggested that families with a child with ADHD have lower family coherence

(i.e. poor co-operation) and experience greater conflict than controls (788, 789). These families had specific deficits in problem solving, communication, affective responsiveness and involvement (788).

Studies finding an increased rate of family dysfunction have primarily been conducted with male children with ADHD. One study that examined reports of family functioning in families with girls with ADHD found similar patterns to those for families of boys with ADHD (785). Compared to controls, the girls with ADHD were more likely to report greater disruption in their family environment (785).

Studies of family functioning have largely compared families with a child with ADHD to families with non-affected children. Little is known about the relative effects on family functioning of having a child with ADHD compared to a child with another mental health difficulty. In the first study of its type, researchers explored family functioning in families with a child with ADHD, families with a child with emotional disorders (mood and anxiety) and control families (790). No significant differences were found between families with a child with ADHD compared to families with a child with emotional disorders; however, both of these groups reported significant deficits in family functioning compared to control families (790). These families reported higher levels of stress and lower levels of social support and quality of life compared to families with non-affected children (790). Families with a child with ADHD are also at greater risk of family adversity (396, 784, 791, 792). One study found significantly more family adversity in families with a child with ADHD compared to controls, and significantly higher adversity in children with the combined subtype compared to the inattentive subtype, even after controlling for comorbid ODD and CD (396). Family adversity was associated with socio-economic status, parental psychopathology (maternal and paternal), marital conflict and stressful events (396). These findings are particularly important as family adversity has been shown to predict ADHD diagnosis and persistence in a longitudinal sample of children at a 4-year follow-up (793).

Financial impact on families

The financial cost of raising a child or adolescent with ADHD is an important issue for many families. It has been consistently documented that there are higher medical costs for children with ADHD than children without the disorder (794-798). Estimates of exact costs vary considerably across studies depending on the types of services included and whether services related to the treatment of comorbid disorders are included (798). A recent review of studies of the economic impact of ADHD in children and adolescents estimated that the average annual cost of ADHD treatment and medical services was approximately \$2,636 (798). When costs related to parental work loss, juvenile delinquency and education were considered, the average annual cost rose to \$14,576 (798).

In addition to the costs directly related to the child with ADHD, family members also have significantly higher costs associated with medical care and work loss, compared to family members with children without the disorder (\$2,728 versus \$1,440) (773). It is unclear whether these large differences in cost are due to the increased pressures of having a child with ADHD in the family, or to the familial link between ADHD and other disorders.

The financial cost of having a child or adolescent with ADHD has yet to be explored in an Australian sample. An important Australian study of the use of health- and school-based services in a national sample of children and adolescents with ADHD found that 28.1% of the children with ADHD had attended a health- or schoolbased service in the previous 6 months (234). They were also more likely to have attended a health-based service (22.6%) than a school-based service (17.1%), although this difference was non-significant (234). They frequently attended paediatricians and family doctors, and counselling was the service most often used by children with ADHD in schools and in health services. Children with the combined subtype of ADHD were significantly more likely to attend services than children with the predominantly inattention or predominantly hyperactive/impulsive subtype of ADHD (234). This is consistent with other research that suggests children with ADHD combined type experience greater difficulties than children with the other subtypes, and commonly present to clinic-based services.

15.1.2 Parent-child interactions

Many studies have explored the patterns of interaction between parents and children with ADHD. These studies have consistently identified greater adversity and more difficulties in the style of interaction than experienced by parents and non-affected children (e.g. (175, 799-801)). Most of this research has focused on mother-child interactions, but a growing number of studies have considered the important role of fathers (800, 802-804).

An early study (805) observed the interactions of unaffected and hyperactive children with their mothers in both structured tasks and free play. Mothers of hyperactive children spent significantly more time attempting to direct, control and structure their child's play in both settings. Other studies also suggest that mothers of children with ADHD are more directive (804, 806), more negative (175, 807) and less responsive to child-initiated interactions compared with mothers from control samples (806, 808).

While similar patterns have been observed for fathers of children with ADHD (804), several studies have reported minor differences. Fathers were found to make fewer demands than mothers in dyadic interactions and increase their demands during interactions with the mother and child (802). Mothers of children with ADHD see themselves as more likely than fathers to give in to child misbehaviour, with less efficient coping methods to anticipate and avoid problems during interactions with their child (808).

Several studies have explored the behaviour of the child with ADHD in mother-child interactions during free play and structured tasks. Compared to controls, children with ADHD were found to be more negative (799, 804, 809), less compliant (175, 802, 810) and more competitive (799, 804). Research has also found increased conflict in the interactions between mothers and children with ADHD (787, 799), and one study found a link between this conflict and the number of ADHD symptoms in the child (787).

Impact of comorbidity

It is unclear whether the difficulties in parent-child interactions can be attributed to ADHD symptomatology alone. A number of studies have considered the potential role of comorbid externalising disorders such as ODD or CD. When compared to non-affected controls, parent-child interactions in ADHD-only groups, comorbid ADHD-ODD groups (803) and comorbid ADHD-CD groups (811) are more negative and have more non-compliant child behaviours. Reports from studies addressing whether these difficulties are related to ADHD, comorbid externalising disorders or a combination of the two are conflicting.

Some studies have suggested that difficulties in parent–child interactions are more commonly associated with comorbid ADHD-ODD and ADHD-CD than with ADHD alone (803, 811).

In contrast, one recent study found that while there was significantly less warmth, engagement and communication in interactions between parents and children with ADHD during problem-solving tasks, the presence or absence of a comorbid externalising disorder had no significant impact on these interactions (800).

There is also evidence to suggest that externalising and internalising disorders in combination may lead to additional difficulties in parent-child interactions. In one

study, mothers of children with comorbid externalising/internalising problems displayed less warmth and were more hostile and controlling than mothers of children with pure externalising difficulties (812). In addition, children with comorbid problems were more hostile and non-compliant during interactions with their mothers (812).

Other research suggests that symptoms of ADHD may be less important than those of related externalising disorders in certain domains of interaction. One recent study investigated the interactions between mothers and children with ADHD, ADHD plus ODD, and controls without ADHD or ODD (813). Mothers of children with ADHD plus ODD were less responsive and more over-reactive than mothers of control children without ADHD or ODD (813). However, mothers of children with ADHD only were not significantly more responsive than mothers of children with ADHD plus ODD nor significantly less responsive than mothers of control children. These findings suggest that ODD symptoms may be a greater predictor of maternal responsiveness and over-reactivity than ADHD symptoms. Similarly, maternal responsiveness in parent-child interactions has been reported to be the result of the child's CD symptomatology rather than symptoms of ADHD (814).

The heterogeneous outcomes observed in studies addressing whether difficulties in parent-child interaction are related to ADHD, externalising disorders or a combination of the two may be explained by factors such as symptom severity and differences in study design. Several studies suggest that problems in parent-child interactions may in fact be due to increased symptomatology rather than categorical diagnoses (787, 800). It has also been suggested that the quality of parent-child interactions is influenced by the type of interaction being observed. When compared with problem-solving tasks, sessions of game playing have been associated with more positive interactions between parents and children, irrespective of whether the child had ADHD only or ADHD plus a comorbid externalising disorder (307).

15.1.3 Impact on parents/caregivers

The impact of caring for a child with ADHD on the mental health and wellbeing of parents/caregivers is well documented. It has repeatedly been reported that the often adverse behaviours demonstrated by children with ADHD result in increased levels of stress, low parenting self-efficacy, low self-esteem, marital discord and depression in parents. The majority of studies have focused on the impact on mothers, but the few studies examining the impact of ADHD on fathers are just as important.

Parental stress

Parents of children diagnosed with ADHD frequently experience elevated stress in their parenting role (807, 815-819). It has been suggested that the high levels of stress are due, at least in part, to the increased care-taking demands on these parents (805).

Parents often see the behaviours associated with ADHD (hyperactivity/impulsivity, difficulty in sustaining attention) as both disruptive and annoying, and this may add to their stress levels. A study of 40 mothers of hyperactive children found that their levels of stress were significantly higher than those of mothers of non-affected children (820). Child characteristics such as distractibility and degree of bother accounted for over 74% of the variance between the groups (820).

In a study looking at both maternal and paternal parenting stress, 20 sets of parents and children with ADHD were compared (817). Mothers identified their child with ADHD as more stressful than did fathers; however, although this difference was significant, it was minimal and accounted for only a small proportion of the variance (817). The researcher concluded that fathers and mothers of children with

ADHD experienced comparable levels of parenting stress, but mothers were more likely to perceive child characteristics as more stressful (817). The study had no control group, so it is unclear whether this self-reported stress would differ significantly from that experienced by parents with non-affected children.

Both maternal psychopathology and child oppositional-defiant behaviour have been shown to be significant predictors of parenting stress for parents of children with ADHD (815). Other factors, such as ADHD severity, child health status and maternal health status were found to be less important (815). Similarly, parent stress has been significantly associated with maternal depression and severity of child psychopathology (aggression, conduct problems, hyperactivity) (818). The level of parenting stress has also been shown to be associated with the severity of the child's behavioural disturbance, and the level of control that parents perceived themselves as having over the child's behaviour (821).

Parenting self-efficacy and self-esteem

Parenting self-efficacy has been defined as "an estimation of the degree to which parents perceive themselves as capable of the varied tasks associated with this highly demanding role" (822). Parents who have high parenting self-efficacy are better equipped to deal with the behaviours often associated with children diagnosed with ADHD (823). A study involving 100 mothers and 57 fathers found that parents of children with ADHD reported low parenting efficacy, which was further associated with poorer treatment outcomes for children with ADHD (823). Mothers reported lower parenting efficacy than fathers (823).

The same study also found low levels of self-esteem in both mothers and fathers of children with ADHD, with maternal self-esteem being lower than paternal self-esteem (823). Parents with low self-esteem were inclined to reduce their effective participation in treatments designed to assist their children (823).

An earlier study found that parents of hyperactive children saw themselves as significantly less knowledgeable and skilled than parents of children unaffected by hyperactivity. They also reported that they gained less sense of value and comfort from their parenting roles than the parents of unaffected children (820).

Parental depression

Several studies have identified a possible genetic link between ADHD and major depressive disorders through examining children with ADHD and their first-degree relatives (824, 825). In a study of 73 ADHD probands and 264 of their first-degree relatives, and a control sample of 26 probands and 92 relatives, 33% of the ADHD probands had a major affective disorder (bipolar disorder, major depressive disorder, or dysthymia), compared to 4% of controls. Major depression was the most prominent affective disorder, with 21% of ADHD probands meeting diagnostic criteria. In addition, the first-degree relatives of probands with ADHD were at greater risk of having any affective disorder than the relatives of the control group (825). Similar results were obtained in a subsequent follow-up study (824).

Marital discord

Within the literature investigating the marital relationships of parents of children with ADHD, the majority of studies suggest an association between ADHD and marital discord. In these studies, parents of children with ADHD report less marital satisfaction and increased levels of conflict, compared to parents of non-affected children (799, 801, 816, 826, 827). However, there are exceptions and some studies have found no significant difference in marital discord between parents of children with ADHD and parents of controls (807, 828, 829). Rates of divorce and separation are also not consistently higher in families of children with ADHD than in families with unaffected children (786, 807, 829).
15.1.4 Siblings of children with ADHD

Research on the siblings of children with ADHD and their relationship with their brother or sister with ADHD is limited, although interest in this area is expanding. Increasingly, texts are highlighting the importance of considering the potential impact of ADHD on non-affected siblings (830, 831), and several studies have investigated the neuropsychological and psychosocial consequences of having a sibling with ADHD. Research in this area is particularly important as the sibling relationship is potentially the longest lasting of all relationships (832) and is known to facilitate childhood social development (833).

Sibling characteristics

Several early studies examined the intellectual functioning and the prevalence of developmental disorders in biological siblings of children with ADHD. One study found that while non-affected siblings of children with ADHD had greater intellectual impairment and school failure than a control sample, these differences were not significant (183). Both ADHD and learning disorders were also found to be more prevalent in siblings of children with ADHD, compared to siblings of children with Down syndrome (834) and normal controls (824, 835, 836).

The search for endophenotypes to determine specific genetic links that contribute to ADHD has led many researchers to investigate the neuropsychological functioning of children with ADHD and their non-affected siblings. As siblings share much of their genetic make-up, it is thought that siblings of a person with ADHD may share some of the genes that predispose individuals to ADHD, though they do not develop the disorder. One study investigated measures of attention, memory and executive functioning in siblings (with and without ADHD) of children with ADHD and siblings of children without ADHD (837). Results showed a similar pattern of neuropsychological functioning in unaffected siblings of children with ADHD and the siblings of control children, suggesting that some neuropsychological deficits in siblings are primarily associated with ADHD status (837). The exception was verbal learning, where siblings of children with ADHD were significantly more impaired than those in the control group (837).

Studies have also found similar patterns of impairment on measures of response inhibition for ADHD probands and non-affected siblings (838, 839). Most recently, a study of neuropsychological deficits associated with ADHD looked at a sample of non-identical twins of whom one had ADHD, and control twins with no ADHD (840). Those without ADHD themselves but with a twin with ADHD showed impairment on many of the same measures as their co-twin with ADHD (e.g. measures of executive function, processing speed and response variability) (840). As twins with ADHD and their non-affected co-twins were both significantly more impaired than controls, these findings suggest that measures of executive functioning, processing speed and response variability may be significant endophenotypes in characterising ADHD.

Differences between mother-child interaction with a child with ADHD and with a non-affected sibling have also been investigated. In one such study, boys with hyperactivity showed more conflicting behaviours associated with parental requests compared to their non-affected siblings, and had more severe behaviour management problems in both free play and structured settings (810).

Sibling relationships and the impact of having a brother or sister with ADHD

A recent study investigated sibling relationships among children with ADHD and children in a control sample (841). The study compared reports from three informants (mother, child with ADHD, and sibling), and considered the potential influence of comorbid internalising and externalising problems (841). Findings

suggest greater conflict in sibling relationships of children with ADHD compared to controls, with mothers and children reporting similar amounts of conflict in the sibling relationship. While no significant differences were found between the groups in warmth/closeness, mothers reported more warmth/closeness in the sibling relationship than children with ADHD. Comorbid internalising and externalising problems both predicted less warmth/closeness, and externalising problems were associated with higher levels of conflict in the sibling relationship (841). Comorbid internalising and externalising behaviours accounted for many of the difficulties in the sibling relationships, highlighting the importance of considering the potential influence of comorbidity.

The severity of behavioural symptoms in children with ADHD was associated with the level of conflict in their relationship with siblings (787). Furthermore, younger siblings developed significant active coping strategies due to their frequent responsibilities in caretaking for their sibling with ADHD. In another study, mothers of children with ADHD reported less warmth/closeness, fewer intimate relationships and more arguments within the sibling relationship, compared to the reports of mothers of non-affected siblings (842). Compared to controls, siblings of children with ADHD in this study were seen as having increased feelings of resentment towards their ADHD brother or sister (842).

These studies have built on Kendall's qualitative research (843), which was the first published study to investigate the experiences of siblings of a child (in this study, a brother) with ADHD and to examine the impact of the disorder on their lifestyle and relationship with their sibling. The study found disruption to be the major complaint of non-ADHD siblings, due to the symptoms and externalising behaviours of the child with ADHD (843). Siblings experienced disruption in three ways: victimisation, caretaking, and sorrow/loss. They reported feeling victimised by their sibling with ADHD through "... overt acts of physical violence, verbal aggression, and manipulation and control" (843). Siblings were expected by parents to look after and to protect their brother with ADHD, and because of this the majority viewed negatively the responsibility of caring for their brother with ADHD. Non-ADHD siblings reported anxiousness, worry and sorrow due to the absence of a normal family environment (843).

Kendall et al (843) also reported that parents often overlooked or minimised the disruption felt by siblings of a child with ADHD. Non-ADHD siblings described not obtaining help from parents when they were experiencing difficulty, and often being ignored and distrusted. They managed disruption in one of three ways: retaliatory aggression, accommodation, or avoidance (843).

A study of internalising symptoms in twins and non-twin siblings of children with ADHD found that, overall, twins with ADHD were more impaired than non-twin siblings (844). Non-ADHD twins who had a twin with ADHD combined subtype showed increased symptoms of both generalised anxiety and separation anxiety compared to controls (844). Generalised anxiety was also high for non-ADHD twins whose twin had ADHD-predominantly inattentive type.

In a recent qualitative study (830) siblings reported that the child with ADHD was responsible for creating considerable discord within the family, with disruption often created by extreme moodiness, verbal abuse, intimidation and physical violence. Siblings reported multiple episodes of disruption that would escalate and cause significant discord in family relationships (830).

Annoyance and resentment were the feelings most commonly reported by siblings when speaking of their interactions and relationship with their brother or sister with ADHD (830). There are a number of reasons for these feelings: the child with ADHD exerting control over the family, parents giving in to the child, parents being lenient towards the child, perceived violations of trust, and the child with ADHD embarrassing the sibling in front of his or her peers (830). Siblings reported various

coping strategies to manage their feelings and reduce tension and discord: talking to their mothers if they wanted information about ADHD or advice on coping with the disruptive behaviour; reflecting on the experiences of other family members; avoiding situations where conflict was likely; trying to persuade their sibling with ADHD to comply with parental requests; and feigning compliance with parental demands (830).

Overall, findings from both quantitative and qualitative studies are consistent and suggest that siblings of children with ADHD experience more difficulties than controls and have more problems in their sibling relationships. The consistency of these findings is notable, as some studies relied on reports from the mother and some from the child, and consistent reporting by children and parents is unusual (845). Nevertheless, more research is needed to obtain a complete picture of the experience of siblings of children with ADHD. For example, only one study to date (841) has considered whether results may be attributable to comorbid disorders, and differences between ADHD subtypes are yet to be explored.

15.2 Adult ADHD and its impact on the family

Recommendations

190. Treatment planning for adults with ADHD needs to include strategies for daily living and additional support during life transitions such as changes in career or family situation.

✓ Recommended best practice based on clinical experience and expert opinion

191. Treatment planning for children with ADHD needs to take into account whether the parent/caregiver has symptoms of ADHD, as parents/caregivers with ADHD symptoms may need additional support to implement the program with their child successfully.

 \checkmark Recommended best practice based on clinical experience and expert opinion

192. When an adult is diagnosed with ADHD, the clinician should consider the impact on their partner and family as additional support may be needed.
 ✓ Recommended best practice based on clinical experience and expert opinion

15.2.1 Adult ADHD and relationships

The difficulties in peer relationship and family functioning that often affect children and adolescents with ADHD frequently extend into adulthood and may continue regardless of age or role (parent or child).

In a longitudinal study that followed individuals with ADHD over 13 years, young adults (aged 19–25 years) reported having fewer friends and more difficulty in keeping friends than controls. Their current dating relationships were of lower quality, and they were more likely than controls to report arguments with friends (222).

Adults with ADHD also report higher rates of separation and divorce (154, 223). In an exploration of marital and family functioning (224), adults with ADHD reported poorer marital adjustment and more family dysfunction than controls. There were no significant differences for their spouses, compared to controls, on overall marital adjustment and family functioning, but marital adjustment scores were more impaired for spouses of people with ADHD. Compared to their spouses, adults with ADHD had a more negative view of the health of their marriages and families.

15.2.2 Adult ADHD and parenting

Parenting by individuals with ADHD is a relatively new area of research, and one that is important given the high heritability of ADHD.

Parents with ADHD may face problems in parenting effectively due to their own difficulties paying attention, being organised and/or managing their own impulsive and distractible behaviour.

A review of the clinical implications of ADHD in parents of a child with ADHD has drawn attention to some of the possible negative and positive attributes of parents with ADHD (846). Adults with ADHD may have difficulty maintaining attention while supervising children and difficulty in concentrating on their child's actions and behaviour. They may also find it difficult to defuse a volatile situation and calm the child down, as they may over-react and escalate the situation (846).

One study of parenting a child with ADHD found that mothers with ADHD themselves monitored their child's behaviour significantly less than mothers without the disorder (847). They were also less efficient in disciplining their child and more impaired in identifying strategies for effective child rearing (847).

Parents with ADHD may also have limited confidence in their parenting abilities. Mothers with high levels of ADHD report lower parenting self-esteem, a more external parenting locus of control and less effective disciplinary styles, compared to mothers with fewer ADHD symptoms (848).

Harvey et al (849) investigated the association between parental ADHD symptomatology and parent-child behaviour among 46 mothers and 26 fathers of children with ADHD. Mothers and fathers with symptoms of inattention, and fathers with symptoms of impulsivity, were more likely to report lax parenting, both before and after parenting programs, and fathers with symptoms of impulsivity were also more likely to report being over-reactive with their child (849). Mothers who reported the most inattention were most negative in their interactions with their child.

Research has also demonstrated how having ADHD may help a parent to understand their child with ADHD. A study that explored whether the similarity between mothers and children with ADHD would help or hinder parenting found that maternal ADHD symptoms appeared to ameliorate the negative effects of child ADHD symptoms on parenting (850). In addition, when mothers and children both had high levels of ADHD symptoms, the mother's response to the child was significantly more positive and affectionate (850).

One critical aspect of parenting children with ADHD is the role of parents in the treatment of the child's ADHD. For parents with ADHD themselves this may pose several difficulties. Problems with following instructions, listening and completing tasks may affect the ability of parents with ADHD to adhere to their child's treatment program, or to participate in parenting programs (846). In addition, a study of parenting programs for preschoolers with ADHD found that maternal ADHD symptoms limited the improvement shown by children with ADHD (851). This study highlights the importance of treatment for parents with ADHD who are participating in treatments for their children with ADHD.

At the same time, parents with ADHD may act as a positive role model for a child with ADHD, showing insight into ADHD symptoms, being available to provide information, modelling coping strategies and being involved in a treatment program themselves (846).

15.3 Children and adolescents with ADHD in out-of-home care

Recommendations				
193.	All children and adolescents in out-of-home care should receive a medical assessment that includes a developmental and mental health assessment. Recommended best practice based on clinical experience and expert opinion 			
194.	Special care needs to be taken in establishing or refuting a diagnosis of ADHD in children in out-of-home care, because of the high risks of misdiagnosis and the equally high risks of doing harm by ceasing established therapy for children already at high risk of long-term disadvantage and disability. Recommended best practice based on clinical experience and expert opinion 			
195.	When a child with a diagnosis of ADHD is placed in out-of-home care, every effort should be made to maintain continuity of treatment and to ensure that foster carers are educated about the special needs of the child. Recommended best practice based on clinical experience and expert opinion 			
196.	The professionals in Australia's welfare system who are responsible for the care of children and adolescents should be educated about ADHD. Recommended best practice based on clinical experience and expert opinion 			
197.	Research on the management and care of children and adolescents with ADHD in out-of-home care is needed.			
Children and adolescents in out-of-home care are a small but significant sub- population in Australia. Children and adolescents can be placed in out-of-home care in response to a court order or voluntarily. The number of persons aged below 17 years living in out-of-home care in Australia at 30 lune 2007 was 28 441 including				

7,892 Indigenous children and adolescents (852). The total number of children and adolescents in out-of-home care has risen dramatically in the last decade, increasing 102% from 1997, when the number was 14,078.

The majority of children and adolescents in out-of-home care at 30 June 2007 were in home-based care (95%), in either foster care (50%), relative/kinship care (44%), or in some other type of home-based care (1%). Four percent of children in out-of-home care were living in residential care (852).

Children in out-of-home care are a vulnerable, at-risk group with special needs (853). A high proportion of children and adolescents in out-of-home care are shown to have mental health problems. International studies report levels of mental health problems (including inattention and hyperactivity) among children and adolescents in welfare systems that are much higher than those reported in the general population (854-857).

A high prevalence of ODD and ADHD has also been found in adoptive populations in the US. A study of 808 adopted youth (4–18 years) found that residing in foster homes prior to being adopted was significantly more prevalent for children with ADHD and ODD (858). Further, residing in multiple foster homes increased the risk of exhibiting ADHD/ODD and ODD symptomatology, but not solely ADHD behaviours (858).

Australian studies of children and adolescents in out-of-home care also report a high prevalence of mental health problems that include ADHD symptoms such as inattention and hyperactivity (579, 859, 860).

15.3.1 Services for children and adolescents with ADHD in out-of-home care

All children and adolescents in out-of-home care need medical assessment, including mental health assessment, development of a healthcare plan and a

transferable health record, to underpin continuity of care (853). This is particularly important given the high levels of mental health problems in this population.

Children and adolescents in out-of-home care pose particular problems in assessment and management because of the issues of differential diagnosis of comorbidity and of alternative causes of symptoms. In addition, there is inherent bias in some systems against a diagnosis of ADHD and it is important that the professionals in Australia's welfare system who are responsible for the care of children and adolescents are educated about ADHD. As Australia relies heavily on volunteer caregivers to provide homes for children and adolescents in the welfare system, researchers also support the need for high-quality professional support for volunteer caregivers (579).

When families with a child with suspected ADHD come to the attention of child welfare services, it is critical that a diagnosis is sought through a specialist clinician. As discussed in section 15.2 Adult ADHD and its impact on the family, page 190, parents of a child with ADHD may have symptoms of ADHD themselves and may need support and appropriate referral.

The treatment and care needs of children and adolescents with ADHD in out-ofhome care are likely to differ from those of the general community. For example, the needs of this group are complicated by the issues associated with adjustment to out-of-home care and the possible adversities such as abuse, neglect, family dysfunction and parent psychopathology that have resulted in their placement (861).

Research data on the treatment and care of children and adolescents with ADHD and other psychiatric disorders in out-of-home care are sparse. Studies in the US have focused on medication or service use for children with ADHD in foster care, but are yet to address treatment outcomes (862, 863).

One study of 326 children and adolescents in home-based foster care in Adelaide between August 2004 and January 2006 found a high prevalence of mental health problems including inattention (579). Only half of those living in home-based foster care who were identified by caregivers as needing professional help had received it during the previous 6 months (579). Notably, there was also a high proportion of children and adolescents in the community control group who had not accessed professional care. These findings suggest that barriers to accessing care were generalised and not specific to individuals in foster care.

More research is required to determine the incidence of ADHD amongst children and adolescents in out-of-home care and to examine strategies for their management and care that can be used to inform policy and practice decisions.

15.4 Issues for families living with ADHD that have not been addressed in these Guidelines

There are a number of important issues for Australian families living with ADHD that cannot be adequately addressed in these Guidelines because of the reliance upon scientific evidence in the process of guideline development. These issues which concern consumers and their supporters warrant acknowledgement by clinical and health service planners. They include the frustration, false starts, and wasted time and money that is often involved in finding appropriate help, because of the widespread scepticism and lack of understanding among clinicians about ADHD and its management. The problem often resurfaces in the transition period between, for example, seeing a paediatrician and moving to the care of a psychiatrist as the child grows older. It frequently happens that a diagnosis made by one clinician is disputed by another, or by school staff or a social worker. Sceptics within the professions sometimes combine with promoters of non-scientific treatments and people associated with socio-religious cults to foster disrespect for the science behind the understanding of ADHD. Satirising of ADHD and sensationalised media reporting create a particularly onerous stigma that greatly exacerbates the burden of the condition. Parents in higher income brackets may be accused of seeking ADHD medication to give their child an unfair advantage at school, while those in lower income brackets may be suspected of inventing or causing their child's problems by abuse or neglect. Such beliefs tacitly underlie some treatment approaches, and only serve to discourage people from seeking help and to disempower those who do.

Adults face particular difficulties in accessing and affording services and treatments. Psychiatrists in the public health system do not treat ADHD in adults, except in some remote areas of Australia. Few psychiatrists and psychologists are familiar with ADHD, and adults can find themselves misdiagnosed, inappropriately medicated and on a very expensive merry-go-round. They may also find inconsistencies with diagnostic procedures and dosage rates from state to state and difficulty renewing prescriptions when travelling or holidaying across states. Of great concern for adolescents and adults is holidaying, or travelling overseas on business, with stimulant medication, particularly in countries where stimulants of any kind are banned. Difficulties also arise when adults with ADHD and depression seek life insurance or insurance for loss of income. The stigma of being labelled ADHD prevents many adults from being diagnosed and treated for this debilitating condition.

16.1 ADHD and risk of delinquency and criminal behaviour

Although the vast majority of individuals with ADHD will never become involved with crime, research indicates a consistent association between ADHD and delinquency, criminal behaviour and recidivism. Identifying the role of ADHD in crime, however, is a complex task, particularly given that ADHD often occurs with other psychiatric conditions, including depression, anxiety disorder, substance use disorder, CD, and anti-social personality disorder (864). In addition, factors such as IQ and socio-economic status may also have an impact upon the risk of later delinquency or crime. A meta-analysis of studies assessing a link between ADHD and crime and delinquency (212) and several subsequent longitudinal studies that have followed cohorts with ADHD (209, 865) or cohorts drawn from other population groups (213, 866-870) support an association between ADHD and offending.

The meta-analysis by Pratt et al (2002) (212) considered 48 studies and found a statistically significant overall association of ADHD with crime and delinquency. The overall effect size was 0.155 (CI 0.135-0.175, p < 0.01), indicating that ADHD is an important risk factor for crime and delinquency.

Long-term follow-up studies of individuals with ADHD support these findings. Satterfield et al (865, 871) have followed a cohort of 179 clinically referred hyperactive Caucasian boys and 75 community controls in the US over 30 years, from childhood to mid-adulthood. A substantial subgroup of the hyperactive boys became serious adult offenders, with high rates of arrests (44%), convictions (29%) and incarcerations (26%). The authors concluded that hyperactive boys are at risk for antisocial behaviour as adults, and that socio-economic status, IQ and a history of childhood conduct problems were related to an antisocial outcome.

Barkley et al (209) conducted a 13-year prospective follow-up of children diagnosed with hyperactivity (n = 147) and a matched control group (n = 73). Participants had a mean age of 20-21 at the most recent follow-up. Both self-report and official arrest records showed significantly higher rates of crime in the hyperactive group compared to controls. For example, 27% of the hyperactive group compared to 11% of the control group had a felony arrest.

A prospective study of 230 males and 75 females in the San Francisco area showed that symptoms of hyperactivity/impulsivity alone, and in combination with CD, predict both official arrests and a high level of self-reported crime by males, but not females. However, symptoms of inattention seem to be largely unrelated to adult criminal activity (211).

Several long-term follow-up studies from other population groups have been conducted:

- Analysis of data from the 1970 British Cohort Study (with over 10,000 participants) (213) found that 7.4% of the participants met criteria for ADHD at age 10, and at age 30 these participants were significantly more likely than those without ADHD to face negative outcomes in education, economic status, housing, relationships, health and crime. By age 30, compared to their unaffected peers, men with childhood ADHD were 13.1% more likely to report minor offending and 7.8% more likely to report persistent offending. Women with childhood ADHD were 3.8% more likely to report minor offending.
- A German prospective study of 321 8-year-old children followed up to the age of 25 years (866) looked at three different life courses of delinquency, namely episodic, juvenile delinquency, continued juvenile delinquency in young adulthood and late-starting delinquency. The study found that CD/ADHD was

significantly associated both with continued juvenile delinquency and with latestarting delinquency.

- A UK longitudinal study of 86 women raised in institutional care (868) found that rates of conviction were 39%, compared to 7% in a community control group. The study found a strong link between antisocial behaviour, institutional rearing, hyperactivity and CD in childhood, and offending in adulthood.
- A US longitudinal study of 622 African American youth (867) found that, compared to those who had never committed an offence, those who reported committing a felony offence in adolescence were more likely to have had ADHD symptoms in late childhood.
- A Finnish longitudinal study (869) looked at childhood predictors for crime in late adolescence in a follow-up sample of 2,713 boys born in 1981. Over the 4 years from age 16 to 20, 22.2% of boys committed at least one criminal offence other than a minor traffic violation and 4% committed more than five criminal acts. Factors at age 8 that independently predicted more than five offences were living in a non-intact family, low parental education level, parent reports of conduct problems and teacher reports of hyperactivity problems.
- A longitudinal study sample of 754 adoptees found ADHD was a significant predictor of later illegal behaviour, arrest, jail stay or felony conviction (870). ADHD retained significance in predicting adult disruptive behaviour, arrest, jail stays and felony conviction after controlling for other variables such as CD and substance abuse.

16.2 Prevalence of ADHD and comorbidities among offenders

Recommendation

198. As ADHD and ADHD symptoms are common in individuals entering the justice system, screening for ADHD may be indicated in this population.
 ✓ Recommended best practice based on clinical experience and expert opinion

16.2.1 Prevalence studies

Mental health problems are common in juvenile offenders and in adult prisoners. In many cases, individuals will have more than one psychiatric disorder. For example, a large study of juvenile detainees in the USA (872) found that 66.3% of male detainees and 73.8% of female detainees met diagnostic criteria for one or more psychiatric disorder.

It is clear from prevalence studies that the rates of ADHD among both adult and juvenile offenders are considerably higher than in the general population, although rates vary across the studies (see Table 7).

Variations in prevalence rates stem primarily from differences in the methods used to assess ADHD in the incarcerated populations, with differences among the available studies being in the types of rating scales and symptom cut-offs used. The most frequently used rating scale has been the Wender Utah Rating Scale, which does not strictly correspond with the DSM-IV classification of symptoms (873). In addition, comparisons between countries are difficult, as different ages are used to differentiate juvenile and adult offenders, prisoners may be segregated by age, sex or degree of security risk, and different legal systems may be more or less likely to impose custodial sentences.

Many offenders with ADHD meet criteria for additional psychiatric conditions. For example, the Ottweiler Study in Germany looked at 129 male prisoners aged 15–28 years (average age 19.5) and a group of matched controls. The prevalence of ADHD in the offender group was 45% compared to 7.5% in the control group.

Twenty-five of the prisoners with ADHD (19.4% of the total group) also had another psychiatric condition (alcoholism/substance use disorder, externalising disorder and/or personality disorder) (864). A Norwegian study of 82 adult prisoners aged 19–57 years found that 46% met the criteria for childhood ADHD (assessed using the Wender Utah Rating Scale) and 30% met the criteria for adult ADHD (assessed using the Brawon Attention Deficit Disorder Scale – BADD) (874). Reading disorders and personality disorders were more prevalent amongst those prisoners with childhood or current symptoms of ADHD.

Estimates of the prevalence of ADHD among juvenile offender populations also vary, but again are consistently higher than the prevalence in the general population. Two overseas studies were identified:

- A Dutch study (875) found that, among 108 young people aged 12–18 years who were brought before the Juvenile Court of the district of The Hague in 1993 and who were either required, or volunteered, to be assessed, 14% had ADHD as measured by DSM-III criteria (a further 70 neither volunteered nor were required to be assessed).
- An Irish study compared three groups of adolescent boys: 30 in juvenile detention schools, average age 14.9 years; 20 awaiting an initial appointment at an adolescent psychiatry clinic; and 30 from the community. Among the 83% of boys in the offender group who met diagnostic criteria for at least one psychological disorder (compared to 60% in the mental health group), 10.71% had ADHD, based on the Diagnostic Interview Schedule for Children Version IV (876).

Across several published Australian studies of juvenile offenders, ADHD prevalence rates have ranged from 13% to 46% (see Table 7). In a survey of 802 young people aged 12–21 years on community orders in New South Wales, 144 (19%) reported a diagnosis of ADHD (877). This prevalence may have been underestimated as over 1,000 of 1,900 eligible participants were excluded from the study due to refusal to participate or being uncontactable or exclusion criteria such as "too violent or disruptive" (877). In a study of 100 teenage girls (13.5–19 years, mean age 16.5 years) in a female juvenile detention centre in Sydney, 13% met DSM-IV criteria for ADHD (878, 879). In 589 screening assessments among youths completing Community Service Orders in South Australia, 25% reported being previously diagnosed with ADHD (A Putnins, personal communication based on data from the South Australian Adolescent Psychosocial Screening program in July 2008).

In Tasmania, of the 53 young people (12–18 years) admitted to a youth detention centre in Tasmania over a 14-month period who agreed to be assessed (63% of all admissions), 46% scored positive for ADHD on the Adolescent Psychology Scale. Psychiatric comorbidity in this group (and in the total sample) was common, and included CD, major depressive disorder, dysthymia, mood disorder, anxiety, post traumatic stress disorder, adjustment disorder, somatisation disorder and substance abuse disorder (880).

Table 7. Prevalence of ADHD arr	nongst offender	populations
---------------------------------	-----------------	-------------

Reference	Location	Measures	Group	Age (years)	Prevalence of ADHD
Rosler et al 2004 (864) & Retz et al 2004 (881)	Germany (Ottweiler Study)	Wender Utah Rating Scale, & DSM-IV criteria	129 male inmates	15–28	45% current ADHD
Young et al 2003 (873)	England	Wender Utah Rating Scale	69 offenders with personality disorder	18–60	78% childhood ADHD 6% current ADHD
Eyestone & Howell 1994 (882)	USA	Wender Adult/Child list of problems	102 male inmates	16–64	25.5% current ADHD 21.5% childhood ADHD
Gunter et al 2008 (883)	USA	Mini International Neuropsychiatric Interview Plus (Mini-Plus)	264 male and 56 female inmates	Mean age 31.1	22% current ADHD
Minor et al 2003 (884)	USA	Wender Utah Rating Scale	112 male and 103 female inmates	Mean age 32.4	38% of the males and 41% of the females met the criteria for ADHD
Rasmussen et al 2001 (874)	Norway	Wender Utah Rating Scale & Brown Attention Deficit Disorder Scale (BADD)	82 male inmates	19–57	46% childhood ADHD 30% current ADHD
Gudjonsson et al 2008 (885)	Iceland	DSM-IV Checklist criteria for ADHD Wender Utah Rating Scale	90 male inmates	19–56	50% childhood ADHD (of whom 31% had current ADHD symptoms)
Vitelli 1996 (886)	Canada	Wender Utah Rating Scale	100 inmates (gender not reported)	17–70	67% childhood ADHD
Doreleijers et al 2000 (875)	Netherlands	DSM-III criteria	108 at Juvenile Court	12–18	14% current ADHD
Hayes et al 2007 (876)	Ireland	Diagnostic Interview Schedule for Children Version IV	30 males in juvenile detention 30 controls	Mean age 14.9	Offenders: 83% current ADHD Controls: 10.71% current ADHD
Kenny et al 2007 (877)	Australia (NSW)	Adolescent Psychopathology Scale – Short Form (APS-SF)	683 males and 119 females on community orders (largely self-selected)	12–21	19%
Dixon et al 2004 (878), Dixon et al 2005 (879)	Australia (NSW)	Schedule for Affective Disorders and Schizophrenia for School-Age Children – Present and Lifetime Version	100 female inmates 100 age- and SES-matched controls	13.5– 19	Offenders: 13 Controls: 1% (p <0.001)

		(K-SADS-PL)			
Bickel et al 2002 (880)	Australia (Tasmania)	Adolescent Psychology Scale	43 male and 7 female inmates	12–18	46%

16.2.2 Screening for ADHD within the justice system

The high prevalence of ADHD among people in prisons and juvenile detention centres, and those placed on community orders, raises the important possibility of screening those in the justice system for ADHD. The goal of screening would be to identify individuals who may benefit from a comprehensive ADHD assessment and, if ADHD is diagnosed, treatment.

Further impetus for screening for ADHD in people entering the justice system comes from the link between ADHD and an increased risk of suicidal behaviour (887). A South Australian study of 900 incarcerated young offenders found that ADHD was one of the factors (even after controlling for prior suicide attempt status) that predicted suicidal behaviour (888). ADHD was also found to be a significant predictor of suicidal behaviours amongst juveniles detained in an Austrian pre-trial detention facility (889).

Regular assessments are needed, not just at the point of entry into the criminal justice or prison system as the mental health needs of offenders change over time. For example, in a study of 3,058 offenders (890), the prevalence of DSM-IV attention deficit or disruptive behaviour disorders (ADHD, CD, and ODD) were found to vary across age groups. Among 15–17 year olds, prevalence was 23.3% (14/60); in 18–21 year olds, 11.3% (34/300); and among those aged 22 years and older, 2.1% (56/2698).

Self-report questionnaires have been suggested as one possible screening tool as they are fast and cost-effective (891). However, self-reporting of ADHD symptoms in this population may be poor, as research suggests that individuals with severe behavioural and psychiatric disorders may minimise symptoms of psychopathology on self-report questionnaires (892). Consequently, it would be important to seek third-party reports to confirm self-reported assessments. Rating scales that consider ADHD symptoms have been designed for assessing adult prisoners. One example is the Conners' Adult ADHD Rating Scales (CAARS) Use in Correctional Settings Supplement.

The Secure Care Psychosocial Screening Tool (SECAPS) is a standardised instrument developed in Australia and used routinely in South Australia for initial brief assessment of young people shortly after admission to juvenile detention. It covers a number of psychological and social need areas, including numeracy, literacy, intellectual functioning, social relationships, mood, self-harm, ADHD signs, anger and aggression, and substance use (893). A recidivism risk index has also been developed, based on the SECAPS, and its predictive accuracy was found to be high when the scale was validated on recidivism status 6 months after first release from secure care in a South Australian study of 458 juvenile offenders (894).

Difficulties in diagnosis may arise from the associated comorbidities. For example, Young et al (873) noted that ADHD may be under-diagnosed or misidentified in offenders with personality disorder as it may be difficult to discriminate retrospectively between childhood CD and ADHD.

16.3 Impact of treatment for ADHD on rates of offending and recidivism

Recommendation

199. More research is needed to determine whether treatment of ADHD can reduce the risk of crime and recidivism.

By intervening early and effectively with children and young people who have known and treatable risk factors, it may be possible to deflect them from offending. Extrapolation of current research that shows treatment of ADHD leads to reductions in substance use and aggressive behaviour, both of which are closely related to offending, suggests that treatment of ADHD is likely to be of benefit in reducing recidivism (for review see (895)). To date, however, there is no clear data to show that treatment of ADHD reduces offending and recidivism. Whether treatment of ADHD in childhood can divert a child from later criminal activity, and which treatments might achieve this, remain unclear, with few studies addressing these specific questions. Although numerous studies have followed up individuals with ADHD to look at the impact of ADHD on later crime, there is a paucity of information on the impact of treatment.

There is also the possibility of treating offenders diagnosed with ADHD to prevent recidivism. There are a number of rehabilitation programs targeted at offenders that have been found to have a positive impact on recidivism (896). However, no studies were identified that specifically looked at the efficacy of these programs for individuals with ADHD.

Self-control is viewed as a key psychological determinant of offending (897). Accordingly, impulsiveness and stimulus seeking are common targets of many cognitive skills programs for offenders that aim to reduce recidivism. As a result, these programs may be particularly relevant to individuals with ADHD.

Cognitive skills programs for offenders are largely based on CBT and social skills training. Such programs, for example "Think First", "Thinking for Change" and "Offending Is Not The Only Choice", have been either implemented or piloted in all Australian States and Territories (898). Cognitive skills programs focus on the way people think about their behaviour and target problem-solving skills, moral reasoning, cognitive style, self-control, perspective taking and victim awareness (898). Research is needed to determine whether these programs could be beneficial for individuals with ADHD.

One international example is a Wraparound program for juveniles with mental health problems that has been developed in the USA (899). In addition, a prosocial competence training program "Reasoning and Rehabilitation" has been described (900). This program has been trialled in a number of countries and meta-analysis has shown a significant decrease in recidivism for program participants compared to controls (900). A version of Reasoning and Rehabilitation has been developed specifically for ADHD. This program, called "Reasoning and Rehabilitation 2 for ADHD in Youths and Adults" is currently being trialled.

To optimise program success with individual offenders there is a need to take into account individual differences (responsivity factors) that affect learning styles. These differences can include literacy, intellectual ability, motivation and psychiatric status. They can also extend to factors particularly associated with ADHD such as distractibility and impulsiveness. Accordingly, it is important that programs are designed to directly engage offenders with ADHD to ensure these individuals are able to benefit fully from the programs that are offered.

16.4 Particular vulnerabilities of people with ADHD in the justice system

Recommendations

- 200. Officers of the legal and justice systems should be made aware of the potential vulnerabilities and needs of people with ADHD. ✓ Recommended best practice based on clinical experience and expert opinion
- 201. Proper assessment and diagnosis of ADHD is important for identifying symptoms that may make an individual vulnerable to standard police and court procedures. Psychoeducational and neuropsychological assessment may be useful in this situation to assess individual vulnerability in comparison to population standards.

Recommended best practice based on clinical experience and expert opinion

- 202. Special provisions may be appropriate for some individuals with ADHD (e.g. regular breaks, repetition of important information). Recommended best practice based on clinical experience and expert opinion

16.4.1 Vulnerabilities

The British Association for Psychopharmacology, in their evidenced-based guidelines for management of ADHD in adolescents and adults (388), and several research groups (212, 873, 885, 901), have identified a number of ways in which people with ADHD may be vulnerable at all stages of the justice system:

- Inattention and impulsiveness may compromise performance in a police interview.
- Individuals with ADHD may be more compliant during police interrogation (885).
- In arranging for an adult family member to support a young person with ADHD, it should be remembered that they may have undiagnosed or untreated ADHD, and in some cases this may compromise their capacity to provide adequate support and advice.
- People with ADHD can also be vulnerable in court due to inattention and other symptoms, which could have implications for conviction and sentencing.
- In detention, individuals with ADHD may have particular difficulty in conforming to the expectations of this environment, such as attending to instructions or taking on tasks that require extended periods of attention.
- People with ADHD symptoms who are incarcerated have been found to be more disruptive (verbal aggression, damage to property) than a non-ADHD control group (873).

Several studies have looked at whether individuals with ADHD may be more suggestible than people without the disorder. Research addressing the potential vulnerabilities of people with ADHD during police interrogation in a referred clinic population found that those with ADHD were no more suggestible than a matched control group; however, the ADHD group gave more answers of "don't know" (902). A similar study in a prison population found again that individuals with ADHD were not more suggestible than controls, but they did show greater compliance and claimed to have made more false confessions (885). The increased compliance was found to be significantly related to both current and childhood symptoms of ADHD (885).

16.4.2 Addressing these vulnerabilities

Special provisions for individuals with ADHD during police interrogation or in court

In Australia the *Disability Discrimination Act 1992* (Cth) makes it unlawful to discriminate directly or indirectly against a person with a disability. People appropriately diagnosed with ADHD are entitled to reasonable accommodations during police interrogation or in the courtroom (as in the classroom or workplace).

Some individuals with ADHD coming in for police interrogation or attending court may require special consideration. This may apply particularly to those not currently treated for ADHD and those with severe symptoms. Accordingly, it is important to identify and properly diagnose ADHD. Psychoeducational and neuropsychological assessment might offer a measure of vulnerability and provide a path to allow special provisions for the defendant (901).

Suggestions for how police or court procedures may assist an individual with ADHD in participating fully include (901, 903):

- regular breaks
- avoiding lengthy questions and complex language
- ensuring that important information is put across directly and simply
- repetition of important information
- additional time to reply to questions
- non-confrontational communication.

Treatment of ADHD within prisons

Within the prison system there are many difficulties associated with the treatment of individuals who may have ADHD. For those diagnosed with ADHD, the use of stimulant medications with their potential for abuse may be problematic in a prison environment. Newer medications, such as ATX, with little or no potential for abuse may provide a preferable alternative. (For discussion of the abuse potential of ADHD medications, see section 8.9 ADHD and substance misuse, page 124.) Research addressing the impact of treatments for ADHD within prisons or other correctional facilities is sparse. A single case study from Britain reports a significant improvement in an individual in secure detention following diagnoses of ADHD and subsequent treatment with stimulant medication (904).

16.4.3 Conclusions

There are many gaps in our current state of knowledge about individuals with ADHD and their relationship to the justice system. Some important gaps occur in relation to the possibility of effective early intervention to prevent criminality, and also, in relation to reliable means of assessment of ADHD, and practicable interventions for those with ADHD who find themselves in secure detention.

The opportunity to reduce future offending and imprisonment, and to improve lives, deserves to be given high priority by researchers, clinicians, courts and policy-makers.

CHAPTER 17. ECONOMIC CONSIDERATIONS IN ADHD

17.1 Economic burden of ADHD

17.1.1 What is the cost to the community of ADHD internationally?

There has been limited analysis of the economic impact of ADHD, most of it undertaken in the United States. No studies have assessed the full economic impact of ADHD on society. A limited number of studies have assessed the cost of therapy.

A 2007 review identified 13 studies examining the economic impact of ADHD in children and adolescents using a Cost of Illness Model that included estimates of the costs of medical care, special education, disciplinary costs, parental work loss and juvenile justice (798). Annual costs were estimated to range from \$12,005 to \$17,458 per individual. Using a prevalence of 5%, this means an annual economic cost of ADHD in the USA of between \$36 and \$52 billion (798).

ADHD is associated with a loss of productivity in the workplace (see Chapter 13. ADHD in the Workplace, page 176). For example, the recent WHO World Mental Health Survey Initiative found that ADHD was associated with an average annual reduction in work quantity (21.7 excess days) and quality (13.6 excess days) compared to workers without ADHD (199). The researchers estimated 143.8 million lost days of productivity associated with ADHD each year in the 10 surveyed countries from North and South America, Europe and the Middle East. Longitudinal studies show that individuals with ADHD have completed less education and have lower ranking occupations, compared with controls (769) (198).

A 2005 review of the literature on the economic costs and potential economic benefits of treating ADHD identified 22 relevant publications (905). Compared to matched controls, there were additional annual costs (in 2004 US dollars) of US\$503 to US\$1,343 for children with ADHD, and additional costs of US\$4,929 to US\$5,651 for adults with ADHD. Cost-effectiveness ratios for treatment with MPH, which is a cost-effective treatment option, ranged from US\$15,509 to US\$27,766 per quality-adjusted life year gained (905).

Direct medical costs for youth with ADHD were found to be approximately double those of youth without ADHD, with costs being highest for those with comorbid psychiatric disorders and those being treated with stimulant medication (906). A subsequent study found increased health service costs associated with ADHD, and decreased use of Emergency Department facilities with stimulant therapy for ADHD (797, 907).

In a study of 143,561 patients in disease cohorts in a managed care plan, the annual economic cost for adults with ADHD was less than for diabetes or depression, but equivalent to seasonal allergy (908).

Children with ADHD aged 2–10 years in an integrated healthcare system were found to use significantly more services in the 2 years both pre and post diagnosis than children without ADHD (909).

17.1.2 What is the cost to the community of ADHD in Australia?

Stimulant medications subsidised by the Pharmaceutical Benefits Scheme in Australia do not appear in the top 100 drugs by volume or by cost.

In Western Australia, during the period August 2003 to December 2004, there were a total of 19,062 patients with at least one record of a dispensed stimulant prescription (910). For these patients, a total of 178,369 prescriptions were dispensed. The average number of prescriptions for a stimulant medication dispensed per patient was 9.4, with a range of 1 to 100. The New South Wales Clinical Excellence Commission reported that there were 5.015 "defined daily doses" per 1,000 population per day for 4–17 year olds in Australia for the 12 months from 30 June 2006 (460). Figure 1 shows the number of prescriptions for medication used to treat ADHD in Australia for the years 2003 to 2007. These data record both private and Pharmaceutical Benefits Scheme prescriptions.



Figure 1: Prescriptions of medication used in treatment of ADHD in Australia

17.2 Cost-effectiveness of treatment

17.2.1 Cost-effectiveness of treatment versus no treatment

Analysis of cost-effectiveness of treatments has been conducted for the MTA study, which considered 579 children aged 7–9.9 years and compared 14 months of treatment with medication, behavioural treatment, combined medication and behavioural treatment, or community care (911). All groups showed symptom improvement, with medication and combined medication and behavioural therapy showing the greatest improvements and being equally efficacious at 14 months. There was, however, a four-fold variation in treatment costs, with medication being the least and combined treatment the most expensive. Medication was only marginally more costly than community care.

In an extension of this study, it was noted that the most cost-effective therapy for comorbidities was likely to be behaviour therapy, with or without medication, and that this would yield greater long-term benefits to policy-makers prepared to pay the higher costs (912).

King et al's (443) systematic review of treatment costs in the USA and Canada concluded that medication therapy was superior to no medication therapy, that there were no significant differences in terms of efficacy or side effects between medications, and that the additional benefits from behavioural therapy were uncertain. The cost-effectiveness of therapy was dependent upon medication costs, and economic modelling suggested DEX as the optimal first-line treatment followed by MPH-IR, with ATX as the third-line agent. For patients in whom stimulants are contraindicated, ATX was the preferred agent.

A comparison of MPH with placebo demonstrated that the medication was "reasonably" cost-effective in the short term, but data were lacking on long-term benefits (913).

A large US study of adults found MPH to be a lower cost option than MAS or ATX over the 6-month period following initiation of treatment (914). This cost benefit

was for drug costs and total medical and drug costs. The study controlled for substance abuse, depression and other comorbidities.

Overall, DEX provides the most cost-effective treatment option for ADHD in Australia, although current Australian pricing structures may artificially influence this conclusion.

17.3 Conclusions

The limited data available indicate that ADHD poses an economic burden on families and the community. It is not possible to estimate the full magnitude of this burden in Australia. Similarly, it is not clear whether treating ADHD provides an economic advantage, particularly as the full societal costs, including educational costs and lost employment opportunities, have not been well assessed.

ABBREVIATIONS AND ACRONYMS

ABI	acquired brain injury
ACCC	Australian Competition and Consumer Commission
ACRRM	Australian College of Rural and Remote Medicine
ADHD	Attention deficit hyperactivity disorder
ADRAC	Adverse Drug Reactions Advisory Committee
AISRS	Adult ADHD Investigator Symptom Rating Scale
APS	Australian Psychological Society
ATX	Atomoxetine
AUD	alcohol-use disorder
BSI	Brief Symptom Inventory
CAHE	Centre for Allied Health Evidence
CALD	culturally and linguistically diverse
СВТ	cognitive behavioural therapy
CD	conduct disorder
CGI	Clinical Global Impression
CI	confidence interval
CLAM	Conners' Loney & Milich Scale
CPRS-R	Conners' Parent Rating Scale – Revised
CPRS	Conners' Parent Rating Scale
CPRS-H	Conners' Parent Rating Subscale - Hyperactivity
CTRS-R	Conners' Teacher Rating Scale – Revised
CTRS	Conners' Teacher Rating Scale
DALY	disability adjusted life year
DEX	dexamphetamine sulphate
DoHA	Department of Health and Ageing
DSM-III	<i>Diagnostic and Statistical Manual of Mental Disorders</i> (3rd edition)
DSM-IV	<i>Diagnostic and Statistical Manual of Mental Disorders</i> (4th edition)
ECG	Electrocardiogram
EEG	Electroencephalogram
GAR	Guidelines Assessment Registrar
ICD-10	International Statistical Classification of Diseases and Related Health Problems (10th revision)
ICER	incremental cost-effectiveness ratio
ITT	intention-to-treat
MAS	mixed amphetamine salts

MAS-ER	mixed amphetamine salts – extended-release
MeSH	Medical Subject Headings
MPH	Methylphenidate
MPH-ER	methylphenidate – extended-release
MPH-IR	methylphenidate – immediate-release
MPH-MLR	methylphenidate – multilayer-release
MTA	Multimodal Treatment Study of ADHD
NHMRC	National Health and Medical Research Council
ODD	oppositional defiant disorder
PATS	Preschool ADHD Treatment Study
PBS	Pharmaceutical Benefits Scheme
PICO	Population, intervention, comparator, outcome
RACGP	Royal Australian College of General Practitioners
RACP	Royal Australasian College of Physicians
RANZP	Royal Australian and New Zealand College of Psychiatrists
RCT	Randomised controlled trial
SCT	Sluggish combined type
SKAMP	Swanson, Kotkin, Agler, M-Flynn and Pelham (scale)
SNAP	Swanson, Nolan and Pelham (scale)
SST	social skills training
TGA	Therapeutic Goods Administration

GLOSSARY

Allied health professionals	Health professionals other than doctors and nurses. Allied health professionals who may be involved with people with ADHD include (but are not limited to) occupational therapists, social workers, psychologists, speech pathologists and audiologists.	
Audiology	The study of hearing.	
Behavioural optometry	Behavioural optometry is a system of eye care that emphasises visual training as a way to improve the way a patient uses his or her eyes. Behavioural optometry aims to treat the whole patient, not just correct his or her vision.	
Biofeedback/ neurofeedback	A technique in which people learn to control involuntary body processes such as breathing rate, blood pressure and body temperature by responding to feedback from electrodes that monitor changing body conditions.	
Bipolar disorder	A psychiatric disorder where the individual has episodes of mania (or hypomania) alone or with depressive episodes at other times. Can occur as mixed episodes and with varying degrees of intensity.	
Clinician or health professional	A professional such as a GP, nurse, psychologist or psychiatrist employed in clinical practice.	
Comorbidity	The occurrence of more than one mental health disorder at the same time.	
Conduct disorder	A disruptive behaviour disorder of childhood or adolescence characterised by a persistent pattern of conduct in which the rights of others or age- appropriate societal norms or rules are violated. Behaviours may include bullying, cruelty to animals or people, and law-breaking activities such as shoplifting, vandalism and deliberately lighting fires.	
Differential diagnosis	An alternative diagnosis that could be made on the basis of observed signs and reported symptoms.	
Executive function	A cluster of high-order skills, which include selective attention, behavioural planning and response inhibition, the manipulation of information in problem-solving tasks, and the regulation of behaviour.	
Efficacy	The extent to which an intervention or treatment produces favourable outcomes under ideally controlled conditions, such as a randomised controlled trial.	
Epilepsy	A disorder characterised by recurrent unprovoked seizures with sudden onset.	
Extended-release medications	Extended-release medications are designed to release the medication over a longer period of time. The advantage is that the extended-release medications can often be taken less frequently than immediate-	

	release formulations of the same medication.
Homeopathy	A system of therapy where people are treated with highly dilute doses of substances thought to be capable of causing specific diseases, and that produce the same symptoms in a healthy individual.
Immediate-release medications	Immediate-release medications release the full dose immediately. Multiple doses may be needed to maintain the effect of the drug.
Intellectual disability	Limitations in intellectual functioning. Difficulty in learning and performing certain daily life skills.
Meta-analysis	A statistical synthesis that combines the results from a set of comparable studies, identified in a systematic review, that have investigated the same research question.
Multimodal therapy	Treatment strategies that typically combine medication and psychosocial interventions such as behaviour therapy, psychoeducation, counselling or support. For children and adolescents, educational interventions that are designed to help with learning difficulties can also be included.
Neuroimaging	The production of images of the brain by non- invasive techniques. Examples of neuroimaging techniques include magnetic resonance imaging (MRI), functional magnetic resonance imaging (fMRI), diffusion tensor imaging (DTI), positron emission tomography (PET) and single photon emission computed tomography (SPECT).
Neuropsychological tests	Neuropsychological tests are specifically designed tasks used to measure brain function. They can be used to examine a range of mental processes from simple motor performance to complex reasoning and problem solving.
Neurophysiological tests	Neurophysiological tests measure the electrical activity of the brain from the skull surface. Electrophysiological measures such as event-related potentials (ERP) and electroencephalography (EEG) are examples of neurophysiological tests. EEG maps the electrical activity of specific brain regions. ERPs provide information on electrical activity taking place in the brain in response to stimuli.
Open label studies	A clinical trial in which the researchers and the participants in a research study know the treatment the participant is using.
Oppositional defiant disorder	A type of disruptive behaviour disorder characterised by a recurrent pattern of defiant, hostile, disobedient and negativistic behaviour directed toward those in authority, including such actions as defying the requests or rules of adults, deliberately annoying others, arguing, spitefulness and vindictiveness that occur much more frequently than would be expected on the basis of age and developmental stage.

Parent/carer	The parent(s) or person(s) that take legal responsibility for the preschooler, child or adolescent and provides direct care. This includes birth parents, step-parents, adopted parents, foster parents, legal guardians, custodial parents or other appropriate primary caregivers.
Parenting programs	Parenting programs (or parent-training programs) aim to teach parents/caregivers strategies for managing disruptive behaviour in their child and improving parent-child relationships. Parenting programs make use of behaviour modification techniques such as structured reward systems and discipline techniques. They also commonly involve cognitive behavioural techniques which require identifying problem behaviours, analysing their cause, developing a consistent response and modifying it on the basis of feedback.
Pervasive developmental disorders	A group of disorders that includes autism, Rett syndrome, Asperger's syndrome, Childhood Disintegrative Disorder, and Pervasive Developmental Disorder Not Otherwise Specified (PDD NOS). PDDs are characterised by delays in the development of socialisation and communication skills. Symptoms may include problems with using and understanding language; difficulty relating to people, objects and events; unusual play with toys and other objects; difficulty with changes in routine or familiar surroundings; and repetitive body movements or behaviour patterns.
Psychosocial interventions	A therapeutic intervention using cognitive, cognitive- behavioural, behavioural or supportive interventions that is carried out by mental health professionals.
Psychotropic medications	Medications that primarily act on the central nervous system and affect the mind or mood or other mental processes.
Randomised controlled trial	An experimental comparison study in which participants are allocated to treatment/intervention or control/placebo groups using a random mechanism. Participants have an equal chance of being allocated to an intervention or control group and therefore allocation bias is eliminated.
Recidivism	A relapse into prior criminal or delinquent behaviour especially following conviction and punishment.
Research question	Specific and clearly defined questions concerning key areas of interest which are addressed in the systematic review.
Respite	Any support service that offers both individuals and their carers a break or change from their routine for anything from a few hours to a few weeks.

Sensory integration A form of occupational therapy in which special exercises are used to strengthen the patient's sense of touch, sense of balance, and sense of where the body and its parts are in space. The process of systematically locating, appraising and Systematic review synthesising evidence from scientific studies in order to obtain a reliable overview. Tic disorders Tic disorders are characterised by the persistent presence of tics, which are abrupt, repetitive involuntary movements and sounds. Tics may be simple (using only a few muscles or simple sounds) or complex (using many muscle groups or full words and sentences). A neurological disorder characterised by repetitive, Tourette syndrome stereotyped involuntary movements and sounds. A family centred planning process for building Wraparound constructive relationships and support networks among families, teachers and other caregivers to benefit the inclusion or facilitate the progress of youth with emotional and behavioural problems.

REFERENCES

- 1. National Health and Medical Research Council (NHMRC). NHMRC additional levels of evidence and grades of recommendations for developers of guidelines. Stage 2 Consultation 2008-2010. 2008.
- National Health and Medical Research Council (NHMRC). A guide to the development, implementation and evaluation of clinical practice guidelines. 1998.
- 3. National Health and Medical Research Council (NHMRC). How to review the evidence:systematic identification and review of the scientific literature. 1999.
- 4. National Health and Medical Research Council (NHMRC). How to use the evidence: assessment and application of scientific evidence. 2000.
- 5. National Health and Medical Research Council (NHMRC). NHMRC standards and procedures for externally developed guidelines. 2005.
- 6. Biederman J, Faraone S. Attention-deficit hyperactivity disorder. *Lancet* 2005; 366:237-248.
- 7. Graetz B, Sawyer M, Hazell P, Arney F, Baghurst P. Validity of DSM-IV ADHD subtypes in a nationally representative sample of Australian children and adolescents. *J Am Acad Child Adolesc Psychiatry* 2001; 40:1410-1417.
- Kroes M, Kalff AC, Kessels AG, Steyaert J, Feron FJ, van Someren AJ, Hurks PP, Hendriksen JG, van Zeben TM, Rozendaal N, Crolla IF, Troost J, Jolles J, Vles JS. Child psychiatric diagnoses in a population of Dutch schoolchildren aged 6 to 8 years. J Am Acad Child Adolesc Psychiatry 2001; 40:1401-9.
- 9. Rohde LA, Biederman J, Busnello EA, Zimmermann H, Schmitz M, Martins S, Tramontina S. ADHD in a school sample of Brazilian adolescents: a study of prevalence, comorbid conditions, and impairments. *J Am Acad Child Adolesc Psychiatry* 1999; 38:716-22.
- 10. Barbaresi WJ, Katusic SK, Colligan RC, Pankratz VS, Weaver AL, Weber KJ, Mrazek DA, Jacobsen SJ. How common is attention-deficit/hyperactivity disorder? Incidence in a population-based birth cohort in Rochester, Minn. *Arch Pediatr Adolesc Med* 2002; 156:217-24.
- Szatmari P, Offord DR, Boyle MH. Correlates, associated impairments and patterns of service utilization of children with attention deficit disorder: findings from the Ontario Child Health Study. J Child Psychol Psychiatry 1989; 30:205-17.
- 12. Wilens TE, Biederman J, Brown S, Tanguay S, Monuteaux MC, Blake C, Spencer TJ. Psychiatric comorbidity and functioning in clinically referred preschool children and school-age youths with ADHD. *J Am Acad Child Adolesc Psychiatry* 2002; 41:262-268.
- 13. Spencer T. ADHD and comorbidity in childhood. *J Clin Psychiatry* 2006; 67 Suppl 8:27-31.
- Dykman R, PT. A. Attention deficit disorder and specific reading disability: Separate but often overlapping disorders In: Shaywitz S and Shaywitz B, eds. Attention deficit disorder comes of age: Toward the twenty-first century. Austin Tx: Pro-Ed, 1992:165-184.
- Gillberg C, Kadesjo B. Attention-deficit/hyperactivity disorder and developmental coordination disorder In: Brown T, ed. Attention deficit disorder and comorbidities in children, adolescents and adults. Washington DC: American Psychiatric Press, 2000:393-406.
- 16. Weiss G, Hechtman L. Hyperactive children grown-up: ADHD in children, adolescents, and adults. New York: The Guilford Press, 1993.
- 17. Barkley RA, Cox D. A review of driving risks and impairments associated with attention-deficit/hyperactivity disorder and the effects of stimulant medication on driving performance. J Safety Res 2007; 38:113-28.

- Crichton A. An inquiry into the nature and origin of mental derangement: on attention and its diseases. J Attention Disord 2008; 12:200-4; discussion 205-6.
- 19. Still G. The Coulstonian lectures on some abnormal psychical conditions in children. *Lancet* 1902; 1:1008-1012, 1077-1082, 1163-1168.
- 20. Ebaugh F. Neuropsychiatric sequelae of acute epidemic encephalitis in children. *Am J Dis Child* 1923; 25:89-97.
- 21. Strauss A, Lehtinen L. Psychopathology and education of the brain-injured child. Grune and Stratton: New York, 1947.
- 22. American Psychiatric Association. Diagnostic and Statistical Manual of Mental Disorders. Second Edition (DSM-II) Washington, DC, 1968.
- 23. Douglas V. Stop, look and listen: The problem of sustained attention and impulse control in hyperactive and normal children. *Can J Behavioural Science* 1972; 4:259-282.
- 24. American Psychiatric Association. Diagnostic and statistical manual of mental disorders, third edition (DSM-III) Washington DC: American Psychiatric Association, 1980.
- 25. Bradley C. The behavior of children receiving Benzedrine. *Am J Psychiatry* 1937; 94:577-585.
- 26. Castellanos FX, Tannock R. Neuroscience of attention-deficit/hyperactivity disorder: the search for endophenotypes. *Nat Rev Neurosci* 2002; 3:617-28.
- 27. Levy F, Hay DA. Attention, Genes and ADHD. Hove: Brunner Routledge, 2001.
- 28. Todd RD. Genetics of attention deficit/hyperactivity disorder: are we ready for molecular genetic studies? *Am J Med Genet* 2000; 96:241–243.
- 29. Levy F, Hay DA, McStephen M, Wood C, Waldman I. Attention-deficit hyperactivity disorder: a category or a continuum? Genetic analysis of a large-scale twin study. J Am Acad Child Adolesc Psychiatry 1997; 36:737-44.
- Asherson P, Kuntsi J, Taylor E. Unravelling the complexity of attention-deficit hyperactivity disorder: a behavioural genomic approach. Br J Psychiatry 2005; 187:103-5.
- 31. Barkley RA. Behavioral inhibition, sustained attention, and executive functions: constructing a unifying theory of ADHD. *Psychol Bull* 1997; 121:65-94.
- 32. American Psychiatric Association. Diagnostic and statistical manual of mental disorders, fourth edition (DSM-IV) Washington DC: American Psychiatric Association, 1994.
- 33. McBurnett K. The diagnosis and how we got there, In: McBurnett K and Pfiffner L, eds. Attention deficit hyperactivity disorder: Concepts, controversies, new directions. New York: Informa Healthcare, 2008:1-8.
- 34. Barkley RA, Biederman J. Toward a broader definition of the age-of-onset criterion for attention-deficit hyperactivity disorder. J Am Acad Child Adolesc Psychiatry 1997; 36:1204-10.
- 35. McBurnett K, Pfiffner LJ, Frick PJ. Symptom properties as a function of ADHD type: an argument for continued study of sluggish cognitive tempo. *J Abnorm Child Psychol* 2001; 29:207-13.
- 36. World Health Organization. International Classification of Diseases (ICD-10) 10th Edition. Geneva: World Health Organisation 1992.
- Santosh P. Multi-modaltreatment study of ADHD (MTA): impact of classificatory system on pharmacological interventions. American Academy of Child and Adolescent Psychiatry 49th Annual Meeting; Symposium 22. San Francisco, USA, 2002.
- 38. Schachar R, Chen S, Crosbie J, Goos L, Ickowicz A, Charach A. Comparison of the predictive validity of hyperkinetic disorder and attention deficit hyperactivity disorder. *J Can Acad Child Adolesc Psychiatry* 2007; 16:90-100.
- Gozal D, Molfese D. Attention Deficit Hyperactivity Disorder: From Genes to patients. New Jersey: Humana Press Inc, 2005.
- 40. Fitzgerald M, Bellgrove M, Gill M. Handbook of Attention Deficit Hyperactivity Disorder. England: John Wiley & Sons, 2007.

- 41. Rohde LA. Is there a need to reformulate attention deficit hyperactivity disorder criteria in future nosologic classifications? *Child Adolesc Psychiatr Clin N Am* 2008; 17:405-20, x.
- 42. Santosh PJ, Taylor E, Swanson J, Wigal T, Chuang S, Davies M, Greenhill L, Newcorn J, Arnold LE, Jensen P, Vitiello B, Elliott G, Hinshaw S, Hechtman L, Abikoff H, Pelham W, Hoza B, Molina B, Wells K, Epstein J, Posner M. Refining the diagnoses of inattention and overactivity syndromes: A reanalysis of the Multimodal Treatment study of attention deficit hyperactivity disorder (ADHD) based on ICD-10 criteria for hyperkinetic disorder. *Clinical Neuroscience Research* 2005; 5:307-314.
- 43. McBurnett K, Pfiffner L. Attention deficit hyperactivity disorder: concepts, controversies, new directions. New York: Informa Healthcare, 2008.
- 44. Gomez R. Item response theory analyses of the parent and teacher ratings of the DSM-IV ADHD rating scale. *J Abnorm Child Psychol* 2008; 36:865-85.
- 45. Rasmussen ER, Neuman RJ, Heath AC, Levy F, Hay DA, Todd RD. Replication of the latent class structure of Attention-Deficit/Hyperactivity Disorder (ADHD) subtypes in a sample of Australian twins. *J Child Psychol Psychiatry* 2002; 43:1018-28.
- 46. Levy F, McStephen M, Hay DA. The diagnostic genetics of ADHD symptoms and subtypes, In: Levy F and Hay DA, eds. Attention, Genes and ADHD. Hove: Brunner-Routledge, 2001:35-57.
- 47. Levy F, Hay DA, Bennett K, McStephen M. Gender differences in ADHD subtype comorbidity. J Am Acad Child Adolesc Psychiatry 2004; 44:368-376.
- 48. Hay DA, McStephen M, Levy F. The developmental genetics of ADHD symptoms and subtypes, In: Levy F and Hay DA, eds. Attention, Genes and ADHD. Hove: Brunner-Routledge, 2001:35-57.
- 49. Stawicki JA, Nigg JT, von Eye A. Family psychiatric history evidence on the nosological relations of DSM-IV ADHD combined and inattentive subtypes: new data and meta-analysis. *J Child Psychol Psychiatr* 2006; 47:935-45.
- 50. Martin J, McDougall M, Hay DA, Piek J. ADHD: Different measures, different definitions, different interpretations. *Manuscript submitted* 2009.
- 51. Swanson JM, Kinsbourne M, Nigg J, Lanphear B, Stefanatos GA, Volkow N, Taylor E, Casey BJ, Castellanos FX, Wadhwa PD. Etiologic subtypes of attention-deficit/hyperactivity disorder: brain imaging, molecular genetic and environmental factors and the dopamine hypothesis. *Neuropsychol Rev* 2007; 17:39-59.
- 52. Willcutt EG, Doyle AE, Nigg J, Faraone SV, Pennington BF. A meta-analytic review of the executive functioning theory of ADHD. *Biol Psychiatry* 2005; 57:1336-1346.
- 53. Albayrak O, Friedel S, Schimmelmann BG, Hinney A, Hebebrand J. Genetic aspects in attention-deficit/hyperactivity disorder. *J Neural Transm* 2008; 115:305-15.
- 54. Sprich S, Biederman J, Crawford MH, Mundy E, Faraone SV. Adoptive and biological families of children and adolescents with ADHD. J Am Acad Child Adolesc Psychiatry 2000; 39:1432-7.
- 55. Cadoret RJ, Stewart MA. An adoption study of attention deficit/hyperactivity/aggression and their relationship to adult antisocial personality. *Compr Psychiatry* 1991; 32:73-82.
- 56. Cantwell DP. Genetics of hyperactivity. *J Child Psychol Psychiatry* 1975; 16:261-4.
- Faraone SV, Perlis RH, Doyle AE, Smoller JW, Goralnick JJ, Holmgren MA, Sklar P. Molecular genetics of attention-deficit/hyperactivity disorder. *Biol Psychiatry* 2005; 57:1313-23.
- 58. Smalley SL, McCracken J, McGough J. Refining the ADHD phenotype using affected sibling pair families. *Am J Med Genet* 2001; 105:31-3.
- 59. Faraone SV, Doyle AE, Lasky-Su J, Sklar PB, D'Angelo E, Gonzalez-Heydrich J, Kratochvil C, Mick E, Klein K, Rezac AJ, Biederman J. Linkage analysis of

attention deficit hyperactivity disorder. Am J Med Genet B Neuropsychiatr Genet 2007.

- 60. Fisher SE, Francks C, McCracken JT, McGough JJ, Marlow AJ, MacPhie IL, Newbury DF, Crawford LR, Palmer CG, Woodward JA, Del'Homme M, Cantwell DP, Nelson SF, Monaco AP, Smalley SL. A genomewide scan for loci involved in attention-deficit/hyperactivity disorder. *Am J Hum Genet* 2002; 70:1183-96.
- 61. Bakker SC, van der Meulen EM, Buitelaar JK, Sandkuijl LA, Pauls DL, Monsuur AJ, van 't Slot R, Minderaa RB, Gunning WB, Pearson PL, Sinke RJ. A wholegenome scan in 164 Dutch sib pairs with attention-deficit/hyperactivity disorder: suggestive evidence for linkage on chromosomes 7p and 15q. *Am J Hum Genet* 2003; 72:1251-60.
- 62. Arcos-Burgos M, Castellanos FX, Konecki D, Lopera F, Pineda D, Palacio JD, Rapoport JL, Berg K, Bailey-Wilson J, Muenke M. Pedigree disequilibrium test (PDT) replicates association and linkage between DRD4 and ADHD in multigenerational and extended pedigrees from a genetic isolate. *Mol Psychiatry* 2004; 9:252-9.
- 63. Mick E, Faraone SV. Genetics of attention deficit hyperactivity disorder. *Child Adolesc Psychiatr Clin N Am* 2008; 17:261-84
- 64. Martin N, McDougall M, Hay DA. What are the key directions in the genetics of attention deficit hyperactivity disorder? *Curr Opin Psychiatry* 2008; 21:356-61.
- 65. Laucht M, Skowronek MH, Becker K, Schmidt MH, Esser G, Schulze TG, Rietschel M. Interacting effects of the dopamine transporter gene and psychosocial adversity on attention-deficit/hyperactivity disorder symptoms among 15-year-olds from a high-risk community sample. *Arch Gen Psychiatry* 2007; 64:585-90.
- 66. Bastain TM, Lewczyk CM, Sharp WS, James RS, Long RT, Eagen PB, Ebens CL, Meck JM, Chan WY, Sidransky E, Rapoport JL, Castellanos FX. Cytogenetic abnormalities in attention-deficit/hyperactivity disorder. *J Am Acad Child Adolesc Psychiatry* 2002; 41:806-10.
- 67. Linnet KM, Dalsgaard S, Obel C, Wisborg K, Henriksen TB, Rodriguez A, Kotimaa A, Moilanen I, Thomsen PH, Olsen J, Jarvelin M-R. Maternal lifestyle factors in pregnancy risk of attention deficit hyperactivity disorder and associated behaviors: review of the current evidence. *Am J Psychiatry* 2003; 160:1028-1040.
- 68. Rodriguez A, Bohlin G. Are maternal smoking and stress during pregnancy related to ADHD symptoms in children? *J Child Psychol Psychiatry* 2005; 46:246-54.
- 69. Schmitz M, Denardin D, Laufer Silva T, Pianca T, Hutz MH, Faraone S, Rohde LA. Smoking during pregnancy and attention-deficit/hyperactivity disorder, predominantly inattentive type: a case-control study. J Am Acad Child Adolesc Psychiatry 2006; 45:1338-45.
- 70. Thapar A, Fowler T, Rice F, Scourfield J, van den Bree M, Thomas H, Harold G, Hay D. Maternal smoking during pregnancy and attention deficit hyperactivity disorder symptoms in offspring. *Am J Psychiatry* 2003; 160:1985-1990.
- 71. Kahn R. Role of dopamine transporter genotype and maternal prenatal smoking in childhood hyperactive-impulsive, inattentive, and oppositional behaviors. *J Pediatr* 2003; 143:104-110.
- 72. Neuman RJ, Lobos E, Reich W, Henderson CA, Sun LW, Todd RD. Prenatal smoking exposure and dopaminergic genotypes interact to cause a severe ADHD subtype. *Biol Psychiatry* 2007; 61:1320-8.
- 73. Vaglenova J, Birru S, Pandiella NM, Breese CR. An assessment of the long-term developmental and behavioral teratogenicity of prenatal nicotine exposure. *Behav Brain Res* 2004; 150:159-70.
- 74. Ward C, Lewis S, Coleman T. Prevalence of maternal smoking and environmental tobacco smoke exposure during pregnancy and impact on birth weight: retrospective study using Millennium Cohort. BMC Public Health 2007; 7:81.

- 75. Streissguth AP, Bookstein FL, Barr HM, Sampson PD, O'Malley K, Young JK. Risk factors for adverse life outcomes in fetal alcohol syndrome and fetal alcohol effects. *J Dev Behav Pediatr* 2004; 25:228-38.
- 76. Fryer SL, McGee CL, Matt GE, Riley EP, Mattson SN. Evaluation of psychopathological conditions in children with heavy prenatal alcohol exposure. *Pediatrics* 2007; 119:e733-741.
- 77. Bhatara V, Loudenberg R, Ellis R. Association of attention deficit hyperactivity disorder and gestational alcohol exposure: an exploratory study. *J Attention Disord* 2006; 9:515-22.
- Knopik VS, Heath AC, Jacob T, Slutske WS, Bucholz KK, Madden PAF, Waldron M, Martin NG. Maternal alcohol use disorder and offspring ADHD: disentangling genetic and environmental effects using a children-of-twins design. *Psychol Med* 2006; 36:1461-71.
- 79. Knopik VS, Sparrow EP, Madden PAF, Bucholz KK, Hudziak JJ, Reich W, Slutske WS, Grant JD, McLaughlin TL, Todorov A, Todd RD, Heath AC. Contributions of parental alcoholism, prenatal substance exposure, and genetic transmission to child ADHD risk: a female twin study. *Psychol Med* 2005; 35:625-35.
- Talge NM, Neal C, Glover V. Antenatal maternal stress and long-term effects on child neurodevelopment: how and why? J Child Psychol Psychiatry 2007; 48:245-61.
- 81. Banerjee TD, Middleton F, Faraone SV. Environmental risk factors for attentiondeficit hyperactivity disorder. *Acta Paediatrica* 2007; 96:1269-74.
- 82. Braun J, Kahn R, Froehlich T, Auinger P, Lanphear B. Exposures to environmental toxicants and attention deficit hyperactivity disorder in U.S. children *Environ Health Perspective* 2006; 114:1904-1909.
- 83. Bhutta AT, Cleves MA, Casey PH, Cradock MM, Anand KJ. Cognitive and behavioral outcomes of school-aged children who were born preterm: a metaanalysis. *Jama* 2002; 288:728-37.
- 84. Hultman CM, Torrang A, Tuvblad C, Cnattingius S, Larsson JO, Lichtenstein P. Birth weight and attention-deficit/hyperactivity symptoms in childhood and early adolescence: a prospective Swedish twin study. J Am Acad Child Adolesc Psychiatry 2007; 46:370-7.
- 85. Lehn H, Derks EM, Hudziak JJ, Heutink P, van Beijsterveldt TC, Boomsma DI. Attention problems and attention-deficit/hyperactivity disorder in discordant and concordant monozygotic twins: evidence of environmental mediators. J Am Acad Child Adolesc Psychiatry 2007; 46:83-91.
- 86. Nigg JT, Breslau N. Prenatal smoking exposure, low birth weight, and disruptive behavior disorders. J Am Acad Child Adolesc Psychiatry 2007; 46:362-9.
- 87. Mick E, Biederman J, Prince J, Fischer MJ, Faraone SV. Impact of low birth weight on attention-deficit hyperactivity disorder. *J Dev Behav Pediatr* 2002; 23:16-22.
- Botting N, Powls A, Cooke RW, Marlow N. Attention deficit hyperactivity disorders and other psychiatric outcomes in very low birthweight children at 12 years. J Child Psychol Psychiatry 1997; 38:931-41.
- 89. Valdimarsdottir M, Hrafnsdottir AH, Magnusson P, Gudmundsson OO. [The frequency of some factors in pregnancy and delivery for Icelandic children with ADHD]. *Laeknabladid* 2006; 92:609-14.
- 90. St Sauver JL, Barbaresi WJ, Katusic SK, Colligan RC, Weaver AL, Jacobsen SJ. Early life risk factors for attention-deficit/hyperactivity disorder: a populationbased cohort study. *Mayo Clin Proc* 2004; 79:1124-31.
- 91. Lou HC. Etiology and pathogenesis of attention-deficit hyperactivity disorder (ADHD): significance of prematurity and perinatal hypoxic-haemodynamic encephalopathy. *Acta Paediatr* 1996; 85:1266-71.
- 92. Gerring JP, Brady KD, Chen A, Vasa R, Grados M, Bandeen-Roche KJ, Bryan RN, Denckla MB. Premorbid prevalence of ADHD and development of secondary ADHD after closed head injury. J Am Acad Child Adolesc Psychiatry 1998; 37:647-54.

- 93. Max JE, Lindgren SD, Knutson C, Pearson CS, Ihrig D, Welborn A. Child and adolescent traumatic brain injury: correlates of injury severity. *Brain Inj* 1998; 12:31-40.
- 94. Catroppa C, Anderson V. Children's attentional skills 2 years post-traumatic brain injury. *Dev Neuropsychol* 2003; 23:359-73.
- 95. Ewing-Cobbs L, Prasad M, Fletcher JM, Levin HS, Miner E, Eisenberg H. Attention after pediatric traumatic brain injury: A multidimensional assessment. *Child Neuropsychol* 1998; 4:35–48.
- 96. Willmott C, Anderson V, Anderson P. Attention following pediatric head injury: a developmental perspective. *Dev Neuropsychol* 2000; 17:361-79.
- 97. Anderson VA, Catroppa C, Haritou F, Morse S, Pentland L, Rosenfeld J, Stargatt R. Predictors of acute child and family outcome following traumatic brain injury in children. *Pediatr Neurosurg* 2001; 34:138-48.
- 98. Catroppa C, Anderson VA, Morse SA, Haritou F, Rosenfeld JV. Children's attentional skills 5 years post-TBI. J Pediatr Psychol 2007; 32:354-69.
- 99. Weiss RE, Stein MA, Trommer B, Refetoff S. Attention-deficit hyperactivity disorder and thyroid function. *J Pediatr* 1993; 123:539-45.
- 100. Alvarez-Pedrerol M, Ribas-Fito N, Torrent M, Julvez J, Ferrer C, Sunyer J. TSH concentration within the normal range is associated with cognitive function and ADHD symptoms in healthy preschoolers. *Clinical Endocrinology* 2007; 66:890-8.
- 101. Hauser P, Zametkin AJ, Martinez P, Vitiello B, Matochik JA, Mixson AJ, Weintraub BD. Attention deficit-hyperactivity disorder in people with generalized resistance to thyroid hormone. *N Engl J Med* 1993; 328:997-1001.
- 102. Spencer T, Biederman J, Wilens T, Guite J, Harding M. ADHD and thyroid abnormalities: a research note. *J Child Psychol Psychiatry* 1995; 36:879-85.
- 103. Schnoll R, Burshteyn D, Cea-Aravena J. Nutrition in the treatment of attention-deficit hyperactivity disorder: a neglected but important aspect. *Appl Psychophysiol Biofeedback* 2003; 28:63-75.
- 104. McCann D, Barrett A, Cooper A, Crumpler D, Dalen L, Grimshaw K, Kitchin E, Lok K, Porteous L, Prince E, Sonuga-Barke E, Warner JO, Stevenson J. Food additives and hyperactive behaviour in 3-year-old and 8/9-year-old children in the community: a randomised, double-blinded, placebo-controlled trial. *Lancet* 2007; 370:1560-7.
- 105. Rowe KS, Rowe KJ. Synthetic food coloring and behavior: a dose response effect in a double-blind, placebo-controlled, repeated-measures study. *J Pediatr* 1994; 125:691-8.
- 106. Bateman B, Warner JO, Hutchinson E, Dean T, Rowlandson P, Gant C, Grundy J, Fitzgerald C, Stevenson J. The effects of a double blind, placebo controlled, artificial food colourings and benzoate preservative challenge on hyperactivity in a general population sample of preschool children. *Arch Dis Child* 2004; 89:506-11.
- 107. Pollock I, Warner JO. Effect of artificial food colours on childhood behaviour. *Arch Dis Child* 1990; 65:74-7.
- 108. Wolraich ML, Wilson DB, White JW. The effect of sugar on behavior or cognition in children. A meta-analysis. *JAMA* 1995; 274:1617-21.
- 109. Arnold LE, DiSilvestro RA. Zinc in attention-deficit/hyperactivity disorder. J Child Adolesc Psychopharmacol 2005; 15:619-27.
- 110. Millichap JG, Yee MM, Davidson SI. Serum ferritin in children with attentiondeficit hyperactivity disorder. *Pediatr Neurol* 2006; 34:200-3.
- 111. Rutter M, Cox A, Tupling C, Berger M, Yule W. Attainment and adjustment in two geographical areas. I--The prevalence of psychiatric disorder. *Br J Psychiatry* 1975; 126:493-509.
- 112. Biederman J, Milberger S, Faraone SV, Kiely K, Guite J, Mick E, Ablon JS, Warburton R, Reed E, Davis SG. Impact of adversity on functioning and comorbidity in children with attention-deficit hyperactivity disorder. *J Am Acad Child Adolesc Psychiatry* 1995; 34:1495-503.

- 113. Famularo R, Kinscherff R, Fenton T. Psychiatric diagnoses of maltreated children: preliminary findings. J Am Acad Child Adolesc Psychiatry 1992; 31:863-7.
- 114. McLeer SV, Callaghan M, Henry D, Wallen J. Psychiatric disorders in sexually abused children. J Am Acad Child Adolesc Psychiatry 1994; 33:313-9.
- 115. Stevens SE, Sonuga-Barke EJ, Kreppner JM, Beckett C, Castle J, Colvert E, Groothues C, Hawkins A, Rutter M. Inattention/Overactivity following early severe institutional deprivation: presentation and associations in early adolescence. J Abnorm Child Psychol 2008; 36:385-98.
- 116. Roy P, Rutter M, Pickles A. Institutional care: risk from family background or pattern of rearing? *J Child Psychol Psychiatr* 2000; 41:139-149.
- 117. Roy P, Rutter M, Pickles A. Institutional care: associations between overactivity and lack of selectivity in social relationships. J Child Psychol Psychiatry 2004; 45:866-73.
- 118. Teicher MH, Andersen SL, Polcari A, Anderson CM, Navalta CP, Kim DM. The neurobiological consequences of early stress and childhood maltreatment. *Neurosci Biobehav Rev* 2003; 27:33-44.
- 119. Millichap JG. Etiologic Classification of Attention-Deficit/Hyperactivity Disorder. *Pediatrics* 2008; 121:e358-365.
- 120. Kieling C. Neurobiology of attention deficit hyperactivity disorder. *Child Adolesc Psychiatr Clin N Am* 2008; 17:285-307
- 121. Dickstein SG, Bannon K, Xavier Castellanos F, Milham MP. The neural correlates of attention deficit hyperactivity disorder: an ALE meta-analysis. *J Child Psychol Psychiatr* 2006; 47:1051-62.
- 122. Valera EM, Faraone SV, Murray KE, Seidman LJ. Meta-analysis of structural imaging findings in attention-deficit/hyperactivity disorder. *Biol Psychiatry* 2007; 61:1361-1369.
- 123. Paloyelis Y, Mehta MA, Kuntsi J, Asherson P. Functional MRI in ADHD: a systematic literature review. *Expert Rev Neurother* 2007; 7:1337-56.
- 124. Krain AL, Castellanos FX. Brain development and ADHD. *Clin Psychol Rev* 2006; 26:433-44.
- 125. Faraone SV, Sergeant J, Gillberg C, Biederman J. The worldwide prevalence of ADHD: is it an American condition? *World Psychiatry* 2003; 2:104-113.
- 126. Skounti M, Philalithis A, Galanakis E. Variations in prevalence of attention deficit hyperactivity disorder worldwide. *Eur J Pediatr* 2007; 166:117-23.
- 127. Polanczyk G, de Lima MS, Horta BL, Biederman J, Rohde LA. The worldwide prevalence of ADHD: a systematic review and metaregression analysis. *Am J Psychiatr* 2007; 164:942-8.
- 128. Taylor E, Sandberg S. Hyperactive behavior in English schoolchildren: a questionnaire survey. J Abnorm Child Psychol 1984; 12:143-55.
- 129. Taylor AC. The overactive child. London: The Spastics Society, 1986.
- 130. American Psychiatric Association. Diagnostic and statistical manual of mental disorders, third edition, revised (DSM-III-R) Washington DC: American Psychiatric Association, 1987.
- 131. August GJ, Realmuto GM, MacDonald AW, 3rd, Nugent SM, Crosby R. Prevalence of ADHD and comorbid disorders among elementary school children screened for disruptive behavior. *J Abnorm Child Psychol* 1996; 24:571-95.
- 132. Achenbach TM, McConaughy SH, Howell CT. Child/adolescent behavioral and emotional problems: implications of cross-informant correlations for situational specificity. *Psychol Bull* 1987; 101:213-32.
- 133. Egger HL, Kondo D, Angold A. The epidemiology and diagnostic issues in preschool attention-deficit/hyperactivity disorder A review. *Infants and Young Children* 2006; 19:109-122.
- 134. Weiss G, Hechtman L, Milroy T, Perlman T. Psychiatric status of hyperactives as adults: a controlled prospective 15-year follow-up of 63 hyperactive children. *J Am Acad Child Psychiatry* 1985; 24:211-20.

- 135. Mannuzza S, Klein RG, Bonagura N, Malloy P, Giampino TL, Addalli KA. Hyperactive boys almost grown up. V. Replication of psychiatric status. *Arch Gen Psychiatry* 1991; 48:77-83.
- 136. Gittelman R, Mannuzza S, Shenker R, Bonagura N. Hyperactive boys almost grown up. I. Psychiatric status. *Arch Gen Psychiatry* 1985; 42:937-47.
- 137. Faraone SV, Biederman J, Mick E. The age-dependent decline of attention deficit hyperactivity disorder: a meta-analysis of follow-up studies. *Psychol Med* 2006; 36:159-65.
- 138. Kessler RC, Adler L, Barkley R, Biederman J, Conners CK, Demler O, Faraone SV, Greenhill LL, Howes MJ, Secnik K, Spencer T, Ustun TB, Walters EE, Zaslavsky AM. The prevalence and correlates of adult ADHD in the United States: results from the National Comorbidity Survey Replication. Am J Psychiatry 2006; 163:716-23.
- 139. Fayyad J, De Graaf R, Kessler R, Alonso J, Angermeyer M, Demyttenaere K, De Girolamo G, Haro JM, Karam EG, Lara C, Lepine JP, Ormel J, Posada-Villa J, Zaslavsky AM, Jin R. Cross-national prevalence and correlates of adult attention-deficit hyperactivity disorder. *Br J Psychiatry* 2007; 190:402-409.
- 140. Kessler RC, Adler LA, Barkley R, Biederman J, Conners CK, Faraone SV, Greenhill LL, Jaeger S, Secnik K, Spencer T, Ustun TB, Zaslavsky AM. Patterns and predictors of attention-deficit/hyperactivity disorder persistence into adulthood: results from the National Comorbidity Survey Replication. *Biol Psychiatry* 2005; 57:1442-1451.
- 141. Rutter M, Taylor E. Child and adolescent psychiatry, fourth edition: Blackwell Publishing, 2002.
- 142. American Psychiatric Association. Diagnostic and statistical manual of mental disorders, fourth edition, revised (DSM-IV-TR) Washington DC: American Psychiatric Association, 2000.
- 143. Barkley RA. Attention-Deficit Hyperactivity Disorder A Handbook for Diagnosis and Treatment, 3rd Edition. New York Guilford Press, 2006.
- 144. Angold A, Costello EJ, Erkanli A. Comorbidity. J Child Psychol Psychiatr 1999; 40:57-87.
- 145. Loeber R, Burke JD, Lahey BB, Winters A, Zera M. Oppositional defiant and conduct disorder: a review of the past 10 years, part I. J Am Acad Child Adolesc Psychiatry 2000; 39:1468-84.
- 146. Freeman R, Consortium TSID, Volume 16 SJ, 2007 Tic disorders and ADHD: answers from a world-wide clinical dataset on Tourette syndrome *Eur Child Adolesc Psychiatry* 2007; 16 Supp 9:I/15-I/23.
- 147. Levy F, Hay DA, Bennett KS, McStephen M. Gender differences in ADHD subtype comorbidity. J Am Acad Child Adolesc Psychiatry 2005; 44:368-76.
- 148. Semrud-Clikeman M, Biederman J, Sprich-Buckminster S, Lehman BK, Faraone SV, Norman D. Comorbidity between ADDH and learning disability: a review and report in a clinically referred sample. J Am Acad Child Adolesc Psychiatry 1992; 31:439-48.
- 149. Willcutt EG, Doyle AE, Nigg JT, Faraone SV, Pennington BF. Validity of the executive function theory of attention-deficit/hyperactivity disorder: a metaanalytic review. *Biol Psychiatry* 2005; 57:1336-46.
- 150. Corkum P, Moldofsky H, Hogg-Johnson S, Humphries T, Tannock R. Sleep problems in children with attention-deficit/hyperactivity disorder: impact of subtype, comorbidity, and stimulant medication. J Am Acad Child Adolesc Psychiatry 1999; 38:1285-93.
- 151. van der Heijden KB, Smits MG, Gunning WB. Sleep-related disorders in ADHD: a review. *Clin Pediatr (Phila)* 2005; 44:201-10.
- 152. Piek JP, Pitcher TM, Hay DA. Motor coordination and kinaesthesis in boys with attention deficit-hyperactivity disorder. *Dev Med Child Neurol* 1999; 41:159-65.
- 153. Woodward LJ, Fergusson DM, Horwood LJ. Driving outcomes of young people with attentional difficulties in adolescence. J Am Acad Child Adolesc Psychiatry 2000; 39:627-34.

- 154. Biederman J, Faraone SV, Spencer T, Wilens T, Norman D, Lapey KA, Mick E, Lehman BK, Doyle A. Patterns of psychiatric comorbidity, cognition, and psychosocial functioning in adults with attention deficit hyperactivity disorder. *Am J Psychiatry* 1993; 150:1792-8.
- 155. Biederman J, Newcorn J, Sprich S. Comorbidity of attention deficit hyperactivity disorder with conduct, depressive, anxiety, and other disorders. *Am J Psychiatry* 1991; 148:564-77.
- 156. Spencer TJ, Biederman J, Mick E. Attention-deficit/hyperactivity disorder: diagnosis, lifespan, comorbidities, and neurobiology. *Ambul Pediatr* 2007; 7:73-81.
- 157. Applegate B, Lahey BB, Hart EL, Biederman J, Hynd GW, Barkley RA, Ollendick T, Frick PJ, Greenhill L, McBurnett K, Newcorn JH, Kerdyk L, Garfinkel B, Waldman I, Shaffer D. Validity of the age-of-onset criterion for ADHD: a report from the DSM-IV field trials. *J Am Acad Child Adolesc Psychiatry* 1997; 36:1211-21.
- 158. Faraone SV, Biederman J, Spencer T, Mick E, Murray K, Petty C, Adamson JJ, Monuteaux MC. Diagnosing adult attention deficit hyperactivity disorder: Are late onset and subthreshold diagnoses valid? *Am J Psychiatr* 2006; 163:1720-1729.
- 159. Willoughby MT, Curran PJ, Costello EJ, Angold A. Implications of early versus late onset of attention-deficit/hyperactivity disorder symptoms. *J Am Acad Child Adolesc Psychiatry* 2000; 39:1512-9.
- 160. Hart EL, Lahey BB, Loeber R, Applegate B, Frick PJ. Developmental change in attention-deficit hyperactivity disorder in boys: a four-year longitudinal study. J Abnorm Child Psychol 1995; 23:729-49.
- 161. Biederman J, Mick E, Faraone SV. Age-dependent decline of symptoms of attention deficit hyperactivity disorder: impact of remission definition and symptom type. *Am J Psychiatry* 2000; 157:816-8.
- 162. Murphy K, Barkley RA. Attention deficit hyperactivity disorder adults: comorbidities and adaptive impairments. *Compr Psychiatry* 1996; 37:393-401.
- 163. Hinshaw S, Owens E, Sami N, Fargeon S. Prospective follow-up of girls with attention-deficit/hyperactivity disorder into adolescence: Evidence for continuing cross-domain impairment. *J Consult Clin Psychol* 2006; 74:489-499.
- 164. Price TS, Simonoff E, Asherson P, Curran S, Kuntsi J, Waldman I, Plomin R. Continuity and change in preschool ADHD symptoms: longitudinal genetic analysis with contrast effects. *Behavior Genetics* 2005; 35:121-32.
- 165. Rietveld MJ, Hudziak JJ, Bartels M, van Beijsterveldt CE, Boomsma DI. Heritability of attention problems in children: longitudinal results from a study of twins, age 3 to 12. *J Child Psychol Psychiatry* 2004; 45:577-88.
- 166. Larsson JO, Larsson H, Lichtenstein P. Genetic and environmental contributions to stability and change of ADHD symptoms between 8 and 13 years of age: a longitudinal twin study. *J Am Acad Child Adolesc Psychiatry* 2004; 43:1267-75.
- 167. Taylor E, Chadwick O, Heptinstall E, Danckaerts M. Hyperactivity and conduct problems as risk factors for adolescent development. *J Am Acad Child Adolesc Psychiatry* 1996; 35:1213-26.
- 168. Hill JC, Schoener EP. Age-dependent decline of attention deficit hyperactivity disorder. *Am J Psychiatry* 1996; 153:1143-6.
- 169. Biederman J, Mick E, Faraone SV. Normalized functioning in youths with persistent attention-deficit/hyperactivity disorder. *J Pediatr* 1998; 133:544-51.
- 170. Lahey BB, Pelham WE, Stein MA, Loney J, Trapani C, Nugent K, Kipp H, Schmidt E, Lee S, Cale M, Gold E, Hartung CM, Willcutt E, Baumann B. Validity of DSM-IV attention-deficit/hyperactivity disorder for younger children. J Am Acad Child Adolesc Psychiatry 1998; 37:695-702.
- 171. Campbell SB, Ewing LJ. Follow-up of hard-to-manage preschoolers: adjustment at age 9 and predictors of continuing symptoms. J Child Psychol Psychiatry 1990; 31:871-89.

- 172. Pierce EW, Ewing LJ, Campbell SB. Diagnostic status and symptomatic behavior of hard-to-manage preschool children in middle childhood and early adolescence. J Clin Child Psychol 1999; 28:44-57.
- 173. Lee SS, Lahey BB, Owens EB, Hinshaw S. Few Preschool Boys and Girls with ADHD are Well-Adjusted During Adolescence *J Abnorm Child Psychol* 2008; 36:373-383.
- 174. Massetti GM, Lahey BB, Pelham WE, Loney J, Ehrhardt A, Lee SS, Kipp H. Academic achievement over 8 years among children who met modified criteria for attention-deficit/hyperactivity disorder at 4–6 years of age *J Abnorm Child Psychol* 2008; 36:399-410.
- 175. DuPaul GJ, McGoey KE, Eckert TL, VanBrakle J. Preschool children with attention-deficit/hyperactivity disorder: impairments in behavioral, social, and school functioning. J Am Acad Child Adolesc Psychiatry 2001; 40:508-15.
- 176. Rappley MD, Mullan PB, Alvarez FJ, Eneli IU, Wang J, Gardiner JC. Diagnosis of attention-deficit/hyperactivity disorder and use of psychotropic medication in very young children. *Arch Pediatr Adolesc Med* 1999; 153:1039-45.
- 177. Tannock R. Attention deficit hyperactivity disorder: advances in cognitive, neurobiological, and genetic research. *J Child Psychol Psychiatry* 1998; 39:65-99.
- 178. Hinshaw SP. Academic underachievement, attention deficits, and aggression: comorbidity and implications for intervention. *J Consult Clin Psychol* 1992; 60:893-903.
- 179. McGee R, Prior M, Williams S, Smart D, Sanson A. The long-term significance of teacher-rated hyperactivity and reading ability in childhood: Findings from two longitudinal studies. *J Child Psychol Psychiatr* 2002; 43:1004-1017.
- 180. Barkley RA. Attention-deficit hyperactivity disorder: A handbook for diagnosis and treatment. New York: The Guilford Press, 1998.
- 181. Barry TD, Lyman RD, Klinger LG. Academic underachievement and attentiondeficit/hyperactivity disorder: The negative impact on symptom severity on school performance. J Sch Psychol 2002; 40:259-283.
- 182. DuPaul GJ, Stoner G. ADHD in the schools: Assessment and intervention strategies. New York: The Guilford Press, 1994.
- 183. Faraone SV, Biederman J, Lehman BK, Spencer T, Norman D, Seidman LJ, Kraus I, Perrin J, Chen WJ, Tsuang MT. Intellectual performance and school failure in children with attention deficit hyperactivity disorder and in their siblings. *J Abnorm Psychol* 1993; 102:616-23.
- 184. Carlson CL, Lahey BB, Neeper R. Direct assessment of the cognitive correlates of attention deficit disorders with and without hyperactivity. *J Psychopathol Behav Assess* 1986; 8:69-86.
- 185. Marshall RM, Hynd GW, Handwerk MJ, Hall J. Academic underachievement in ADHD subtype. *J Learn Disabil* 1997; 30:635-642.
- 186. Hoza B. Peer functioning in children with ADHD. Ambul Pediatr 2007; 7:101-6.
- 187. Landau S, Milich R. Social communication patterns of attention-deficitdisordered boys. J Abnorm Child Psychol 1988; 16:69-81.
- 188. Bagwell CL, Molina BS, Pelham WE, Jr., Hoza B. Attention-deficit hyperactivity disorder and problems in peer relations: predictions from childhood to adolescence. J Am Acad Child Adolesc Psychiatry 2001; 40:1285-92.
- 189. Whalen CK, Henker B. The social profile of attention-deficit hyperactivity disorder: five fundamental facets. *Child Adolesc Psychiatr Clin N Am* 1992; 1:395-410.
- 190. Kitchens SA, Rosen LA, Braaten EB. Differences in anger, aggression, depression, and anxiety between ADHD and non-ADHD children. *J Attention Disord* 1999; 3:77-83.
- 191. DiScala C, Lescohier I, Barthel M, Li G. Injuries to children with attention deficit hyperactivity disorder. *Pediatrics* 1998; 102:1415-21.

- 192. Holtkamp K, Konrad K, Muller B, Heussen N, Herpertz S, Herpertz-Dahlmann B, Hebebrand J. Overweight and obesity in children with Attention-Deficit/Hyperactivity Disorder. *Int J Obes Relat Metab Disord* 2004; 28:685-9.
- 193. Agranat-Meged AN, Deitcher C, Goldzweig G, Leibenson L, Stein M, Galili-Weisstub E. Childhood obesity and attention deficit/hyperactivity disorder: a newly described comorbidity in obese hospitalized children. *Int J Eat Disord* 2005; 37:357-9.
- 194. Barkley RA. Major life activity and health outcomes associated with attentiondeficit/hyperactivity disorder. *J Clin Psychiatry* 2002; 63 Suppl 12:10-5.
- 195. Heiligenstein E, Guenther G, Levy A, Savino F, Fulwiler J. Psychological and academic functioning in college students with attention deficit hyperactivity disorder. *J Am Coll Health* 1999; 47:181-5.
- 196. Kessler RC, Adler L, Ames M, Barkley RA, Birnbaum H, Greenberg P, Johnston JA, Spencer T, Ustun TB. The prevalence and effects of adult attention deficit/hyperactivity disorder on work performance in a nationally representative sample of workers. *J Occup Environ Med* 2005; 47:565-72.
- 197. Biederman J, Faraone SV. The effects of attention-deficit/hyperactivity disorder on employment and household income. *Medscape General Medicine* 2006; 8:12.
- 198. Barkley RA, Fischer M, Smallish L, Fletcher K. Young adult outcome of hyperactive children: adaptive functioning in major life activities. J Am Acad Child Adolesc Psychiatry 2006; 45:192-202.
- 199. de Graaf R, Kessler RC, Fayyad J, ten Have M, Alonso J, Angermeyer M, Borges G, Demyttenaere K, Gasquet I, de Girolamo G, Haro JM, Jin R, Karam EG, Ormel J, Posada-Villa J. The prevalence and effects of Adult Attention-Deficit/hyperactivity Disorder (ADHD) on the performance of workers: Results from the WHO World Mental Health Survey Initiative. *Occup Environ Med* 2008:oem.2007.038448.
- 200. Toner M, O'Donoghue T, Houghton S. Living in chaos and striving for control: How adults with Attention deficit hyperactivity disorder deal with their disorder. *Int J Disabil Dev Educ* 2006; 53:247-261.
- 201. Jerome L, Segal A, Habinski L. What we know abouy ADHD and driving risk: Literatute review, meta-anlysis and critique. J Can Acad Child Adolesc Psychiatry 2006; 15:105-125.
- 202. Molina BS, Flory K, Hinshaw SP, Greiner AR, Arnold LE, Swanson JM, Hechtman L, Jensen PS, Vitiello B, Hoza B, Pelham WE, Elliott GR, Wells KC, Abikoff HB, Gibbons RD, Marcus S, Conners CK, Epstein JN, Greenhill LL, March JS, Newcorn JH, Severe JB, Wigal T. Delinquent behaviour and emerging substance use in the MTA at 36 months: Prevalence, course and treatment effects. J Am Acad Child Adolesc Psychiatry 2007; 46:1027-1039.
- 203. Milberger S, Biederman J, Faraone SV, Chen L, Jones J. ADHD is associated with early initiation of cigarette smoking in children and adolescents. *J Am Acad Child Adolesc Psychiatry* 1997; 36:37-44.
- 204. Molina BS, Pelham WE. Substance use, substance abuse, and LD among adolescents with a childhood history of ADHD. *J Learn Disabil* 2001; 34:333-42, 351.
- 205. Biederman J, Wilens T, Mick E, Faraone SV, Weber W, Curtis S, Thornell A, Pfister K, Jetton JG, Soriano J. Is ADHD a risk factor for psychoactive substance use disorders? Findings from a four-year prospective follow-up study. J Am Acad Child Adolesc Psychiatry 1997; 36:21-9.
- 206. Biederman J, Monuteaux MC, Mick E, Spencer T, Wilens TE, Silva JM, Snyder LE, Faraone SV. Young adult outcome of attention deficit hyperactivity disorder: a controlled 10-year follow-up study. *Psychol Med* 2006; 36:167-79.
- 207. Elkins IJ, McGue M, Iacono WG. Prospective effects of attentiondeficit/hyperactivity disorder, conduct disorder, and sex on adolescent substance use and abuse. *Arch Gen Psychiatry* 2007; 64:1145-52.

- 208. Molina BS, Pelham WE, Jr. Childhood predictors of adolescent substance use in a longitudinal study of children with ADHD. *J Abnorm Psychol* 2003; 112:497-507.
- 209. Barkley RA, Fischer M, Smallish L, Fletcher K. Young adult follow-up of hyperactive children: antisocial activities and drug use. *J Child Psychol Psychiatry* 2004; 45:195-211.
- Milberger S, Biederman J, Faraone SV, Wilens T, Chu MP. Associations between ADHD and psychoactive substance use disorders. Findings from a longitudinal study of high-risk siblings of ADHD children. *Am J Addict* 1997; 6:318-29.
- 211. Babinski LM, Hartsough CS, Lambert NM. Childhood conduct problems, hyperactivity-impulsivity, and inattention as predictors of adult criminal activity. *J Child Psychol Psychiatry* 1999; 40:347-55.
- 212. Pratt TC, FT, Blevins K, Daigle L, Unnever J. The relationship of attention deficit hyperactivity disorder to crime and delinquency: a meta-analysis. *Int J Police Sci Management* 2002; 4:344-360.
- 213. Brassett-Grundy A, Butler N. Prevalence and adult outcomes of Attention-Deficit/Hyperactivity Disorder: evidence from a 30-year prospective longitudinal study. BG occasional paper: no. 2, London: Bedford Group for Lifecourse and Statistical Studies, Institute of Education, 2004.
- 214. Brassett-Grundy A, Butler N. Attention-Deficit/Hyperactivity Disorder: an overview and review of the literature relating to the correlates and lifecourse outcomes for males and females. BG occasional paper: no. 1, London: Bedford Group for Lifecourse and Statistical Studies, Institute of Education, 2004.
- 215. Altfas JR. Prevalence of attention deficit/hyperactivity disorder among adults in obesity treatment. *BMC Psychiatry* 2002; 2:9.
- 216. Fleming JP, Levy LD, Levitan RD. Symptoms of attention deficit hyperactivity disorder in severely obese women. *Eat Weight Disord* 2005; 10:e10-3.
- 217. Levy LD, Fleming JP, Klar D. Treatment of refractory obesity in severely obese adults following management of newly diagnosed attention deficit hyperactivity disorder. *Int J Obes (Lond)* 2009; 33:326-34.
- 218. Waring ME, Lapane KL. Overweight in children and adolescents in relation to attention-deficit/hyperactivity disorder: results from a national sample. *Pediatrics* 2008; 122:e1-6.
- Mustillo S, Worthman C, Erkanli A, Keeler G, Angold A, Costello EJ. Obesity and psychiatric disorder: developmental trajectories. *Pediatrics* 2003; 111:851-9.
- 220. Pagoto SL, Curtin C, Lemon SC, Bandini LG, Schneider KL, Bodenlos JS, Ma Y. Association between adult attention deficit/hyperactivity disorder and obesity in the US population. *Obesity (Silver Spring)* 2009; 17:539-44.
- 221. Cortese S, Angriman M, Maffeis C, Isnard P, Konofal E, Lecendreux M, Purper-Ouakil D, Vincenzi B, Bernardina BD, Mouren MC. Attention-deficit/hyperactivity disorder (ADHD) and obesity: a systematic review of the literature. *Crit Rev Food Sci Nutr* 2008; 48:524-37.
- 222. Fischer M, Barkley R. Young adult outcomes of children with hyperactivity: Leisure, financial, and social activities. *Int J Disabil Dev Educ* 2006; 53:229-245.
- 223. Murphy KR, Barkley RA, Bush T. Young adults with attention deficit hyperactivity disorder: subtype differences in comorbidity, educational, and clinical history. *J Nerv Ment Dis* 2002; 190:147-57.
- 224. Eakin L, Minde K, Hechtman L, Ochs E, Krane E, Bouffard R, Greenfield B, Looper K. The marital and family functioning of adults with ADHD and their spouses. J Attention Disord 2004; 8:1-10.
- 225. Flory K, Molina BS, Pelham WE, Jr., Gnagy E, Smith B. Childhood ADHD predicts risky sexual behavior in young adulthood. *J Clin Child Adolesc Psychol* 2006; 35:571-7.
- 226. Davidson MA. Literature review: ADHD in adults: A review of the literature. J Attention Disord 2008; 11:628-641.
- 227. Secnik K, Swensen A, Lage MJ. Comorbidities and costs of adult patients diagnosed with attention-deficit hyperactivity disorder. *Pharmacoeconomics* 2005; 23:93-102.
- 228. Mikami AY, Hinshaw SP, Patterson KA, Lee JC. Eating pathology among adolescent girls with attention-deficit/hyperactivity disorder. J Abnorm Psychol 2008; 117:225-35.
- 229. Biederman J, Ball SW, Monuteaux MC, Surman CB, Johnson JL, Zeitlin S. Are girls with ADHD at risk for eating disorders? Results from a controlled, five-year prospective study. *J Dev Behav Pediatr* 2007; 28:302-7.
- 230. Barkley R, Murphy KR, Fischer M. ADHD in adults: what the science says. New York: Guildford Press, 2008.
- 231. Ramsay JR, Rostain AL. Adult ADHD research: current status and future directions. *J Attention Disord* 2008; 11:624-7.
- 232. Angold A, Messer SC, Stangl D, Farmer EM, Costello EJ, Burns BJ. Perceived parental burden and service use for child and adolescent psychiatric disorders. *Am J Public Health* 1998; 88:75-80.
- 233. Bussing R, Zima BT, Gary FA, Garvan CW. Barriers to detection, help-seeking, and service use for children with ADHD symptoms. *J Behav Health Serv Res* 2003; 30:176-89.
- 234. Sawyer MG, Rey JM, Arney FM, Whitham JN, Clark JJ, Baghurst PA. Use of health and school-based services in Australia by young people with attention-deficit/hyperactivity disorder. *J Am Acad Child Adolesc Psychiatry* 2004; 43:1355-63.
- 235. Sayal K, Goodman R, Ford T. Barriers to the identification of children with attention deficit/hyperactivity disorder. *J Child Psychol Psychiatr* 2006; 47:744-750.
- 236. Cassidy LJ, Jellinek MS. Approaches to recognition and management of childhood psychiatric disorders in pediatric primary care. *Pediatr Clin North Am* 1998; 45:1037-52.
- 237. Simkin DR. Adolescent substance use disorders and comorbidity. *Pediatr Clin North Am* 2002; 49:463-77.
- 238. Asherson P, Chen W, Craddock B, Taylor E. Adult attention-deficit hyperactivity disorder: recognition and treatment in general adult psychiatry. *Br J Psychiatry* 2007; 190:4-5.
- 239. Feifel D, MacDonald K. Attention-deficit/hyperactivity disorder in adults: recognition and diagnosis of this often-overlooked condition. *Postgrad Med* 2008; 120:39-47.
- 240. Quinn PO. Attention-deficit/hyperactivity disorder and its comorbidities in women and girls: an evolving picture. *Curr Psychiatry Rep* 2008; 10:419-23.
- 241. American Academy of Child and Adolescent Psychiatry (AACAP). Practice parameter for the assessment and treatment of children and adolescents with attention-deficit/hyperactivity disorder Washington DC, 2007.
- 242. American Academy of Pediatrics. Committee on Quality Improvement. Subcommittee on Attention-Deficit/Hyperactivity Disorder. Clinical practice guideline: diagnosis and evaluation of the child with attentiondeficit/hyperactivity disorder. American Academy of Pediatrics. *Pediatrics* 2000; 105:1158-70.
- 243. American Academy of Pediatrics CoQI, Subcommittee on Attention-Deficit/Hyperactivity Disorder. Clinical practice guideline: treatment of the school-aged child with attention-deficit/hyperactivity disorder. *Pediatrics* 2001; 108:1033-44.
- 244. National Institute for Health and Clinical Excellence (NICE). Attention deficit hyperactivity disorder. Diagnosis and management of ADHD in children, young people and adults. , 2008.

- 245. Weiler MD, Bellinger DK, Simmons EK, Rappaport LK, Urion DK, Mitchell WJ, Bassett NJ, Burke PJ, Marmor J, Waber D. Reliability and validity of a DSM-IV based ADHD screener. *Child Neuropsychol* 2000; 6:3-23.
- 246. Kessler RC, Adler L, Ames M, Demler O, Faraone S, Hiripi E, Howes MJ, Jin R, Secnik K, Spencer T, Ustun TB, Walters EE. The World Health Organization Adult ADHD Self-Report Scale (ASRS): a short screening scale for use in the general population. *Psychol Med* 2005; 35:245-56.
- 247. Shaw K, Wagner I, Eastwood H, Mitchell G. A qualitative study of Australian GPs' attitudes and practices in the diagnosis and management of attentiondeficit/hyperactivity disorder (ADHD). *Fam Pract* 2003; 20:129-34.
- 248. Shaw KA, Mitchell GK, Wagner IJ, Eastwood HL. Attitudes and practices of general practitioners in the diagnosis and management of attentiondeficit/hyperactivity disorder. *J Paediatr Child Health* 2002; 38:481-6.
- 249. Barkley RA. What may be in store for DSM-V. The ADHD Report 2007; 15:1-7.
- 250. Demanez L. Differential diagnosis of attention and auditory processing disorders, In: McBurnett K and Pfiffner L, eds. Attention deficit hyperactivity disorder: Concepts, controversies, new directions. New York: Informa Healthcare, 2008:1-8.
- 251. Rasmussen ER, Neuman RJ, Heath AC, Levy F, Hay DA, Todd RD. Familial clustering of latent class and DSM-IV defined attention-deficit/hyperactivity disorder (ADHD) subtypes. *J Child Psychol Psychiatry* 2004; 45:589-98.
- 252. Todd RD, Joyner CA, Heath AC, Neuman RJ, Reich W. Reliability and stability of a semistructured DSM-IV interview designed for family studies. *J Am Acad Child Adolesc Psychiatry* 2003; 42:1460-8.
- 253. Barkley RA, Murphy KR. Attention deficit hyperactivity disorder: A clinical workbook (3rd ed.). New York: The Guilford Press, 2006.
- 254. Biederman J, Monuteaux MC, Kendrick E, Klein KL, Faraone SV. The CBCL as a screen for psychiatric comorbidity in paediatric patients with ADHD. *Arch Dis Child* 2005; 90:1010-5.
- 255. Collett BR, Ohan JL, Myers KM. Ten-year review of rating scales. V: scales assessing attention-deficit/hyperactivity disorder. J Am Acad Child Adolesc Psychiatry 2003; 42:1015-37.
- 256. Dreyer BP. The diagnosis and management of Attention-Deficit/Hyperactivity Disorder in preschool children: The state of our knowledge and practice. *Curr Probl Pediatr Adolesc Health Care* 2006; 36:6-30.
- 257. Gleason MM, Egger HL, Emslie GJ, Greenhill LL, Kowatch RA, Lieberman AF, Luby JL, Owens J, Scahill LD, Scheeringa MS, Stafford B, Wise B, Zeanah CH. Psychopharmacological treatment for very young children: contexts and guidelines. J Am Acad Child Adolesc Psychiatry 2007; 46:1532-72.
- 258. Zeanah C, Boris N, Heller S, Hinshaw-Fuselier S, Larrieu J, Lewis M, Palomino R, Rovaris M, Valliere J. Relationship assessment in infant mental health. *Infant Mental Health J* 1997; 18:182-197.
- 259. Smidts DP, Oosterlaan J. How common are symptoms of ADHD in typically developing preschoolers? A study on prevalence rates and prenatal/demographic risk factors. *Cortex* 2007; 43:710-7.
- 260. Boris NW, Fueyo M, Zeanah CH. The clinical assessment of attachment in children under five. J Am Acad Child Adolesc Psychiatry 1997; 36:291-3.
- 261. Conners CK, Erhardt D, Sparrow E. Conners' adults rating scales. North Tonawada NY: Multi-Health Systems, 1999.
- 262. Wender P, Reimherr F, Wood D. Attention deficit disorder ('Minimal Brain Dysfunction') in adults: A replication study of diagnosis and drug treatment. *Arch Gen Psychiatry* 1981; 38:449–456.
- 263. Ward M, Wender P, Reimherr F. The Wender Utah Rating Scale: An aid in the retrospective diagnosis of childhood Attention Deficit Hyperactivity Disorder. *Am J Psychiatry* 1993; 150:885-890.
- 264. Brown T. Brown Attention-Deficit Disorder Scales Manual. San Antonio: The Psychological Corporation, 1996.

- 265. Copeland T. Copeland Symptom Checklist for Adult Attention Deficit Disorders. Atlanta, GA: Southeastern Psychological Institute, 1989. 1989.
- 266. DuPaul GJ, Power TJ, McGoey KE, Ikeda MJ, Anastopoulos AD. Reliability and validity of parent and teacher ratings of attention-deficit/ hyperactivity disorder symptoms. *J Psychoeduc Assess* 1998; 16:55-68.
- 267. Kooij JJ, Buitelaar JK, van den Oord EJ, Furer JW, Rijnders CA, Hodiamont PP. Internal and external validity of attention-deficit hyperactivity disorder in a population-based sample of adults. *Psychol Med* 2005; 35:817-27.
- 268. Biederman J, Petty C, Fried R, Doyle A, Mick E, Aleardi M, Monuteaux M, Seidman L, Spencer T, Faneuil A, Holmes L, Faraone S. Utility of an abbreviated questionnaire to identify individuals with ADHD at risk for functional impairments. *J Psychiatr Res* 2008; 42:304-410.
- 269. Biederman J, Petty CR, Fried R, Fontanella J, Doyle AE, Seidman LJ, Faraone SV. Can self-reported Behavioral scales assess executive function deficits? A controlled study of adults with ADHD. J Nerv Ment Dis 2007; 195:240-246.
- 270. Sattler JM. Assessment of children. Behavioural and clinical applications (4th Ed.). San Diego, 2002.
- 271. Cohen NJ, Menna R, Vallance DD, Barwick MA, Im N, Horodezky NB. Language, social cognitive processing, and behavioral characteristics of psychiatrically disturbed children with previously identified and unsuspected language impairments. *J Child Psychol Psychiatry* 1998; 39:853-64.
- 272. McGrath LM, Hutaff-Lee C, Scott A, Boada R, Shriberg LD, Pennington BF. Children with comorbid speech sound disorder and specific language impairment are at increased risk for attention-deficit/hyperactivity disorder. J Abnorm Child Psychol 2008; 36:151-63.
- 273. Mangeot SD, Miller LJ, McIntosh DN, McGrath-Clarke J, Simon J, Hagerman RJ, Goldson E. Sensory modulation dysfunction in children with attention-deficithyperactivity disorder. *Dev Med Child Neurol* 2001; 43:399-406.
- 274. Schaaf RC, Miller LJ. Occupational therapy using a sensory integrative approach for children with developmental disabilities. *Ment Retard Dev Disabil Res Rev* 2005; 11:143-8.
- 275. Chu S, Reynolds F. Occupational therapy for children with attention deficit hyperactivity disorder (ADHD), part 2: a multicentre evaluation of an assessment and treatment package. *British Journal of Occupational Therapy* 2007; 70:439-48.
- 276. Miller LJ, Anzalone ME, Lane SJ, Cermak SA, Osten ET. Concept evolution in sensory integration: a proposed nosology for diagnosis. *Am J Occup Ther* 2007; 61:135-40.
- 277. Dunn W, Bennett D. Patterns of sensory processing in children with attention deficit hyperactivity disorder. *Occup Ther J Res* 2002; 22:4-15.
- 278. Chermak GD, Somers EK, Seikel JA. Behavioral signs of central auditory processing disorder and attention deficit hyperactivity disorder. *J Am Acad Audiol* 1998; 9:78-84; quiz 85.
- 279. Cook JR, Mausbach T, Burd L, Gascon GG, Slotnick HB, Patterson B, Johnson RD, Hankey B, Reynolds BW. A preliminary study of the relationship between central auditory processing disorder and attention deficit disorder. *J Psychiatry Neurosci* 1993; 18:130-7.
- 280. Bloom J, Hynd GW. Dysfunctions of attention, learning, and central auditory processing: What's the difference?, In: McBurnett K and Pfiffner L, eds. Attention deficit hyperactivity disorder: Concepts, controversies, new directions. New York: Informa Healthcare, 2008:1-8.
- 281. Lecavalier L. Behavioral and emotional problems in young people with pervasive developmental disorders: relative prevalence, effects of subject characteristics, and empirical classification. *J Autism Dev Disord* 2006; 36:1101-14.
- 282. Conklin HM, Khan RB, Reddick WE, Helton S, Brown R, Howard SC, Bonner M, Christensen R, Wu S, Xiong X, Mulhern RK. Acute neurocognitive response to

methylphenidate among survivors of childhood cancer: a randomized, doubleblind, cross-over trial. *J Pediatr Psychol* 2007; 32:1127-39.

- 283. Bakker K, Waugh M-C. Stimulant use in paediatric acquired brain injury: Evaluation of a protocol. *Brain Impairment* 2000; 1:29-36.
- 284. Lahey BB, Pelham WE, Loney J, Kipp H, Ehrhardt A, Lee SS, Willcutt EG, Hartung CM, Chronis A, Massetti G. Three-year predictive validity of DSM-IV attention deficit hyperactivity disorder in children diagnosed at 4-6 years of age. *Am J Psychiatr* 2004; 161:2014-20.
- 285. Willcutt EG, Hartung CM, Lahey BB, Loney J, Pelham WE. Utility of behavior ratings by examiners during assessments of preschool children with attention-deficit/hyperactivity disorder. *J Abnorm Child Psychol* 1999; 27:463-72.
- 286. Fabiano GA, Pelham WE, Jr., Waschbusch DA, Gnagy EM, Lahey BB, Chronis AM, Onyango AN, Kipp H, Lopez-Williams A, Burrows-Maclean L. A practical measure of impairment: psychometric properties of the impairment rating scale in samples of children with attention deficit hyperactivity disorder and two school-based samples. J Clin Child Adolesc Psychol 2006; 35:369-85.
- 287. Gathje RA, Lewandowski LJ, Gordon M. The role of ipairment in the diagnosis of ADHD. J Attention Disord 2008; 11:529-537.
- 288. Mota VL, Schachar RJ. Reformulating attention-deficit/hyperactivity disorder according to signal detection theory. *J Am Acad Child Adolesc Psychiatry* 2000; 39:1144-51.
- 289. Winters NC, Collett BR, Myers KM. Ten-year review of rating scales, VII: scales assessing functional impairment. *J Am Acad Child Adolesc Psychiatry* 2005; 44:309-38; discussion 339-42.
- 290. Gowers SG, Harrington RC, Whitton A, Beevor A, Lelliott P, Jezzard R, Wing JK. Health of the Nation Outcome Scales for Children and Adolescents (HoNOSCA). Glossary for HoNOSCA score sheet. *Br J Psychiatry* 1999; 174:428-31.
- 291. Pirkis JE, Burgess PM, Kirk PK, Dodson S, Coombs TJ, Williamson MK. A review of the psychometric properties of the Health of the Nation Outcome Scales (HoNOS) family of measures. *Health Qual Life Outcomes* 2005; 3:76.
- 292. Schorre BE, Vandvik IH. Global assessment of psychosocial functioning in child and adolescent psychiatry. A review of three unidimensional scales (CGAS, GAF, GAPD). *Eur Child Adolesc Psychiatry* 2004; 13:273-86.
- 293. Steinhausen HC, Metzke CW. Global measures of impairment in children and adolescents: results from a Swiss community survey. *Aust N Z J Psychiatry* 2001; 35:282-6.
- 294. Loney J, Ledolter J, Kramer JR, Volpe RJ. Retrospective ratings of ADHD symptoms made at young adulthood by clinic-referred boys with ADHD-Related problems, their brothers without ADHD, and control participants. *Psychol Assess* 2007; 19:269-280.
- 295. Mannuzza S, Klein RG, Klein DF, Bessler A, Shrout P. Accuracy of adult recall of childhood attention deficit hyperactivity disorder. *Am J Psychiatry* 2002; 159:1882-8.
- 296. Barkley RA, Fischer M, Smallish L, Fletcher K. The persistence of attentiondeficit/hyperactivity disorder into young adulthood as a function of reporting source and definition of disorder. *J Abnorm Psychol* 2002; 111:279-89.
- 297. Murphy P, Schachar R. Use of self-ratings in the assessment of symtoms of attention deficit hyperactivity disorder in adults. *Am J Psychiatry* 2000; 157:1156-1159.
- 298. Murray DW, Kollins SH, Hardy KK, Abikoff HB, Swanson JM, Cunningham C, Vitiello B, Riddle MA, Davies M, Greenhill LL, McCracken JT, McGough JJ, Posner K, Skrobala AM, Wigal T, Wigal SB, Ghuman JK, Chuang SZ. Parent versus teacher ratings of Attention-Deficit/Hyperactivity Disorder symptoms in the preschoolers with Attention-Deficit/Hyperactivity Disorder Treatment Study (PATS). J Child Adolesc Psychopharmacol 2007; 17:605-20.

- 299. Amador-Campos JA, Forns-Santacana M, Guardia-Olmos J, Pero-Cebollero M. DSM-IV attention deficit hyperactivity disorder symptoms: Agreement between informants in prevalence and factor structure at different ages. J Psychopathol Behav Assess 2006; 28:23-31.
- 300. Gomez R. Australian parent and teacher ratings of the DSM-IV ADHD symptoms: Differential symptom functioning and parent-teacher agreement and differences. J Attention Disord 2007; 11:17-26.
- 301. Mitsis EM, McKay KE, Schulz KP, Newcorn JH, Halperin JM. Parent-teacher concordance for DSM-IV attention-deficit/hyperactivity disorder in a clinic-referred sample. *J Am Acad Child Adolesc Psychiatry* 2000; 39:308-13.
- 302. Power TJ, Andrews TJ, Eiraldi RB, Doherty BJ, Ikeda MJ, DuPaul GJ, Landau S. Evaluating Attention Deficit Hyperactivity Disorder using multiple informants: The incremental utility of combining teacher with parent reports. *Psychol Assess* 1998; 10:250-260.
- 303. Antrop I, Roeyers H, Oosterlaan J, Van Oost P. Agreement between parent and teacher ratings of disruptive behavior disorders in children with clinically diagnosed ADHD. *J Psychopathol Behav Assess* 2002; 24:67-73.
- 304. Sherman DK, McGue MK, Iacono WG. Twin concordance for attention deficit hyperactivity disorder: a comparison of teachers' and mothers' reports. *Am J Psychiatr* 1997; 154:532-5.
- 305. Sprafkin J, Gadow KD, Nolan EE. The utility of the DSM-IV-Referenced Screening Instrument for attention deficit/hyperactivity disorder. *J Emot Behav Disord* 2001; 9:182-191.
- 306. Wolraich ML, Lambert EW, Bickman L, Simmons T, Doffing MA, Worley KA. Assessing the impact of parent and teacher agreement on diagnosing attention-deficit hyperactivity disorder. *J Dev Behav Pediatr* 2004; 25:41-47.
- 307. Tripp G, Schaughency EA, Clarke B. Parent and teacher rating scales in the evaluation of attention-deficit hyperactivity disorder: Contribution to diagnosis and differential diagnosis in clinically referred children. *J Dev Behav Pediatr* 2006; 27:209-218.
- 308. van der Oord S, Prins PJM, Oosterlaan J, Emmelkamp PMG. The association between parenting stress, depressed mood and informant agreement in ADHD and ODD. *Behav Res Ther* 2006; 44:1585-95.
- 309. Rohde LA, Biederman J, Knijnik MP, Ketzer C, Chachamovich E, Vieira GM, al. e. Exploring different information sources for DSM–IV ADHD diagnoses in Brazilian adolescents. J Attention Disord 1999; 3:91–96.
- 310. Hope TL, Adams C, Reynolds L, Powers D, Perez RA, Kelley ML. Parent vs. self-report: Contributions toward diagnosis of adolescent psychopathology. J Psychopathol Behav Assess 1999; 21:349-363.
- 311. Cantwell DP, Lewinsohn PM, Rohde P, Seeley JR. Correspondence between adolescent report and parent report of psychiatric diagnostic data. J Am Acad Child Adolesc Psychiatry 1997; 36:610-9.
- 312. Young S. The YAQ-S and YAQ-I: the development of self and informant questionnaires reporting on current adult ADHD symptomatology, comorbid and associated problems. *Pers Indiv Differ* 2004; 36:1211-1223.
- 313. Kooij JJS, Boonstra AM, Swinkels SHN, Bekker EM, de Noord I, Buitelaar JK. Reliability, validity, and utility of instruments for self-report and informant report concerning symptoms of ADHD in adult patients. *J Attention Disord* 2008; 11:445-458.
- 314. Zucker M, Morris MK, Ingram SM, Morris RD, Bakeman R. Concordance of selfand informant ratings of adults' current and childhood attentiondeficit/hyperactivity disorder symptoms. *Psychol Assess* 2002; 14:379-389.
- 315. Downey KK, Stelson FW, Pomerleau OF, Giordani B. Adult attention deficit hyperactivity disorder: psychological test profiles in a clinical population. J Nerv Ment Dis 1997; 185:32-8.

- 316. Faraone SV, Monuteaux MC, Biederman J, Cohan SL, Mick E. Does parental ADHD bias maternal reports of ADHD symptoms in children? *J Consult Clin Psychol* 2003; 71:168-175.
- 317. Chilcoat HD, Breslau N. Does psychiatric history bias mothers' reports? An application of a new analytic approach. *J Am Acad Child Adolesc Psychiatry* 1997; 36:971-9.
- 318. Chi TC, Hinshaw SP. Mother-child relationships of children with ADHD: the role of maternal depressive symptoms and depression-related distortions. *J Abnorm Child Psychol* 2002; 30:387-400.
- 319. Baumann BL, Pelham WE, Lang AR, Jacob RG, Blumenthal JD. The impact of maternal depressive symptomatology on ratings of children with ADHD and child confederates. *J Emot Behav Disord* 2004; 12:90-98.
- 320. Conners CK, Sitarenios G, Parker JD, Epstein JN. Revision and restandardization of the Conners Teacher Rating Scale (CTRS-R): factor structure, reliability, and criterion validity. *J Abnorm Child Psychol* 1998; 26:279-91.
- 321. Conners CK, Sitarenios G, Parker JD, Epstein JN. The revised Conners' Parent Rating Scale (CPRS-R): factor structure, reliability, and criterion validity. *J Abnorm Child Psychol* 1998; 26:257-68.
- 322. Achenbach TM, Becker A, Dopfner M, Heiervang E, Roessner V, Steinhausen HC, Rothenberger A. Multicultural assessment of child and adolescent psychopathology with ASEBA and SDQ instruments: research findings, applications, and future directions. J Child Psychol Psychiatry 2008; 49:251-75.
- 323. Crijnen AA, Achenbach TM, Verhulst FC. Problems reported by parents of children in multiple cultures: the Child Behavior Checklist syndrome constructs. *Am J Psychiatry* 1999; 156:569-74.
- 324. Brewis AA, Meyer MC, Schmidt KL. Does school, compared to home, provide a unique adaptive context for children's ADHD-associated behaviors? A cross-cultural test. *Cross Cult Res* 2002; 36:303-320.
- 325. Luk ESL, Leung PW, Ho T-H. Cross-cultural / ethnic aspects of childhood hyperactivity In: Sandberg S, ed. Hyperactivity and attantion disorders of childhood (2nd Ed): Cambridge University Press, 2002:64-98.
- 326. Baydala L, Sherman J, Rasmussen C, Wikman E, Janzen H. ADHD characteristics in Canadian Aboriginal children *J Attention Disord* 2006; 9:642-647.
- 327. Calver J, Preen D, Bulsara M, Sanfilippo F. Stimulant prescribing for the treatment of ADHD in Western Australia: socioeconomic and remoteness differences. *Med J Aust* 2007; 186:124-127.
- 328. Australian Institute of Criminology. Australian crime: facts and figures. Canberra: Australian Institute of Criminology., 2005.
- 329. Isaacs D. Attention-deficit/hyperactivity disorder: are we medicating for social disadvantage? (For). *J Paediatr Child Health* 2006; 42:544-7.
- 330. Boyle MH, Offord DR, Hofmann HG, Catlin GP, Byles JA, Cadman DT, Crawford JW, Links PS, Rae-Grant NI, Szatmari P. Ontario Child Health Study. I. Methodology. Arch Gen Psychiatry 1987; 44:826-31.
- 331. Pelham WE, Jr., Fabiano GA, Massetti GM. Evidence-based assessment of attention deficit hyperactivity disorder in children and adolescents. J Clin Child Adolesc Psychol 2005; 34:449-76.
- 332. Forbes GB. Clinical utility of the Test of Variables of Attention (TOVA) in the diagnosis of attention-deficit/hyperactivity disorder. *J Clin Psychol* 1998; 54:461-76.
- 333. Preston AS, Fennell EB, Bussing R. Utility of a CPT in diagnosing ADHD among a representative sample of high-risk children: A cautionary study: Child Neuropsychology Vol 11(5) Oct 2005, 459-469, 2005.
- 334. Schatz AM, Ballantyne AO, Trauner DA. Sensitivity and specificity of a computerized test of attention in the diagnosis of Attention-Deficit/Hyperactivity Disorder. *Assessment* 2001; 8:357-65.

- 335. Mayes SD, Calhoun SL. The Gordon Diagnostic System and WISC-III Freedom from Distractibility Index: validity in identifying clinic-referred children with and without ADHD. *Psychol Rep* 2002; 91:575-87.
- 336. Mayes SD, Calhoun SL, Crowell EW. Clinical validity and interpretation of the Gordon Diagnostic System in ADHD assessments. *Child Neuropsychol* 2001; 7:32-41.
- 337. Rielly NE, Cunningham CE, Richards JE, Elbard H, Mahoney WJ. Detecting attention deficit hyperactivity disorder in a communications clinic: Diagnostic utility of the Gordon Diagnostic System. J Clin Exp Neuropsychol 1999; 21:685-700.
- 338. Dewey D, Kaplan BJ, Crawford SG, Fisher GC. Predictive accuracy of the wide range assessment of memory and learning in children with attention deficit hyperactivity disorder and reading difficulties. *Developmental Neuropsychology* 2001; 19:173-89.
- 339. Romine CB, Lee D, Wolfe ME, Homack S, George C, Riccio CA. Wisconsin Card Sorting Test with children: a meta-analytic study of sensitivity and specificity. *Arch Clin Neuropsychol* 2004; 19:1027-41.
- 340. Homack S, Riccio CA. A meta-analysis of the sensitivity and specificity of the Stroop Color and Word Test with children. *Arch Clin Neuropsychol* 2004; 19:725-43.
- 341. Inoue K, Nadaoka T, Oiji A, Morioka Y, Totsuka S, Kanbayashi Y, Hukui T. Clinical evaluation of attention-deficit hyperactivity disorder by objective quantitative measures. *Child Psychiatry & Human Development* 1998; 28:179-88.
- 342. Grodzinsky GM, Barkley RA. Predictive power of frontal lobe tests in the diagnosis of attention deficit hyperactivity disorder. *Clin Neuropsychol* 1999; 13:12-21.
- 343. Solanto MV, Etefia K, Marks DJ. The utility of self-report measures and the continuous performance test in the diagnosis of ADHD in adults. *CNS Spectrums* 2004; 9:649-59.
- 344. Quinn CA. Detection of malingering in assessment of adult ADHD. Arch Clin Neuropsychol 2003; 18:379-95.
- 345. Oyler RF, Rosenhagen KM, Michal ML. Sensitivity and specificity of Keith's Auditory Continuous Performance Test. *Lang Speech Hear Serv Sch* 1998; 29:180-5.
- 346. McGee RA, Clark SE, Symons DK. Does the Conners' Continuous Performance Test aid in ADHD diagnosis? *J Abnorm Child Psychol* 2000; 28:415-24.
- 347. Youngwirth SD, Harvey EA, Gates EC, Hashim RL, Friedman-Weieneth JL. Neuropsychological abilities of preschool-aged children who display hyperactivity and/or oppositional-defiant behavior problems. *Child Neuropsychol* 2007; 13:422-43.
- 348. Doyle A, Biederman J, Siedman L, Weber W, Faraone S. Diagnostic efficiency of neuropsychological test scores for discriminating boys with and without attention deficit hyperactivity disorder. *J Consult Clin Psychol* 2000; 68:477-488.
- 349. Berlin L, Bohlin G, Nyberg L, Janols L. How well do measures of inhibition and other executive functions discriminate between children with ADHD and controls. *Child Neuropsychol* 2004; 10:1-13.
- 350. Pineda DA, Puerta IC, Aguirre DC, Garcia-Barrera MA, Kamphaus RW. The role of neuropsychologic tests in the diagnosis of attention deficit hyperactivity disorder. *Pediatric Neurology* 2007; 36:373-381.
- 351. Perugini EM, Harvey EA, Lovejoy DW, Sandstrom K, Webb AH. The predictive power of combined neuropsychological measures for attentiondeficit/hyperactivity disorder in children. *Child Neuropsychol* 2000; 6:101-14.
- 352. Katz LJ, Wood DS, Goldstein G, Auchenbach RC, Geckle M. The utility of neuropsychological tests in evaluation of Attention-Deficit/ Hyperactivity Disorder (ADHD) versus depression in adults. Assessment 1998; 5:45-52.

- 353. Walker AJ, Shores EA, Trollor JN, Lee T, Sachdev PS. Neuropsychological functioning of adults with attention deficit hyperactivity disorder. *J Clin Exp Neuropsychol* 2000; 22:115-24.
- 354. American Psychiatric Association. DSM-V externalizing disorders research planning conference. <u>http://dsm5.org/conference9.cfm</u> 2007.
- 355. Nigg JT, Blaskey LG, Huang-Pollock CL, Rappley MD. Neuropsychological executive functions and DSM-IV subtypes *J Am Acad Child Adolesc Psychiatry* 2002; 41:59–66.
- 356. Hinshaw SP, Carte ET, Sami N, Treuting JJ, Zupan BA. Preadolescent girls with attention-deficit/hyperactivity disorder: II. Neuropsychological performance in relation to subtypes and individual classification. *J Consult Clin Psychol* 2002; 70:1099–1111.
- 357. Huang-Pollock CL, Nigg JT, Halperin JM. Single dissociation findings of ADHD deficits in vigilance but not anterior or posterior attention systems. *Neuropsychology* 2006; 20:420–429.
- 358. Solanto MV, Gilbert SN, Raj A, Zhu J, Pope-Boyd S, Stepak B, Vail L, Newcorn JH. Neurocognitive functioning in AD/HD, predominantly inattentive and combined subtypes. *J Abnorm Child Psychol* 2007; 35:729-744.
- 359. Chhabildas N, Pennington BF, Willcutt EG. A comparison of the neuropsychological profiles of the DSM-IV subtypes of ADHD J Abnorm Child Psychol 2001; 29:529-540.
- 360. Smith JL, Johnstone SJ, Barry RJ. Aiding diagnosis of attentiondeficit/hyperactivity disorder and its subtypes: discriminant function analysis of event-related potential data. J Child Psychol Psychiatr 2003; 44:1067-1075.
- 361. Coolidge FL, Starkey MT, Cahill BS. Comparison of a parent-rated DSM-IV measure of attention-deficit/hyperactivity disorder and quantitative EEG parameters in an outpatient sample of children. *J Clin Neurophysiol* 2007; 24:348-51.
- 362. Monastra VJ, Lubar JF, Linden M, VanDeusen P, Green G, Wing W, Phillips A, Fenger T. Assessing attention deficit hyperactivity disorder via quantitative electroencephalography: An initial validation study. *Neuropsychology* 1999; 13:424-433.
- 363. Monastra VJ, Lubar JF, Linden M. The development of a quantitative electroencephalographic scanning process for attention deficit-hyperactivity disorder: Reliability and validity studies. *Neuropsychology* 2001; 15:136-144.
- 364. Magee CA, Clarke AR, Barry RJ, McCarthy R, Selikowitz M. Examining the diagnostic utility of EEG power measures in children with attention deficit/hyperactivity disorder. *Clinical Neurophysiology* 2005; 116:1033-1040.
- 365. Kovatchev B, Cox D, Hill R, Reeve R, Robeva R, Loboschefski T. A psychophysiological marker of attention deficit/hyperactivity disorder (ADHD)-defining the EEG consistency index. *Appl Psychophysiol Biofeedback* 2001; 26:127-40.
- 366. Quintana H, Snyder SM, Purnell W, Aponte C, Sita J. Comparison of a standard psychiatric evaluation to rating scales and EEG in the differential diagnosis of attention-deficit/hyperactivity disorder. *Psychiatry Research* 2007; 152:211-22.
- 367. Hastings JE, Barkley RA. A review of psychophysiological research with hyperkinetic children. J Abnorm Child Psychol 1978; 6:413-47.
- 368. Barry RJ, Clarke AR, Johnstone SJ. A review of electrophysiology in attentiondeficit/hyperactivity disorder: I. Qualitative and quantitative electroencephalography. *Clinical Neurophysiology* 2003; 114:171-183.
- 369. Penberthy JK, Kalbfleisch ML, Quigg M, Loboschefski T, Cox D, Runyon C, Kovatchev B. Electroencephalographic profiles of children with symptoms of attention deficit hyperactivity disorder: A review of the literature *Current Pediatric Reviews* 2006; 2:17-32.
- 370. Loo SK, Barkley RA. Clinical utility of EEG in attention deficit hyperactivity disorder. *Applied Neuropsychology* 2005; 12:64-76.

- 371. Bresnahan SM, Barry RJ. Specificity of quantitative EEG analysis in adults with attention deficit hyperactivity disorder. *Psychiatry Research* 2002; 112:133-144.
- 372. Clarke AR, Barry RJ, McCarthy R, Selikowitz M, Croft RJ. EEG differences between good and poor responders to methylphenidate in boys with the inattentive type of attention-deficit/hyperactivity disorder. *Clin Neurophysiol* 2002; 113:1191-8.
- 373. Chabot RJ, di Michele F, Prichep L, John ER. The clinical role of computerized EEG in the evaluation and treatment of learning and attention disorders in children and adolescents. *J Neuropsychiatry Clin Neurosci* 2001; 13:171-86.
- 374. Suffin SC, Emory WH. Neurometric subgroups in attentional and affective disorders and their association with pharmacotherapeutic outcome. *Clin Electroencephalogr* 1995; 26:76-83.
- 375. Chabot RJ, Serfontein G. Quantitative electroencephalographic profiles of children with attention deficit disorder. *Biol Psychiatry* 1996; 40:951-63.
- 376. Banaschewski T, Brandeis D, Heinrich H, Albrecht B, Brunner E, Rothenberger A. Association of ADHD and conduct disorder--brain electrical evidence for the existence of a distinct subtype. *J Child Psychol Psychiatr* 2003; 44:356-76.
- 377. Rothenberger A, Banaschewski T, Heinrich H, Moll GH, Schmidt MH, van't Klooster B. Comorbidity in ADHD-children: effects of coexisting conduct disorder or tic disorder on event-related brain potentials in an auditory selective-attention task. *Eur Arch Psychiatr Clin Neurosci* 2000; 250:101-10.
- 378. Yordanova J, Heinrich H, Kolev V, Rothenberger A. Increased event-related theta activity as a psychophysiological marker of comorbidity in children with tics and attention-deficit/hyperactivity disorders. *Neuroimage* 2006; 32:940-55.
- 379. Clarke AR, Barry RJ, McCarthy R, Selikowitz M. EEG analysis of children with attention-deficit/hyperactivity disorder and comorbid reading disabilities. *J Learn Disabil* 2002; 35:276-85.
- 380. Clarke AR, Barry RJ, McCarthy R, Selikowitz M. Children with attentiondeficit/hyperactivity disorder and comorbid oppositional defiant disorder: an EEG analysis. *Psychiatry Res* 2002; 111:181-90.
- 381. Boutros N, Fraenkel L, Feingold A. A four-step approach for developing diagnostic tests in psychiatry: EEG in ADHD as a test case. *J Neuropsychiatry* 2005; 17:455-64.
- 382. Bush G, Valera EM, Seidman LJ. Functional neuroimaging of attentiondeficit/hyperactivity disorder: A review and suggested future directions. *Biol Psychiatry* 2005; 57:1273-1284.
- 383. Hesslinger B, Tebartz van Elst L, Nyberg E, Dykierek P, Richter H, Berner M, Ebert D. Psychotherapy of attention deficit hyperactivity disorder in adults--a pilot study using a structured skills training program. *Eur Arch Psychiatr Clin Neurosci* 2002; 252:177-84.
- 384. Concannon PE, Tang YP. Management of attention deficit hyperactivity disorder: A parental perspective. *J Paediatr Child Health* 2005; 41:625-630.
- 385. Department of Health. Western Australian Stimulany Regulatory Scheme 2006 Annual Report: Pharmaceutical Services Bransh, Health Protection Group, Department of Health, Western Australia, 2007.
- 386. Wolraich ML, Wibbelsman CJ, Brown TE, Evans SW, Gotlieb EM, Knight JR, Ross EC, Shubiner HH, Wender EH, Wilens T. Attention-deficit/hyperactivity disorder among adolescents: a review of the diagnosis, treatment, and clinical implications. *Pediatrics* 2005; 115:1734-46.
- 387. Rabiner DL, Anastopoulos AD, Costello J, Hoyle RH, Swartzwelder HS. Adjustment to college in students with ADHD. J Attention Disord 2008; 11:689-699.
- 388. Nutt DJ, Fone K, Asherson P, Bramble D, Hill P, Matthews K, Morris KA, Santosh P, Sonuga-Barke E, Taylor E, Weiss M, Young S. Evidence-based guidelines for management of attention-deficit/hyperactivity disorder in

adolescents in transition to adult services and in adults: recommendations from the British Association for Psychopharmacology. *J Psychopharmacol* 2007; 21:10-41.

- 389. Wolf L. College students with AD/HD and other hidden disabilities: outcomes and interventions. *Annals New York Academy of Sciences* 2001; 931:385-395.
- 390. Faraone SV, Spencer TJ, Montano CB, Biederman J. Attentiondeficit/hyperactivity disorder in adults: a survey of current practice in psychiatry and primary care. *Arch Intern Med* 2004; 164:1221-6.
- 391. Buckmaster L. Research Brief no. 2 200405 Medication for Attention Deficit/Hyperactivity Disorder (ADHD): an analysis by Federal Electorate (200103): Commonwealth of Australia, 2004.
- 392. Rothstein J, Heazlewood R, Fraser M. Health of Aboriginal and Torres Strait Islander children in remote Far North Queensland: findings of the Paediatric Outreach Service. *Med J Aust* 2007; 186:519-21.
- 393. Zubrick S, Silburn S, Lawrence D, Mitrou F, Dalby R, Blair E, Griffin J, Milroy H, DeMaio J, Cox A, Li J. The Western Australian Aboriginal Child Health Survey: The social and emotional wellbeing of Aboriginal children and young people. Perth: Curtin University of Technology and Telethon Institute for Child Health Research, 2005.
- 394. Reid R, Hakendorf P, Prosser B. Use of psychostimulant medication for ADHD in South Australia. J Am Acad Child Adolesc Psychiatry 2002; 41:906-13.
- 395. Prosser B, Reid R. Changes in use of psychostimulant medication for ADHD in South Australia (1990-2006). *Aust New Zeal J Psychiatr* 2009; 43:340 347.
- 396. Counts CA, Nigg JT, Stawicki JA, Rappley MD, Von Eye A. Family adversity in DSM-IV ADHD combined and inattentive subtypes and associated disruptive behavior problems. *J Am Acad Child Adolesc Psychiatry* 2005; 44:690-698.
- 397. Visser SN, Lesesne CA. Mental Health in the United States: Prevalence of Diagnosis and Medication Treatment for Attention-Deficit/Hyperactivity Disorder - United States, 2003. *Morbidity & Mortality Weekly Report* 2005; 54:842-847.
- 398. Bussing R, Zima BT, Belin TR. Differential access to care for children with ADHD in special education programs. *Psychiatr Serv* 1998; 49:1226-9.
- 399. Bussing R, Gary FA, Mills TL, Garvan CW. Parental explanatory models of ADHD: gender and cultural variations. *Soc Psychiatry Psychiatr Epidemiol* 2003; 38:563-75.
- 400. Sayal K, Taylor E, Beecham J, Byrne P. Pathways to care in children at risk of attention-deficit hyperactivity disorder. *Br J Psychiatry* 2002; 181:43-48.
- 401. Education and Health Standing Committee of the Western Australian Legislative Assembly. Attention deficit hyperactivity disorder in Western Australia. Perth, Western Australia: State Law Publisher, 2004.
- 402. Thapar A, Thapar A. Is primary care ready to take on Attention Deficit Hyperactivity Disorder? *BMC Fam Pract* 2002; 3:7.
- 403. Epstein JN, Rabiner D, Johnson DE, Fitzgerald DP, Chrisman A, Erkanli A, Sullivan KK, March JS, Margolis P, Norton EC, Conners CK. Improving attention-deficit/hyperactivity disorder treatment outcomes through use of a collaborative consultation treatment service by community-based pediatricians: a cluster randomized trial. *Arch Pediatr Adolesc Med* 2007; 161:835-40.
- 404. Power TJ, Mautone JA, Manz PH, Frye L, Blum NJ. Managing attentiondeficit/hyperactivity disorder in primary care: a systematic analysis of roles and challenges. *Pediatrics* 2008; 121:e65-72.
- 405. Special Review Committee. Attention Deficit Hyperativity Disorder in Children and Adolescents in New South Wales-2007. Sydney: Clinical Excellence Commission, deprtment of Health, Giovernment of New South Wales, 2007.
- 406. Scottish Intercollegiate Guidelines Network (SIGN). Attention deficit and hyperkinetic disorders in children and young people: A national clinical guideline. 2001.

- 407. National Collaborating Centre for Mental Health Commissioned by the National Institute for Health and Clinical Excellence. Diagnosis and management of ADHD in children, young people and adults. Draft guidelines. 2008.
- 408. The MTA Cooperative Group. A 14-month randomized clinical trial of treatment strategies for attention-deficit/hyperactivity disorder. *Arch Gen Psychiatry* 1999; 56:1073 -1086.
- 409. The MTA Cooperative Group. National Institute of Mental Health Multimodal Treatment Study of ADHD follow-up: 24-month outcomes of treatment strategies for attention-deficit/hyperactivity disorder. *Pediatrics* 2004; 113:754-61.
- 410. Jensen PS, Arnold LE, Swanson JM, Vitiello B, Abikoff HB, Greenhill LL, Hechtman L, Hinshaw SP, Pelham WE, Wells KC, Conners CK, Elliott GR, Epstein JN, Hoza B, March JS, Molina BS, Newcorn JH, Severe JB, Wigal T, Gibbons RD, Hur K. 3-Year Follow-up of the NIMH MTA study. J Am Acad Child Adolesc Psychiatry 2007; 46:989-1001.
- 411. Pelham WE, Fabiano GA. Evidence-Based Psychosocial Treatments for Attention-Deficit/Hyperactivity Disorder. J Clin Child Adolesc Psychol 2008; 37:184 - 214.
- 412. Kendall J, Leo MC, Perrin N, Hatton D. Modeling ADHD child and family relationships *West J Nurs Res* 2005; 27:500-518.
- 413. Kendall J, Shelton K. A typology of management styles in families with children with ADHD *J Fam Nurs* 2003; 9:257-280.
- 414. Mrug S, Hoza B, Gerdes A. Children with Attention-Defecit/Hyperactivity Disorder: Peer relationships and peer-oriented interventions *New Directions for Child and Adolescent Development* 2001; 91:51-78.
- 415. Carr A. Evidence-based practice in family therapy and systemic consultationChild-focused problems *J Fam Ther* 2000; 22:29-60.
- 416. Ramsay JR, Rostain AL. Psychosocial treatments for attention deficit/hyperactivity disorder in adults: Current evidence and future directions. *Professional Psychology: Research and Practice* 2007; 38:338-346
- 417. Goldstein S. Coaching as a treatment for ADHD. *J Attention Disord* 2005; 9:379-81.
- 418. Bor W, Sanders MR, Markie-Dadds C. The effects of the Triple P-Positive Parenting Program on preschool children with co-occurring disruptive behavior and attentional/hyperactive difficulties. J Abnorm Child Psychol 2002; 30:571-87.
- 419. Sonuga-Barke EJ, Daley D, Thompson M, Laver-Bradbury C, Weeks A. Parentbased therapies for preschool attention-deficit/hyperactivity disorder: a randomized, controlled trial with a community sample. J Am Acad Child Adolesc Psychiatry 2001; 40:402-8.
- 420. Sonuga-Barke EJ, Thompson M, Daley D, Laver-Bradbury C. Parent training for Attention Deficit/Hyperactivity Disorder: is it as effective when delivered as routine rather than as specialist care? *Br J Clin Psychol* 2004; 43:449-57.
- 421. Wyatt Kaminski J, Valle LA, Filene JH, Boyle CL. A meta-analytic review of components associated with parent training program effectiveness. *J Abnorm Child Psychol* 2008; 36:567-89.
- 422. Corcoran J, Dattalo P. Parent involvement in treatment for ADHD: A metaanalysis of the published studies. *Research on Social Work Practice* 2006; 16:561-570.
- 423. Hoath FE, Sanders MR. A feasibility study of enhanced group Triple P Positive Parenting Program for parents of children with attention deficit hyperactivity disorder. *Behaviour Change* 2002; 19:191-206.
- 424. Pfiffner LJ, Mikami AY, Huang-Pollock C, Easterlin B, Zalecki C, McBurnett K. A randomized, controlled trial of integrated home-school behavioral treatment for ADHD, predominantly inattentive type. J Am Acad Child Adolesc Psychiatry 2007; 46:1041-1050.

- 425. van den Hoofdakker BJMA, van der Veen-Mulders LMA, Sytema SPD, Emmelkamp PMGPD, Minderaa RBPD, Nauta MHPD. Effectiveness of behavioral parent training for children with ADHD in routine clinical practice: A randomized controlled study. J Am Acad Child Adolesc Psychiatry 2007; 46:1263-1271.
- 426. de Boo GM, Prins PJM. Social incompetence in children with ADHD: possible moderators and mediators in social-skills training. *Clinical Psychology Review* 2007; 27:78-97.
- 427. Bjornstad G, Montgomery P. Family therapy for attention-deficit disorder or attention-deficit/hyperactivity disorder in children and adolescents. *Cochrane database of systematic reviews (Online)* 2005:Issue 2 Art. No. :CD005042. DOI: 10.1002/14651858.CD005042.pub2.
- 428. Gonzalez LO, Sellers EW. The effects of a stress-management program on self-concept, locus of control, and the acquisition of coping skills in school-age children diagnosed with attention deficit hyperactivity disorder. *J Child Adolesc Psychiatr Nurs* 2002; 15:5-15.
- 429. Chronis AM, Fabiano GA, Gnagy EM, Onyango AN, Pelham Jr. WE, Lopez-Williams A, Chacko A, Wymbs BT, Coles EK, Seymour KE. An evaluation of the summer treatment program for children with attention-deficit/hyperactivity disorder using a treatment withdrawal design *Behavior Therapy* 2004; 35:561-585.
- 430. Arnold LE, Chuang S, Davies M, Abikoff HB, Conners CK, Elliott GR, Greenhill LL, Hechtman L, Hinshaw SP, Hoza B, Jensen PS, Kraemer HC, Langworthy-Lam KS, March JS, Newcorn JH, Pelham WE, Severe JB, Swanson JM, Vitiello B, Wells KC, Wigal T. Nine months of multicomponent behavioral treatment for ADHD and effectiveness of MTA fading procedures. J Abnorm Child Psychol 2004; 32:39-51.
- 431. Wells K, Chi T, Hinshaw S, Epstein J, Pfiffner L, Nebel-Schwalm M, Owens E, Arnold L, Abikoff H, Conners C, Elliott G, Greenhill L, Hechtman L, Hoza B, Jensen P, March J, Newcorn J, Pelham W, Severe J, Swanson J, Vitiello B, Wigal T. Treatment-related changes in objectively measured parenting behaviors in the multimodal treatment study of children with attention-deficit/hyperactivity disorder. J Consult Clin Psychol 2006; 74:649-57.
- 432. Stevenson CS, Whitmont S, Bornholt L, Livesey D, Stevenson RJ. A cognitive remediation programme for adults with Attention Deficit Hyperactivity Disorder. *Aust N Z J Psychiatry* 2002; 36:610-6.
- 433. Stevenson CS, Stevenson RJ, Whitmont S. A self directed psychosocial intervention with minimal therapist contact for adults with attention deficit hyperactivity disorder. *Clin Psychol Psychother* 2003; 10:93-101.
- 434. Wiggins D, Singh K, Getz H, Hutchins D. Effects of brief group intervention for adults with attention deficit/hyperactivity disorder. *J Mental Health Counselling* 1999; 21:82-92.
- 435. Philipsen A, Richter H, Peters J, Alm B, Sobanski E, Colla M, Munzebrock M, Scheel C, Jacob C, Perlov E, Tebartz van Elst L, Hesslinger B. Structured group psychotherapy in adults with attention deficit hyperactivity disorder: results of an open multicentre study. *J Nerv Ment Dis* 2007; 195:1013-9.
- 436. Virta M, Vedenpaa A, Gronroos N, Chydenius E, Partinen M, Vataja R, Kaski M, Iivanainen M. Adults with ADHD benefit from cognitive behaviorally oriented group rehabilitation: A study of 29 participants. *J Attention Disord* 2008; in press.
- 437. Solanto MV, Marks DJ, Mitchell KJ, Wasserstein J, Kofman MD. Development of a new psychosocial treatment for adult ADHD. J Attention Disord 2008; 11:728-36.
- 438. Young S, Bramham J, Gray K, Rose E. The experience of receiving a diagnosis and treatment of ADHD in adulthood: a qualitative study of clinically referred patients using interpretative phenomenological analysis. *J Attention Disord* 2008; 11:493-503.

- 439. Barkley R, Edwards G, Laneri M, Fletcher K, Metevia L. The efficacy of problem-solving communication training alone, behavior management training alone, and their combination for parent-adolescent conflict in teenagers with ADHD and ODD. J Consult Clin Psychol 2001; 69:926-41
- 440. Jensen PS, Hinshaw SP, Kraemer HC, Lenora N, Newcorn JH, Abikoff HB, March JS, Arnold LE, Cantwell DP, Conners CK, Elliott GR, Greenhill LL, Hechtman L, Hoza B, Pelham WE, Severe JB, Swanson JM, Wells KC, Wigal T, Vitiello B. ADHD comorbidity findings from the MTA study: comparing comorbid subgroups. J Am Acad Child Adolesc Psychiatry 2001; 40:147-58.
- 441. March JS, Swanson JM, Arnold LE, Hoza B, Conners CK, Hinshaw SP, Hechtman L, Kraemer HC, Greenhill LL, Abikoff HB, Elliott LG, Jensen PS, Newcorn JH, Vitiello B, Severe J, Wells KC, Pelham WE. Anxiety as a predictor and outcome variable in the multimodal treatment study of children with ADHD (MTA). J Abnorm Child Psychol 2000; 28:527-41.
- 442. Gazarian M, Kelly M, McPhee JR, Graudins LV, Ward RL, Campbell TJ. Off-label use of medicines: consensus recommendations for evaluating appropriateness. *Med J Aust* 2006; 185:544-8.
- 443. King S, Griffin S, Hodges Z, Weatherly H, Asseburg C, Richardson G, Golder S, Taylor E, Drummond M, Riemsma R. A systematic review and economic model of the effectiveness and cost-effectiveness of methylphenidate, dexamfetamine and atomoxetine for the treatment of attention deficit hyperactivity disorder in children and adolescents. *Health Technology Assessment* 2006; 10:iii-iv, xiii-146.
- 444. Carlson GA, Dunn D, Kelsey D, Ruff D, Ball S, Ahrbecker L, Allen AJ. A pilot study for augmenting atomoxetine with methylphenidate: safety of concomitant therapy in children with attention-deficit/hyperactivity disorder. *Child Adolesc Psychiatry Ment Health* 2007; 1:10.
- 445. Greenhill L, Kollins S, Abikoff H, McCracken J, Riddle M, Swanson J, McGough J, Wigal S, Wigal T, Vitiello B, Skrobala A, Posner K, Ghuman J, Cunningham C, Davies M, Chuang S, Cooper T. Efficacy and safety of immediate-release methylphenidate treatment for preschoolers with ADHD. J Am Acad Child Adolesc Psychiatry 2006; 45:1284-93.
- 446. Musten LM, Firestone P, Pisterman S, Bennett S, Mercer J. Effects of methylphenidate on preschool children with ADHD: cognitive and behavioral functions. J Am Acad Child Adolesc Psychiatry 1997; 36:1407-15.
- 447. Wilens TE, McBurnett K, Bukstein O, McGough J, Greenhill L, Lerner M, Stein MA, Conners CK, Duby J, Newcorn J, Bailey CE, Kratochvil CJ, Coury D, Casat C, Denisco MJ, Halstead P, Bloom L, Zimmerman BA, Gu J, Cooper KM, Lynch JM. Multisite controlled study of OROS methylphenidate in the treatment of adolescents with attention-deficit/hyperactivity disorder. Arch Pediatr Adolesc Med 2006; 160:82-90.
- 448. Spencer T, Wilens T, Biederman J, Weisler R, Read S, Pratt R. Efficacy and safety of mixed amphetamine salts extended release (Adderall XR) in the management of attention-deficit/hyperactivity disorder in adolescent patients: a 4-week, randomized, double-blind, placebo-controlled, parallel-group study. *Clin Ther* 2006; 28:266-79.
- 449. Faraone SV, Spencer T, Aleardi M, Pagano C, Biederman J. Meta-analysis of the efficacy of methylphenidate for treating adult attention-deficit/hyperactivity disorder. *J Clin Psychopharmacol* 2004; 24:24-9.
- 450. Biederman J, Mick E, Surman C, Doyle R, Hammerness P, Harpold T, Dunkel S, Dougherty M, Aleardi M, Spencer T. A randomized, placebo-controlled trial of OROS methylphenidate in adults with attention-deficit/hyperactivity disorder. *Biol Psychiatry* 2006; 59:829-35.
- 451. Jain U, Hechtman L, Weiss M, Ahmed TS, Reiz JL, Donnelly GA, Harsanyi Z, Darke AC. Efficacy of a novel biphasic controlled-release methylphenidate formula in adults with attention-deficit/hyperactivity disorder: results of a

double-blind, placebo-controlled crossover study. *J Clin Psychiatry* 2007; 68:268-77.

- 452. Kooij JJ, Burger H, Boonstra AM, Van der Linden PD, Kalma LE, Buitelaar JK. Efficacy and safety of methylphenidate in 45 adults with attentiondeficit/hyperactivity disorder. A randomized placebo-controlled double-blind cross-over trial. *Psychol Med* 2004; 34:973-82.
- 453. Spencer T, Biederman J, Wilens T, Doyle R, Surman C, Prince J, Mick E, Aleardi M, Herzig K, Faraone S. A large, double-blind, randomized clinical trial of methylphenidate in the treatment of adults with attentiondeficit/hyperactivity disorder. *Biol Psychiatry* 2005; 57:456-63.
- 454. Taylor F, Russo J. Efficacy of modafinil compared to dextroamphetamine for the treatment of attention deficit hyperactivity disorder in adults. *J Child Adolesc Psychopharmacol* 2000; 10:311-20.
- 455. Taylor FB, Russo J. Comparing guanfacine and dextroamphetamine for the treatment of adult attention-deficit/hyperactivity disorder. *J Clin Psychopharmacol* 2001; 21:223-8.
- 456. Paterson R, Douglas C, Hallmayer J, Hagan M, Krupenia Z. A randomised, double-blind, placebo-controlled trial of dexamphetamine in adults with attention deficit hyperactivity disorder. *Aust N Z J Psychiatry* 1999; 33:494-502.
- 457. Spencer T, Biederman J, Wilens T, Faraone S, Prince J, Gerard K, Doyle R, Parekh A, Kagan J, Bearman SK. Efficacy of a mixed amphetamine salts compound in adults with attention-deficit/hyperactivity disorder. *Arch Gen Psychiatry* 2001; 58:775-82.
- 458. Weisler R, Biederman J, Spencer TJ, Wilens TE, Faraone SV, Chrisman AK, Read SC, Tulloch SJ. Mixed amphetamine salts extended-release in the treatment of adult ADHD: a randomized, controlled trial. *CNS spectrums* 2006; 11:625-639.
- 459. Pelham WE, Gnagy EM, Burrows-Maclean L, Williams A, Fabiano GA, Morrisey SM, Chronis AM, Forehand GL, Nguyen CA, Hoffman MT, Lock TM, Fielbelkorn K, Coles EK, Panahon CJ, Steiner RL, Meichenbaum DL, Onyango AN, Morse GD. Once-a-day Concerta methylphenidate versus three-times-daily methylphenidate in laboratory and natural settings. *Pediatrics* 2001; 107:E105.
- 460. Clinical Excellence Commission. Attention deficit hyperactivity disorder in children and adolescents in New South Wales: Final report of the special review. <u>www.cec.health.nsw.gov.au</u>. 2007.
- 461. Martins S, Tramontina S, Polanczyk G, Eizirik M, Swanson JM, Rohde LA. Weekend holidays during methylphenidate use in ADHD children: a randomized clinical trial. J Child Adolesc Psychopharmacol 2004; 14:195-206.
- 462. Spencer TJ, Faraone SV, Biederman J, Lerner M, Cooper KM, Zimmerman B. Does prolonged therapy with a long-acting stimulant suppress growth in children with ADHD? J Am Acad Child Adolesc Psychiatry 2006; 45:527-37.
- 463. Firestone P, Musten LM, Pisterman S, Mercer J, Bennett S. Short-term side effects of stimulant medication are increased in preschool children with attention-deficit/hyperactivity disorder: a double-blind placebo-controlled study. J Child Adolesc Psychopharmacol 1998; 8:13-25.
- 464. Wigal T, Greenhill L, Chuang S, McGough J, Vitiello B, Skrobala A, Swanson J, Wigal S, Abikoff H, Kollins S, McCracken J, Riddle M, Posner K, Ghuman J, Davies M, Thorp B, Stehli A. Safety and tolerability of methylphenidate in preschool children with ADHD. J Am Acad Child Adolesc Psychiatry 2006; 45:1294-303.
- 465. Michelson D, Adler L, Spencer T, Reimherr FW, West SA, Allen AJ, Kelsey D, Wernicke J, Dietrich A, Milton D. Atomoxetine in adults with ADHD: two randomized, placebo-controlled studies. *Biol Psychiatry* 2003; 53:112-20.
- 466. Michelson D, Buitelaar JK, Danckaerts M, Gillberg C, Spencer TJ, Zuddas A, Faries DE, Zhang S, Biederman J. Relapse prevention in pediatric patients with

ADHD treated with atomoxetine: a randomized, double-blind, placebocontrolled study. J Am Acad Child Adolesc Psychiatry 2004; 43:896-904.

- 467. Buitelaar JK, Michelson D, Danckaerts M, Gillberg C, Spencer TJ, Zuddas A, Faries DE, Zhang S, Biederman J. A randomized, double-blind study of continuation treatment for attention-deficit/hyperactivity disorder after 1 year. *Biol Psychiatry* 2007; 61:694-9.
- 468. Wang Y, Zheng Y, Du Y, Song DH, Shin Y-J, Cho SC, Kim BN, Ahn DH, Marquez-Caraveo ME, Gao H, Williams DW, Levine LR. Atomoxetine versus methylphenidate in paediatric outpatients with attention deficit hyperactivity disorder: a randomized, double-blind comparison trial. *Aust New Zeal J Psychiatr* 2007; 41:222-30.
- 469. Sangal RB, Owens J, Allen AJ, Sutton V, Schuh K, Kelsey D. Effects of atomoxetine and methylphenidate on sleep in children with ADHD. *Sleep* 2006; 29:1573-85.
- 470. Newcorn JH, Kratochvil CJ, Allen AJ, Casat CD, Ruff DD, Moore RJ, Michelson D, Atomoxetine/Methylphenidate Comparative Study G. Atomoxetine and Osmotically Released Methylphenidate for the Treatment of Attention Deficit Hyperactivity Disorder: Acute Comparison and Differential Response. *Am J Psychiatry* 2008:appi.ajp.2007.05091676.
- 471. Wigal SB, McGough JJ, McCracken JT, Biederman J, Spencer TJ, Posner KL, Wigal TL, Kollins SH, Clark TM, Mays DA, Zhang Y, Tulloch SJ. A laboratory school comparison of mixed amphetamine salts extended release (Adderall XR) and atomoxetine (Strattera) in school-aged children with attention deficit/hyperactivity disorder. J Attention Disord 2005; 9:275-89.
- 472. The Pharmaceutical Benefits Scheme (PBS). Atomoxetine Strattera for attention deficit hyperactivity disorder, 2005.
- 473. Eli Lilly and Company (USA). Strattera (atomoxetine) Prescribing Information.
- 474. Wilens TE, Newcorn JH, Kratochvil CJ, Gao H, Thomason CK, Rogers AK, Feldman PD, Levine LR. Long-term atomoxetine treatment in adolescents with attention-deficit/hyperactivity disorder. *J Pediatr* 2006; 149:112-9.
- 475. Adler L, Dietrich A, Reimherr FW, Taylor LV, Sutton VK, Bakken R, Allen AJ, Kelsey D. Safety and tolerability of once versus twice daily atomoxetine in adults with ADHD. *Ann Clin Psychiatry* 2006; 18:107-13.
- 476. Palumbo DR, Sallee FR, Pelham WE, Jr., Bukstein OG, Daviss WB, McDermott MP. Clonidine for attention-deficit/hyperactivity disorder: I. Efficacy and tolerability outcomes. *J Am Acad Child Adolesc Psychiatry* 2008.
- 477. Daviss WB, Patel NC, Robb AS, McDermott MP, Bukstein OG, Pelham WE, Jr., Palumbo D, Harris P, Sallee FR. Clonidine for attention-deficit/hyperactivity disorder: II. ECG changes and adverse events analysis. J Am Acad Child Adolesc Psychiatry 2008.
- 478. Biederman J, Swanson J, Wigal S, Kratochvil C, Boellner S, Earl C, Jiang J, Greenhill L. Efficacy and safety of modafinil film-coated tablets in children and adolescents with attention-deficit/hyperactivity disorder: results of a randomized, double-blind, placebo-controlled, flexible-dose study. *Pediatrics* 2005; 116:e777-84.
- 479. Biederman J, Swanson JM, Wigal SB, Boellner SW, Earl CQ, Lopez FA, Modafinil ASG. A comparison of once-daily and divided doses of modafinil in children with attention-deficit/hyperactivity disorder: a randomized, doubleblind, and placebo-controlled study. *J Clin Psychiatr* 2006; 67:727-35.
- 480. Wigal SB, Biederman J, Swanson JM, Yang R, Greenhill LL. Efficacy and safety of modafinil film-coated tablets in children and adolescents with or without prior stimulant treatment for attention-deficit/hyperactivity disorder: pooled analysis of 3 randomized, double-blind, placebo-controlled studies. *Prim Care Companion J Clin Psychiatry* 2006; 8:352-60.
- 481. Swanson JM, Greenhill LL, Lopez FA, Sedillo A, Earl CQ, Jiang JG, Biederman J. Modafinil film-coated tablets in children and adolescents with attentiondeficit/hyperactivity disorder: results of a randomized, double-blind, placebo-

controlled, fixed-dose study followed by abrupt discontinuation. *J Clin Psychiatr* 2006; 67:137-47.

- 482. Rugino T, Samsock T. Modafinil in children with attention-deficit hyperactivity disorder. *Pediatric neurology*. 2003; 29:136-42.
- 483. Greenhill LL, Biederman J, Boellner SW, Rugino TA, Sangal RB, Earl CQ, Jiang JG, Swanson JM. A randomized, double-blind, placebo-controlled study of modafinil film-coated tablets in children and adolescents with attention-deficit/hyperactivity disorder. J Am Acad Child Adolesc Psychiatry 2006; 45:503-11.
- 484. Amiri S, Mohammadi MR, Mohammadi M, Nouroozinejad GH, Kahbazi M, Akhondzadeh S. Modafinil as a treatment for Attention-Deficit/Hyperactivity Disorder in children and adolescents: A double blind, randomized clinical trial. *Prog Neuropsychopharmacol Biol Psychiatry* 2007.
- 485. Simon P, Hemet C, Ramassamy C, Costentin J. Non-amphetaminic mechanism of stimulant locomotor effect of modafinil in mice. *Eur Neuropsychopharmacol* 1995; 5:509-14.
- 486. Morgan RE, Crowley JM, Smith RH, LaRoche RB, Dopheide MM. Modafinil improves attention, inhibitory control, and reaction time in healthy, middle-aged rats. *Pharmacol Biochem Behav* 2007; 86:531-41.
- 487. Minzenberg MJ, Carter CS. Modafinil: A Review of Neurochemical Actions and Effects on Cognition. *Neuropsychopharmacology* 2007.
- 488. Food and Drug Administration (FDA). NDA 20-717 PROVIGIL® (modafinil) Tablets. FDA Approved Labeling dated August 17, 2007, 2007.
- 489. Rubinstein S, Malone MA, Roberts W, Logan WJ. Placebo-controlled study examining effects of selegiline in children with attention-deficit/hyperactivity disorder. J Child Adolesc Psychopharmacol 2006; 16:404-15.
- 490. Mohammadi MR, Ghanizadeh A, Alaghband-Rad J, Tehranidoost M, Mesgarpour B, Soori H. Selegiline in comparison with methylphenidate in attention deficit hyperactivity disorder children and adolescents in a doubleblind, randomized clinical trial. J Child Adolesc Psychopharmacol 2004; 14:418-25.
- 491. Akhondzadeh S, Tavakolian R, Davari-Ashtiani R, Arabgol F, Amini H. Selegiline in the treatment of attention deficit hyperactivity disorder in children: a double blind and randomized trial. *Progress in Neuro-Psychopharmacology & Biological Psychiatry* 2003; 27:841-5.
- 492. Biederman J, Melmed RD, Patel A, McBurnett K, Konow J, Lyne A, Scherer N. A randomized, double-blind, placebo-controlled study of guanfacine extended release in children and adolescents with attention-deficit/hyperactivity disorder. *Pediatrics* 2008; 121:e73-84.
- 493. Shytle RD, Silver AA, Wilkinson BJ, Sanberg PR. A pilot controlled trial of transdermal nicotine in the treatment of attention deficit hyperactivity disorder. *World J Biol Psychiatr* 2002; 3:150-5.
- 494. Wilens TE, Spencer TJ, Biederman J, Girard K, Doyle R, Prince J, Polisner D, Solhkhah R, Comeau S, Monuteaux MC, Parekh A. A controlled clinical trial of bupropion for attention deficit hyperactivity disorder in adults. *Am J Psychiatr* 2001; 158:282-288.
- 495. Wilens TE, Haight BR, Horrigan JP, Hudziak JJ, Rosenthal NE, Connor DF, Hampton KD, Richard NE, Modell JG. Bupropion XL in adults with attentiondeficit/hyperactivity disorder: a randomized, placebo-controlled study. *Biol Psychiatry* 2005; 57:793-801.
- 496. Kuperman S, Perry PJ, Gaffney GR, Lund BC, Bever-Stille KA, Arndt S, Holman TL, Moser DJ, Paulsen JS. Bupropion SR vs. methylphenidate vs. placebo for attention deficit hyperactivity disorder in adults. *Ann Clin Psychiatry* 2001; 13:129-34.
- 497. Reimherr FW, Hedges DW, Strong RE, Marchant BK, Williams ED. Bupropion SR in adults with ADHD: A short-term, placebo-controlled trial: Neuropsychiatric Disease And Treatment Vol 1(3) 2005, 245-251, 2005.

- 498. Armenteros JL, Lewis JE, Davalos M. Risperidone augmentation for treatmentresistant aggression in attention-deficit/hyperactivity disorder: a placebocontrolled pilot study. J Am Acad Child Adolesc Psychiatry 2007; 46:558-65.
- 499. Aman MG, Binder C, Turgay A. Risperidone effects in the presence/absence of psychostimulant medicine in children with ADHD, other disruptive behavior disorders, and subaverage IQ. J Child Adolesc Psychopharmacol 2004; 14:243-54.
- 500. Adverse Drug Reactions Advisory Committee. Use of SSRI antidepressants in children and adolescents. Available from http://www.tga.gov.au/adr/adrac_ssri.htm. 2004.
- 501. Royal Australian and New Zealand College of Psychiatrists, Royal Australian College of General Practitioners, Royal Australasian College of Physicians. Clinical guidance on the use of antidepressant medications in children and adolescents. Available at <u>http://www.racgp.org.au/Content/NavigationMenu/ClinicalResources/RACGPGui</u> <u>delines/Antidepressantmedications/20050509antidepressantguidelines.pdf.</u>
- 2005.
 502. Diamond I, Tannock R, Schachar R. Response to methylphenidate in children with ADHD and comorbid anxiety. J Am Acad Child Adolesc Psychiatry 1999; 38:402-409.
- 503. Geller D, Donnelly C, Lopez F, Rubin R, Newcorn J, Sutton V, Bakken R, Paczkowski M, Kelsey D, Sumner C. Atomoxetine treatment for pediatric patients with attention-deficit/hyperactivity disorder with comorbid anxiety disorder. J Am Acad Child Adolesc Psychiatry 2007; 46:1119-27.
- 504. Kolko DJ, Bukstein OG, Barron J. Methylphenidate and behavior modification in children with ADHD and comorbid ODD or CD: main and incremental effects across settings. *J Am Acad Child Adolesc Psychiatry* 1999; 38:578-86.
- 505. Hazell PL, Stuart JE. A randomized controlled trial of clonidine added to psychostimulant medication for hyperactive and aggressive children. J Am Acad Child Adolesc Psychiatry 2003; 42:886-894.
- 506. Biederman J, Spencer TJ, Newcorn JH, Gao H, Milton DR, Feldman PD, Witte MM. Effect of comorbid symptoms of oppositional defiant disorder on responses to atomoxetine in children with ADHD: a meta-analysis of controlled clinical trial data. *Psychopharmacology (Berl)* 2007; 190:31-41.
- 507. Newcorn JH, Spencer TJ, Biederman J, Milton DR, Michelson D. Atomoxetine treatment in children and adolescents with attention-deficit/hyperactivity disorder and comorbid oppositional defiant disorder. J Am Acad Child Adolesc Psychiatry 2005; 44:240-248.
- 508. Hazell P, Zhang S, Wolaczyk T, Barton J, Johnson M, Zuddas A, Danckaerts M, Ladikos A, Benn D, Yoran-Hegesh R, Zeiner P, Michelson D. Comorbid oppositional defiant disorder and the risk of relapse during 9 months of atomoxetine treatment for attention-deficit/hyperactivity disorder. *Eur Child Adolesc Psychiatry* 2006; 15:105-10.
- 509. Bangs ME, Emslie GJ, Spencer TJ, Ramsey JL, Carlson C, Bartky EJ, Busner J, Duesenberg DA, Harshawat P, Kaplan SL, Quintana H, Allen AJ, Sumner CR. Efficacy and safety of atomoxetine in adolescents with attentiondeficit/hyperactivity disorder and major depression. J Child Adolesc Psychopharmacol 2007; 17:407-20.
- 510. Findling RL, Short EJ, McNamara NK, Demeter CA, Stansbrey RJ, Gracious BL, Whipkey R, Manos MJ, Calabrese JR. Methylphenidate in the treatment of children and adolescents with bipolar disorder and attentiondeficit/hyperactivity disorder. J Am Acad Child Adolesc Psychiatry 2007; 46:1445-53.
- 511. Scheffer RE, Kowatch RA, Carmody T, Rush AJ. Randomized, placebocontrolled trial of mixed amphetamine salts for symptoms of comorbid ADHD in pediatric bipolar disorder after mood stabilization with divalproex sodium. *Am J Psychiatry* 2005; 162:58-64.

- 512. Plioplys SMD, Dunn DWMD, Caplan RMD. 10-year research update review: Psychiatric problems in children with epilepsy. *J Am Acad Child Adolesc Psychiatry* 2007; 46:1389-1402.
- 513. Torres AR, Whitney J, Gonzalez-Heydrich J. Attention-deficit/hyperactivity disorder in pediatric patients with epilepsy: review of pharmacological treatment. *Epilepsy Behav* 2008; 12:217-33.
- 514. Gucuyener K, Erdemoglu AK, Senol S, Serdaroglu A, Soysal S, Kockar AI. Use of methylphenidate for attention-deficit hyperactivity disorder in patients with epilepsy or electroencephalographic abnormalities. *J Child Neurol* 2003; 18:109-12.
- 515. Feldman H, Crumrine P, Handen BL, Alvin R, Teodori J. Methylphenidate in children with seizures and attention-deficit disorder. *Am J Dis Child* 1989; 143:1081-6.
- 516. Gross-Tsur V, Manor O, van der Meere J, Joseph A, Shalev RS. Epilepsy and attention deficit hyperactivity disorder: is methylphenidate safe and effective? *J Pediatr* 1997; 130:670-4.
- 517. Hemmer SA, Pasternak JF, Zecker SG, Trommer BL. Stimulant therapy and seizure risk in children with ADHD. *Pediatr Neurol* 2001; 24:99-102.
- 518. Wernicke JF, Holdridge KC, Jin L, Edison T, Zhang S, Bangs ME, Allen AJ, Ball S, Dunn D. Seizure risk in patients with attention-deficit-hyperactivity disorder treated with atomoxetine. *Dev Med Child Neurol* 2007; 49:498-502.
- 519. Hernandez A, Barragan P. Efficacy of atomoxetine treatment in children with ADHD and epilepsy. *Epilepsia* 2005; 46 (Suppl 6):718
- 520. Gadow K, Sverd J, Nolan E, Sprafkin J, Schneider J. Immediate-release methylphenidate for ADHD in children with comorbid chronic multiple tic disorder. *J Am Acad Child Adolesc Psychiatry* 2007; 46:840-848.
- 521. Allen AJ, Kurlan RM, Gilbert DL, Coffey BJ, Linder SL, Lewis DW, Winner PK, Dunn DW, Dure LS, Sallee FR, Milton DR, Mintz MI, Ricardi RK, Erenberg G, Layton LL, Feldman PD, Kelsey DK, Spencer TJ. Atomoxetine treatment in children and adolescents with ADHD and comorbid tic disorders. *Neurology* 2005; 65:1941-9.
- 522. Tourette's Syndrome Study Group. Treatment of ADHD in children with tics: a randomized controlled trial. *Neurology* 2002; 58:527-36.
- 523. Palumbo D, Spencer T, Lynch J, Co-Chien H, Faraone S. Emergence of tics in children with ADHD: impact of once-daily OROS methylphenidate therapy. *J Child Adolesc Psychopharmacol* 2004; 14:185-194.
- 524. Law S, Schachar R. Do typical clinical doses of methylphenidate cause tics in children treated for attention-deficit hyperactivity disorder? *J Am Acad Child Adolesc Psychiatry* 1999; 38:944-951.
- 525. Evans SW, Pelham WE, Smith BH, Bukstein O, Gnagy EM, Greiner AR, Altenderfer L, Baron-Myak C. Dose-response effects of methylphenidate on ecologically valid measures of academic performance and classroom behavior in adolescents with ADHD. *Exp Clin Psychopharmacol* 2001; 9:163-75.
- 526. Wolraich ML, Greenhill LL, Pelham W, Swanson J, Wilens T, Palumbo D, Atkins M, McBurnett K, Bukstein O, August G. Randomized, controlled trial of oros methylphenidate once a day in children with attention-deficit/hyperactivity disorder. *Pediatrics* 2001; 108:883-92.
- 527. Spencer TJ, Sallee FR, Gilbert DL, Dunn DW, McCracken JT, Coffey BJ, Budman CL, Ricardi RK, Leonard HL, Allen AJ, Milton DR, Feldman PD, Kelsey DK, Geller DA, Linder SL, Lewis DW, Winner PK, Kurlan RM, Mintz M. Atomoxetine treatment of ADHD in children with comorbid tourette syndrome. J Attention Disord 2008; 11:470-81.
- 528. Roessner V, Robatzek M, Knapp G, Banaschewski T, Rothenberger A. Firstonset tics in patients with attention-deficit-hyperactivity disorder: impact of stimulants. *Dev Med Child Neurol* 2006; 48:616-21.
- 529. Lowe TL, Cohen DJ, Detlor J, Kremenitzer MW, Shaywitz BA. Stimulant medications precipitate Tourette's syndrome. *JAMA* 1982; 247:1729-31.

- 530. Golden GS. The relationship between stimulant medication and tics. *Pediatr Ann* 1988; 17:405-6, 408.
- 531. Kurlan R. Tourette's syndrome: are stimulants safe? *Curr Neurol Neurosci Rep* 2003; 3:285-8.
- 532. Lipkin PH, Goldstein IJ, Adesman AR. Tics and dyskinesias associated with stimulant treatment in attention-deficit hyperactivity disorder. *Arch Pediatr Adolesc Med* 1994; 148:859-61.
- 533. Gadow KD, Sverd J, Sprafkin J, Nolan EE, Ezor SN. Efficacy of methylphenidate for attention-deficit hyperactivity disorder in children with tic disorder. *Arch Gen Psychiatry* 1995; 52:444-55.
- 534. Castellanos F, Giedd JN EJ, Marsh, WL RG, Hamburger SD, et al. Controlled stimulant treatment of ADHD and comorbid Tourette's syndrome: effects of stimulant and dose. *J Am Acad Child Adolesc Psychiatry* 1997; 36:589–596.
- 535. Gadow KD, Sverd J, Sprafkin J, Nolan EE, Grossman S. Long-term methylphenidate therapy in children with comorbid attention-deficit hyperactivity disorder and chronic multiple tic disorder. *Arch Gen Psychiatry* 1999; 56:330-6.
- 536. Scahill L, Chappell PB, Kim YS, Schultz RT, Katsovich L, Shepherd E, Arnsten AF, Cohen DJ, Leckman JF. A placebo-controlled study of guanfacine in the treatment of children with tic disorders and attention deficit hyperactivity disorder. *Am J Psychiatry* 2001; 158:1067-74.
- 537. Grizenko N, Bhat M, Schwartz G, Ter-Stepanian M, Joober R. Efficacy of methylphenidate in children with attention-deficit hyperactivity disorder and learning disabilities: a randomized crossover trial. *J Psychiatr Neurosci* 2006; 31:46-51.
- 538. McInnes A, Bedard AC, Hogg-Johnson S, Tannock R. Preliminary evidence of beneficial effects of methylphenidate on listening comprehension in children with attention-deficit/hyperactivity disorder. J Child Adolesc Psychopharmacol 2007; 17:35-49.
- 539. Francis S, Fine J, Tannock R. Methylphenidate selectively improves story retelling in children with attention deficit hyperactivity disorder. *J Child Adolesc Psychopharmacol* 2001; 11:217-28.
- 540. Keulers EHH, Hendriksen JGM, Feron FJM, Wassenberg R, Wuisman-Frerker MGF, Jolles J, Vles JSH. Methylphenidate improves reading performance in children with attention deficit hyperactivity disorder and comorbid dyslexia: an unblinded clinical trial. *Eur J Paediatr Neurol* 2007; 11:21-8.
- 541. Pearson DA, Santos CW, Roache JD, Casat CD, Loveland KA, Lachar D, Lane DM, Faria LP, Cleveland LA. Treatment effects of methylphenidate on behavioral adjustment in children with mental retardation and ADHD. J Am Acad Child Adolesc Psychiatry 2003; 42:209-16.
- 542. Aman M, Buican B, Arnold LE. Methylphenidate treatment in children with borderline IQ and mental retardation: Analysis of three aggregated studies. *J Child Adolesc Psychopharmacol* 2003; 13:29-40.
- 543. Handen BL, Johnson CR, Lubetsky M. Efficacy of methylphenidate among children with autism and symptoms of attention-deficit hyperactivity disorder. J Autism Dev Disord 2000; 30:245-255.
- 544. Posey DJ, Aman MG, McCracken JT, Scahill L, Tierney E, Arnold LE, Vitiello B, Chuang SZ, Davies M, Ramadan Y, Witwer AN, Swiezy NB, Cronin P, Shah B, Carroll DH, Young C, Wheeler C, McDougle CJ. Positive effects of methylphenidate on inattention and hyperactivity in pervasive developmental disorders: an analysis of secondary measures. *Biol Psychiatry* 2007; 61:538-44.
- 545. Research Units on Pediatric Psychopharmacology Autism Network. Randomized, controlled, crossover trial of methylphenidate in pervasive developmental disorders with hyperactivity. *Arch Gen Psychiatry* 2005; 62:1266-74.

- 546. Arnold LE, Aman MG, Cook AM, Witwer AN, Hall KL, Thompson S, Ramadan Y. Atomoxetine for hyperactivity in autism spectrum disorders: placebo-controlled crossover pilot trial. J Am Acad Child Adolesc Psychiatry 2006; 45:1196-205.
- 547. Hazell P. Drug therapy for attention-deficit/hyperactivity disorder-like symptoms in autistic disorder. *J Paediatr Child Health* 2007; 43:19-24.
- 548. Wolraich ML, McGuinn L, Doffing M. Treatment of attention deficit hyperactivity disorder in children and adolescents: safety considerations. *Drug Saf* 2007; 30:17-26.
- 549. Lim JR, Faught PR, Chalasani NP, Molleston JP. Severe liver injury after initiating therapy with atomoxetine in two children. *J Pediatr* 2006; 148:831-4.
- 550. Stojanovski SD, Casavant MJ, Mousa HM, Baker P, Nahata MC. Atomoxetineinduced hepatitis in a child. *Clin Toxicol* 2007; 45:51-5.
- 551. Swanson J, Greenhill L, Wigal T, Kollins S, Stehli A, Davies M, Chuang S, Vitiello B, Skrobala A, Posner K, Abikoff H, Oatis M, McCracken J, McGough J, Riddle M, Ghuman J, Cunningham C, Wigal S. Stimulant-related reductions of growth rates in the PATS. J Am Acad Child Adolesc Psychiatry 2006; 45:1304-13.
- Lisska MC, Rivkees SA. Daily methylphenidate use slows the growth of children: a community based study. J Pediatr Endocrinol Metab 2003; 16:711-8.
- 553. Pliszka SR, Matthews TL, Braslow KJ, Watson MA. Comparative effects of methylphenidate and mixed salts amphetamine on height and weight in children with attention-deficit/hyperactivity disorder. J Am Acad Child Adolesc Psychiatry 2006; 45:520-6.
- 554. Swanson JM, Hinshaw SP, Arnold LE, Gibbons RD, Marcus S, Hur K, Jensen PS, Vitiello B, Abikoff HB, Greenhill LL, Hechtman L, Pelham WE, Wells KC, Conners CK, March JS, Elliott GR, Epstein JN, Hoagwood K, Hoza B, Molina BS, Newcorn JH, Severe JB, Wigal T. Secondary evaluations of MTA 36-month outcomes: propesinty score and growth mixture model analysis. J Am Acad Child Adolesc Psychiatry 2007; 46:1003-1013.
- 555. Charach A, Figueroa M, Chen S, Ickowicz A, Schachar R. Stimulant treatment over 5 years: effects on growth. *J Am Acad Child Adolesc Psychiatry* 2006; 45:415-21.
- 556. Faraone SV, Biederman J, Monuteaux M, Spencer T. Long-term effects of extended-release mixed amphetamine salts treatment of attention-deficit/hyperactivity disorder on growth. *J Child Adolesc Psychopharmacol* 2005; 15:191-202.
- 557. Poulton A, Cowell CT. Slowing of growth in height and weight on stimulants: a characteristic pattern. *J Paediatr Child Health* 2003; 39:180-5.
- 558. Zachor DA, Roberts AW, Hodgens JB, Isaacs JS, Merrick J. Effects of long-term psychostimulant medication on growth of children with ADHD. *Res Dev Disabil* 2006; 27:162-74.
- 559. Faraone SV, Giefer E. Long-term effects of methylphenidate transdermal delivery system treatment of ADHD on growth. *J Am Acad Child Adolesc Psychiatry* 2007; 46:1138-1147.
- 560. Kramer JR, Loney JAN, Ponto LB, Roberts MA, Grossman S. Predictors of adult height and weight in boys treated with methylphenidate for childhood behavior problems. *J Am Acad Child Adolesc Psychiatry* 2000; 39:517.
- 561. Kratochvil CJ, Wilens TE, Greenhill LL, Gao H, Baker KD, Feldman PD, Gelowitz DL. Effects of long-term atomoxetine treatment for young children with attention-deficit/hyperactivity disorder. *J Am Acad Child Adolesc Psychiatry* 2006; 45:919-27.
- 562. Spencer TJ, Kratochvil CJ, Sangal RB, Saylor KE, Bailey CE, Dunn DW, Geller DA, Casat CD, Lipetz RS, Jain R, Newcorn JH, Ruff DD, Feldman PD, Furr AJ, Allen AJ. Effects of Atomoxetine on growth in children with Attention-Deficit/Hyperactivity Disorder following up to five years of treatment. J Child Adolesc Psychopharmacol 2007; 17:689-700.

- 563. Spencer TJ, Newcorn JH, Kratochvil CJ, Ruff D, Michelson D, Biederman J. Effects of atomoxetine on growth after 2-year treatment among pediatric patients with attention-deficit/hyperactivity disorder. *Pediatrics* 2005; 116:e74-80.
- 564. Poulton A. Growth on stimulant medication; clarifying the confusion: a review. *Arch Dis Child* 2005; 90:801-6.
- 565. Villalba L. Follow up review of AERS search identifying cases of sudden death occurring with drugs used for the treatment of Attention Deficit Hyperactivity Disorder (ADHD): US Food and Drug Administration, 2006.
- 566. Gelperin K, Benoit S, Pamer C. Review of AERS data from marketed safety experience during stimulant therapy: death, sudden death, cardiovascular SAEs (including stroke): Department of Health and Human Services, Public Health Service, Food and Drug Administration, Center for Drug Evaluation and Reseaerch, 2004.
- 567. Wilens T, Hammerness PG, Biederman J, Kwon A, Spencer TJ, Clark S, Scott M, Podolski A, Ditterline JW, Morris MC, Moore H. Blood pressure changes associated with medication treatment of adults with attention-deficit/hyperactivity disorder. *J Clin Psychiatry* 2005; 66:253-259.
- 568. Samuels JA, Franco K, Wan F, Sorof JM. Effect of stimulants on 24-h ambulatory blood pressure in children with ADHD: a double-blind, randomized, cross-over trial. *Pediatric Nephrology* 2006; 21:92-5.
- 569. Donner R, Michaels MA, Ambrosini PJ. Cardiovascular effects of mixed amphetamine salts extended release in the treatment of school-aged children with attention-deficit/hyperactivity disorder. *Biol Psychiatry* 2007; 61:706-12.
- 570. Findling RL, Biederman J, Wilens TE, Spencer TJ, McGough JJ, Lopez FA, Tulloch SJ. Short- and long-term cardiovascular effects of mixed amphetamine salts extended release in children. *J Pediatr* 2005; 147:348-54.
- 571. Wilens T, Spencer TJ, Biederman J. Short- and long-term cardiovascular effects of mixed amphetamine salts extended-release in adolescents with ADHD. *CNS Spectrums* 2005; 10 (Suppl 15):22-30.
- 572. Wernicke J, Faries D, Girod D, Brown J, Gao H, Kelsey D, Quintana H, Lipetz R, Michelson D, Heiligenstein J. Cardiovascular effects of atomoxetine in children, adolescents, and adults. *Drug Safety* 2003; 26:729-40.
- 573. Winterstein AG, Gerhard T, Shuster J, Johnson M, Zito JM, Saidi A. Cardiac safety of central nervous system stimulants in children and adolescents with attention-deficit/hyperactivity disorder. *Pediatrics* 2007; 120:e1494-501.
- 574. Vetter VL, Elia J, Erickson C, Berger S, Blum N, Uzark K, Webb CL. Cardiovascular monitoring of children and adolescents with heart disease receiving stimulant drugs: a scientific statement from the American Heart Association Council on Cardiovascular Disease in the Young Congenital Cardiac Defects Committee and the Council on Cardiovascular Nursing. *Circulation* 2008; 117:2407-23
- 575. The Phamraceutical Benefits Scheme (PBS). Extended-release methylphenidate (Concerta) for attention deficit hyperactivity disorder. 2007.
- 576. Sawyer MG, Arney FM, Baghurst PA, Clark JJ, Graetz BW, Kosky RJ, Nurcombe B, Patton GC, Prior MR, Raphael B, Rey JM, Whaites LC, Zubrick SR. The mental health of young people in Australia: key findings from the child and adolescent component of the national survey of mental health and well-being. *Aust N Z J Psychiatry* 2001; 35:806-14.
- 577. Cosgrave EM, Robinson J, Godfrey KA, Yuen HP, Killackey EJ, Baker KD, Buckby JA, Yung AR. Outcome of suicidal ideation and behavior in a young, help-seeking population over a 2-year period. *Crisis* 2007; 28:4-10.
- 578. McKelvey RS, Pfaff JJ, Acres JG. The relationship between chief complaints, psychological distress, and suicidal ideation in 15-24-year-old patients presenting to general practitioners. *Med J Aust* 2001; 175:550-2.

- 579. Sawyer MG, Carbone JA, Searle AK, Robinson P. The mental health and wellbeing of children and adolescents in home-based foster care. *Med J Aust* 2007; 186:181-4.
- 580. Paxton G, Cranswick N. Acute suicidality after commencing atomoxetine *J Paediatr Child Health* 2008; 44:596-598.
- 581. Mosholder A. Psychiatric adverse events in clinical trials of drugs for attention deficit hyperactivity disorder (ADHD): Food and Drug Administration, 2006.
- 582. Bangs ME, Tauscher-Wisniewski S, Polzer J, Zhang S, Acharya N, Desaiah D, Trzepacz PT, Allen AJ. Meta-analysis of suicide-related behavior events in patients treated with atomoxetine. *J Am Acad Child Adolesc Psychiatry* 2008; 47:209-218.
- 583. Bangs M, Tauscher-Wisniewski S, Polzer J, Zhang S, Acharya N, Desaiah D, Allen A. Meta-analysis of suicide-related events in atomoxetine-treated patients (Eli Lilly & Company), 2006.
- 584. Wilens TE, Adler LA, Adams J, Sgambati S, Rotrosen J, Sawtelle R, Utzinger L, Fusillo S. Misuse and diversion of stimulants prescribed for ADHD: A systematic review of the literature. *J Am Acad Child Adolesc Psychiatry* 2008; 47:21-31.
- 585. Drug and Alcohol Office; Government of Western Australia. Statistical Bulletin No. 37, Summary of West Australian Results Australian School Students' Alcohol & Drug Survey, 2005, 2007.
- 586. Miller J, Lang A. ASSAD Drug Report 2005
- http://www.dao.health.wa.gov.au/Publications/tabid/99/DMXModule/427/Default.as px?EntryId=693. 2007.
- 587. Stevens M. Response to Calver et al. on the WA regulatory scheme for stimulants. *Aust New Zeal J Publ Health* 2008; 32:182-182.
- 588. Calver J, Sanfilippo F, Preen D, Bulsara M. Prescribed stimulant use by Western Australians with Attention Deficit Hyperactivity Disorder (ADHD): does amount dispensed exceed the expected authorised use? *Aust New Zeal J Publ Health* 2007; 31:533-9.
- 589. Molina BSG, Pelham WE, Gnagy EM, Thompson AL, Marshal MP. Attention-DeficitHyperactivity Disorder risk for heavy drinking and alcohol use disorder is age specific. *Alcohol Clin Exp Res* 2007; 31:643-654.
- 590. Fischer M, Barkley RA, Smallish L, Fletcher K. Young adult follow-up of hyperactive children: Self-reported psychiatric disorders, comorbidity, and the role of childhood conduct problems and teen CD. J Abnorm Child Psychol 2002; 30:463–475.
- 591. Lambert NM, Hartsough CS. Prospective study of tobacco smoking and substance dependencies among samples of ADHD and non-ADHD participants. *J Learn Disabil* 1998; 31:533-44.
- 592. Tercyak KP, Lerman C, Audrain J. Association of attention-deficit/hyperactivity disorder symptoms with levels of cigarette smoking in a community sample of adolescents. *J Am Acad Child Adolesc Psychiatry* 2002; 41:799-805.
- 593. Pomerleau OF, Downey KK, Stelson FW, Pomerleau CS. Cigarette smoking in adult patients diagnosed with attention deficit hyperactivity disorder. J Subst Abuse 1995; 7:373-8.
- 594. Rodriguez D, Tercyak KP, Audrain-McGovern J. Effects of inattention and hyperactivity/impulsivity symptoms on development of nicotine dependence from mid adolescence to young adulthood. *J Pediatr Psychol* 2007.
- 595. Wilens T, Adamson J, Sgambati S, Whitley J, Santry A, Monuteaux MC, Biederman J. Do individuals with ADHD self-medicate with cigarettes and substances of abuse? Results from a controlled family study of ADHD. *Am J Addict* 2007; 16:14-23.
- 596. Kollins S. Abuse liability of medications used to treat attentiondeficit/hyperactivity disorder (ADHD). *Am J Addict* 2007; 16:35-44.
- 597. Shaw K, Mitchell G, Hilton D. Are stimulants addictive in children? What the evidence says. *Aust Fam Physician* 2000; 29:1202-1204.

- 598. Spencer TJ, Biederman J, Ciccone PE, Madras BK, Dougherty DD, Bonab AA, Livni E, Parasrampuria DA, Fischman AJ. PET study examining pharmacokinetics, detection and likeability, and dopamine transporter receptor occupancy of short- and long-acting oral methylphenidate. *Am J Psychiatry* 2006; 163:387-95.
- 599. Heil SH, Holmes HW, Bickel WK, Higgins ST, Badger GJ, Laws HF, Faries DE. Comparison of the subjective, physiological, and psychomotor effects of atomoxetine and methylphenidate in light drug users. *Drug Alcohol Depend* 2002; 67:149-56.
- 600. Wee S, Woolverton WL. Evaluation of the reinforcing effects of atomoxetine in monkeys: comparison to methylphenidate and desipramine. *Drug Alcohol Depend* 2004; 75:271-6.
- 601. Natioanl Prescribing Service. Rational Assessment of Drugs and Research. Atomoxetine (Strattera) for attention deficit hyperactivity disorder. 2007.
- 602. Myrick H, Malcolm R, Taylor B, LaRowe S. Modafinil: preclinical, clinical, and post-marketing surveillance--a review of abuse liability issues. *Ann Clin Psychiatry* 2004; 16:101-9.
- 603. Szobot CM, Rohde LA, Bukstein O, Molina BSG, Martins C, Ruaro P, Pechansky F. Is attention-deficit/hyperactivity disorder associated with illicit substance use disorders in male adolescents? A community-based case-control study. *Addiction* 2007; 102:1122-1130.
- 604. Clure C, Brady KT, Saladin ME, Johnson D, Waid R, Rittenbury M. Attentiondeficit/hyperactivity disorder and substance use: symptom pattern and drug choice. *Am J Drug Alcohol Abuse* 1999; 25:441-8.
- 605. Horner BR, Scheibe KE. Prevalence and implications of attention-deficit hyperactivity disorder among adolescents in treatment for substance abuse. J Am Acad Child Adolesc Psychiatry 1997; 36:30-6.
- 606. Wise BK, Cuffe SP, Fischer T. Dual diagnosis and successful participation of adolescents in substance abuse treatment. *J Subst Abuse Treat* 2001; 21:161-165.
- 607. Kidorf M, Disney ER, King VL, Neufeld K, Beilenson PL, Brooner RK. Prevalence of psychiatric and substance use disorders in opioid abusers in a community syringe exchange program. *Drug Alcohol Depend* 2004; 74:115-22.
- 608. Levin FR, Evans SM, Kleber HD. Prevalence of adult attention-deficit hyperactivity disorder among cocaine abusers seeking treatment. *Drug Alcohol Depend* 1998; 52:15-25.
- 609. Luty J, Sarkhel A, O'Gara C, Umoh O. Prevalence of childhood attention deficit hyperactivity disorder in opiate-dependent adults. *Int J Psychiatr Clin Pract* 2007; 11:157-162.
- 610. Schubiner H, Tzelepis A, Milberger S, Lockhart N, Kruger M, Kelley BJ, Schoener EP. Prevalence of attention-deficit/hyperactivity disorder and conduct disorder among substance abusers. J Clin Psychiatry 2000; 61:244-51.
- 611. Yewers TM, Hay DA, Barton A. Attention deficit hyperactivity disorder and severity of drug use in a sample of adult male drug users *Australian Psychologist* 2005; 40:109-117.
- 612. Wilens TE, Faraone S, Biederman J, Gunawardene S. Does stimulant therapy of attention-deficit/hyperactivity disorder beget later substance abuse? A meta-analytic review of the literature. *Pediatrics* 2003; 111:179-185.
- 613. Faraone SV, Biederman J, Wilens TE, Adamson J. A naturalistic study of the effects of pharmacotherapy on substance use disorders among ADHD adults. *Psychol Med* 2007; 37:1743-1752.
- 614. Goksoyr PK, Nottestad JA. The burden of untreated ADHD among adults: The role of stimulant medication. *Addict Behav* 2008; 33:342-6.
- 615. Katusic SK, Barbaresi WJ, Colligan RC, Weaver AL, Leibson CL, Jacobsen SJ. Psychostimulant treatment and risk for substance abuse among young adults with a history of attention-deficit/hyperactivity disorder: A population-based, birth cohort study. J Child Adolesc Psychopharmacol 2005; 15:764-776.

- 616. Biederman J, Monuteaux MC, Spencer T, Wilens TE, Macpherson HA, Faraone SV. Stimulant therapy and risk for subsequent substance use disorders in male adults with ADHD: a naturalistic controlled 10-year follow-up study. *Am J Psychiatry* 2008; 165:597-603.
- 617. Chilcoat HD, Breslau N. Pathways from ADHD to early drug use. J Am Acad Child Adolesc Psychiatry 1999; 38:1347-54.
- 618. Schubiner H, Saules KK, Arfken CL, Johanson CE, Schuster CR, Lockhart N, Edwards A, Donlin J, Pihlgren E. Double-blind placebo-controlled trial of methylphenidate in the treatment of adult ADHD patients with comorbid cocaine dependence. *Exp Clin Psychopharmacol* 2002; 10:286-94.
- 619. Carpentier PJ, de Jong CAJ, Dijkstra BAG, Verbrugge CAG, Krabbe PFM. A controlled trial of methylphenidate in adults with attention deficit/hyperactivity disorder and substance use disorders. *Addiction* 2005; 100:1868-74.
- 620. Levin FR, Evans SM, Brooks DJ, Kalbag AS, Garawi F, Nunes EV. Treatment of methadone-maintained patients with adult ADHD: double-blind comparison of methylphenidate, bupropion and placebo. *Drug Alcohol Depend* 2006; 81:137-48.
- 621. Levin FR, Evans SM, Brooks DJ, Garawi F. Treatment of cocaine dependent treatment seekers with adult ADHD: double-blind comparison of methylphenidate and placebo. *Drug Alcohol Depend* 2007; 87:20-9.
- 622. Wilens TE, Biederman J, Mick E, Faraone SV, Spencer T. Attention deficit hyperactivity disorder (ADHD) is associated with early onset substance use disorders. *J Nerv Ment Dis* 1997; 185:475-82.
- 623. Carroll KM, Rounsaville BJ. History and significance of childhood attention deficit disorder in treatment-seeking cocaine abusers. *Compr Psychiatry* 1993; 34:75-82.
- 624. Latimer WW, Ernst J, Hennessey J, Stinchfield RD, Winters KC. Relapse among adolescent drug abusers following treatment: the role of probable ADHD status. *J Child Adolesc Subst Abuse* 2004; 13:1-16.
- 625. Levin FR, Evans SM, Vosburg SK, Horton T, Brooks D, Ng J. Impact of attention-deficit hyperactivity disorder and other psychopathology on treatment retention among cocaine abusers in a therapeutic community. *Addict Behav* 2004; 29:1875-82.
- 626. Wilens TE, Biederman J, Mick E. Does ADHD affect the course of substance abuse? Findings from a sample of adults with and without ADHD. *Am J Addict* 1998; 7:156-63.
- 627. Kolpe M, Carlson GA. Influence of attention-deficit/hyperactivity disorder symptoms on methadone treatment outcome. *Am J Addict* 2007; 16:46-8.
- 628. Wilens TE, Monuteaux MC, Snyder LE, Moore H, Whitley J, Gignac M. The clinical dilemma of using medications in substance-abusing adolescents and adults with attention-deficit/hyperactivity disorder: what does the literature tell us? J Child Adolesc Psychopharmacol 2005; 15:787-98.
- 629. Mariani J, Levin F. Treatment strategies for co-occurring ADHD and substance use disorders. *Am J Addict* 2007; 16:45-56.
- 630. Wilens T, Fusillo S. When ADHD and substance use disorders intersect: Relationship and treatment implications. *Curr Psychiatry Rep* 2007; 9:408-414.
- 631. The World Anti-Doping Agency. The World Anti-Doping Code: The 2008 prohibited list international standard, 2008.
- 632. Orchard JW, Fricker PA, White SL, Burke LM, Healey DJ. 4. The use and misuse of performance-enhancing substances in sport. *Med J Aust* 2006; 184:132-6.
- 633. Corrigan B. Attention deficit hyperactivity disorder in sport: a review. Int J Sports Med 2003; 24:535-40.
- 634. Pelham WE, Jr., McBurnett K, Harper GW, Milich R, Murphy DA, Clinton J, Thiele C. Methylphenidate and baseball playing in ADHD children: who's on first? *J Consult Clin Psychol* 1990; 58:130-3.

- 635. Vickers JR, ST, Brown L. Gaze pursuit and arm control of adolescent males diagnosed with attention deficit hyperactivity disorder (ADHD) and normal controls: evidence of a dissociation in processing visual information of short and long duration. *J Sports Sci* 2002; 20:201-16.
- 636. Fabiano G, Pelham Jr. W, Gnagy E, Burrows-MacLean L, Coles E, Chacko A, Wymbs B, Walker K, Arnold F, Garefino A, Keenan J, Onyango A, Hoffman M, Massetti G, Robb J. The single and combined effects of multiple intensities of behavior modification and methylphenidate for children with attention deficit hyperactivity disorder in a classroom setting. *Sch Psychol Rev* 2007; 36:195 217.
- 637. Van der Oord S, Prins PJ, Oosterlaan J, Emmelkamp PM. Efficacy of methylphenidate, psychosocial treatments and their combination in school-aged children with ADHD: A meta-analysis. *Clin Psychol Rev* 2007; In press.
- 638. Pelham WE, Burrows-Maclean L, Gnagy EM, Fabiano GA, Coles EK, Tresco KE, Chacko A, Wymbs BT, Wienke AL, Walker KS, Hoffman MT. Transdermal methylphenidate, behavioral, and combined treatment for children with ADHD. *Exp Clin Psychopharmacol* 2005; 13:111-26.
- 639. Chacko A, Pelham WE, Gnagy EM, Greiner A, Vallano G, Bukstein O, Rancurello M. Stimulant medication effects in a summer treatment program among young children with attention-deficit/hyperactivity disorder. J Am Acad Child Adolesc Psychiatry 2005; 44:249-57.
- 640. Abikoff H, Hechtman L, Klein RG, Weiss G, Fleiss K, Etcovitch J, Cousins L, Greenfield B, Martin D, Pollack S. Symptomatic improvement in children with ADHD treated with long-term methylphenidate and multimodal psychosocial treatment. J Am Acad Child Adolesc Psychiatry 2004; 43:802-11.
- 641. Abikoff H, Hechtman L, Klein RG, Gallagher R, Fleiss K, Etcovitch J, Cousins L, Greenfield B, Martin D, Pollack S. Social functioning in children with ADHD treated with long-term methylphenidate and multimodal psychosocial treatment. J Am Acad Child Adolesc Psychiatry 2004; 43:820-9.
- 642. Klein RG, Abikoff H, Hechtman L, Weiss G. Design and rationale of controlled study of long-term methylphenidate and multimodal psychosocial treatment in children with ADHD. J Am Acad Child Adolesc Psychiatry 2004; 43:792-801.
- 643. Majewicz-Hefley A, Carlson JS. A meta-analysis of combined treatments for children diagnosed with ADHD. J Attention Disord 2007; 10:239-50.
- 644. van der Oord S, Prins PJ, Oosterlaan J, Emmelkamp PM. Does brief, clinically based, intensive multimodal behavior therapy enhance the effects of methylphenidate in children with ADHD? *Eur Child Adolesc Psychiatry* 2007; 16:48-57.
- 645. Safren SA, Otto MW, Sprich S, Winett CL, Wilens TE, Biederman J. Cognitivebehavioral therapy for ADHD in medication-treated adults with continued symptoms. *Behav Res Ther* 2005; 43:831-42.
- 646. Wilens TE, McDermott SP, Biederman J, Abrantes A, Hahesy A, Spencer TJ. Cognitive therapy in the treatment of adults with ADHD: A systematic chart review of 26 cases. J Cognit Psychother 1999; 13:215-226.
- 647. The MTA Cooperative Group. Moderators and mediators of treatment response for children with attention-deficit/ hyperactivity disorder: The multimodal treatment study of children with attention-deficit/hyperactivity disorder. *Arch Gen Psychiatry* 1999; 56:1088-1096.
- 648. Owens EB, Hinshaw SP, Kraemer HC, Arnold LE, Abikoff HB, Cantwell DP, Conners CK, Elliott G, Greenhill LL, Hechtman L, Hoza B, Jensen PS, March JS, Newcorn JH, Pelham WE, Severe JB, Swanson JM, Vitiello B, Wells KC, Wigal T. Which treatment for whom for ADHD? Moderators of treatment response in the MTA. J Consult Clin Psychol 2003; 71:540-52.
- 649. Rieppi R, Greenhill LL, Ford RE, Chuang S, Wu M, Davies M, Abikoff HB, Arnold LE, Conners CK, Elliott GR, Hechtman L, Hinshaw SP, Hoza B, Jensen PS, Kraemer HC, March JS, Newcorn JH, Pelham WE, Severe JB, Swanson JM,

Vitiello B, Wells KC, Wigal T. Socioeconomic status as a moderator of ADHD treatment outcomes. J Am Acad Child Adolesc Psychiatry 2002; 41:269-77.

- 650. Odom S, Brantlinger E, Gersten R, Horner R, Thompson B, Harris K. Research in special education: Scientific methods and evidence-based practices. *Exceptional Children* 2005; 71:137-148.
- 651. McArthur G. Does What Works Clearinghouse work? A brief review of Fast ForWord *Australas J Spec Educ* 2008; 32:101-108.
- 652. Carter M, Wheldall K. Why can't a teacher be more like a scientist? Science, pseudoscience and the art of teaching. *Australas J Spec Educ* 2008; 32:5-21.
- 653. Mosteller F, Boruch R. Evidence matters: Randomized trials in educational research. Washington, DC: Brookings Institute, 2002.
- 654. Kauffman JM. Are we all postmodernists now? *Behavioral Disorders* 1998; 23:149-152.
- 655. Sasso GM. The retreat from knowledge and inquiry from special education. J Spec Educ 2001; 34 178.
- 656. Yates G. Roadblocks to scientific thinking in educational decision making Australas J Spec Educ 2008; 32:125 – 137
- 657. Giangreco M, Taylor M. "Scientifically based research" and qualitative inquiry. *Research and Practice for Persons with Severe Disabilities* 2003; 28:133-137.
- 658. Shaddock AJ, Giorcelli L, Smith S. Inclusive practice: A user-friendly guide for teachers. Canberra: Department of Education, Science & Training. Commonwealth of Australia, 2007.
- 659. Antshel KM, Faraone SV, Stallone K, Nave A, Kaufmann FA, Doyle A, Fried R, Seidman L, Biederman J. Is attention deficit hyperactivity disorder a valid diagnosis in the presence of high IQ? Results from the MGH Longitudinal Family Studies of ADHD. J Child Psychol Psychiatry 2007; 48:687-94.
- 660. Kos JM, Richdale AL, Jackson MS. Knowledge of attention-deficit/hyperactivity disorder: A comparison of in-service and pre-service teachers. *Psychol Schools* 2004; 41:517-526.
- 661. Beckle B. Knowledge and attitudes about Attention-Deficit Hyperactivity Disorder (ADHD): A comparison between practicing teachers and undergraduate education students. *J Attention Disord* 2004; 7:151-161.
- 662. West J, Taylor M, Houghton S, Hudyma S. A comparison of teachers' and parents' knowledge and beliefs about attention-deficit/hyperactivity disorder (ADHD). *Sch Psychol Int* 2005; 26:192-208.
- 663. Sciutto MJ, Terjesen MD, Bender Frank AS. Teachers' knowledge and misconceptions of attention-deficit/hyperactcity disorder. *Psychol Schools* 2000; 37:115-122.
- 664. Kos JM, Richdale AL, Hay DA. Children with Attention Deficit Hyperactivity Disorder and their teachers: A review of the literature. *Int J Disabil Dev Educ* 2006; 53:147-160.
- 665. DEST. Department of Education Science and Training. A class act: Inquiry into the status of the teaching profession. Canberra: Australian Government Printing Service. 1998.
- 666. DEST. Department of Education Science and Training. Senate inquiry into the education of students with disabilities. Canberra: AGPS. 2002.
- 667. DEST. Department of Education Science and Training. Australia's teachers: Australia's future. Canberra. 2003.
- 668. Ford J. Educational supports for students with disabilities and significant behavioural challenges: Teacher perceptions. *Australas J Spec Educ* 2008; 31:109-127.
- 669. Ramsey G. Quality matters. Revitalising teaching: Critical times, critical choices. Report of the review of teacher education, New South Wales. Sydney: Department of Education and Training New South Wales. 2000.
- 670. Vinson T. Inquiry into the provision of public education in NSW. Sydney: NSW Teachers' Federation & Federation of P&C Associations of NSW. 2002.

- 671. Pearce M, Forlin C. Challenges and potential solutions for enabling inclusion in secondary schools. *Australas J Spec Educ* 2005; 29:93-106.
- 672. Efron D, Sciberras E, Hassall P. Are schools meeting the needs of students with ADHD? *Australian Journal of Education, in press.* 2008.
- 673. Rudland N, Kemp C. The professional reading habits of teachers: Implications for student learning. *Australas J Spec Educ* 2004; 28:4-17.
- 674. Boardman AG, Arguelles ME, Vaughn S, Hughes MT, Klinger J. Special education teachers' views of research-based practices. *J Spec Educ* 2005; 39:168-180.
- 675. Spira EG, Fischel JE. The impact of preschool inattention, hyperactivity and impulsivity on social and academic development: A review. *J Child Psychol Psychiatr* 2005; 46:755-773.
- 676. Deutscher BF, R.R. Attention deficit hyperactivity disorder in very young children: Early signs and interventions. *Infants & Young Children* 2002; 14:24-32.
- 677. Cantwell D, Baker L. Association between attention deficit hyperactivity disorder and learning disorders *J Learn Disabil* 1991; 24:88-95.
- 678. DuPaul G, Stoner G. ADHD in the schools: Assessment and intervention strategies. New York: Guilford, 2003.
- 679. Tannock R, Martinussen R. Reconceptualizing ADHD. *Educational Leadership* 2001; 59:20-25.
- 680. Loe IM, Feldman HM. Academic and educational outcomes of children with ADHD. *Ambul Pediatr* 2007; 7:82-90.
- 681. Tannock R, Martinussen R. Promising practices in education for students with ADHD. *Orbit* 2007; 37:32.
- 682. Baddeley AD. Working memory. Oxford: Clarendon Press, 2000.
- 683. Gathercole SE, Alloway TP. Practitioner review: Short-term and working memory impairments in neurodevelopmental disorders: diagnosis and remedial support. *J Child Psychol Psychiatr* 2006; 47:4-15.
- 684. Alloway TP, Gathercole SE, Pickering SJ. Verbal and visuospatial short-term and working memory in children: are they separable? *Child Dev* 2006; 77:1698-716.
- 685. Barkley RA. Attention-deficit/hyperactivity disorder, In: Barkley RA and Mash EJ, eds. Child Psychopathology (2nd ed). New York Guilford Press 2003:75-143.
- 686. Re AM, Pedron M, Cornoldi C. Expressive writing difficulties in children described as exhibiting ADHD symptoms. *J Learn Disabil* 2007; 40:244-255.
- 687. Douglas VI. Cognitive deficits in children with attention deficit hyperactivity disorder: A long-term follow-up. *Canadian Psychology* 2005; 46:23–31.
- 688. Tannock RM, Schachar R. Executive dysfunction as an underlying mechanism of behavior and language problems in attention deficit hyperactivity disorder, In: Beitchman JH, Cohen NJ, Konstantareas MM, and Tannock RM, eds. Language, learning and behavior problems. New York: Cambridge University Press, 1996:128–154.
- 689. DuPaul GJ, Jitendra AK, Tresca KE, Junod RE, Volpe R, Lutz JG. Children with attention hyperactivity deficit disorder: Are there gender differences in school functioning? *Sch Psychol Rev* 2006; 35:292-308.
- 690. Evans SW, Serpell ZN, Schultz B, Pastor DA. Cumulative benefits of secondary school-based treatment of students with Attention Deficit Hyperactivity Disorder. *Sch Psychol Rev* 2007; 36:256-273.
- 691. Robin A. ADHD in adolescents: Diagnosis and treatment. New York: Guildford Press, 1998.
- 692. Frazier TW, Youngstrom EA, Glutting JJ, Watkins MW. ADHD and achievement: meta-analysis of the child, adolescent, and adult literatures and a concomitant study with college students. *J Learn Disabil* 2007; 40:49-65.

- 693. Richards TL, Rosen LA, Ramirez CA. Psychological functioning differences among college students with confirmed ADHD, ADHD by self-report only, and without ADHD. J Coll Student Dev 1999; 40 299-304.
- 694. Weyandt LL, Iwaszuk W, Fulton K, Ollerton M, Beatty N, Fouts H, Schepman S, Greenlaw C. The internal restlessness scale: performance of college students with and without ADHD. *J Learn Disabil* 2003; 36:382-9.
- 695. Brown TE. Attention Deficit Disorder: An unfocused mind in children and adults. New Haven, CT: Yale Uni Press, 2005.
- 696. Pearce M. The inclusive secondary school teacher, In: Forlin C and Lian M-G, eds. Reform, Inclusion & Teacher Education: Towards a New Era of Special Education in the Asia-Pacific Region. Abingdon: Routledge., 2008.
- 697. Eber L, Sugai G, Smith C, Scott T. Wraparound and positive behavioural interventions and supports in schools. *J Emot Behav Disord* 2002; 10:171-180.
- 698. DuPaul GJ, Eckert TL. The effects of school-based interventions for attention deficit hyperactivity disorder: A meta-analysis. *Sch Psychol Rev* 1997; 26:5-27.
- 699. Barkley RA, Shelton TL, Crosswait C, Moorehouse M, Fletcher K, Barrett S, Jenkins L, Metevia L. Multi-method psycho-educational intervention for preschool children with disruptive behavior: preliminary results at post-treatment. J Child Psychol Psychiatr 2000; 41:319-32.
- 700. Miranda A, Presentacion MJ. Efficacy of cognitive-behavioural therapy in the treatment of children with ADHD, with and without aggressiveness *Psychol Schools* 2000; 37:169-182.
- 701. Miranda A, Presentacion MJ, Soriano M. Effectiveness of a school-based multicomponent program for the treatment of children with ADHD. *J Learn Disabil* 2002; 35:546-62.
- 702. Miranda A, Jarque S, Rosel J. Treatment of children with ADHD: psychopedagogical program at school versus psychostimulant medication. *Psicothema* 2006; 18:335-41.
- 703. Klingberg T, Fernell E, Olesen PJ, Johnson M, Gustafsson P, Dahlstrom K, Gillberg CG, Forssberg H, Westerberg H. Computerized training of working memory in children with ADHD--a randomized, controlled trial. J Am Acad Child Adolesc Psychiatry 2005; 44:177-86.
- 704. Jitendra A, DuPaul G, Volpe R, Tresco K, Vile Junod R, Lutz G, Cleary K, Flammer-Rivera L, Mannella M. Consultation-based academic intervention for children with attention deficit hyperactivity disorder: school functioning outcomes. Sch Psychol Rev 2007; 36:217 - 237.
- 705. Hoza B, Mrug S, Pelham WE, Jr., Greiner AR, Gnagy EM. A friendship intervention for children with Attention-Deficit/Hyperactivity Disorder: preliminary findings. *J Attention Disord* 2003; 6:87-98.
- 706. Strayhorn JM, Jr., Bickel DD. Reduction in children's symptoms of attention deficit hyperactivity disorder and oppositional defiant disorder during individual tutoring as compared with classroom instruction. *Psychol Rep* 2002; 91:69-80.
- 707. DuPaul GJ, Ervin RA, Hook CL, McGoey KE. Peer tutoring for children with attention deficit hyperactivity disorder: effects on classroom behavior and academic performance. J Appl Behav Anal 1998; 31:579-92.
- 708. Meyer K, Kelley K. Improving homework in adolescents with Attention-Deficit/Hyperactivity Disorder: Self v parent monitoring of homework behaviours. *Child Fam Behav Ther* 2007; 29:25-42.
- 709. Bradshaw CP, Reinke WM, Brown LD, Bevans KB, Leaf PJ. Implementation of school-wide Positive Behavioral Interventions and Supports (PBIS) in elementary schools: Observations from a randomized trial. *Education & Treatment of Children* 2008; 31:1-26.
- 710. Cohen R, Kincaid D, Childs KE. Measuring school-wide positive behavior support implementation: Development and validation of the benchmarks of quality. *Journal of Positive Behavior Interventions* 2007; 9:203-213.

- 711. Harvey MT, Lewis-Palmer T, Horner RH, Sugai G. Trans-situational interventions: Generalization of behavior support across school and home environments. *Behavioral Disorders* 2003; 28:299-312.
- 712. Gureasko-Moore S, DuPaul GJ, White GP. Self-management of classroom preparedness and homework: Effects on school functioning of adolescents with Attention Deficit Hyperactivity Disorder. *Sch Psychol Rev* 2007; 36:647-664.
- 713. Fenstermacher K, Olympia D, Sheridan SM. Effectiveness of a computerfacilitated interactive social skills training program for boys with attention deficit hyperactivity disorder. *Sch Psychol Q* 2006; 21:197-224.
- 714. Spencer VG. Peer tutoring and students with emotional or behavioural disorders: A review of the literature. *Behavioral Disorders* 2006; 31:204-222.
- 715. Tournaki N, Criscitiello E. Using peer tutoring as an effective part of behavior management. *Teaching Exceptional Children* 2003; 36:22-29.
- 716. Glomb N, Buckley L, Minskoff E, Rogers S. The learning leaders mentoring program for children with ADHD and learning disabilities. *Preventing School Failure* 2006; 50:31-35.
- 717. Shelton TL, Barkley RA, Crosswait C, Moorehouse M, Fletcher K, Barrett S, Jenkins L, Metevia L. Multimethod psychoeducational intervention for preschool children with disruptive behavior: two-year post-treatment follow-up. J Abnorm Child Psychol 2000; 28:253-66.
- 718. South M, Lim A. Use of complementary and alternative medicine in children: Too important to ignore. *J Paediatr Child Health* 2003; 39:573-574.
- 719. Sinha D, Efron D. Complementary and alternative medicine use in children with attention deficit hyperactivity disorder. J Paediatr Child Health 2005; 41:23-6.
- 720. Stubberfield T, Parry T. Utilization of alternative therapies in attention-deficit hyperactivity disorder. *J Paediatr Child Health* 1999; 35:450-3.
- 721. Schmidt MH, Mocks P, Lay B, Eisert HG, Fojkar R, Fritz-Sigmund D, Marcus A, Musaeus B. Does oligoantigenic diet influence hyperactive/conduct-disordered children--a controlled trial. *Eur Child Adolesc Psychiatry* 1997; 6:88-95.
- 722. Pelsser LM, Frankena K, Toorman J, Savelkoul HF, Pereira RR, Buitelaar JK. A randomised controlled trial into the effects of food on ADHD. *Eur Child Adolesc Psychiatry* 2008.
- 723. Rojas NL, Chan E. Old and new controversies in the alternative treatment of attention-deficit hyperactivity disorder. *Ment Retard Dev Disabil Res Rev* 2005; 11:116-30.
- 724. Cormier E, Harrison Elder J. Diet and child behaviour problems: Fact or fiction. *Pediatric Nursing* 2007; 33:138-143.
- 725. Connors CK, Goyette GH, Southwick DS, Lees JM, Andrulonis PA. Food additives and hyperkinesis: A double blind experiment. *Pediatrics* 1976; 58:154-166.
- 726. Harley JP, Ray RS, Tomasi L, Eichman PL, Matthews CG, Chun R, Cleeland CS, Traisman E. Hyperkinesis and food additives: testing the Feingold hypothesis. *Pediatrics* 1978; 61:818-28.
- 727. Kavale KA, Forness SR. Hyperactivity and diet treatment: a meta-analysis of the Feingold hypothesis. *J Learn Disabil* 1983; 16:324-30.
- 728. Egger J, Carter CM, Graham PJ, Gumley D, Soothill JF. Controlled trial of oligoantigenic treatment in the hyperkinetic syndrome. *Lancet* 1985; 1:540-5.
- 729. Carter CM, Urbanowicz M, Hemsley R, Mantilla L, Strobel S, Graham PJ, Taylor
 E. Effects of a few food diet in attention deficit disorder. Arch Dis Child 1993;
 69:564-8.
- 730. Voigt RG, Llorente AM, Jensen CL, Fraley JK, Berretta MC, Heird WC. A randomized, double-blind, placebo-controlled trial of docosahexaenoic acid supplementation in children with attention-deficit/hyperactivity disorder. *J Pediatr* 2001; 139:189-96.
- 731. Hirayama S, Hamazaki T, Terasawa K. Effect of docosahexaenoic acidcontaining food administration on symptoms of attention-deficit/hyperactivity

disorder - a placebo-controlled double-blind study. *Eur J Clin Nutr* 2004; 58:467-73.

- 732. Stevens L, Zhang W, Peck L, Kuczek T, Grevstad N, Mahon A, Zentall SS, Arnold LE, Burgess JR. EFA supplementation in children with inattention, hyperactivity, and other disruptive behaviors. *Lipids* 2003; 38:1007-21.
- 733. Sinn N, Bryan. Effect of supplementation with polyunsaturated fatty acids and micronutrients on ADHD-related problems with attention and behavior. *J Dev Behav Pediatr* 2007; 28:82-91.
- 734. Richardson AJ, Puri BK. A randomized double-blind, placebo-controlled study of the effects of supplementation with highly unsaturated fatty acids on ADHD-related symptoms in children with specific learning difficulties. *Prog Neuropsychopharmacol Biol Psychiatry* 2002; 26:233-9.
- 735. National Health and Medical Research Council (NHMRC). Nutrient Reference Values for Australia and New Zealand Including Recommended Dietary Intakes, 2006.
- 736. Richardson AJ. Omega-3 fatty acids in ADHD and related neurodevelopmental disorders. *Int Rev Psychiatry* 2006; 18:155-72.
- 737. Tsalamanios E, Yanni AE, Koutsari C. Omega-3 fatty acids: Role in the prevention and treatment of psychiatric disorders. *Curr Psychiatr Rev* 2006; 2 215-234.
- 738. Breuner CC. Complementary medicine in pediatrics: A review of acupuncture, homeopathy, massage, and chiropractic therapies. *Curr Probl Pediatr Adolesc Health Care* 2002; 32:353-384.
- 739. Riccio CA, French CL. The status of empirical support for treatments of attention deficits. *Clinical Neuropsychologist* 2004; 18:528-558.
- 740. Strehl U, Leins U, Goth G, Klinger C, Hinterberger T, Birbaumer N. Selfregulation of slow cortical potentials: a new treatment for children with attention-deficit/hyperactivity disorder. *Pediatrics* 2006; 118:e1530-40.
- 741. Heinrich H, Gevensleben H, Freisleder FJ, Moll GH, Rothenberger A. Training of slow cortical potentials in attention-deficit/hyperactivity disorder: evidence for positive behavioral and neurophysiological effects. *Biol Psychiatry* 2004; 55:772-5.
- 742. Monastra VJ, Monastra DM, George S. The effects of stimulant therapy, EEG biofeedback, and parenting style on the primary symptoms of attention-deficit/hyperactivity disorder. *Appl Psychophysiol Biofeedback* 2002; 27:231-49.
- 743. Fuchs T, Birbaumer N, Lutzenberger W, Gruzelier JH, Kaiser J. Neurofeedback treatment for attention-deficit/hyperactivity disorder in children: a comparison with methylphenidate. *Appl Psychophysiol Biofeedback* 2003; 28:1-12.
- 744. Rossiter T. The effectiveness of neurofeedback and stimulant drugs in treating AD/HD: part II. Replication. *Appl Psychophysiol Biofeedback* 2004; 29:233-43.
- 745. Banaschewski T, Brandeis D. Annotation: What electrical brain activity tells us about brain function that other techniques cannot tell us a child psychiatric perspective. J Child Psychol Psychiatr 2007; 48:415-435.
- 746. Altunc U, Pittler MH, Ernst E. Homeopathy for childhood and adolescence ailments: systematic review of randomized clinical trials. *Mayo Clinic Proceedings* 2007; 82:69-75.
- 747. Dantas F, Rampes H. Do homeopathic medicines provoke adverse effects? A systematic review. *Br Homeopath J* 2000; 89 Suppl 1:S35-8.
- 748. Rutherford R, Nicolson RI, Arnold E. Significant reduction in symptoms of attention deficit in learning-disbled children and adults following excersise based treatment. American Psycholological Association Convention, 2006.
- 749. Bishop DVM. Curing dyslexia and attention-deficit hyperactivity disorder by training motor co-ordination: Miracle or myth? *J Paediatr Child Health* 2007; 43:653-655.

- 750. Jensen P, Kenny D. The effects of yoga on the attention and behavior of boys with Attention-Deficit/hyperactivity Disorder (ADHD). *J Attention Disord* 2004; 7:205-216.
- 751. Harrison LJ, Manocha R, Rubia K. Sahaja Yoga Meditation as a family treatment programme for children with attention deficit-hyperactivity disorder. *Clin Child Psychol Psychiatry* 2004; 9:479-497.
- 752. Field TM, Quintino O, Hernandez-Reif M, Koslovsky G. Adolescents with attention deficit hyperactivity disorder benefit from massage therapy. *Adolescence* 1998; 33:103-8.
- 753. Khilnani S, Field T, Hernandez-Reif M, Schanberg S. Massage therapy improves mood and behavior of students with attention-deficit/hyperactivity disorder. *Adolescence* 2003; 38:623-38.
- 754. Hernandez-Reif M, Field TM, Thimas E. Attention Deficit Hyperactivity Disorder: Benefits from Tai Chi. *Journal of Bodywork and Movement Therapies* 2001; 5:120-123.
- 755. Ekeland E, Heian F, Hagen K, Abbott J, Nordheim L. Exercise to improve selfesteem in children and young people. *Cochrane database of systematic reviews (Online)* 2004.
- 756. Larun L, Nordheim LV, Ekeland E, Hagen KB, Heian F. Exercise in prevention and treatment of anxiety and depression among children and young people. *Cochrane database of systematic reviews (Online)* 2006; 3:CD004691.
- 757. Brown GT, Rodger S, Brown A, Roever C. A comparison of Canadian and Australian paediatric occupational therapists. *Occup Ther Int* 2005; 12:137-61.
- 758. Miller LJ, Coll JR, Schoen SA. A randomized controlled pilot study of the effectiveness of occupational therapy for children with sensory modulation disorder. *Am J Occup Ther* 2007; 61:228-38.
- 759. ACTIVE Inc. "Do What You Want To Do". A moderated panel discussion with six Melbourne teenagers who were taking stimulant medication for ADHD. Audio cassette produced by ACTIVE Inc, Melbourne., 1997.
- 760. Nadeau KG. Career choices and workplace challenges for individuals with ADHD. J Clin Psychol 2005; 61:549-63.
- Hervey AS, Epstein JN, Curry JF. Neuropsychology of adults with attentiondeficit/hyperactivity disorder: A meta-analytic review. *Neuropsychology* 2004; 18:485-503.
- 762. Young S, Morris R, Toone B, Tyson C. Planning ability in adults with attentiondeficit/hyperactivity disorder *Neuropsychology* 2007; 215:581-589.
- 763. Canu WH, Newman ML, Morrow TL, Pope DLW. Social appraisal of adult ADHD: Stigma and influences of the beholder's big five personality traits. *J Attention Disord* 2008; 11:700-710.
- 764. Swensen A, Birnbaum HG, Ben Hamadi R, Greenberg P, Cremieux PY, Secnik K. Incidence and costs of accidents among attention-deficit/hyperactivity disorder patients. *J Adolesc Health* 2004; 35:346 e1-9.
- 765. Reynolds SH. Attention deficit disorder in adults: the missing link in the chronic accident repeater syndrome *Professional Safety* 1997; 42:20-26.
- 766. Kessler RC, Lane M, Stang PE, Van Brunt DL. The prevalence and workplace costs of adult attention deficit hyperactivity disorder in a large manufacturing firm. *Psychol Med. in press* 2008.
- 767. Coetzer G, Richmond L. An empirical analysis of the relationship between adult attention deficit and efficacy for working in teams. *Team Performance Management* 2007; 1/2:5-20.
- 768. Biederman J, Mick E, Fried R, Aleardi M, Potter A, Herzig K. A simulated workplace experience for nonmedicated adults with and without ADHD. *Psychiatric Services* 2005; 56:1617-20.
- 769. Mannuzza S, Klein RG, Bessler A, Malloy P, Hynes ME. Educational and occupational outcome of hyperactive boys grown up. J Am Acad Child Adolesc Psychiatry 1997; 36:1222-7.

- 770. Biederman J, Faraone SV, Spencer T. Functional impairments in adults with self-reports of diagnosed ADHD: a controlled study of 1001 adults in the community. *J Clin Psychiatry* 2006; 67:524-540.
- 771. Attention Deficit Disorder Association. ADHD in the workplace. Available at http://www.add.org/pdf/AdultADHDReport.pdf, 2006.
- 772. Young S, Toone B, Tyson C. Comorbidity and psychosocial profile of adults with attention deficit hyperactivity disorder. *Pers Indiv Differ* 2003; 35:743-755.
- 773. Swensen AR, Birnbaum HG, Secnik K, Marynchenko M, Greenberg P, Claxton A. Attention-deficit/hyperactivity disorder: increased costs for patients and their families. *J Am Acad Child Adolesc Psychiatry* 2003; 42:1415-23.
- 774. Schweitzer JB, Lee DO, Hanford RB, Zink CF, Ely TD, Tagamets MA, Hoffman JM, Grafton ST, Kilts CD. Effect of methylphenidate on executive functioning in adults with attention-deficit/hyperactivity disorder: normalization of behavior but not related brain activity. *Biol Psychiatry* 2004; 56:597-606.
- 775. Simpson D, Plosker GL. Spotlight on atomoxetine in adults with attentiondeficit hyperactivity disorder. CNS Drugs 2004; 18:397-401.
- 776. Fallu A, Richard C, Prinzo R, Binder C. Does OROS-methylphenidate improve core symptoms and deficits in executive function? Results of an open-label trial in adults with attention deficit hyperactivity disorder. *Curr Med Res Opin* 2006; 22:2557-66.
- 777. Gerber P. Employment of Adults with Learning Disabilities and ADHD: Reasons for Success and Implications for Resilience. *ADHD Report* 2001; 9:1-5.
- 778. Canu WH. Vocational safety preference of college men with and without attention-deficit/hyperactivity disorder: An exploratory study. *J Coll Counsel* 2007; 10:54-63.
- 779. Austroads Inc. Assessing fitness to drive for commercial and private vehicle drivers. A resource for health professionals in Australia. Sydney Austroads Inc. , 2006.
- 780. Civil Aviation Safety Authority. Designated Aviation Medical Examiner's Handbook. Canberra: Australian Government, 2007.
- 781. Barkley RA, Anderson DL, Kruesi M. A pilot study of the effects of atomoxetine on driving performance in adults with ADHD. *J Attention Disord* 2007; 10:306-16.
- 782. Cox DJ, Punja M, Powers K, Merkel RL, Burket R, Moore M, Thorndike F, Kovatchev B. Manual transmission enhances attention and driving performance of ADHD adolescent males: pilot study. *J Attention Disord* 2006; 10:212-6.
- 783. Cox DJ, Merkel RL, Penberthy JK, Kovatchev B, Hankin CS. Impact of methylphenidate delivery profiles on driving performance of adolescents with attention-deficit/hyperactivity disorder: a pilot study. J Am Acad Child Adolesc Psychiatry 2004; 43:269-75.
- 784. Scahill L, Schwab-Stone M, Merikangas KR, Leckman JF, Zhang H, Kasl S. Psychosocial and clinical correlates of ADHD in a community sample of schoolage children. J Am Acad Child Adolesc Psychiatry 1999; 38:976-984.
- 785. Biederman J, Faraone SV, Mick E, Williamson S, Wilens TE, Spencer TJ, Weber W, Jetton J, Kraus I, Pert J, Zallen B. Clinical correlates of ADHD in females: findings from a large group of girls ascertained from pediatric and psychiatric referral sources. J Am Acad Child Adolesc Psychiatry 1999; 38:966-75.
- 786. Brown RT, Pacini JN. Perceived family functioning, marital status, and depression in parents of boys with attention deficit disorder. *J Learn Disabil* 1989; 22:581-587.
- 787. Smith AJ, Brown RT, Bunke V, Blount RL, Christophersen E. Psychosocial adjustment and peer competence of siblings of children with attention-deficit/hyperactivity disorder. *J Attention Disord* 2002; 5:165-177.
- 788. Ghanizadeh A, Shams F. Children's perceived parent-child relationships and family functioning in attention-deficit/hyperactivity disorder. *Child Fam Behav Ther* 2007; 29:1-11.

- 789. Niederhofer H, Hackenberg B, Lanzendorfer K, Staffen W, Mair A. Family coherence and ADHD. *Psychol Rep* 2002; 91:123-126.
- 790. Lange G, Sheerin D, Carr A, Dooley B, Barton V, Marshall D, Mulligan A, Lawlor M, Belton M, Doyle M. Family factors associated with attention deficit hyperactivity disorder and emotional disorders in children. *J Fam Ther* 2005; 27:76-96.
- 791. Biederman J, Milberger S, Faraone SV, Kiely K, Guite J, Mick E, Ablon S, Warburton R, Reed E. Family-environment risk factors for attention-deficit hyperactivity disorder: A test of Rutter's indicators of adversity. *Arch Gen Psychiatry* 1995; 52:464-470.
- 792. Biederman J, Faraone SV, Monuteaux MC. Differential effect of environmental adversity by gender: Rutter's index of adversity in a group of boys and girls with and without ADHD. *Am J Psychiatr* 2002; 159:1556-1562.
- 793. Biederman J, Faraone S, Milberger S, Curtis S, Chen L, Marrs A, Ouellette C, Moore P, Spencer T. Predictors of persistence and remission of ADHD into adolescence: Results from a four-year prospective follow-up study. *J Am Acad Child Adolesc Psychiatry* 1996; 35:343-351.
- 794. Birnbaum HG, Kessler RC, Lowe SW, Secnik K, Greenberg PE, Leong SA, Swensen AR. Costs of attention deficit-hyperactivity disorder (ADHD) in the US: excess costs of persons with ADHD and their family members in 2000. *Curr Med Res Opin* 2005; 21:195-206.
- 795. Chan E, Zhan C, Homer CJ. Health care use and costs for children with attention-deficit /hyperactivity disorder: National estimates from the medical expenditure panel survey. *Arch Pediatr Adolesc Med* 2002; 156:504-511.
- 796. Guevara J, Lozano P, Wickizer T, Mell L, Gephart H. Utilization and cost of health care services for children with attention-deficit/hyperactivity disorder. *Pediatrics* 2001; 108:71-78.
- 797. Leibson CL, Katusic SK, Barbaresi WJ, Ransom JBS, O'Brien PC. Use and costs of medical care for children and adolescents with and without attention-deficit/hyperactivity disorder. *JAMA* 2001; 285:60-66.
- 798. Pelham WE, Foster EM, Robb JA. The economic impact of attentiondeficit/hyperactivity disorder in children and adolescents. *Ambul Pediatr* 2007; 7:121-31.
- 799. Befera MS, Barkley RA. Hyperactive and normal girls and boys: Mother-child interaction, parent psychiatric status and child psychopathology. *J Child Psychol Psychiatr* 1985; 26:439-452.
- 800. Tripp G, Schaughency EA, Langlands R, Mouat K. Family interactions in children with and without ADHD. J Child Fam Stud 2007; 16:385-400.
- 801. Johnston C, Mash EJ. Families of children with attention-deficit/hyperactivity disorder: review and recommendations for future research. *Clin Child Fam Psychol Rev* 2001; 4:183-207.
- 802. Buhrmester D, Camparo L, Christensen A, Gonzalez LS, Hinshaw SP. Mothers and fathers interacting in dyads and triads with normal and hyperactive sons. *Developmental Psychology* 1992; 28:500-509.
- 803. Johnston C. Parent characteristics and parent-child interactions in families of nonproblem children and ADHD children with higher and lower levels of oppositional-defiant behavior. *J Abnorm Child Psychol* 1996; 24:85-104.
- 804. Tallmadge J, Barkley RA. The interactions of hyperactive and normal boys with their fathers and mothers. *J Abnorm Child Psychol* 1983; 11:565-579.
- 805. Cunningham CE, Barkley RA. The interactions of normal and hyperactive children with their mothers in free play and structured tasks. *Child Development* 1979; 50:217-224.
- 806. Mash EJ, Johnston C. A comparison of the mother-child interactions of younger and older hyperactive and normal children. *Child Development* 1982; 53:1371-1381.
- 807. Barkley RA, Fischer M, Edelbrock C, Smallish L. The adolescent outcome of hyperactive children diagnosed by research criteria-III. Mother-child

interactions, family conflicts and maternal psychopathology. *J Child Psychol Psychiatr* 1991; 32:233-255.

- 808. Keown LJ, Woodward LJ. Early parent-child relations and family functioning of preschool boys with pervasive hyperactivity. *J Abnorm Child Psychol* 2002; 30:541-553.
- 809. Barkley R, Du Paul G, McMurray M. Attention deficit hyperactivity disorder with and without hyperactivity; clinical response to 3 dose levels of methylphenidate. *Paediatrics* 1991; 519-531.
- 810. Tarver-Behring S, Barkley RA, Karlsson J. The mother-child interactions of hyperactive boys and their normal siblings. *Am J Orthopsychiatry* 1985; 55:202-209.
- 811. Gomez R, Sanson AV. Mother-child interactions and noncompliance in hyperactive boys with and without conduct problems. *J Child Psychol Psychiatr* 1994; 35:477-490.
- 812. Granic I, Lamey AV. Combining dynamic systems and multivariate analyses to compare the mother-child interactions of externalizing subtypes. *J Abnorm Child Psychol* 2002; 30:265-283.
- 813. Seipp CM, Johnston C. Mother-son interactions in families of boys with attention-deficit/hyperactivity disorder with and without oppositional behavior. *J Abnorm Child Psychol* 2005; 33:87-98.
- 814. Johnston C, Murray C, Hinshaw SP, Pelham WEJ, Hoza B. Responsiveness in interactions of mothers and sons with ADHD: Relations to maternal and child characteristics. *J Abnorm Child Psychol* 2002; 30:77-88.
- 815. Anastopoulos AD, Guevremont DC, Shelton TL, DuPaul GJ. Parenting stress among families of children with attention deficit hyperactivity disorder. J Abnorm Child Psychol 1992; 20:503-520.
- 816. Anastopoulos AD, Shelton TL, DuPaul GJ, Guevremont DC. Parent training for attention-deficit hyperactivity disorder: Its impact on parent functioning. *J Abnorm Child Psychol* 1993; 21:581-596.
- 817. Baker DB. Parenting stress and ADHD: A comparison of mothers and fathers. J *Emot Behav Disord* 1994; 2:46-50.
- 818. Breen MJ, Barkley RA. Child psychopathology and parenting stress in girls and boys having attention deficit disorder with hyperactivity. *J Pediatr Psychol* 1988; 13:265-280.
- 819. Podolski CL, Nigg JT. Parent stress and coping in relation to child ADHD severity and associated child disruptive behavior problems. *J Clin Child Psychol* 2001; 30:503-513.
- 820. Mash EJ, Johnston C. Parental perceptions of child behavior problems, parenting self-esteem, and mothers' reported stress in younger and older hyperactive and normal children. *J Consult Clin Psychol* 1983; 51:86-99.
- 821. Harrison C, Sofronoff K. ADHD and parental psychological distress: Role of demographics, child behavioural characteristics, and parental cognitions. *J Am Acad Child Adolesc Psychiatry* 2002; 41:703-711.
- 822. Coleman PK, Karraker KH. Self-efficacy and parenting quality: Findings and future applications. *Developmental Review* 1998; 18:47-85.
- 823. Hoza B, Owens JS, Pelham WE, Swanson JM, Conners CK, Hinshaw SP, Arnold LE, Kraemer HC. Parent cognitions as predictors of child treatment response in attention-deficit/hyperactivity disorder. J Abnorm Child Psychol 2000; 28:569-83.
- 824. Biederman J, Faraone SV, Keenan K, Benjamin J, Krifcher B, Moore C, Sprich-Buckminster S, Ugaglia K, Jellinek MS, Steingard R, Spencer T, Norman D, Kolodny R, Kraus I, Perrin J, Keller MB, Tsaung MT. Further evidence for familygenetic risk factors in attention deficit hyperactivity disorder: Patterns of comorbidity in probands and relatives in psychiatrically and pediatrically referred samples. Arch Gen Psychiatry 1992; 49:728-738.

- 825. Biederman J, Faraone SV, Keenan K, Tsaung MT. Evidence of familial association between attention deficit disorder and major affective disorder. *Arch Gen Psychiatry* 1991; 48:633-642.
- 826. Murphy KR, Barkley RA. Parents of children with attention-deficit/hyperactivity disorder: Psychological and attentional impairment. *Am J Orthopsychiatry* 1996; 66:93-102.
- 827. Shelton TL, Barkley RA, Crosswait C, Moorehouse M, Fletcher K, Barrett S, Jenkins L, Metevia L. Psychiatric and psychological morbidity as a function of adaptive disability in preschool children with aggressive and hyperactive-impulsive inattentive behavior. *J Abnorm Child Psychol* 1998; 26:475-494.
- 828. Camparo LB, Christensen A, Buhrmester D, Hinshaw SP. System functioning in families with ADHD and non ADHD sons. *Personal Relationships* 1994; 1:301–308.
- 829. Faraone SV, Biederman J, Keenan K, Tsaung MT. Separation of DSM-III attention deficit disorder and conduct disorder: Evidence from a family genetic study of American child psychiatric patients. *Psychol Med* 1991; 21:109-121.
- 830. Houghton S, Carroll A, Taylor M, O'Donoghue T. From traditional to ecological: Understanding attention deficit disorders through quantitative and qualitative research. New York: Nove Science Publishers, 2006.
- 831. Whalen CK, Henker B. The child with attention-deficit/hyperactivity disorder in family contexts, In: Quay HC and Hogan AE, eds. Handbook of disruptive behavior disorders. New York: Kluwer Academic / Plenum Publishers, 1999:139-155.
- 832. Crnic KA, Leconte JM. Understanding sibling needs and influences, In: Fewell RR and Vadasy PF, eds. Families of handicapped children: Needs and supports accross the lifespan. Austin, TX: PRO-ED, 1986:75-98.
- 833. Pike A, Coldwell J, Dunn JF. Sibling relationships in early/middle childhood: links with individual adjustment. *J Fam Psychol* 2005; 19:523-32.
- 834. Roizen NJ, Blondis TA, Irwin M, Rubinoff A, Kieffer J, Stein MA. Psychiatric and developmental disorders in families of children with attention-deficit hyperactivity disorder. *Arch Pediatr Adolesc Med* 1996; 150:203-208.
- 835. Biederman J, Munir K, Knee D, Habelow W, Armentano M, Autor S, al e. A family study of patients with attention deficit disorder and normal controls. J *Psychiatr Res* 1986; 20:263-274.
- 836. Faraone SV, Biederman J, Chen WJ, Milberger S, Warburton R, Tsaung MT. Genetic heterogeneity in attention-deficit hyperactivity disorder (ADHD): Gender, psychiatric comorbidity, and maternal ADHD. J Abnorm Psychol 1995; 104:334-345.
- 837. Seidman LJ, Biederman J, Monuteaux MC, Weber W. Neuropsychological functioning in nonreferred siblings of children with attention deficit/hyperactivity disorder. *J Abnorm Psychol* 2000; 109:252-265.
- 838. Schachar RJ, Crosbie J, Barr CL, Ornstein tJ, Kennedy J, Malone M, Roberts W, Ickowicz A, Tannock R, Chen S, Pathare T. Inhibition of motor responses in siblings concordant and discordant for attention deficit hyperactivity disorder. *Am J Psychiatr* 2005; 162:1076-1082.
- 839. Slaats-Willemse D, Swaab-Barneveld H, de Sonneville L, van der Meulen E, Buitelaar J. Deficient response inhibition as a cognitive endophenotype of ADHD. J Am Acad Child Adolesc Psychiatry 2003; 42:1242-1248.
- 840. Bidwell LC, Willcutt EG, DeFries JC, Pennington BF. Testing for neuropsychological endophenotypes in siblings discordant for attentiondeficit/hyperactivity disorder. *Biol Psychiatry* 2007; 62:991-998.
- 841. Mikami AY, Pfiffner L. Sibling relationships among children with ADHD. J Attention Disord 2008; 11:482-492.
- 842. Stone KL. An investigation of sibling relationships of children with AD/HD and their older siblings. (Doctoral dissertation, University of Kentucky), 2000.
- 843. Kendall J. Sibling accounts of attention deficit hyperactivity disorder (ADHD). Family Process 1999; 38:117-136.

- 844. McDougall MR, Hay DA, Bennett KS. Having a co-twin with attention-deficit hyperactivity disorder. *Twin Res Hum Genet* 2006; 9:148-54.
- 845. Hartman CA, Auerbach J, Erol N, Fonseca AC, Mellenbergh GJ, Gideon J, Novik TS, Oosterlaan J, Roussos AC, Shalev RS, Zilber N, Sergeant JA. Syndrome dimensions of the child behavior checklist and the teacher report form: A critical emperical evaluation. *J Child Psychol Psychiatr* 1999; 40:1095-1116.
- 846. Weiss M, Hechtman L, Weiss G. ADHD in parents. J Am Acad Child Adolesc Psychiatry 2000; 39:1059-61.
- 847. Murray C, Johnston C. Parenting in mothers with and without attentiondeficit/hyperactivity disorder. *J Abnorm Psychol* 2006; 115:52-61.
- 848. Banks T, Ninowski JE, Mash EJ, Semple DL. Parenting behavior and cognitions in a comminuty sample of mothers with and without symptoms of attentiondeficit/hyperactivity disorder. *J Child Fam Stud* 2008; 17:28-43.
- 849. Harvey E, Danforth JS, Eberhardt McKee T, Ulaszek WR, Friedman JL. Parenting of children with attention-deficit/hyperactivity disorder (ADHD): The role of parental ADHD symptomatology. *J Attention Disord* 2003; 7:31-42.
- 850. Psychogiou L, Daley DM, Thompson MJ, Sonuga-Barke EJS. Do maternal attention-deficit/hyperactivity disorder symptoms exacerbate or ameliorate the negative effect of child attention-deficit/hyperactivity disorder symptoms on parenting? *Dev Psychopathol* 2008; 20:121-137.
- 851. Sonuga-Barke E, Daley D, Thompson M. Does maternal ADHD reduce the effectiveness of parent training for preschool children's ADHD? J Am Acad Child Adolesc Psychiatry 2002; 41:696-702.
- 852. Australian Institute of Health and Welfare. Child protection Australia 2006–07. Child welfare series no. 43. Cat. no. CWS 31. Canberra: AIHW. 2008.
- 853. Royal Australasian College of Physicians (RACP). Health of children in "out- ofhome" care. Sydney: RACP 2006.
- 854. Pilowsky D. Psychopathology among children placed in family foster care. *Psychiatr Serv* 1995; 46:906-10.
- 855. Armsden G, Pecora P, Payne V, Szatkiewicz J. Children placed in long-term foster care: an intake profile using the child behavior checklist/4-18. *. J Emot Behav Disord* 2000; 8:49-64.
- 856. Leslie LK, Landsverk J, Ezzet-Lofstrom R, Tschann JM, Slymen DJ, Garland AF. Children in foster care: factors influencing outpatient mental health service use. *Child Abuse Negl* 2000; 24:465-76.
- 857. Horwitz S, Owens P, Simms M. Specialized assessments for children in foster care. *Pediatrics* 2000; 106:59–66.
- 858. Simmel C, Brooks D, Barth RP, Hinshaw SP. Externalizing symptomatology among adoptive youth: prevalence and preadoption risk factors. *J Abnorm Child Psychol* 2001; 29:57-69.
- 859. Barber JG, Delfabbro PH, Cooper LL. The predictors of unsuccessful transition to foster care. J Child Psychol Psychiatry 2001; 42:785-90.
- 860. Tarren-Sweeney M, Hazell P. Mental health of children in foster and kinship care in New South Wales, Australia. *J Paediatr Child Health* 2006; 42:89-97.
- 861. Racusin R, Maerlender AC, Jr., Sengupta A, Isquith PK, Straus MB. Psychosocial treatment of children in foster care: a review. *Community Ment Health J* 2005; 41:199-221.
- 862. dosReis S, Owens PL, Puccia KB, Leaf PJ. Multimodal treatment for ADHD among youths in three Medicaid subgroups: disabled, foster care, and low income. *Psychiatr Serv* 2004; 55:1041-8.
- 863. Zito JM, Safer DJ, Sai D, Gardner JF, Thomas D, Coombes P, Dubowski M, Mendez-Lewis M. Psychotropic Medication Patterns Among Youth in Foster Care. *Pediatrics* 2008; 121:e157-163.
- 864. Rosler M, Retz W, Retz-Junginger P, Hengesch G, Schneider M, Supprian T, Schwitzgebel P, Pinhard K, Dovi-Akue N, Wender P, Thome J. Prevalence of attention deficit-/hyperactivity disorder (ADHD) and comorbid disorders in
young male prison inmates. Eur Arch Psychiatr Clin Neurosci 2004; 254:365-71.

- 865. Satterfield JH, Faller KJ, Crinella FM, Schell AM, Swanson JM, Homer LD. A 30year prospective follow-up study of hyperactive boys with conduct problems: adult criminality. J Am Acad Child Adolesc Psychiatry 2007; 46:601-10.
- 866. Lay B, Ihle W, Esser G, Schmidt MH. Juvenile-episodic, continued or adultonset delinquency?: Risk conditions analysed in a cohort of children followed up to the age of 25 years. *Eur J Criminol* 2005; 2:39-66.
- 867. Rabiner D, Coie J, Miller-Johnson S, Boykin A-S, Lochman J. Predicting the persistence of aggressive offending of African American males from adolescence into young adulthood: the importance of peer relations, aggressive behavior, and ADHD symptoms. *J Emot Behav Disord* 2005; 13:131-140.
- 868. Messer J, Maughan B, Quinton D, Taylor A. Precursors and correlates of criminal behaviour in women. *Crim Behav Ment Health* 2004; 14:82-107.
- 869. Sourander A, Elonheimo H, Niemela S, Nuutila A-M, Helenius H, Sillanmaki L, Piha J, Tamminen T, Kumpulainen K, Moilenen I, Almqvist F. Childhood predictors of male criminality: a prospective population-based follow-up study from age 8 to late adolescence. J Am Acad Child Adolesc Psychiatry 2006; 45:578-86.
- 870. Gunter TD, Arndt S, Riggins-Caspers K, Wenman G, Cadoret RJ. Adult outcomes of attention deficit hyperactivity disorder and conduct disorder: Are the risks independent or additive? *Ann Clin Psychiatry* 2006; 18:233-237.
- 871. Satterfield JH, Schell A. A prospective study of hyperactive boys with conduct problems and normal boys: adolescent and adult criminality. *J Am Acad Child Adolesc Psychiatry* 1997; 36:1726-35.
- Teplin LA, Abram KM, McClelland GM, Dulcan MK, Mericle AA. Psychiatric disorders in youth in juvenile detention. *Arch Gen Psychiatry* 2002; 59:1133-43.
- 873. Young S, Gudjonsson G, Lam J. Attention deficit hyperactivity disorder (ADHD) in personality disordered offenders and the association with disruptive behaviour problems. *J Forensic Psychiatr Psychol* 2003; 14.
- 874. Rasmussen K, Almik R, Levander S. Attention Deficit Hyperactivity Disorder, reading disability & personality disorders in a prison population. *J Am Acad Psychiatry Law* 2001; 296:186–93.
- 875. Doreleijers TA, Moser F, Thijs P, van Engeland H, Beyaert FH. Forensic assessment of juvenile delinquents: prevalence of psychopathology and decision-making at court in the Netherlands. *J Adolesc* 2000; 23:263-75.
- 876. Hayes J, O'Reilly G. Emotional Intelligence, Mental Health and Juvenile Delinquency. Cork: Juvenile Mental Health Matters.

http://www.juvenilementalhealthmatters.com/Research%20Reports.html, 2007.

- 877. Kenny D, Nelson P, Butler T, Lennings C, Allerton M, Champion U. New South Wales Young People on Community Orders Health Survey 2003-2006: Key Findings Report. The University of Sydney, p 25-26. 2007.
- 878. Dixon AH, P, Starling J. Psychopathology in female juvenile offenders. *J Child Psychol Psychiatry* 2004; 45:1150 - 8.
- 879. Dixon A, Howie P, Starling J. Trauma exposure, posttraumatic stress, and psychiatric comorbidity in female juvenile offenders. *J Am Acad Child Adolesc Psychiatry* 2005; 44:798 806.
- 880. Bickel R, Campbell A. Mental health of adolescents in custody: the use of the "Adolescent Psychopathology Scale" in a Tasmanian context. *Aust N Z J Psychiatry* 2002; 36:603-9.
- 881. Retz W, Retz-Junginger P, Hengesch G, Schneider M, Thome J, Pajonk FG, Salahi-Disfan A, Rees O, Wender PH, Rosler M. Psychometric and psychopathological characterization of young male prison inmates with and without attention deficit/hyperactivity disorder. *Eur Arch Psychiatry Clin Neurosci* 2004; 254:201-8.

- 882. Eyestone LL, R.J. H. An epidemiological study of attention-deficit hyperactivity disorder and major depression in a male prison population *Bull Am Acad Psychiatry Law* 1994; 22:181-193.
- 883. Gunter TD, Arndt S, Wenman G, Allen J, Loveless P, Sieleni B, Black DW. Frequency of mental and addictive disorders among 320 men and women entering the Iowa prison system: use of the MINI-Plus. *J Am Acad Psychiatry Law* 2008; 36:27-34.
- 884. Minor S. Attention Deficit/Hyperactivity Disorder, Aggression and Incarceration. Santiago, Cuba. Available at http://www.uo.edu.cu/ojs/index.php/stgo/issue/view/98/showToc. 2003.
- 885. Gudjonsson GH, Sigurdsson JF, Bragason OO, Newton AK, Einarsson E. Interrogative suggestibility, compliance and false confessions among prisoners and their relationship with attention deficit hyperactivity disorder (ADHD) symptoms. *Psychol Med* 2008:1-8.
- 886. Vitelli R. Prevalence of childhood conduct and attention deficit hyperactivity disorders in adult maximum-security inmates. *Int J Offender Ther Comp Criminol* 1996; 40:263-271.
- 887. James A, Lai F, Dahl C. Attention deficit hyperactivity disorder and suicide: a review of possible associations. *Acta Psychiatrica Scandinavica* 2004; 110:408-415.
- 888. Putnins AL. Correlates and predictors of self-reported suicide attempts among incarcerated youths. *Int J Offender Ther Comp Criminol* 2005; 49:143-57.
- 889. Plattner B, The SSL, Kraemer HC, Williams RP, Bauer SM, Kindler J, Feucht M, Friedrich MH, Steiner H. Suicidality, psychopathology, and gender in incarcerated adolescents in Austria. *J Clin Psychiatr* 2007; 68:1593-600.
- 890. Fazel M, Langström N, Grann M, Fazel S. Psychopathology in adolescent and young adult criminal offenders (15-21 years) in Sweden. *Soc Psychiatry Psychiatr Epidemiol* 2008; 43:319-324.
- 891. Young S. Forensic Aspects of ADHD In: Fitzgerald M, Bellgrove M, and Gill M, eds. Handbook of Attention Deficit Hyperactivity Disorder. London: John Wiley & Sons, 2007:95.
- 892. Breuk RE, Clauser CA, Stams GJ. The validity of questionnaire self-report of psychopathology and parent-child relationship quality in juvenile delinquents with psychiatric disorders. *J Adolesc* 2007; 30:761-771.
- 893. Putnins AL. Substance use by South Australian young offenders. Information Bulletin No 19, July 2001. Office of Crime Statistics 2001.
- 894. Putnins AL. Assessing recidivism risk among young offenders. Aust New Zeal J Criminol 2005; 38:324-339.
- 895. Barry LM, Gaines T. Attention Deficit Hyperactivity Disorder: Intervention as crime prevention. *Journal of Behavior Analysis of Offender and Victim Treatment and Prevention* 2008; 1:154-170.
- 896. Day A, Howells K. Psychological treatments for rehabilitating offenders: evidence-based practice comes of age *Aust Psychol* 2002; 37:39-47.
- 897. Gottfredson MR, Hirschi T. A general theory of crime. Palo Alto, CA: Stanford University Press, 1990.
- 898. Howells K, Heseltine K, Sarre R, Davey L, Day A. Correctional Offender Rehabilitation Programs: The National Picture in Australia. Report for Criminology Research Council: Forensic Psychology Research Group, Centre for Applied Psychological Research, The University of South Australia 2004.
- 899. Pullmann MD, Kerbs J, Koroloff N, Veach-white E, Gaylor R, Sieler D. Juvenile offenders with mental health needs: Reducing recidivism using Wraparound. *Crime & Delinquency* 2006; 52:375-397.
- 900. Tong JLS, Farrington D. How effective is the "Reasoning and Rehabilitation" programme in reducing reoffending? A meta-analysis of evaluations in four countries. *Psychology, Crime and Law* 2006 12:3-24.

- 901. Gudjonsson G, Young S. An overlooked vulnerability in a defendant: Attention deficit hyperactivity disorder and a miscarriage of justice. *Legal and Criminological Psychology* 2006; 11:211-218.
- 902. Gudjonsson G, Young S, Bramham J. Interrogative suggestibility in adults diagnosed with attention-deficit hyperactivity disorder (ADHD): a potential vulnerability during police questioning. *Pers Indiv Differ* 2007; 43:737–745.
- 903. Goldstein S. Attention Deficit Hyperactivity Disorder: Implications for the criminal justice system. *FBI Law Enforcement Bulletin* 1997; 66:11, <u>http://www.fbi.gov/publications/leb/1997/june973.htm</u>
- 904. Young S, Harty M. Treatment issues in a personality disordered offender: a case of attention deficit hyperacticity disorder in secure psychiatric services. J *Forensic Psychiatr* 2001; 12:158-167.
- 905. Matza L, Paramore C, Prasad M. A review of the economic burden of ADHD. *Cost Effectiveness and Resource Allocation* 2005; 3:9.
- 906. Leibson CL, Long KH. Economic implications of attention-deficit hyperactivity disorder for healthcare systems. *Pharmacoeconomics* 2003; 21:1239-62.
- 907. Leibson CL, Barbaresi WJ, Ransom J, Colligan RC, Kemner J, Weaver AL, Katusic SK. Emergency department use and costs for youth with attentiondeficit/hyperactivity disorder: associations with stimulant treatment. *Ambul Pediatr* 2006; 6:45-53.
- 908. Hinnenthal JA, Perwien AR, Sterling KL. A comparison of service use and costs among adults with ADHD and adults with other chronic diseases. *Psychiatr Serv* 2005; 56:1593-9.
- 909. Ray GT, Levine P, Croen LA, Bokhari FA, Hu TW, Habel LA. Attentiondeficit/hyperactivity disorder in children: excess costs before and after initial diagnosis and treatment cost differences by ethnicity. *Arch Pediatr Adolesc Med* 2006; 160:1063-9.
- 910. Pharmaceutical Services Branch; Department of Health. Stimulant Prescribing and Usage Patterns for the Treatment of ADHD in Western Australia (1 August 2003 – 31 December 2004) 2005.
- 911. Jensen PS, Garcia JA, Glied S, Crowe M, Foster M, Schlander M, Hinshaw S, Vitiello B, Arnold LE, Elliott G, Hechtman L, Newcorn JH, Pelham WE, Swanson J, Wells K. Cost-effectiveness of ADHD treatments: findings from the multimodal treatment study of children with ADHD. Am J Psychiatry 2005; 162:1628-36.
- 912. Foster EM, Jensen PS, Schlander M, Pelham WE, Jr., Hechtman L, Arnold LE, Swanson JM, Wigal T. Treatment for ADHD: is more complex treatment costeffective for more complex cases? *Health Serv Res* 2007; 42:165-82.
- 913. Gilmore A, Milne R. Methylphenidate in children with hyperactivity: review and cost-utility analysis. *Pharmacoepidemiol Drug Saf* 2001; 10:85-94.
- 914. Wu EQ, Birnbaum HG, Zhang HF, Ivanova JI, Yang E, Mallet D. Health care costs of adults treated for attention-deficit/hyperactivity disorder who received alternative drug therapies. *J Manag Care Pharm* 2007; 13:561-9.